FUDAN UNIVERSITY AND UNIVERSITÉ D'ANGERS

 $\begin{array}{c} {\rm Ann\acute{e}} \ 2011 \\ {\rm N^o} \ {\rm d'order} \ 1159 \end{array}$

Dynamics of McMullen maps and Thurston-type theorems for rational maps with rotation domains

THESIS

Speciality : Mathematics

By WANG Xiaoguang

June 13, 2011

Jury:

John H. HUBBARD	Cornell University, US	President
Kevin M. PILGRIM	Indiana University, US	Examinator
QIU Weiyuan	Fudan University, China	Advisor
Mary REES	University of Liverpool, England	Reviewer
Mitsuhiro Shishikura	Kyoto University, Japan	Reviewer
Tan Lei	Université d'Angers, France	Advisor
YIN Yongcheng	Zhejiang University, China	Examinator
Wu Shengjian	Peking University, China	Reviewer

LAREMA, Université d'Angers, France

Acknowledgement

First, I would like to thank Professors Mary Rees, Mitsuhiro Shishikura and Wu Shengjian for their acceptance to be the reporters of my thesis. Mary pointed out many typos and gave me many comments which highly improved the thesis. Mitsu showed me an example of Herman map which gave me a better understanding of rational maps with Herman rings. Wu provided me with many helpful comments.

It my great honor to thank Professors John Hubbard, Kevin Pilgrim and Yin Yongcheng to be the members of the jury.

I owe my deepest gratitude to my advisors, Professors Qiu Weiyuan and Tan Lei, for all their guidance and constant help for the past six years, during my master and Ph.D study at Fudan University and Université d'Angers.

It is Professor Qiu Weiyuan who leads me to complex dynamics at Fudan University. He has attractive personality. He is always a role model for me. Every time I have any kind of problems, he is always there. He shares generously with me his experience on how to deal with problems both in mathematics and in life. He corrected every mistake and typo of my first paper and gave me very good advice on academic writings. He suggested me to think about the dynamics of McMullen maps and provided me patient guidance every time I came up with difficulties. I am grateful also for the financial support provided through his grants.

I have moved to France for a joint Ph.D program since November 2009 and Prof. Tan Lei became my advisor at Université d'Angers. Tan Lei is, undoubtedly, very very nice and considerate. Her help makes my stay in France easier and memorable. I should say it's an utmost pleasure and pleasant time to be with her. I would never forget the first time we traveled together from Paris to Angers. On the train, I told her that I didn't know what the absence of Thurston obstructions really means. Then she explained to me by some examples that the algebraic conditions on $\lambda < 1$ is essentially equivalent to the Grötzsch inequality. I am greatly impressed by her effective way of absorbing and imparting knowledge. She introduces to me many interesting mathematics and encourages me to attend many academic conferences. I am indebted to her for her translating many French documents for me, which enables me to concentrate on mathematical research. She has served to me as an example of what it is like to be a successful mathematician as well as a charming person.

The thesis would not have been possible without the essential and generous help of Prof. Yin Yongcheng. I am co-directed by him during my stay at Fudan University. He has a great influence on me to be a researcher in mathematics. I have benefited a lot not only from fruitful discussions and pleasant conversations with him, but also from his charming personality and elegant manner. I will never forget his wonderful lectures and his way of thinking about mathematical problems, from which I find the beauty of mathematics.

Prof. Cui Guizhen is one of the nicest and most talent mathematicians I have ever known. I have learned a lot from his optimistic attitude toward life and his way of handling mathematical problems. His ideas, manuscripts, as well as his great mathematical works joint with Tan Lei have a big influence on the thesis' work. I am grateful for his support and help.

My basic knowledge on quasiconformal mapping and Teichmuller space comes from Prof. Chen Jixiu, who is a one of the best educators at Fudan University. I express my thanks to Prof. Jiang Yunping for his wonderful lectures on holomorphic motions. It's my pleasure to thank Prof. Qiao Jianyong for helpful discussions, which gave birth to Chaper 4 of the thesis. Thanks go to Pascale Roesch for helpful discussions, her hospitality and constant help. Part of the thesis is based on her beautiful work. I am grateful to Lin Wei for all his help and support at Fudan University. Discussions and conversations with Zhang Gaofei is helpful and enjoyable. I would like to thank Peng Wenjuan for all her help during my stay in France and Beijing.

I am indebted to many people for their kind help. I would like to mention: Xavier Buff, Caroline Chalumeau, Loïc Chaumont, Arnaud Chéritat, Bob Devaney, Adam Epstein, Núria Fagella, Linda Keen, Alexandra Le Petitcorps, Jean-Jacques Loeb, Carsten Petersen, Feliks Przytycki, Shen Weixiao, Wang Yuefei, Wang Yun, Michel Zinsmeister, and many others.

Special thanks to Ou Yafei, Zhang Pan, Zhao Weixia and Lv Jing for their essential helps.

I never forget many mathematical people I have ever met: Cheng Tao, Chen Tao, Fan Jinhua, Fang Caiyun, Gao Junyang, Gao Meina, Liu Rongli, Wang Zhe, Xiao Yingqing, Xie Lan, Yang Fei, Yang Jinghua, Zeng Jinsong, Zhai Yu, Antonio Garijo, Sebastein Godillion, Manjula, Tomasz, Yauhen Mikulich, etc.

I thank Chinese Scholarship Council, Chinese Academy of Science, CODY, Fudan University, Institut Henri Poincaré, Université d'Angers and Université Paul Sabatier for all kinds of support.

Finally, my thanks go to my parents and my dearest friends for their emotional support over the years.

Contents

A	Acknowledgement					
1	Ma	in theorems	1			
	1.1	Abstract	-			
	1.2	Dynamics of McMullen maps	4			
	1.3	Decomposition theorem and Thurston-type theorems	L			
	1.4	Other topics	,			
2	Background materials					
	2.1	Spherical derivative	(
	2.2	The modulus of an annulus	1(
	2.3	Distortion Theorems	13			
	2.4	Quasiconformal maps	16			
	2.5	Holomorphic motion	17			
	2.6	Extremal quasiconformal conjugacy	18			
3	On	On meromorphic invariant line fields				
	3.1	Introduction	23			
	3.2	Proof of the Main Theorem	25			
4	AN	Von-escape Locus	31			
	4.1	Introduction	31			
	4.2	Critical points and capture domains	33			
	4.3	Proof of Theorem 4.1.1	36			
	4.4	Proof of Theorem 4.1.2	48			
5	Dyı	namics of McMullen maps	51			
	5.1	Introduction	51			
	5.2	Preliminaries and Notations	53			
	5.3	Cut Rays in the Dynamical Plane	58			
		5.3.1 A Cantor set on the unit circle	58			
		5.3.2 Cut rays	61			
	5.4	Puzzles, Graphs and Tableaux	69			
		5.4.1 The Yoccoz Puzzle	69			
		5.4.2 Admissible graphs	71			
		5.4.3 Modified puzzle piece	78			
		$5.4.4$ Tableaux \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	80			
	5.5	Renormalizations	83			

		5.5.1 Periodic critical tableau implies renormalization	84
	FC	5.5.2 Properties of renormalizations	80
	0.0 F 7	A Criterion of Local Connectivity	81
	ə. (The boundary ∂B_{λ} is a Jordan curve	92
		5.7.1 No periodic critical tableau case	95
		5.7.2 Peol asso	90
		5.7.4 Local connectivity implies higher regularity	90 100
		5.7.5 Corollaries	100
	5.8	Local connectivity of the Julia set $J(f_{\lambda})$	102
	0.0	1000010010100000000000000000000000000	101
6	Dec	omposition Theorem and Thurston-type Theorems	107
	6.1	Introduction	107
	6.2	Decompositions of Herman maps	114
		6.2.1 Marked disk extension	118
	6.3	Combinatorial part	123
	6.4	Surgery part: Gluing holomorphic models	128
		6.4.1 Rational realizations can descend	129
		6.4.2 Promotion of rational realizations when $\Gamma = \emptyset$	131
	0 5	6.4.3 Promotion of rational realizations when $\Gamma \neq \emptyset$	133
	6.5	Decomposition part II	150
		6.5.1 The hole-filling operator	151
		6.5.2 Surface puzzle of constant complexity	152 155
	66	0.5.3 Marked disk extension	155 156
	0.0	Combinatorial part II	$150 \\ 157$
		6.6.2 Reduction of no Thurston obstructions	107 161
	67	Surgery part II	101 164
	6.8	Proof of Theorem 6.1.2 the first two parts	167
	6.9	Analytic part	168
	0.0	6.9.1 Bational-like maps	168
		6.9.2 Renormalizations of rational maps	171
		6.9.3 Herman-Siegel renormalization	175
		6.9.4 Q.c-equivalence vs Möbius conjugation	176
	6.10	Applications	178
	2	6.10.1 Characterization of hyperbolic rational maps	178
		6.10.2 Characterization of rational maps with Siegel disks	179
		6.10.3 Characterization of rational maps with Herman rings .	180
	6.11	No Thurston obstructions vs rational realization	181

Chapter 1 Main theorems

1.1 Abstract

The thesis mainly consists of two subjects:

• Dynamics of McMullen maps.

In this part, we study the local connectivity of Julia sets for rational maps. We develop Yoccoz puzzle techniques to study McMullen maps and show that the boundary of the basin of infinity is always a Jordan curve if the Julia set is not a Cantor set. This give a positive answer to a question posed by Devaney. We also show the Julia set of McMullen maps is locally connected except some special cases.

• Thurston's theory on characterization of rational maps and extensions. For this subject, we establish a 'Decomposition Theorem':

Every non-parabolic branched covering can be decomposed along a stable multicurve into finitely many Siegel maps or Thurston maps, such that the combinatorics and rational realizations of these resulting maps essentially dominate the original one.

These resulting maps can be considered as the renormalizations of the original map. The motivation to establish such a theorem is to prove a Thurston-type theorem for rational maps with Herman rings. The Decomposition Theorem implies:

Thurston-type theorems for rational maps with Herman rings can be reduced to Thurston-type theorems for rational maps with Siegel disks.

According to Shishikura, a rational map with Herman rings admits finitely many rational maps, with Siegel disks or without rotation domains, as renormalizations. The Decomposition Theorem extends this philosophy beyond rational maps.

The Decomposition Theorem enables us to extend Thurston's Theorem to many poscritically infinite cases and give characterizations of rational maps with attracting cycles, Siegel disks and Herman rings. On the other hand, it allows us to construct many branched coverings without Thurston obstructions but not equivalent to rational maps. Besides these two parts, the thesis also includes two short articles. One concerns the classification of rational maps admitting meromorphic line fields (Chapter 3) while the other concerns the parameter plane of a special family of rational maps (Chapter 4).

All of these parts are self-contained.

1.2 Dynamics of McMullen maps

The local connectivity of Julia sets for rational maps is a central problem in complex dynamical systems. It is well studied for classical type of rational maps, for example: hyperbolic and semihyperbolic maps, geometrically finite maps, see [CJY],[M1],[TY]. In polynomial case, it is also known a lot, see [DH2],[GS],[Kiwi],[Ly],[M2]. For quadratic polynomials, Yoccoz proved that the Julia set is locally connected provided that all periodic points are repelling and the map is not infinitely renormalizable, see [Hu],[M2]. Douady exhibited striking example of infinitely renormalizable quadratic polynomial with non locally connected Julia set, see [M2]. For general polynomial with connected Julia sets and without irrationally neutral cycles, Kiwi shows in [Kiwi] that the local connectivity of Julia set is equivalent to the non existence of wandering continua.

The powerful tool to study the local connectivity of Julia sets for polynomials is the so-called 'Branner-Hubbard-Yoccoz puzzle' techniques, which is introduced by Branner-Hubbard and Yoccoz, [BH]. It has a natural way of construction, which is induced by finite periodic external rays together with an equipotential curve.

However, for general rational maps, things are different. The construction of Yoccoz puzzle becomes quite involved, even impossible. Up to now, the only known rational maps which admit Yoccoz puzzle structures are the cubic Newton maps, whose Yoccoz puzzles are constructed by Roesch. In [Ro1], by Yoccoz puzzle techniques, Roesch shows striking differences between rational maps and polynomials. The method also leads to the local connectivity of Julia sets except some specific cases.

In this part, we use Yoccoz puzzle to study another family of rational maps, known as McMullen maps, of the form

$$f_{\lambda}: z \mapsto z^n + \lambda/z^n, \ \lambda \in \mathbb{C}^* = \mathbb{C} \setminus \{0\}, \ n \ge 3.$$

Dynamics of this family have been studied by Devaney and his group, see [D1],[D2],[DK],[DLU].

The difference of the Yoccoz puzzles between cubic Newton maps ([Ro1]) and McMullen maps is as follows: For cubic Newton maps, the ingredient of the Yoccoz puzzle is an converging ray that intersects the Julia set in a countably many points while for McMullen maps, the element to construct Yoccoz puzzle is a Jordan curve that intersects the Julia set in a Cantor set of points. This kind of Jordan curve is induced by some particular angle and can be viewed as an extention of the corresponding external ray.

We denote by B_{λ} the immediate attractive basin of ∞ . The topology of ∂B_{λ} is of special interest. Based on Yoccoz puzzle techniques and combinatorial and topological analysis, we prove:

Theorem 1.2.1. (Cantor or Jordan) For any $n \geq 3$ and any complex parameter λ , if the Julia set $J(f_{\lambda})$ is not a Cantor set, then ∂B_{λ} is a Jordan curve.

This affirmatively answers a question posed by Devaney at the Snowbird Conference on the 25th Birthday of the Mandelbrot set, see [DK]. For the higher regularity of ∂B_{λ} , we show that ∂B_{λ} is a quasicircle except two special cases.

Theorem 1.2.2. Suppose the Julia set $J(f_{\lambda})$ is not a Cantor set, then ∂B_{λ} is a quasicircle if it contains neither parabolic point nor recurrent critical point.

Here, a recurrent critical point c on the Julia set of a rational map f is a critical point such that $c \in \omega(c)$, where $\omega(c)$ is the ω -limit set of c, defined as $\{z \in \overline{\mathbb{C}}; \text{ there exist } n_k \to \infty \text{ such that } z = \lim f^{n_k}(c)\}$. In fact, we can show that if ∂B_{λ} contains a parabolic point, then ∂B_{λ} is not a quasicircle by Leau-Fatou-Flower Theorem ([M2]). The question whether ∂B_{λ} is a quasicircle when ∂B_{λ} contains a recurrent critical point is still unknown.

For the topology of the Julia set, we show

Theorem 1.2.3. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then $J(f_{\lambda})$ is locally connected in either of the following cases:

- 1. The critical orbit does not accumulate on the boundary ∂B_{λ} .
- 2. The map f_{λ} is neither renormalizable nor *-renormalizable.
- 3. The parameter λ is real and positive.

Here are the definitions of renormalization and *-renormalization: If there exist a critical point c of f_{λ} , an integer $p \geq 1$ and two disks U and V containing c such that

$$\varepsilon f_{\lambda}^p : U \to V$$

is a quadratic like map whose Julia set is connected (here $\varepsilon \in \{\pm 1\}$ is a symbol), then we say f_{λ} is *p*-renormalizable at *c* if $\varepsilon = 1$ and f_{λ} is *p*-*-renormalizable at *c* if $\varepsilon = -1$.

Theorem 1.2.3 implies that the Julia set is locally connected except some special cases. In fact, it's stronger than the following statement:

Theorem 1.2.4. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then $J(f_{\lambda})$ is locally connected if the critical orbit does not accumulate on the boundary ∂B_{λ} .

Theorem 1.2.4 is an analogue of Roesch's Theorem [Ro1]:

Theorem 1.2.5. (Roesch) A genuine cubic Newton map, without Siegel disks, has a locally connected Julia set provided that the orbit of the non-fixed critical point does not accumulate on the boundary of any invariant basin of attraction.

1.3 Decomposition theorem and Thurston-type theorems

Let $f: S^2 \to S^2$ be an orientation preserving branched covering of degree at leat two. We denote by $\deg(f, x)$ the local degree of f at $x \in S^2$. The critical set Ω_f of f is defined by

$$\Omega_f = \{ x \in S^2; \deg(f, x) > 1 \},\$$

and the postcritical set P_f of f is defined by

$$P_f = \overline{\bigcup_{n \ge 1} f^n(\Omega_f)}.$$

We say that f is postcritically finite (also called 'critically finite') if P_f is a finite set. Such a map is always called a Thurston map. For a Thurston map, we define a function $\nu_f : S^2 \to \mathbb{N} \cup \{\infty\}$ in the following way: For each $x \in S^2$, define $\nu_f(x)$ (may be ∞) as the least common multiple of the local degrees deg (f^n, y) for all n > 0 and all $y \in S^2$ such that $f^n(y) = x$. (Notice that $\nu_f(x) = 1$ if $x \notin P_f$). We call $\mathcal{O}_f = (S^2, \nu_f)$ the orbifold of f.

In 1980s, Thurston proved the following theorem:

Theorem 1.3.1. (Thurston) Let $f: S^2 \to S^2$ be a critically finite branched covering. Suppose that \mathcal{O}_f does not have signature (2, 2, 2, 2). Then f is combinatorially equivalent to a rational function R if and only if for any fstable multicurve Γ , we have $\lambda(\Gamma, f) < 1$. The rational function R is unique up to Möbius conjugation.

1.3. DECOMPOSITION THEOREM AND THURSTON-TYPE THEOREMS

The detailed proof of Thurston's theorem is given by Douady and Hubbard [DH1].

Thurston's theorem has connection with a number of related areas such as Teichmuller theory, quasiconformal surgery, dynamics of several complex variables, transversality, group theory, algorithm, etc.

There are many applications of Thurston's theorem. Here is an incomplete list: Douady's proof of monotonicity of entropy for unimodal maps [Dou2], Kiwi's characterization of polynomial laminations [Kiwi] (using previous work of Bielefield-Fisher-Hubbard [BFH] and Poirier [Poi]), Mikulich's classification of postcritically finite Newton maps, McMullen's work on rational quotients [McM1], Pilgrim-Tan's cut-and-paste surgery along arcs ([PT1]), Rees' descriptions of parameter spaces [Rees2], Rees, Shishikura and Tan's studies on matings of polynomials ([Rees1],[ST], [Tan1], [Tan2]), ...

Over the years, there are several various attempts to generalize Thurston's theorem beyond postcritically finite rational maps. For example, David Brown [Bro], supported by the previous work of Hubbard and Schleicher [HS], has succeeded in extending the theory to the uni-critical polynomials with an infinite postcritical set (but always with a connected Julia set), and pushed it even further to the infinite degree case, namely the exponential maps. We would also like to mention a recent work of Hubbard-Schleicher-Shishikura [HSS] extending Thurston's theorem to postcritically finite exponential maps. Cui-Tan[CT1], and Jiang-Zhang [JZ], independently, using different methods, extend Thurston's theorem to hyperbolic rational maps. Furthermore, Cui and Tan [CT2] extend Thurston's theorem to geometrically finite rational maps. Meanwhile, Zhang [Zh2] extends Thurston's theorem to a class of rational maps with Siegel disks.

In this work, we aim to extend Thurston's theorem to a large class of branched covering, namely 'non-parabolic' branched covering. Roughly speaking, a 'non-parabolic' branched covering is a proper branched covering for which each critical point either has finite orbit or is attracted to an attracting cycle, or is eventually mapped to the closure of some rotation domain (rotation disk or rotation annulus, formal definition can be found in Section 6.1). A non-parabolic map with rotation disks and poscritically finite outside the closure of these rotation disks is called a Siegel map. Our main result is

Theorem 1.3.2. (Decomposition Theorem) Let (f, P) be a non-parabolic map, then there exist a (f, P)-stable multicurve Γ and a collection of Siegel maps or Thurston maps, say $\{(h_k, P_k), k \in \Lambda\}$, where Λ is a finite index set, such that

1. (Combinatorial part) (f, P) has no Thurston obstructions if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) has no Thurston obstructions.

2. (Surgery part) (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is q.c-equivalent to a rational map.

3. (Analytic part) (f, P) is q.c-equivalent to a unique rational map up to Möbius conjugation if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) q.c-equivalent to a unique rational map up to Möbius conjugation.

From the viewpoint of 'reduction', the theorem implies that Thurstontype Theorem for every non-parabolic branched covering can be reduced to Thurston-type Theorem for finitely many Siegel type branched coverings. In particular, Thurston-type Theorem for rational maps with Herman rings can be reduced to Thurston-type Theorem for rational maps with Siegel disks.

The 'Decomposition Theorem' provides a mechanism to produce Thurston type Theorems for non-parabolic maps. Thus it has many applications. For example, it can reduce Thurston-type Theorem for hyperbolic maps to Thurston's Theorem for postcritically finite maps (This is the idea of Cui-Tan's work [CT1]) and thus generalizes Cui-Tan and Jiang-Zhang's work. As another application, it enables us to give a characterization of a class of rational maps with Herman rings based on Zhang's work [Zh2], as follows:

Theorem 1.3.3. Let (f, P) be a non-parabolic map, with only one rotation annulus cycle which is of period one and has rotation number of bounded type, and without rotation disk. Then (f, P) is c-equivalent to a rational map (R, Q)if and only if (f, P) has no Thurston obstructions. Moreover, the Lebesgue measure of the Julia set J(R) is zero, and (R, Q) is unique up to Möbius conjugation.

There is no reason to believe that the absence of Thurston obstruction is always equivalent to rational realization for postcritically infinite branched covering, even if the equivalence is true for hyperbolic case ([CT1], [JZ]), some Siegel cases [Zh2] and Herman cases (Theorem 1.3.3). The mating of two quadratic Siegel polynomials $f_{\theta}(z) = z^2 + c_{\theta}$ and $f_{-\theta}(z) = z^2 + c_{-\theta}$, where $c_{\alpha} = \frac{e^{2\pi i \alpha}}{2} (1 - \frac{e^{2\pi i \alpha}}{2})$, provides a non-parabolic map $g = f_{\theta} \sqcup f_{-\theta}$ for which the equivalence is false. As a supplement to the Decomposition Theorem, following the same idea as Shishikura's construction [Sh1] of rational maps with prescribed numbers of non-repelling cycles and Herman rings, we can construct many such examples by surgery:

Theorem 1.3.4. Given nonnegative integers n_A , n_{RD} , n_{RA} , d satisfying

 $n_A + n_{RD} + 2n_{RA} \le 2d - 2, \ 1 \le n_{RA} \le d - 2, \ n_{RD} + n_{RA} \ge 2.$

There exists a non-parabolic map (f, P) of degree d, such that

1. $n_A(f) = n_A, n_{RD}(f) = n_{RD}, n_{RA}(f) = n_{RA}$, and the rotation number of each rotation cycle is of bounded type.

- 2. (f, P) has no Thurston obstructions.
- 3. (f, P) is not c-equivalent to a rational map.

1.4 Other topics

A line field supported on a subset E of the complex sphere $\overline{\mathbb{C}}$ is the Beltrami differential $\mu = \mu(z)d\overline{z}/dz$ supported on E with $|\mu| = 1$. We say μ is measurable if $\mu(z)$ is a measurable function. Let f be a rational map of degree $\deg(f) \geq 2$. We say f admits an invariant line field if there is a measurable line field μ supported on a set in $\overline{\mathbb{C}}$ with positive measure such that $f^*\mu = \mu$ a.e. (refer to [McM1]).

A meromorphic line field is a line field of the form $\mu = \bar{\phi}/|\phi|$, where ϕ is a nonzero meromorphic quadratic differential defined on $\overline{\mathbb{C}}$. We say f admits a meromorphic invariant line there is a meromorphic line field μ such that $f^*\mu = \mu$.

In Chapter 3, we classify the rational maps admitting meromorphic line fields:

Theorem 1.4.1. Let f be a rational map of degree $\deg(f) \ge 2$. Then f admits a meromorphic invariant line field if and only if f is conformally conjugate to one of the following maps:

1. Integral Lattès map.

2. Power map $z \mapsto z^d$, for $d \in \mathbb{Z}$ and $|d| \ge 2$.

3. $\pm T_n, n \geq 2$, where T_n is the n-th Chebyshef polynomial defined by $T_n(2\cos z) = 2\cos(nz)$.

In Chapter 4, We consider a family of rational maps

$$T_{\lambda}(z) = \left(\frac{z^2 + \lambda - 1}{2z + \lambda - 2}\right)^2,$$

where λ is a complex parameter. This family is indeed the family of renormalization transformations of 2-dimensional diamond-like hierachical Potts models in statistical mechanics. In 1983, Derrida *et al* show that the Yang-Lee zeros of the λ -state Potts model on the diamond hierachical lattice are dense in the Julia set $J(T_{\lambda})$ of the map T_{λ} (See [DDI]). Since then, much interest has been devoted to this family since it exhibits a connection between statistical mechanics and complex dynamics (See [EL],[O],[QG],[QL]).

For this family, note that when $\lambda = 0$, the map T_{λ} degenerates to the quadratic polynomial $T_0(z) = (z+1)^2/4$; when $\lambda \in \mathbb{C}^* = \mathbb{C} \setminus \{0\}$, 1 and ∞ are two supperattracting fixed points for the map T_{λ} while 0 is a critical value.

The non-escape locus \mathcal{M} associated to this family is defined by:

 $\mathcal{M} = \{\lambda \in \mathbb{C}^*; T_{\lambda}^n(0) \not\rightarrow_{n \to \infty} 1 \text{ and } T_{\lambda}^n(0) \not\rightarrow_{n \to \infty} \infty\} \cup \{0\}.$



Figure 1.1: The non-escape locus \mathcal{M}

This part includes two results: For the parameter plane, we have the following:

Theorem 1.4.2. The non-escape locus \mathcal{M} is connected.

For the Julia set, we have the following:

Theorem 1.4.3. If the Julia set $J(T_{\lambda})$ is a quasi-circle, then the Hausdorff dimension of $J(T_{\lambda})$ satisfies:

$$HD(J(T_{\lambda})) \le 1 + |\phi_{\lambda}(0)|^{2/3},$$

where ϕ_{λ} is the Böttcher map of T_{λ} defined near the supperattracting fixed point 1.

The exponent 2/3 in Theorem 1.4.3 is sharp.

This chapter presents some basic knowledge of complex analysis and conformal geometry, which are used in the thesis.

2.1 Spherical derivative

Let $f: \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ be a rational function, $d(\cdot, \cdot)$ be the spherical metric. We define the spherical derivative of f by

$$||f'(z)|| := \lim_{w \to z} \frac{d(f(w), f(z))}{d(w, z)} = \frac{1 + |z|^2}{1 + |f(z)|^2} |f'(z)|.$$

It's obvious that $||f'(\cdot)|| : \overline{\mathbb{C}} \to [0, +\infty)$ is a continuous function. The area element of the sphere is $dS = \left(\frac{2}{1+|z|^2}\right)^2 dx dy$, by calculation

$$\int_{\overline{\mathbb{C}}} \|f'(z)\|^2 dS = \int_{\overline{\mathbb{C}}} f^*(dS) = \deg(f) \int_{\overline{\mathbb{C}}} dS = 4\pi \deg(f).$$

By mean value theorem, for any rational function f, there is $\xi \in \overline{\mathbb{C}}$, such that $\|f'(\xi)\| = \sqrt{\deg(f)}$.

The spherical derivative relates to the normality of rational family:

Theorem 2.1.1. Let \mathcal{F} be a family of rational maps. Then \mathcal{F} is normal if and only if there is a constant $C = C(\mathcal{F})$ such that

$$||f'(z)|| \le C, \ \forall (f,z) \in \mathcal{F} \times \overline{\mathbb{C}}.$$

As an immediate consequence, we have:

Corollary 2.1.1. Let f be a rational map of degree at least two, then the Julia set J(f) is not empty.

Proof. If not, then $\mathcal{F} = \{f^n; n \geq 1\}$ is a normal family. By Theorem 2.1.1, there is a constant C such that for any $n \geq 1$ and any $z \in \overline{\mathbb{C}}$, $\|(f^n)'(z)\| \leq C$. On the other hand, by the mean value theorem, for any $n \geq 1$, there is $z_n \in \overline{\mathbb{C}}$, such that $\|(f^n)'(z_n)\| = \deg(f)^{n/2}$. Contradiction.

2.2 The modulus of an annulus

It's known that a two connected domain is conformally equivalent to $A_R = \{z \in \mathbb{C}; 1 < |z| < R\}, \mathbb{D} \setminus \{0\}, \text{ or } \mathbb{C} \setminus \{0\}$. For the former case, we define the modulus of A_R to be $\frac{1}{2\pi} \log R$. In the latter two cases, we define the moduli of $\mathbb{D} \setminus \{0\}$ and $\mathbb{C} \setminus \{0\}$ to be ∞ . The modulus is a conformal invariant.

Theorem 2.2.1. (McMullen Inequality) Let U, V be two simply connected planner domains, such that $\overline{U} \subset V \neq \mathbb{C}$, and let $A = V \setminus \overline{U}$. Then the modulus of A and the Euclidean areas of U, V satisfy:

$$e^{4\pi \operatorname{mod}(A)} \le \frac{\operatorname{area}(V)}{\operatorname{area}(U)}.$$

Equality holds if and only if ∂U and ∂V are concentric circles.

Proof. We assume A is conformally isomorphic to $A_R = \{z \in \mathbb{C}; 1 < |z| < R\}$, then $\operatorname{mod}(A) = \frac{1}{2\pi} \log R$. Take a conformal map $\varphi : A_R \to A$, preserving the boundary order of A. Suppose that φ has Laurant expansion: $\varphi(z) = \sum_{n \in \mathbb{Z}} a_n z^n$. It follows from area formula that

$$area(V) = \pi \sum_{n \in \mathbb{Z}} n|a_n|^2 R^{2n}, \quad area(U) = \pi \sum_{n \in \mathbb{Z}} n|a_n|^2.$$
$$area(V) - R^2 area(U) = \pi \sum_{n \in \mathbb{Z}} n|a_n|^2 (R^{2n} - R^2) \ge 0.(*)$$

So we have

$$e^{4\pi \operatorname{mod}(A)} = R^2 \le \frac{\operatorname{area}(V)}{\operatorname{area}(U)}$$

From (*) we see that equality holds if and only if for all $n \in \mathbb{Z} \setminus \{0, 1\}$, $a_n = 0$. In this case, $\varphi(z) = a_0 + a_1 z$ and $\partial U, \partial V$ are concentric circles.

Theorem 2.2.2. 1. (Monotonicity) Let A_1, A_2 be two annuli, $A_1 \subset A_2$, and A_1 separates the two boundary curves of A_2 , then $mod(A_1) \leq mod(A_2)$. Equality holds if and only if $A_1 = A_2$.

2. (Subadditivity) Let A_1, A_2, A be annuli. $A_1, A_2 \subset A$, and $A_1 \cap A_2 = \emptyset$. We assume A_1 and A_2 separate the two boundary curves of A. Then $\operatorname{mod}(A_1) + \operatorname{mod}(A_2) \leq \operatorname{mod}(A)$. Equality holds if and only if A_1, A_2, A are standard annuli with $\overline{A_1 \cup A_2} = \overline{A}$.

Proof. We only prove the 'monotonicity'. The proof of the 'subadditivity' follows from the same argument. Since the modulus is conformal invariant, we may assume $A_2 = \{1 < |z| < R\}$. We denote by U, V the simply connected

planar domains bounded by the inner boundary and outer boundary of A_1 . By McMullen Inequality

$$\operatorname{mod}(A_1) \le \frac{1}{4\pi} \log\left(\frac{\operatorname{area}(V)}{\operatorname{area}(U)}\right) \le \frac{1}{4\pi} \log\left(\frac{\pi R^2}{\pi}\right) = \operatorname{mod}(A_2).$$

The left equality holds if and only if the two boundaries of A_1 are concentric circles, and the middle equality implies that $A_1 = A_2$.

In the rest of this section, we introduce *Grötzsch constant* and discuss its relationship with modulus. This constant is called *Grötzsch defect* in Buff and Epstein's paper [BE].

We say that a compact set $K \subset \overline{\mathbb{C}}$ is *equatorial* if $\overline{\mathbb{C}} - K$ consists of two simply connected components, say U and V. Choose two base points p, qwith $p \in U, q \in V$, and then take $\alpha \in Aut(\overline{\mathbb{C}})$, which maps p, q to $0, \infty$ respectively. Let $\varphi_0 : (\mathbb{D}, 0) \to (\alpha(U), 0)$ and $\varphi_\infty : (\overline{\mathbb{C}} - \overline{\mathbb{D}}, \infty) \to (\alpha(V), \infty)$ be two Riemann mappings. Suppose that

$$\varphi_0(z) = \sum_{n \ge 1} a_n z^n, z \in \mathbb{D}; \quad \varphi_\infty(\zeta) = \sum_{n \le 1} b_n \zeta^n, \zeta \in \mathbb{C} - \overline{\mathbb{D}}.$$

We define the *Grötzsch constant* $\gamma(K, p, q)$ of K about p, q by:

$$\gamma(K, p, q) = \frac{1}{2\pi} \log \left| \frac{b_1}{a_1} \right|.$$

One may verify that $\gamma(K, p, q)$ is well-defined (independent of the choices of $\alpha, \varphi_0, \varphi_\infty$).

Example 2.2.1. Let S the unit circle, choose two points $p, q \in \mathbb{C}$, with |p| < 1, |q| > 1, we have

$$\gamma(\mathbb{S}, p, q) = \frac{1}{2\pi} \log\left(\frac{|p-q|^2}{(1-|p|^2)(|q|^2-1)}\right)$$

Theorem 2.2.3. The Grötzsch constant satisfies:

1. For any $\beta \in Aut(\overline{\mathbb{C}}), \ \gamma(\beta(K), \beta(p), \beta(q)) = \gamma(K, p, q).$

2. $\gamma(K, p, q) \ge 0$. $\gamma(K, p, q) = 0$ if and only if there is $\beta \in Aut(\mathbb{C})$, such that β maps $\mathfrak{S}, 0, \infty$ to K, p, q, respectively.

3. If the interior of K is an annulus A, then $\gamma(K, p, q) \ge \text{mod}(A)$. Equality holds if and only if there exist $\beta \in Aut(\overline{\mathbb{C}}), R > 1$, such that β maps $\{1 \le |z| \le R\}, 0, \infty$ to K, p, q, respectively.

4. (Reverse Grötzsch Inequality) For any R > 1, 0 < r < 1, let A(R, r) be the annulus bounded by $\varphi_0(\{|z| = r\})$ and $\varphi_{\infty}(\{|\zeta| = R\})$, A_r be the annulus bounded by $\varphi_0(\{|z|=r\})$ and $\partial \alpha(U)$, A_R be the annulus bounded by $\varphi_{\infty}(\{|\zeta|=R\})$ and $\partial \alpha(V)$. Then for any R > 1, 0 < r < 1, we have

$$\operatorname{mod}(A(R,r)) \le \operatorname{mod}(A_R) + \operatorname{mod}(A_r) + \gamma(K,p,q).$$

Equality holds for some pair (R, r) if and only if there is $\beta \in Aut(\overline{\mathbb{C}}), L \ge 1$ such that β maps $\{1 \le |z| \le L\}, 0, \infty$ to K, p, q, respectively.

Remark 2.2.1. The 'Reverse Grötzsch Inequality' is first introduced in [C], see also [CT1]. In [CT1], Cui Guizhen and Tan Lei use it as an analytic tool to prove a Thurston-type theorem for hyperbolic rational maps. In chaper 6, we will also use it to prove the 'Decomposition Theorem'.

Proof. One may verify 1 by definition, we omit the details. The proofs of 2 and 3 are based on the area formula. Since $\gamma(K, p, q)$ is invariant under Möbius transformation, we may identify K, p, q with $\alpha(K), 0, \infty$. It follows from area formula that

$$area(U) = \pi \sum_{n \ge 1} n|a_n|^2, \quad area(\overline{\mathbb{C}} - V) = \pi \sum_{n \le 1} n|b_n|^2.$$

Since $U \subset \overline{\mathbb{C}} - V$, $\pi |a_1|^2 \leq area(U) \leq area(\overline{\mathbb{C}} - V) \leq \pi |b_1|^2$. This means $|b_1| \geq |a_1|$. Equivalently, $\gamma(K, p, q) \geq 0$. $|b_1| = |a_1|$ if and only if for any $n \geq 2, a_n = 0$, and for any $n \leq -1$, $b_n = 0$. So $\partial U = |a_1| \mathbb{S}, \partial V = |b_1| \mathbb{S} + b_0$. Since $U \cap V = \emptyset$, we have that $b_0 = 0$ and $\partial U = \partial V = |a_1| \mathbb{S}$.

If the interior of K is an annulus A, then it follows from McMullen Inequality that

$$\operatorname{mod}(A) \leq \frac{1}{4\pi} \log \left(\frac{\operatorname{area}(\overline{\mathbb{C}} - V)}{\operatorname{area}(U)} \right) \leq \frac{1}{4\pi} \log \left(\frac{\pi |b_1|^2}{\pi |a_1|^2} \right) = \gamma(K, p, q).$$

Equality $\operatorname{mod}(A) = \gamma(K, p, q)$ holds if and only if the two boundary curves of A are concentric circles, moreover $\varphi_0(z) = a_1 z, \varphi_\infty(\zeta) = b_1 \zeta + b_0$. One can verify that in this case, there is $\beta \in \operatorname{Aut}(\overline{\mathbb{C}})$ such that β maps $\{1 \leq |z| \leq e^{2\pi \operatorname{mod}(A)}\}, 0, \infty$ to K, p, q, respectively.

Here, we give two different proofs of the 'Reverse Grötzsch Inequality'.

The first is based on the McMullen Inequality. The bounded component of $\overline{\mathbb{C}} - \varphi_0(\{|z| = r\})$ is denoted by U_r , the bounded component of $\overline{\mathbb{C}} - \varphi_\infty(\{|\zeta| = R\})$ is denoted by V_R . It follows from area formula that

$$area(U_r) = \pi \sum_{n \ge 1} n|a_n|^2 r^{2n}, \ area(V_R) = \pi \sum_{n \le 1} n|b_n|^2 R^{2n}.$$

By McMullen Inequality,

$$\operatorname{mod}(A(R,r)) \leq \frac{1}{4\pi} \log\left(\frac{\operatorname{area}(V_R)}{\operatorname{area}(U_r)}\right) \leq \frac{1}{4\pi} \log\left(\frac{\pi |b_1|^2 R^2}{\pi |a_1|^2 r^2}\right)$$
$$= \operatorname{mod}(A_R) + \operatorname{mod}(A_r) + \gamma(K, p, q).$$

One may easily verify the condition when the equality holds.

Here is another proof of the 'Reverse Grötzsch Inequality', based on Koebe distortion Theorem: we consider the function

$$f(R,r) = \text{mod}(A(R,r)) - \text{mod}(A_R) - \text{mod}(A_r), \ 0 < r < 1, R > 1.$$

It follows from Grötzsch inequality that $f \ge 0$ and for any $0 < r_2 \le r_1 < 1, R_2 \ge R_1 > 1$, we have $f(R_1, r_1) \le f(R_2, r_2)$. This implies the limit $\lim_{R\to\infty,r\to0} f(R,r)$ exists. By Koebe Theorem, when r is small enough, $\varphi_0(\{|\zeta|=r\})$ looks like a round circle of radius a_1r ; when R is large enough, $\varphi_\infty(\{|\zeta|=R\})$ looks like a round circle of radius b_1R . It turns out that

$$\operatorname{mod}(A(R,r)) = \frac{1}{2\pi} \log \left| \frac{b_1 R (1 + \mathcal{O}(1/R))}{a_1 r (1 + \mathcal{O}(r))} \right|$$

= $\frac{1}{2\pi} \log \frac{R}{r} + \frac{1}{2\pi} \log \left| \frac{b_1}{a_1} \right| + \mathcal{O}(1/R) + \mathcal{O}(r)$

So we have

$$\lim_{R \to \infty, r \to 0} f(R, r) = \frac{1}{2\pi} \log \left| \frac{b_1}{a_1} \right| = \gamma(K, p, q).$$

This means, for any 0 < r < 1, R > 1, $f(R, r) \le \gamma(K, p, q)$. Moreover, the constant $\gamma(K, p, q)$ is sharp.

2.3 Distortion Theorems

Here are some distortion theorems used in the thesis.

Theorem 2.3.1. (Koebe) Let $f : \mathbb{D} \to \mathbb{C}$ be a univalent function. Then for all $z \in \mathbb{D}$,

$$|f'(0)|\frac{|z|}{(1+|z|)^2} \le |f(z) - f(0)| \le |f'(0)|\frac{|z|}{(1-|z|)^2}.$$

Let $U \subsetneq \mathbb{C}$ be a simply connected planar domain and $z \in U$. The shape of U about z is defined by:

Shape
$$(U, z) = \sup_{x \in \partial U} |x - z| / \inf_{x \in \partial U} |x - z|.$$

It's obvious that $\text{Shape}(U, z) = \infty$ if and only if U is unbounded, and Shape(U, z) = 1 if and only if U is a round disk centered at z. In all other cases, $1 < \text{Shape}(U, z) < \infty$.

Let K be a connected and compact subset of U, containing at least two points. For any $z_1, z_2 \in K$, define the turning of K about z_1 and z_2 by:

$$\Delta(K; z_1, z_2) = \operatorname{diam}(K) / |z_1 - z_2|,$$

where diam(·) is the Euclidean diameter. It's obvious that $1 \leq \Delta(K; z_1, z_2) \leq \infty$ and $\Delta(K; z_1, z_2) = \infty$ if and only if $z_1 = z_2$.

Theorem 2.3.2. For $i \in \{1, 2\}$, let (V_i, U_i) be a pair of simply connected planar domains with $\overline{U_i} \subset V_i \subsetneq \mathbb{C}$. $g: V_1 \to V_2$ is a proper holomorphic map of degree D, U_1 is a component of $g^{-1}(U_2)$ and let $d = \deg(g|_{U_1})$. Then

1. We have the following modular distortion:

$$d \operatorname{mod}(V_1 \setminus \overline{U_1}) \le \operatorname{mod}(V_2 \setminus \overline{U_2}) \le D \operatorname{mod}(V_1 \setminus \overline{U_1}).$$

2. Suppose further $mod(V_2 \setminus \overline{U_2}) \ge m > 0$, then

2.1. (Shape distortion) There is a constant $C_0(D,m) > 0$ such that for all $z \in U_1$,

$$\operatorname{Shape}(U_1, z) \leq C_0(D, m) \operatorname{Shape}(U_2, g(z)).$$

2.2. (Turning distortion) There is a constant $C_1(D,m) > 0$ such that for any connected and compact subset K of U_1 with $\#K \ge 2$ and any $z_1, z_2 \in K$,

$$\Delta(K; z_1, z_2) \le C_1(D, m) \Delta(g(K); g(z_1), g(z_2)).$$

Proof. The proof of the first statement (modular distortion) can be found in [KL].

Proof of the Shape distortion. The proof presented here is borrowed from Zhai's Thesis [Zhai]. Fix some point $z \in U_1$, take two Riemann mappings $\phi_1 : (V_1, z) \to (\mathbb{D}, 0)$ and $\phi_2 : (V_2, g(z)) \to (\mathbb{D}, 0)$, then the map $G = \phi_2 \circ g \circ \phi_1^{-1} : \mathbb{D} \to \mathbb{D}$ is a proper map, and G(0) = 0. By modular distortion, we have $\operatorname{mod}(V_1 \setminus \overline{U_1}) \ge m/D$. By Koebe Theorem and Grötzsch Theorem, there are two constants $C_1 = C_1(m/D)$ and $C_2 = C_2(m)$ such that

$$C_1^{-1}$$
Shape $(\phi_1(U_1), 0) \leq$ Shape $(U_1, z) \leq C_1$ Shape $(\phi_1(U_1), 0),$
 C_2^{-1} Shape $(\phi_2(U_2), 0) \leq$ Shape $(U_2, g(z)) \leq C_2$ Shape $(\phi_2(U_2), 0).$

In the following, we will show that there is a constant $C_3 = C_3(D) > 0$ such that

$$\operatorname{Shape}(\phi_1(U_1), 0) \le C_3 \operatorname{Shape}(\phi_2(U_2), 0).$$

Let

$$L_{1} = \max_{x \in \partial \phi_{1}(U_{1})} |x|, \ l_{1} = \min_{x \in \partial \phi_{1}(U_{1})} |x|,$$

$$L_{2} = \max_{y \in \partial \phi_{2}(U_{2})} |y|, \ l_{2} = \min_{y \in \partial \phi_{2}(U_{2})} |y|.$$

By Schwarz Lemma, $l_1 \ge l_2$. If $L_2 \ge \frac{1}{2}$, then

Shape
$$(\phi_1(U_1), 0) = \frac{L_1}{l_1} \le \frac{1}{l_1} \le \frac{2L_2}{l_2} = 2$$
Shape $(\phi_2(U_2), 0)$.

Now we consider the case $L_2 \leq \frac{1}{2}$. Let $\mathbb{D}_{2L_2} = \{z \in \mathbb{C}; |z| < 2L_2\}$ and W be the connected component of $G^{-1}(\mathbb{D}_{2L_2})$ that contains 0. By the maximum modulus principle, W is simply connected. Let $\varphi : (W, 0) \to (\mathbb{D}, 0)$ be a Riemann mapping.

Then the map $F = \frac{1}{2L_2}G \circ \varphi^{-1} : \mathbb{D} \to \mathbb{D}$ is a proper map, with F(0) = 0. Since $F \circ \varphi \circ \phi_1(U_1) = \frac{1}{2L_2}\phi_2(U_2)$ and $\max_{w \in \partial F \circ \varphi \circ \phi_1(U_1)} |w| = \frac{1}{2}$, by the previous argument,

Shape
$$(\varphi \circ \phi_1(U_1), 0) \leq 2$$
Shape $(\phi_2(U_2), 0)$.

Notice that $\operatorname{mod}(W \setminus \phi_1(U_1)) \geq \frac{1}{2\pi} \log 2$, we have that $\operatorname{mod}(W \setminus \overline{U_2}) \geq \frac{1}{2\pi D} \log 2$. By Koebe Theorem and Grötzsch Theorem, there is a constant C = C(D) such that

Shape $(\phi_1(U_1), 0) \leq C$ Shape $(\varphi \circ \phi_1(U_1), 0)$.

Then the conclusion follows immediately.

Proof of the Turning distortion. We assume that $g(z_1) \neq g(z_2)$. For else, $\Delta(g(K); g(z_1), g(z_2)) = \infty$ and the conclusion follows. Let $\rho(x, y)$ be the hyperbolic distance in V_2 and let B_1, B_2 be the hyperbolic disks both centered at $g(z_1)$, with radii $\max_{\zeta \in g(K)} \rho(g(z_1), \zeta)$ and $\rho(g(z_1), g(z_2))$ respectively. Let $\varphi : V_2 \to D$ be the Riemann mapping with $\varphi(g(z_1)) = 0$ and let $W = \varphi(U_2)$. Since $\operatorname{mod}(\mathbb{D} \setminus \overline{W}) = \operatorname{mod}(V_2 \setminus \overline{U}_2) \geq m$, we conclude by Grötzsch Theorem that there is a constant $r(m) \in (0, 1)$ such that $W \subset D_{r(m)}$, here we use D_r to denote the disk $\{z; |z| < r\}$.

Note that $\varphi(B_1), \varphi(B_2)$ are two round disks centered at 0, say D_R and D_r respectively. By Koebe distortion, there exist three constants $C_1(m), C_2(m), C_3(m) > 0$ such that

$$\text{Shape}(B_1, g(z_1)) \le C_1(m), \text{ Shape}(B_2, g(z_1)) \le C_2(m),$$

$$R/r \le C_3(m) \max_{\zeta \in g(K) \cap \partial B_1} |g(z_1) - \zeta| / |g(z_1) - g(z_2)| \le C_3(m) \Delta(g(K); g(z_1), g(z_2))$$

For $i \in \{1, 2\}$, let W_i be the component of $g^{-1}(B_i)$ that contains z_1 . By the Maximum Principle, W_1 and W_2 are simply connected. We may assume that $K \subset \overline{W}_1$ (for else, we can replace B_1 by \widehat{B}_1 , a hyperbolic disk centered at $g(z_1)$ with radius $\varepsilon + \max_{\zeta \in g(K)} \rho(g(z_1), \zeta)$, where ε is a small positive constant, and then let $\varepsilon \to 0^+$). Thus diam $(K) \leq \text{diam}(W_1) \leq 2 \sup_{\zeta \in \partial W_1} |\zeta - z_1|$. Consider the location of z_2 , by Maximum Principle, either $z_2 \in \partial W_2$ or $z_2 \in U_1 \setminus \overline{W}_2$. In either case, $|z_1 - z_2| \ge \inf_{\zeta \in \partial W_2} |\zeta - z_1|$. Thus by Shape distortion,

$$\begin{aligned} \Delta(K;z_1,z_2) &\leq 2 \sup_{\zeta \in \partial W_1} |\zeta - z_1| / \inf_{\zeta \in \partial W_2} |\zeta - z_1| \\ &= 2 \mathrm{Shape}(W_1,z_1) \mathrm{Shape}(W_2,z_1) Q(W_1,W_2,z_1) \\ &\leq C_1(D,m) \mathrm{Shape}(B_1,g(z_1)) \mathrm{Shape}(B_2,g(z_1)) Q(W_1,W_2,z_1) \\ &\leq C_2(D,m) Q(W_1,W_2,z_1) \end{aligned}$$

where $Q(W_1, W_2, z_1) = \inf_{\zeta \in \partial W_1} |\zeta - z_1| / \sup_{\zeta \in \partial W_2} |\zeta - z_1|$. In the following, to finish, we show that there is a constant c(m) > 0 such that

$$Q(W_1, W_2, z_1) \le c(m)\Delta(g(K); g(z_1), g(z_2)).$$

In fact, we just need consider the case $Q(W_1, W_2, z_1) > 1$. In this case, the annulus $W_1 \setminus \overline{W}_2$ contains the round annulus $\{w \in \mathbb{C}; \sup_{\zeta \in \partial W_2} |\zeta - z_1| < |w - z_1| < \inf_{\zeta \in \partial W_1} |\zeta - z_1|\}$. It turns out that

$$\begin{aligned} \frac{1}{2\pi} \log Q(W_1, W_2, z_1) &\leq \mod(W_1 \setminus \overline{W}_2) \leq \mod(B_1 \setminus \overline{B}_2) = \frac{1}{2\pi} \log \frac{R}{r} \\ &\leq \frac{1}{2\pi} \log \left(C_3(m) \Delta(g(K); g(z_1), g(z_2)) \right) \end{aligned}$$

The conclusion follows.

2.4 Quasiconformal maps

Let $f: X \to Y$ be a homeomorphism between two Riemann surfaces. We say that f is a K-quasiconformal map $(K \ge 1)$ if in the distribution sense, $\partial f/\partial \bar{z}, \partial f/\partial z \in L^2_{loc}(X)$, and the Beltrami coefficient

$$\mu_f(z) := \frac{\partial f / \partial \bar{z}}{\partial f / \partial z}$$

satisfies $\|\mu_f\|_{\infty} \le k$, where k = (K-1)/(K+1).

A mapping f is 1-quasiconformal map if and only if f is a conformal map in the normal sense.

Theorem 2.4.1. (Measurable Riemann Mapping Theorem) For any $\mu \in L^{\infty}(\mathbb{C})$ with $\|\mu\|_{\infty} < 1$, there is a unique quasiconformal map $\phi : \mathbb{C} \to \mathbb{C}$, which fixes $0, 1, \infty$ and satisfies $\mu_{\phi} = \mu$.

Moreover, for any $\mu \in L^{\infty}(\mathbb{C})$ with $\|\mu\|_{\infty} \leq 1$, there exist a unique family of quasiconformal maps $\phi_t : \mathbb{C} \to \mathbb{C}, |t| < 1$, which fix $0, 1, \infty$ and satisfy $\mu_{\phi_t} = t\mu$. Then $\phi_t(z)$ is holomorphic with respect to $t \in \mathbb{D}$ for each $z \in \mathbb{C}$.

A quasicircle is the image of the unit circle S under a quasiconformal map.

Theorem 2.4.2. (Ahlfors) A Jordan curve $S \subset \mathbb{C}$ is a quasicircle if and only if there is a constant C > 0, such that for any $p, q \in S$, we have

 $\min\{\operatorname{diam}(S_{pq}^+), \operatorname{diam}(S_{pq}^-)\} \le C|p-q|,$

where diam is the Euclidean diameter and S_{pq}^+, S_{pq}^- are two components of $S - \{p, q\}$.

Given a quasiconformal map $\phi : \mathbb{C} \to \mathbb{C}$, Astala [Ast] showed that the Hausdorff dimension of the quasicircle $\phi(\mathbb{S})$ is less than $1 + \|\mu_{\phi}\|_{\infty}$. He also conjectured that the upper bounded can be improved by $1 + \|\mu_{\phi}\|_{\infty}^2$. This conjecture is resolved by Smirnov [Smi].

Theorem 2.4.3. (Smirnov) Let $\phi : \mathbb{C} \to \mathbb{C}$ be a quasiconformal map, then the Hausdorff dimension of $\phi(\mathbb{S})$ is bounded above by $1 + \|\mu_{\phi}\|_{\infty}^2$.

2.5 Holomorphic motion

Let \mathbb{D} be the unit disk and E be a subset of $\overline{\mathbb{C}}$.

Definition 2.5.1. A map $h : \mathbb{D} \times E \to \overline{\mathbb{C}}$ is called a holomorphic motion of E parameterized by \mathbb{D} and with base point 0 if

- 1. h(0, z) = z for all $z \in E$,
- 2. For every $c \in \mathbb{D}$, $z \mapsto h(c, z)$ is injective on E, and
- 3. For every $z \in E$, $c \mapsto h(c, z)$ is holomorphic for $c \in \mathbb{D}$.

In fact, \mathbb{D} can be replaced by any simply connected hyperbolic domain. Here is the well-known 'Holomorphic Motion Theorem' see [Slo], [GJW]:

Theorem 2.5.1. (Slodkowski) Let E be a compact subset of $\overline{\mathbb{C}}$, $h : \mathbb{D} \times E \to \overline{\mathbb{C}}$ be a holomorphic motion of E parameterized by \mathbb{D} and with base point 0. Then there is a holomorphic motion $H : \mathbb{D} \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$, which extends h. Moreover, for any fixed $c \in \mathbb{D}$, $H(c, \cdot) : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a quasiconformal map, with dilatation

$$K(H(c, \cdot)) \le \frac{1+|c|}{1-|c|}.$$

We can use Slodkowski's Theorem and Smirnov's Theorem to estimate the Hausdorff dimension of Julia set when it is a quasicircle. The following example illustrates how the methodology works. **Example 2.5.1.** Given a quadratic polynomial $f_c(z) = z^2 + c$, suppose that c lies in the cardioid \heartsuit of the Mandelbrot set. Then the Julia set $J(f_c)$ is a quasicircle and $J(f_0) = \mathbb{S}$. By the characterization of stability ([McM1], Theorem 4.2), there is a holomorphic motion $h: \heartsuit \times J(f_0) \to \mathbb{C}$ of $J(f_0) = \mathbb{S}$ parameterized by \heartsuit and with base point 0, such that $h(c, \mathbb{S}) = J(f_c)$ for all $c \in \heartsuit$. By Slodkowski's Theorem, h admits an extension $H: \heartsuit \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$, and for any $c \in \heartsuit$, $H(c, \cdot)$ is a quasiconformal map with dilatation

$$K(H(c, \cdot)) \le \exp(d_{\heartsuit}(0, c)) = \frac{1 + |1 - \sqrt{1 - 4c}|}{1 - |1 - \sqrt{1 - 4c}|}$$

where d_{\heartsuit} is the hyperbolic metric in \heartsuit . Notice that $H(c, \heartsuit) = J(f_c)$ for all $c \in \heartsuit$, it follows from Smirnov's Theorem that the Hausdorff dimension d_c of $J(f_c)$ satisfies:

$$d_c \le 1 + |1 - \sqrt{1 - 4c}|^2, c \in \heartsuit.$$

This implies $d_c = 1 + \mathcal{O}(|c|^2)$ when |c| is small. One may compare this estimate with Ruelle's expansion of d_c when |c| is small ([Ru]):

$$d_c = 1 + \frac{|c|^2}{4\log 2} + o(|c|^2)$$

Example 2.5.2. In chapter 4, we consider the quadratic family:

$$t_{\lambda}(z) = \left(\frac{z-1+\lambda}{z-1}\right)^2, \ \lambda \in \mathbb{C} - \{0\}.$$

By the same method, we can show that when the Julia set $J(t_{\lambda})$ is a quasicircle, the Hausdorff dimension d_{λ} of $J(t_{\lambda})$ satisfies:

$$d_{\lambda} \le 1 + |\phi_{\lambda}(0)|^{2/3},$$

where ϕ_{λ} is the Böttcher map of $T_{\lambda} = t_{\lambda} \circ t_{\lambda}$ defined near the supperattracting fixed point 1. For more details, see the proof of Theorem 4.1.2.

2.6 Extremal quasiconformal conjugacy

Let $f_c(z) = z^2 + c$ and M be the Mandelbrot set, B_c be the Böttcher coordinate of f_c defined in a neighborhood of ∞ . Douady and Hubbard [DH2] showed that the map defined by

$$\Phi: \begin{cases} \mathbb{C} - M \to \mathbb{C} - \overline{\mathbb{D}} \\ c \mapsto B_c(c) \end{cases}$$

is a conformal isomorphism.

Let $\rho(\cdot, \cdot)$ be the hyperbolic distance in $\mathbb{C} - M$. For any $c_1, c_2 \in \mathbb{C} - M$, one may verify that

$$\tanh\left(\frac{\rho(c_1, c_2)}{2}\right) = \left|\frac{\mathrm{Log}\Phi(c_1) - \mathrm{Log}\Phi(c_2)}{\mathrm{Log}\Phi(c_1) + \overline{\mathrm{Log}\Phi(c_2)}}\right|,$$

where $\text{Log} z := \log |z| + \arg z$ and the branch of Log is chosen such that $|\arg \Phi(c_1) - \arg \Phi(c_2)| \leq \pi$. Let $\mathcal{Q}(c_1, c_2)$ be the set of all quasiconformal conjugacies between f_{c_1} and f_{c_2} . That is, for any $\phi \in \mathcal{Q}(c_1, c_2), \phi : \mathbb{C} \to \mathbb{C}$ is a quasiconformal map and $\phi \circ f_{c_1} = f_{c_2} \circ \phi$.

Theorem 2.6.1. Given $c_1, c_2 \in \mathbb{C} - M$, there is $\phi \in \mathcal{Q}(c_1, c_2)$ such that

$$\|\mu_{\phi}\| = \inf\{\|\mu_{\varphi}\|; \varphi \in \mathcal{Q}(c_1, c_2)\} = \tanh\left(\frac{\rho(c_1, c_2)}{2}\right).$$

Proof. Step 1. For any $\varphi \in \mathcal{Q}(c_1, c_2)$,

$$\|\mu_{\varphi}\| \ge \tanh\left(\frac{\rho(c_1, c_2)}{2}\right).$$

We may assume $c_1 \neq c_2$, otherwise the conclusion follows immediately. First notice that $f_{c_1}^*(\mu_{\varphi}) = \mu_{\varphi}$. For $\lambda \in \mathbb{D}$, let φ_{λ} solve the Beltrami equation

$$\frac{\partial \varphi_{\lambda}}{\partial \bar{z}} / \frac{\partial \varphi_{\lambda}}{\partial z} = \lambda \mu_{\varphi} / \|\mu_{\varphi}\|,$$

with $0, 1, \infty$ fixed. φ_{λ} is holomorphic with respect to $\lambda \in \mathbb{D}$ and $\varphi_0 = id$.

The map $\varphi_{\lambda} \circ f_{c_1} \circ \varphi_{\lambda}^{-1}$ is the quadratic polynomial of the form $a(\lambda)z^2 + b(\lambda)$. Let $\phi_{\lambda} = a(\lambda)\varphi_{\lambda}$, then $\phi_{\lambda} \circ f_{c_1} \circ \phi_{\lambda}^{-1}$ is the quadratic polynomial $z^2 + c(\lambda)$, where $c : \mathbb{D} \to \mathbb{C} - M$ is a holomorphic map, with $c(0) = c_1, c(||\mu_{\varphi}||) = c_2$. By Schwarz Lemma, we have

$$\rho(c_1, c_2) = \rho(c(0), c(\|\mu_{\varphi}\|)) \le d_{\mathbb{D}}(0, \|\mu_{\varphi}\|) = \log \frac{1 + \|\mu_{\varphi}\|}{1 - \|\mu_{\varphi}\|}.$$

Equivalently,

$$\|\mu_{\varphi}\| \ge \tanh\left(\frac{\rho(c_1, c_2)}{2}\right).$$

Step 2. For any $c_* \in \mathbb{C} - M$ with $\arg \Phi(c_1) = \arg \Phi(c_*)$, there is $\zeta \in \mathcal{Q}(c_1, c_*)$, such that

$$\|\mu_{\zeta}\| = \tanh\left(\frac{\rho(c_1, c_*)}{2}\right) = \left|\frac{\log|\Phi(c_1)| - \log|\Phi(c_*)|}{\log|\Phi(c_1)| + \log|\Phi(c_*)|}\right|.$$

The proof is based on Thurston algorithm and the fact that $J(f_{c_1})$ has zero Lebesgue measure. Here is the detail:

For $i \in \{1, *\}$, let B_i be the Böttcher coordinate of f_{c_i} , $\Omega_i = \{z \in F(f_{c_i}); |B_i(z)| > |B_i(c_i)| = |\Phi(c_i)|\}$ and $D_i = \{w \in \mathbb{C}; |w| > |\Phi(c_i)|\}$. We define a quasiconformal homeomorphism $\delta : \Omega_1 \to \Omega_*$ such that the following diagram commutes

where α satisfies $|\Phi(c_1)|^{\alpha} = |\Phi(c_*)|$. We can extend δ to a quasiconformal map $\zeta_0 : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$. Then we can get a quasiconformal map ζ_1 such that $f_{c_*} \circ \zeta_1 = \zeta_0 \circ f_{c_1}$ and $\zeta_1|_{\Omega_1} = \delta$. By Thurston algorithm, there is a sequence of quasiconformal maps $\zeta_n : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that $f_{c_*} \circ \zeta_{n+1} = \zeta_n \circ f_{c_1}$ for $n \ge 0$ and ζ_{n+1} is isotopic to ζ_n rel $f_{c_1}^{-n}(\Omega_1)$. One can verify that

For $n \geq 1$, the quasiconformal dilatation $K(\zeta_n)$ of ζ_n is bounded above by $K(\zeta_0)$, so $\{\zeta_n\}$ is a normal family. This implies that there is a limit map $\zeta_{\infty} = \lim \zeta_n$, which is in fact a q.c conjugacy between f_{c_1} and f_{c_*} . Since $J(f_{c_1})$ has zero Lebesgue measure, the Beltrami coefficient of ζ_{∞} satisfies

$$\begin{aligned} \|\mu_{\zeta_{\infty}}\| &= \operatorname{ess.sup}_{z \in F(f_{c_1})} |\mu_{\zeta_{\infty}}(z)| = \lim_{n \to \infty} \operatorname{ess.sup}_{z \in f_{c_1}^{-n}(\Omega_1)} |\mu_{\zeta_{\infty}}(z)| \\ &= \lim_{n \to \infty} \operatorname{ess.sup}_{z \in f_{c_1}^{-n}(\Omega_1)} |\mu_{\zeta_n}(z)| = \operatorname{ess.sup}_{z \in \Omega_1} |\mu_{\zeta_0}(z)| \\ &= \operatorname{ess.sup}_{z \in \Omega_1} |\mu_{\delta}(z)| = \left| \frac{\alpha - 1}{\alpha + 1} \right| = \left| \frac{\log |\Phi(c_1)| - \log |\Phi(c_*)|}{\log |\Phi(c_1)| + \log |\Phi(c_*)|} \right| \end{aligned}$$

Step 3. The proof of the theorem.

We may replace c_2, φ by c_*, ζ_{∞} respectively in Step 1. Then the map $c : \mathbb{D} \to \mathbb{C} - M$ satisfies: $c(0) = c_1, c(\|\mu_{\zeta_{\infty}}\|) = c_*$. By Step 2,

$$\rho(c_1, c_*) = \rho(c(0), c(\|\mu_{\zeta_{\infty}}\|)) = d_{\mathbb{D}}(0, \|\mu_{\zeta_{\infty}}\|).$$

Thus $c : \mathbb{D} \to \mathbb{C} - M$ is a covering map. One can write explicitly

$$c(\lambda) = \Phi^{-1}(\Phi(c_1)^{\frac{1+\lambda}{1-\lambda}}), \lambda \in \mathbb{D}.$$

For any $c_2 \in \mathbb{C} - M$, we can chose $\lambda \in c^{-1}(c_2)$ such that $d_{\mathbb{D}}(0, \lambda) = \rho(c_1, c_2)$. The quasiconformal map ϕ_{λ} conjugate f_{c_1} to f_{c_2} , and

$$\|\mu_{\phi_{\lambda}}\| = |\lambda| = \tanh\left(\frac{\rho(c_1, c_2)}{2}\right),$$

as required.

Remark 2.6.1. Given two quadratic polynomials $f_{c_i}(z) = z^2 + c_i$, i = 1, 2. Suppose that c_1 and c_2 lie in the same hyperbolic component of the Mandelbrot set M and neither of f_{c_i} is postcritically finite. Let $\lambda(c_i)$ be the multiplier of the attracting cycle of f_{c_i} , $\mathcal{Q}(c_1, c_2)$ be the set of all quasiconformal conjugacies between f_{c_1} and f_{c_2} . Then by the same method (based on Thurston algorithm and the fact that $J(f_{c_i})$ has zero Lebesgue measure) as in Theorem 2.6.1, one can show that there is $\phi \in \mathcal{Q}(c_1, c_2)$ such that

$$\|\mu_{\phi}\| = \inf\{\|\mu_{\varphi}\|; \varphi \in \mathcal{Q}(c_1, c_2)\} = \left|\frac{\operatorname{Log}\lambda(c_1) - \operatorname{Log}\lambda(c_2)}{\operatorname{Log}\lambda(c_1) + \overline{\operatorname{Log}\lambda(c_2)}}\right|$$

where the branch of Log is chosen such that $|\arg \lambda(c_1) - \arg \lambda(c_2)| \leq \pi$.

Chapter 3 On meromorphic invariant line fields

3.1 Introduction

A line field supported on a subset E of the complex sphere $\overline{\mathbb{C}}$ is the Beltrami differential $\mu = \mu(z)d\overline{z}/dz$ supported on E with $|\mu| = 1$. We say μ is measurable if $\mu(z)$ is a measurable function. Let f be a rational map of degree $\deg(f) \geq 2$. We say f admits an invariant line field if there is a measurable line field μ supported on a set in $\overline{\mathbb{C}}$ with positive measure such that $f^*\mu = \mu$ a.e. (refer to [McM1]).

We are mostly interested in the invariant line fields which are carried on the Julia sets for rational maps. One example ever known is so called 'integral Lattès map', which is constructed via torus endomorphism. The construction is as follows. Let $X = \mathbb{C}/\Lambda$ be a complex torus, and α be a complex number such that $|\alpha| > 1$ and $\alpha \Lambda \subset \Lambda$. The multiplication by α induces an endomorphism $F: X \to X$. Let $\wp: X \to \overline{\mathbb{C}}$ be the Weierstrass function. Since $\wp(-z) = \wp(z)$, the endomorphism F can induce a rational map $f: \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that $f(\wp(z)) = \wp(F(z))$. Such a map f is called a Lattès map. If α is an integer, then F admits an invariant line field on X. This line field has the form $e^{i\theta} d\bar{z}/dz$ and can descend to an invariant line field for f (see [McM1]). On the other hand, we can verify that if f admits an invariant line field, then this line field can lift to an invariant line field for F and α turns out to be an integer. In this case, we say f is an integral Lattès map.

One of the central problems in complex dynamics is the following:

Conjecture. (No invariant line fields) A rational map f of degree $\deg(f) \ge 2$ carries no invariant line fields on its Julia set, except when f is an integral Lattès map.

The conjecture implies the density of hyperbolic maps in the space Rat_d of all rational maps of degree d (see [McM1]). Much study has been devoted to special families of rational maps, especially quadratic polynomials of the form $f_c(z) = z^2 + c$ for $c \in \mathbb{C}$. However, even for the quadratic family, the conjecture is still unsolved. Fortunately, if we require the line field μ to be 'good', we can classify all the rational maps which leave μ invariant. Here, a 'good' line field means that it can be written in the form $\mu = \overline{\phi}/|\phi|$, where ϕ is a nonzero meromorphic quadratic differential defined on $\overline{\mathbb{C}}$. In this case, formally, we call μ a meromorphic line field, dual to ϕ . Correspondingly, we say f admits a meromorphic invariant line field if $f^*\mu = \mu$.

Now we can formulate our main theorem:

Theorem 3.1.1. Let f be a rational map of degree $\deg(f) \ge 2$. Then f admits a meromorphic invariant line field if and only if f is conformally conjugate to one of the following maps:

1. Integral Lattès map.

2. Power map $z \mapsto z^d$, for $d \in \mathbb{Z}$ and $|d| \ge 2$.

3. $\pm T_n, n \geq 2$, where T_n is the n-th Chebyshef polynomial defined by $T_n(2\cos z) = 2\cos(nz)$.

This theorem is deeply inspired by a theorem in [McM1] which states that if a rational map f admits an invariant line field which is holomorphic on a nonempty open set contained in the Julia set, then f is an integral Lattès map. Moreover, three examples are provided in [McM1]. One is the power map $z \mapsto z^d$, for which the line field is dual to dz^2/z^2 ; another is the integral Lattès map, for which the line field is dual to

$$\frac{dz^2}{(z-p_1)(z-p_2)(z-p_3)(z-p_4)};$$

the third is the quadratic polynomial $f(z) = z^2 - 2$, for which the line field is dual to $dz^2/(z^2 - 4)$. So it is a natural question to figure out whether these are all examples which admit meromorphic invariant line fields. These examples motivate our study.

It is interesting to compare our classification theorem with another trichotomy theorem from the viewpoint of 'permutable maps'. Motivated by [M3], we call a rational map f is permutable if it commutes with another rational map g, $f \circ g = g \circ f$, where both f and g have degree at least two, and no iterate of f is equal to an iterate of g.

Theorem 3.1.2. (Ritt and Eremenko) A rational map f of degree $\deg(f) \geq 2$ is permutable if and only if it is a finite quotient of an affine map; that is if and only if it is either a Lattès, Chebyshef, or power map.

This theorem was first proved by Ritt [Ritt] in 1923, and by Eremenko [Ere] using a quite different method in 1989. For higher dimensional analogues, see [DS].

This chapter has been published as [W]. The same result is obtained by Rempe and van Strien in [RvS] using the orbifold theory, where they use it to prove the absence of invariant line fields on the Julia set of a class of transcendental meromorphic functions.

3.2 Proof of the Main Theorem

First, we need some notations. Let $\mathcal{M}(\mathbb{C})$ be the set of all meromorphic quadratic differentials defined on $\overline{\mathbb{C}}$. For $\phi \in \mathcal{M}(\overline{\mathbb{C}})$, let $\mathcal{Z}(\phi)$ and $\mathcal{P}(\phi)$ be the zero set and the pole set of ϕ respectively. The order of $\phi \in \mathcal{M}(\overline{\mathbb{C}})$ at a point z_0 , denoted by $\operatorname{ord}_{z_0}(\phi)$, is defined as follows. If z_0 is a zero of ϕ of order n, set $\operatorname{ord}_{z_0}(\phi) = n$; if z_0 is a pole of ϕ of order n, set $\operatorname{ord}_{z_0}(\phi) = -n$; for other cases, z_0 is called a regular point of ϕ , set $\operatorname{ord}_{z_0}(\phi) = 0$. For a rational map f, let C(f) be the set of all critical points,

$$P(f) = \overline{\bigcup_{c \in C(f), n > 0} f^n(c)}$$

be the postcritical set. The backward orbit of a point z, under iteration of f is denoted by $\operatorname{orb}^{-}(z) = \bigcup_{n \ge 0} f^{-n}(z)$. Let $\operatorname{deg}(f, z)$ be the local degree of f at z.

Proof of the Main Theorem. Let f be a rational map of degree $\deg(f) \geq 2$ and $\mu = \overline{\phi}/|\phi|$ be a meromorphic invariant line field of f for some $\phi \in \mathcal{M}(\overline{\mathbb{C}})$.

The 'if' part of the theorem is easy to verify. The proof for the 'only if' part is organized in five steps:

Step 1. $f^*(\bar{\phi}/|\phi|) = \bar{\phi}/|\phi|$ if and only if there is a positive constant C such that $f^*\phi = C\phi$. This constant C is uniquely determined by f. Moreover, any other meromorphic invariant line field of f must have the form $e^{i\theta}\mu$ for some $\theta \in \mathbb{R}$.

Note that the relation $f^*(\bar{\phi}/|\phi|) = \bar{\phi}/|\phi|$ is equivalent to

$$f^*\phi/\phi = |f^*\phi|/|\phi|.$$
 (3.1)

This indicates that the well-defined holomorphic map $f^*\phi/\phi: \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ takes only positive value, thus equation (3.1) holds if and only if $f^*\phi/\phi$ is a positive constant by open map theorem.

Now suppose $\mu_i = \phi_i/|\phi_i|$ (i = 1, 2) are two meromorphic invariant line fields for f. Above argument shows $f^*\phi_i = C_i\phi_i$, i = 1, 2. Since ϕ_1/ϕ_2 is a well-defined holomorphic map from $\overline{\mathbb{C}}$ to itself, denoted by R, the relation

$$\frac{f^*\phi_1}{f^*\phi_2} = \frac{C_1\phi_1}{C_2\phi_2}$$

implies that $R \circ f = (C_1/C_2)R$. Comparing the degree of $R \circ f$ and f, we conclude R is a nonzero complex constant, and $C_1 = C_2$. Therefore μ_1 is identical to μ_2 up to a rotation.

From now on, we may write ϕ, C as ϕ_f, C_f , since they are determined by f. To find all rational maps which admit meromorphic invariant line fields is equivalent to find all solutions $(f, \phi_f, C_f) \in \operatorname{Rat}_2^+ \times \mathcal{M}(\overline{\mathbb{C}}) \times \mathbb{R}^+$ to the indeterminate equation

$$f^*\phi_f = C_f \phi_f,$$

where Rat_2^+ is the space of all rational maps of degree at least two. In local coordinate, $\phi_f = \phi_f(z)dz^2$, the indeterminate equation has the form

$$\phi_f(f(z))f'(z)^2 = C_f \phi_f(z).$$
(3.2)

Moreover, for any $z \in \overline{\mathbb{C}}$, comparing the order of $f^*\phi_f$ and ϕ_f at the point z, we have the following identity

$$\operatorname{ord}_{z}(f^{*}\phi_{f}) = \operatorname{deg}(f, z)(2 + \operatorname{ord}_{f(z)}(\phi_{f})) - 2 = \operatorname{ord}_{z}(\phi_{f}).$$
(3.3)

Step 2. $\mathcal{Z}(\phi_f) = \emptyset$.

For else, let $z_0 \in \mathcal{Z}(\phi_f) \neq \emptyset$. We can conclude from equation (3.2) that $\operatorname{orb}^-(z_0) \subset \mathcal{Z}(\phi_f)$. Since $\mathcal{Z}(\phi_f)$ is a discrete subset of $\overline{\mathbb{C}}$, $\#\operatorname{orb}^-(z_0) < \infty$ and $\#f^{-1}(\operatorname{orb}^-(z_0)) \geq \#\operatorname{orb}^-(z_0)$. On the other hand,

$$f^{-1}(\operatorname{orb}^{-}(z_0)) = \bigcup_{n \ge 1} f^{-n}(z_0) \subset \operatorname{orb}^{-}(z_0).$$

So we have $f^{-1}(\operatorname{orb}^-(z_0)) = \operatorname{orb}^-(z_0)$. It is easy to see that all points in $\operatorname{orb}^-(z_0)$ are superattracting periodic points. If $\#\operatorname{orb}^-(z_0) \ge 3$, by Montel's Theorem, the set $\overline{\mathbb{C}}\setminus\operatorname{orb}^-(z_0)$ is completely invariant and lies in the Fatou set F(f). This indicates that the Julia set $J(f) = \emptyset$, which is a contradiction. Thus $\#\operatorname{orb}^-(z_0)$ can only be 1 or 2.

If $\# \operatorname{orb}^-(z_0) = 2$, then f is conformally conjugate to the power map $z \mapsto z^d$, for some $d \in \mathbb{Z}$. But as is known that any meromorphic invariant line field of the power map must be dual to Cdz^2/z^2 (see the previous and Step 1), which has no zeros. So this leads to a contradiction.

If $\# \text{orb}^-(z_0) = 1$, then $f^{-1}(z_0) = \{z_0\}$, $\deg(f, z_0) = \deg(f)$. By identity (3.3), we have

$$\operatorname{ord}_{z_0}(f^*\phi_f) = \operatorname{deg}(f) \left(\operatorname{ord}_{z_0}(\phi_f) + 2 \right) - 2 = \operatorname{ord}_{z_0}(\phi_f).$$

But this is also impossible since $\operatorname{ord}_{z_0}(\phi_f) \ge 1, \deg(f) \ge 2$.

Step 3. f is critically finite, that is $\#P(f) < \infty$. Moreover $P(f) = \mathcal{P}(\phi_f)$.

For any $c \in C(f)$, equation (3.2) implies that $f(c) \in \mathcal{P}(\phi_f)$. For else c will be a zero of ϕ_f , which is already ruled out in step 2. Replacing f by f^n , we have $f^n(c) \in \mathcal{P}(\phi_f)$, thus

$$\bigcup_{n\geq 0} f^n(C(f)) \subset \mathcal{P}(\phi_f).$$

This means f is critically finite, since $\mathcal{P}(\phi_f)$ is a finite set. Moreover $P(f) \subset \mathcal{P}(\phi_f)$.

If $\mathcal{P}(\phi_f) \setminus P(f) \neq \emptyset$, taking $z_0 \in \mathcal{P}(\phi_f) \setminus P(f)$, we have from equation (3.2) that

$$\operatorname{orb}^{-}(z_0) \subset \mathcal{P}(\phi_f), \ \ \#\operatorname{orb}^{-}(z_0) = \infty,$$

which is a contradiction. This ends the proof of step 3.

By the Riemann-Roch theorem,

$$\deg(\phi_f) = \#\mathcal{Z}(\phi_f) - \#\mathcal{P}(\phi_f) = -\#\mathcal{P}(\phi_f) = -4.$$

This means that ϕ_f has four poles (counting the multiplicity). Since f is critically finite, each periodic cycle of f is either repelling or superattracting (See [McM1] or [M1]).

Step 4. If f has no superattracting cycle, then f is an integral Lattès map.

The proof in this step is due to McMullen, compare [McM1]. For completeness, we include it here.

We first show that ϕ_f has four simple poles. That is, up to a constant,

$$\phi_f = \frac{dz^2}{(z - p_1)(z - p_2)(z - p_3)(z - p_4)}.$$

Indeed, if ϕ_f has a pole p_0 of order two or more, that is $\operatorname{ord}_{p_0}(\phi_f) \leq -2$, then we can conclude from the identity (3.3) by induction that for any $z \in \operatorname{orb}^-(p_0)$, $\operatorname{ord}_z(\phi_f) \leq -2$, therefore $\operatorname{orb}^-(p_0) \subset \mathcal{P}(\phi_f)$. The similar argument as in step 2 indicates that $f^{-1}(\operatorname{orb}^-(p_0)) = \operatorname{orb}^-(p_0)$ and $\#\operatorname{orb}^-(p_0) = 1$ or 2. It turns out that f is conjugate to a power map or a polynomial. In either case, f has a superattracting cycle. But this will contradict the assumption.

Now we consider the orbifold \mathcal{O}_f of f. Recall that the orbifold \mathcal{O}_f of the critically finite map f is a pair $(\overline{\mathbb{C}}, N_f)$, where $N_f : \overline{\mathbb{C}} \to \mathbb{N} \cup \{\infty\}$ takes values greater than one only on a discrete set of $\overline{\mathbb{C}}$. It is defined as follows:

(a) $N_f(x) = 1$, when $x \in \overline{\mathbb{C}} \setminus P(f)$,

(b) $N_f(x)$ is the least common multiple of the local degrees in the set $\{\deg(f^n, y); f^n(y) = x, n \ge 1\}$ for $x \in P(f)$,

(c) $N_f(x) = \infty$ if the local degrees in the set $\{ \deg(f^n, y); f^n(y) = x, n \ge 1 \}$ are unbounded.

We claim that $\mathcal{O}_f = (\overline{\mathbb{C}}, (2, 2, 2, 2))$. In fact, it's easy to see that $N_f(x) = 1$ when $x \in \overline{\mathbb{C}} \setminus P(f)$. For $x \in P(f)$, it is obvious $N_f(x) \ge 2$. Let $z \in f^{-n}(x)$ for $n \ge 1$. Note that $(f^n)^* \phi_f = C_f^n \phi_f$, by identity (3.3), we have

$$\operatorname{ord}_z((f^n)^*\phi_f) = \operatorname{deg}(f^n, z)(2 + \operatorname{ord}_x(\phi_f)) - 2 = \operatorname{ord}_z(\phi_f).$$

Since $P(f) = \mathcal{P}(\phi_f)$ and every pole of ϕ_f is simple in this case, $\operatorname{ord}_x(\phi_f) = -1$. Therefore $\operatorname{ord}_z(\phi_f)$ has only two choices 0 or -1 and $\operatorname{deg}(f^n, z)$ can only be 1 or 2. Thus $N_f(x) \leq 2$. The above argument shows that for any $x \in P(f)$, $N_f(x) = 2$. This proves the claim.

By Theorem A.5 in [McM1], if $\mathcal{O}_f = (\overline{\mathbb{C}}, (2, 2, 2, 2))$, then f is a Lattès map. By assumption, f admits a meromorphic invariant line field, so f is an integral Lattès map. Moreover, since ϕ_f is integrable over $\overline{\mathbb{C}}$, we have $C_f = \deg(f)$ from the following identity

$$\int_{\overline{\mathbb{C}}} |f^* \phi_f| = \deg(f) \int_{\overline{\mathbb{C}}} |\phi_f|.$$

Step 5. If f has a superattracting cycle, then f is either conjugate to a power map or conjugate to a Chebyshef polynomial.

Let z_0 be a superattracting point of f with period p. From the identity (3.3), we have

$$\operatorname{ord}_{z_0}((f^p)^*\phi_f) = \operatorname{deg}(f^p, z_0)(2 + \operatorname{ord}_{z_0}(\phi_f)) - 2 = \operatorname{ord}_{z_0}(\phi_f).$$

Since deg $(f^p, z_0) \ge 2$, we have $\operatorname{ord}_{z_0}(\phi_f) = -2$. Thus z_0 is a pole of ϕ_f of order two. Moreover by identity (3.3) and induction, all preimages of z_0 are poles of ϕ_f of order two. There are two possibilities:

(P1) z_0 is a fixed point of f and $f^{-1}(z_0) = \{z_0\}$.

(P2) z_0 is of period two and $f^{-1}(z_0) = \{\zeta\}, f^{-1}(\zeta) = \{z_0\}.$

For (P1), there are two choices for ϕ_f up to a constant:

Case 1.
$$\phi_f = \frac{dz^2}{(z - z_0)^2 (z - z_1)^2},$$

Case 2. $\phi_f = \frac{dz^2}{(z - z_0)^2 (z - z_1)(z - z_2)}.$

In case 1, take $\gamma \in \operatorname{Aut}(\overline{\mathbb{C}})$, the automorphism group of $\overline{\mathbb{C}}$, such that $\gamma(0) = z_1, \ \gamma(\infty) = z_0$. Then $\gamma^* \phi_f = C dz^2/z^2$ for some constant C and $F = \gamma^{-1} \circ f \circ \gamma$ is a polynomial such that $F^*(\gamma^* \phi_f) = C_f \gamma^* \phi_f$. By conjugation, we may assume f is a polynomial and $\phi_f = dz^2/z^2$. The equation $f^* \phi_f = C_f \phi_f$ is equivalent to

$$(f'(z)/f(z))^2 = C_f/z^2.$$
 (3.4)

Comparing the leading coefficients in both sides of (3.4), we have $C_f = \deg(f)^2$. It's easy to find the general polynomial solution $f(z) = Az^d$, where A is a nonzero complex constant and $d = \deg(f)$. In this case, f is conjugate to a power map.

In case 2, take $\gamma \in \operatorname{Aut}(\overline{\mathbb{C}})$, such that $\gamma(\infty) = z_0$, $\gamma(-2) = z_1$, $\gamma(2) = z_2$. It is easy to show $\gamma^* \phi_f = C dz^2 / (z^2 - 4)$ for some constant C and $F = \gamma^{-1} \circ f \circ \gamma$ is a polynomial. Thus as in case 1 we assume that f is a polynomial and $\phi_f = dz^2 / (z^2 - 4)$. The equation $f^* \phi_f = C_f \phi_f$ is equivalent to

$$f'(z)^2(z^2 - 4) = C_f(f(z)^2 - 4).$$
(3.5)

We want to find all polynomial solutions to this equation. First note that f(2) = 2 or -2 if we set z = 2. Comparing the leading coefficients in both sides, we have $C_f = \deg(f)^2$. To solve the equation (3.5), we need a little trick. Let

$$z = w + \frac{1}{w}, \quad f(z) = \varphi(w) + \frac{1}{\varphi(w)},$$

where $\varphi(w)$ is required to be a holomorphic map, it need only to be defined in some open set U in \mathbb{C} . Indeed, since the map $w \mapsto w + w^{-1}$ is locally injective when |w| is large, we can always do this. Calculation shows

$$f'(z) = w \frac{\varphi'(w)}{\varphi(w)} \frac{\varphi(w) - \varphi(w)^{-1}}{w - w^{-1}}.$$

Then the equation (3.5) becomes to

$$w \frac{\varphi'(w)}{\varphi(w)} = \pm n, \quad w \in U,$$

where $n = \deg(f)$. This new equation has general solution $\varphi(w) = Cw^n$ or $\varphi(w) = (Cw^n)^{-1}$ for some undeterminate constant C, so we have

$$f\left(w+\frac{1}{w}\right) = Cw^n + \frac{1}{Cw^n}, \quad w \in U.$$

This relation in fact holds for all $w \in \overline{\mathbb{C}}$ by identity theorem of holomorphic maps. If f(2) = 2, then C = 1, in this case $f(2 \cos z) = 2 \cos(nz)$ if we write $w = e^{iz}$, so $f = T_n$. If f(2) = -2, then C = -1, in this case $f = -T_n$. Therefore in case 2, f is conjugate to a Chebyshef polynomial T_n or $-T_n$.

Up to now, the only remaining case is (P2). In this case, we can easily show that $C_f = \deg(f)^2$, $\phi_f = dz^2/((z-z_0)^2(z-\zeta)^2)$ and f is conjugate to the power map $z \mapsto z^d$, for $d \in \mathbb{Z}$ and $d \leq -2$. We omit the details here.

The proof is completed.

Remark 3.2.1. For convenience, we list all solutions to the indeterminate equation $f^*\phi_f = C_f\phi_f$ in the following table:

f is conjugate to	ϕ_f	C_f	\mathcal{O}_{f}
Integral Lattès map	$\frac{dz^2}{(z-p_1)(z-p_2)(z-p_3)(z-p_4)}$	$\deg(f)$	$(\overline{\mathbb{C}},(2,2,2,2))$
Power map	$\frac{dz^2}{(z-z_0)^2(z-z_1)^2}$	$\deg(f)^2$	$(\overline{\mathbb{C}},(\infty,\infty))$
$\pm Chebyshef polynomial$	$\frac{dz^2}{(z-z_0)^2(z-z_1)(z-z_2)}$	$\deg(f)^2$	$(\overline{\mathbb{C}},(2,2,\infty))$

We can see that for all cases \mathcal{O}_f is a parabolic orbifold, $\sqrt{|\phi_f|}$ is an orbifold metric on \mathcal{O}_f .
4.1 Introduction

It is well known that the famous Mandelbrot set of the quadratic polynomials $f_c(z) = z^2 + c$ is defined by

$$M = \{ c \in \mathbb{C}; f_c^n(0) \text{ remains bounded as } n \to \infty \}.$$

The Mandelbrot set is the connected locus for the quadratic family. It is a central object of study in complex dynamics since it exhibits a rich geometric and combinatorial structure, with many intriguing details and many remaining mysteries. One of most interesting results about M is that it is a connected set, which was obtained by Douady and Hubbard in 1982 by constructing a Riemann mapping from the exterior of M to $\mathbb{C} \setminus \mathbb{D}$, see [DH3]. This result leads to numerous further study of M, especially the study of its topological and combinatorial properties, using the method of 'parameter external rays' and 'puzzle' techniques.

In this chapter, we deal with a family of rational maps

$$T_{\lambda}(z) = \left(\frac{z^2 + \lambda - 1}{2z + \lambda - 2}\right)^2,$$

where λ is a complex parameter. This family is indeed the family of renormalization transformations of 2-dimensional diamond-like hierachical Potts models in statistical mechanics. In 1983, Derrida *et al* show that the Yang-Lee zeros of the λ -state Potts model on the diamond hierachical lattice are dense in the Julia set $J(T_{\lambda})$ of the map T_{λ} (See [DDI]). Since then, much interest has been devoted to this family since it exhibits a connection between statistical mechanics and complex dynamics (See [EL],[O],[QG],[QL]).

For this family, note that when $\lambda = 0$, the map T_{λ} degenerates to the quadratic polynomial $T_0(z) = (z+1)^2/4$; when $\lambda \in \mathbb{C}^* = \mathbb{C} \setminus \{0\}$, 1 and ∞ are two superattracting fixed points for the map T_{λ} while 0 is a critical value.

The non-escape locus \mathcal{M} (an analogue of Mandelbrot set) associated to this family is defined by:

$$\mathcal{M} = \{\lambda \in \mathbb{C}^*; T_{\lambda}^n(0) \not\rightarrow_{n \to \infty} 1 \text{ and } T_{\lambda}^n(0) \not\rightarrow_{n \to \infty} \infty\} \cup \{0\}$$



Figure 4.1: The parameter plane for T_{λ} ($\lambda \in \mathbb{C}$)

Figure 4.1 shows the picture of the non-escape locus \mathcal{M} in the parameter plane for this family. The non-escape locus \mathcal{M} can be identified as the complex plane minus infinitely many 'bubbles', which we will call 'capture domains' formally. An elementary property of \mathcal{M} is that it is compact and symmetric about the real axis. Moreover, the intersection $\mathcal{M} \cap \mathbb{R}$ is contained in the closed interval [0,3], with the boundary points 0 and 3 lying in \mathcal{M} (See [QL]). Many small copies of quadratic Mandelbrot set M are visible in the parameter plane. It is an amazing fact that the small copies of quadratic Mandelbrot set M are dense in \mathcal{M} . This is the a philosophy of 'universality of the Mandelbrot set', which was proved by McMullen, see [McM2].

For the non-escape locus \mathcal{M} , we have the following:

Theorem 4.1.1. The non-escape locus \mathcal{M} is connected.

For the Julia set, we have the following:

Theorem 4.1.2. If the Julia set $J(T_{\lambda})$ is a quasi-circle, then the Hausdorff dimension of $J(T_{\lambda})$ satisfies:

$$HD(J(T_{\lambda})) \le 1 + |\phi_{\lambda}(0)|^{2/3},$$

where ϕ_{λ} is the Böttcher map of T_{λ} defined near the superattracting fixed point 1.

This chapter is organized as follows: in Section 4.2, we discuss the location of the critical points and decompose the parameter plane into the non-escape locus \mathcal{M} plus infinitely many capture domains; in Section 4.3, we parameterize all the capture domains by constructing a Riemann mapping from every capture domain to the unit disk \mathbb{D} and prove Theorem 4.1.1; in Section 4.4, we prove Theorem 4.1.2 by using Smirnov's Theorem and Slodkowski's Theorem.

This chapter has been published as [WQYQG]. I felt depressed when I found that some of our results had already been included in Aspenberg and Yampolsky's paper [AY] after I finished writing this part two years ago. Anyway, there are some differences between our argument and Aspenberg-Yampolsky's, so I include it here as a part of the thesis.

This chapter in fact deals with a special quadratic family. For more discussions of the quadratic family, see also [T] and [Rees3].

4.2 Critical points and capture domains

For $\lambda \in \mathbb{C}^*$, let $C(T_{\lambda})$ be the set of all critical points for T_{λ} . Easy calculation shows

$$T'_{\lambda}(z) = \frac{4(z-1)(z+\lambda-1)(z^2+\lambda-1)}{(2z+\lambda-2)^3}.$$

Thus we have

$$C(T_{\lambda}) = \{1, \infty, 1 - \lambda, \pm \sqrt{1 - \lambda}, 1 - \lambda/2\}.$$

Moreover $T_{\lambda}^{-1}(\infty) = \{\infty, 1 - \lambda/2\}, T_{\lambda}^{-1}(0) = \{\pm \sqrt{1 - \lambda}\}$. Let $\mathcal{A}_{\lambda}(1)$ and $\mathcal{A}_{\lambda}(\infty)$ be the immediate basins of attraction for the superattracting fixed points 1 and ∞ respectively.

First, we introduce an interesting property of the map T_{λ} :

Splitting Principle For $\lambda \in \mathbb{C}^*$, we have $T_{\lambda} = t_{\lambda} \circ t_{\lambda}$, where

$$t_{\lambda}(z) = \left(\frac{z-1+\lambda}{z-1}\right)^2.$$

It's easy to see that t_{λ} permutes 1 and ∞ , $t_{\lambda}^{-1}(\mathcal{A}_{\lambda}(\infty)) = \mathcal{A}_{\lambda}(1)$. The orbits of $\pm \sqrt{1-\lambda}$ and $1-\lambda$ under iterations of T_{λ} in fact lie interlacedly in the same orbit under iterations of t_{λ} :

$$\pm \sqrt{1-\lambda} \mapsto 1-\lambda \mapsto 0 \mapsto (1-\lambda)^2 \mapsto \cdots$$

Lemma 4.2.1. The Julia set $J(T_{\lambda})$ is connected for all $\lambda \in \mathbb{C}$.

Proof. If $\lambda = 0$, $T_0(z) = (z+1)^2/4$ is conformally conjugate to $z \mapsto z^2 + 1/4$, whose Julia set is connected. If $\lambda \neq 0$, by Splitting Principle, $J(T_{\lambda}) = J(t_{\lambda})$. It is known from Yin (See [Yin]) that the Julia set for a quadratic rational map is either connected or a Cantor set, thus $J(t_{\lambda})$ is connected since t_{λ} has two superattracting periodic points 1 and ∞ . **Lemma 4.2.2.** For $\lambda \in \mathbb{C}^*$, the following conditions are equivalent.

1. $J(T_{\lambda})$ is a quasicircle. 2. $\sqrt{1-\lambda} \in \mathcal{A}_{\lambda}(1)$. 3. $-\sqrt{1-\lambda} \in \mathcal{A}_{\lambda}(1)$. 4. $0 \in \mathcal{A}_{\lambda}(1)$. 5. $1-\lambda \in \mathcal{A}_{\lambda}(\infty)$.

6. $1 - \lambda/2 \in \mathcal{A}_{\lambda}(\infty)$.

Proof. First we show $1 \Rightarrow 2+3+4+5+6$. Suppose $J(T_{\lambda})$ is a quasicircle, the Fatou set $F(T_{\lambda})$ decomposes into two completely invariant components $\mathcal{A}_{\lambda}(1)$ and $\mathcal{A}_{\lambda}(\infty)$. It is obvious that $1 - \lambda/2 \in \mathcal{A}_{\lambda}(\infty)$ and $\pm \sqrt{1-\lambda}$ lie in the same Fatou component. By Riemann-Hurwitz formula, $\{\pm \sqrt{1-\lambda}, 0\} \subset \mathcal{A}_{\lambda}(1)$ and $1 - \lambda \in \mathcal{A}_{\lambda}(\infty)$.

It is obvious that $2 \Rightarrow 4$, $3 \Rightarrow 4$ since $T_{\lambda}(\pm \sqrt{1-\lambda}) = \{0\}$ and T_{λ} fixes $\mathcal{A}_{\lambda}(1)$. $5 \Rightarrow 4$ follows from the fact that $t_{\lambda}(\mathcal{A}_{\lambda}(\infty)) = \mathcal{A}_{\lambda}(1)$ and $t_{\lambda}(1-\lambda) = 0$.

Now we show $4 \Rightarrow 1$. Suppose $0 \in \mathcal{A}_{\lambda}(1)$. Since $T_{\lambda}^{-1}(0) = \{\pm \sqrt{1-\lambda}\}$, the pair $\{\pm \sqrt{1-\lambda}\}$ has two possibilities of location: either both lie in $\mathcal{A}_{\lambda}(1)$ or only one lies in $\mathcal{A}_{\lambda}(1)$. By Lemma 4.2.1, every Fatou component is simply connected. We see that the latter is ruled out by Riemann-Hurwitz formula. Thus $\{\pm \sqrt{1-\lambda}\} \subset \mathcal{A}_{\lambda}(1)$. Moreover $C(T_{\lambda}) \cap \mathcal{A}_{\lambda}(1) = \{1, \pm \sqrt{1-\lambda}\}$ and $\mathcal{A}_{\lambda}(1)$ is completely invariant.

Note that $1 - \lambda \in T_{\lambda}^{-1}(\mathcal{A}_{\lambda}(\infty))$, since $T_{\lambda}(1-\lambda) = t_{\lambda}(0) \in \mathcal{A}_{\lambda}(\infty)$. For the critical points $1-\lambda$ and $1-\lambda/2$, either both lie in $\mathcal{A}_{\lambda}(\infty)$ or at most one lies in $\mathcal{A}_{\lambda}(\infty)$. Also by Riemann-Hurwitz formula, the former is the only choice. So we have $\{1 - \lambda, 1 - \lambda/2\} \subset \mathcal{A}_{\lambda}(\infty)$ and $\mathcal{A}_{\lambda}(\infty)$ is completely invariant. Thus $F(T_{\lambda}) = \mathcal{A}_{\lambda}(1) \cup \mathcal{A}_{\lambda}(\infty)$. Since T_{λ} is a hyperbolic map, $J(T_{\lambda})$ is a quasicircle. Thus $4 \Rightarrow 1$.

To conclude, we show $6 \Rightarrow 5$. Suppose $1 - \lambda/2 \in \mathcal{A}_{\lambda}(\infty)$, this means $\mathcal{A}_{\lambda}(\infty)$ is completely invariant since $T_{\lambda}^{-1}(\infty) = \{\infty, 1 - \lambda/2\}$. It turns out that $1 - \lambda \in \mathcal{A}_{\lambda}(\infty)$ by Riemann-Hurwitz formula.

Lemma 4.2.3. For any $\lambda \in \mathbb{C}^*$, we have $0 \notin \mathcal{A}_{\lambda}(\infty)$ and $1 - \lambda \notin \mathcal{A}_{\lambda}(1)$.

Proof. If $0 \in \mathcal{A}_{\lambda}(\infty)$, then any critical point in the set $\{\pm \sqrt{1-\lambda}, 1-\lambda/2\}$ will lie in some component of $T_{\lambda}^{-1}(\mathcal{A}_{\lambda}(\infty))$. Since every Fatou component of $F(T_{\lambda})$ is simply connected by Lemma 4.2.1, this is impossible by Riemann-Hurwitz formula.

Since $t_{\lambda}^{-1}(0) = \{1 - \lambda\}$ and $t_{\lambda}^{-1}(\mathcal{A}_{\lambda}(\infty)) = \mathcal{A}_{\lambda}(1), 0 \notin \mathcal{A}_{\lambda}(\infty)$ indicates $1 - \lambda \notin \mathcal{A}_{\lambda}(1).$

Since t_{λ} permutes 1 and ∞ , we can describe the non-escape locus in another way:

 $\mathcal{M} = \{\lambda \in \mathbb{C}^*; t_{\lambda}^{2n}(0) \not\rightarrow_{n \to \infty} 1 \text{ and } t_{\lambda}^{2n+1}(0) \not\rightarrow_{n \to \infty} 1\} \cup \{0\}$

Now we consider the parameters outside of the non-escape locus \mathcal{M} . Let λ be such a parameter, the critical value 0 is eventually mapped into $\mathcal{A}_{\lambda}(1) \cup \mathcal{A}_{\lambda}(\infty)$ under iterations of t_{λ} . This observation leads to the following definition:

Definition 4.2.1. Let

$$\mathcal{H}_0 = \{ \lambda \in \mathbb{C}^*; 0 \in \mathcal{A}_\lambda(1) \},\$$

$$\mathcal{H}_n = \{\lambda \in \mathbb{C}^*; t_{\lambda}^n(0) \in \mathcal{A}_{\lambda}(1), t_{\lambda}^{n-1}(0) \notin \mathcal{A}_{\lambda}(\infty)\}, n \ge 1,$$

a component of \mathcal{H}_n is called a capture domain of level n for $n \geq 0$.

Lemma 4.2.4. The parameter plane has the following decomposition:

$$\mathbb{C} = \mathcal{M} \sqcup (\sqcup_{n \ge 0} \mathcal{H}_n),$$

where \sqcup denotes the union of mutually disjoint sets.

Proof. First we show that the sets in $\{\mathcal{H}_n; n \geq 0\}$ are mutually disjoint. If not, suppose $\lambda \in \mathcal{H}_n \cap \mathcal{H}_m$ for $m > n \geq 0$. Then by definition

$$t_{\lambda}^{n}(0) \in \mathcal{A}_{\lambda}(1), \ t_{\lambda}^{m}(0) \in \mathcal{A}_{\lambda}(1), \ t_{\lambda}^{m-1}(0) \notin \mathcal{A}_{\lambda}(\infty).$$

We see that $t_{\lambda}^{m-n}(\mathcal{A}_{\lambda}(1)) = \mathcal{A}_{\lambda}(1)$, so m-n is even and $m-n \geq 2$. It follows that $t_{\lambda}^{m-1}(0) = t_{\lambda}^{m-n-1}(t_{\lambda}^{n}(0)) \in \mathcal{A}_{\lambda}(\infty)$. But this is a contradiction.

Now we prove that for any $\lambda \in \mathbb{C} \setminus \mathcal{M}$, λ must lie in \mathcal{H}_n for some $n \geq 0$. Indeed, by definition of \mathcal{M} , for any $\lambda \in \mathbb{C} \setminus \mathcal{M}$, either $t_{\lambda}^{2n}(0) \to 1$ or $t_{\lambda}^{2n+1}(0) \to 1$ as $n \to \infty$. So there is a minimal integer $m \geq 0$ such that $t_{\lambda}^m(0) \in \mathcal{A}_{\lambda}(1)$. If m = 0, then $\lambda \in \mathcal{H}_0$. If m = 1, Lemma 4.2.3 shows that $0 \notin \mathcal{A}_{\lambda}(\infty)$, so $\lambda \in \mathcal{H}_1$. If $m \geq 2$, we can conclude that $t_{\lambda}^{m-1}(0) \notin \mathcal{A}_{\lambda}(\infty)$, for else $t_{\lambda}^{-1}(\mathcal{A}_{\lambda}(\infty)) = \mathcal{A}_{\lambda}(1)$ indicates $t_{\lambda}^{m-2}(0) \in \mathcal{A}_{\lambda}(1)$, which will contradict the choice of m. Thus in this case we also have $\lambda \in \mathcal{H}_m$.

Remark 4.2.1. It is easy to verify that for $n \ge 1$,

$$\mathcal{H}_{2n} = \{ \lambda \in \mathbb{C}^*; T_{\lambda}^n(0) \in \mathcal{A}_{\lambda}(1), T_{\lambda}^{n-1}(0) \notin \mathcal{A}_{\lambda}(1) \}$$

$$= \{ \lambda \in \mathbb{C}^*; T_{\lambda}^{n+1}(1-\lambda) \in \mathcal{A}_{\lambda}(\infty), T_{\lambda}^n(1-\lambda) \notin \mathcal{A}_{\lambda}(\infty) \},$$

$$\mathcal{H}_{2n-1} = \{ \lambda \in \mathbb{C}^*; T_{\lambda}^n(0) \in \mathcal{A}_{\lambda}(\infty), T_{\lambda}^{n-1}(0) \notin \mathcal{A}_{\lambda}(\infty) \}$$

$$= \{ \lambda \in \mathbb{C}^*; T_{\lambda}^n(1-\lambda) \in \mathcal{A}_{\lambda}(1), T_{\lambda}^{n-1}(1-\lambda) \notin \mathcal{A}_{\lambda}(1) \}.$$

Lemma 4.2.5. For any $\lambda \in \mathbb{C}^* \setminus \mathcal{H}_0$, we have

$$\mathcal{A}_{\lambda}(1) \cap C(T_{\lambda}) = \{1\}, \ \mathcal{A}_{\lambda}(\infty) \cap C(T_{\lambda}) = \{\infty\}.$$

Proof. It is obvious that $1 - \lambda/2 \notin \mathcal{A}_{\lambda}(1)$. By Lemma 4.2.2, $\mathcal{A}_{\lambda}(1)$ does not contain $\sqrt{1-\lambda}$ or $-\sqrt{1-\lambda}$ while $\mathcal{A}_{\lambda}(\infty)$ does not contain $1 - \lambda$ or $1 - \lambda/2$. By Lemma 4.2.3, $\{\pm\sqrt{1-\lambda}\} \cap \mathcal{A}_{\lambda}(\infty) = \emptyset$ and $1 - \lambda \notin \mathcal{A}_{\lambda}(1)$. The conclusion follows.

4.3 Proof of Theorem 4.1.1

In this section, we parameterize all the capture domains in the parameter plane. We show that every capture domain is simply connected by constructing a Riemann mapping from the capture domain to the unit disk \mathbb{D} , and this will lead to the connectivity of the non-escape locus \mathcal{M} .

To prove every capture domain of level $n \ge 1$ is conformally equivalent to the unit disk \mathbb{D} , we use the method of quasiconformal surgery and holomorphic motion theorem (See Proposition 4.3.2). The method of quasiconformal surgery is classic, which was first used by Douady and Hubbard to parameterize the hyperbolic components of the quadratic Mandelbrot set M (See [DH2]). This method is developed by Roesch to study the parameter plane of cubic Newton maps and McMullen maps, see [Ro2],[Ro3].

However, this method cannot be applied to \mathcal{H}_0 , since the Julia set for the map in \mathcal{H}_0 is a quasicircle and there is no way to construct quasiconformal deformation. To deal with \mathcal{H}_0 , we divide the proof into several steps. First, for two maps $t_{\lambda_1}, t_{\lambda_2}$ in \mathcal{H}_0 satisfying an 'argument relation', we construct a quasiconformal conjugacy between the two maps elaborately using the philosophy of so called 'bootstrap argument' (See Lemma 4.3.2). This construction shows us an important relation between the Beltrami coefficient of the quasiconformal map and the Green functions of t_{λ_1} and t_{λ_2} . Furthermore, we will see later that this quasiconformal map has an extremal property: its Beltrami coefficient achieves the minimal norm (See the concluding remark of this section).

On the other hand, we show that the essential norm of the Beltrami coefficient can be bounded below by a constant depending on the Poincaré distance between λ_1 and λ_2 in \mathcal{H}_0 (See Lemma 4.3.3). In this way, the Beltrami coefficient acts like a bridge connecting the Green functions with the Poincaré distance. This yields an inequality between the two objects, which play a crucial role in the parameterization for \mathcal{H}_0 (See Proposition 4.3.1). It is an amusing fact that once we prove \mathcal{H}_0 is conformally equivalent to the punctured disk \mathbb{D}^* , we see that the inequality is actually an identity (See the concluding remark of this section). This is again a philosophy of 'bootstrap argument'.

As a consequence, we show the non-escape locus \mathcal{M} has capacity equal to 2.

In the following, we always use \mathbb{D} to denote the unit disk. Let $\mathbb{D}^* = \mathbb{D} \setminus \{0\}$ be the punctured disk and $\mathbb{D}_R = \{z \in \mathbb{C}; |z| < R\}$. For a hyperbolic Riemann surface S, let $d_S(z_1, z_2)$ be the hyperbolic distance for a pair $(z_1, z_2) \in S \times S$.

For $\lambda \in \mathbb{C}^*$, the map T_{λ} has two superattracting fixed points 1 and ∞ . The Green functions $G_{\lambda} : \mathcal{A}_{\lambda}(1) \to (0, \infty]$ and $G_{\lambda}^{\infty} : \mathcal{A}_{\lambda}(\infty) \to (0, \infty]$ are defined as follows:

$$G_{\lambda}(z) = -\lim_{k \to \infty} 2^{-k} \log |T_{\lambda}^{k}(z) - 1|, \ z \in \mathcal{A}_{\lambda}(1),$$

$$G_{\lambda}^{\infty}(z) = \lim_{k \to \infty} 2^{-k} \log |T_{\lambda}^{k}(z)|, \ z \in \mathcal{A}_{\lambda}(\infty).$$

It is known that G_{λ} and G_{λ}^{∞} are continuous and satisfy:

$$G_{\lambda} \circ T_{\lambda}(z) = 2G_{\lambda}(z), \ z \in \mathcal{A}_{\lambda}(1),$$

$$G_{\lambda}^{\infty} \circ T_{\lambda}(z) = 2G_{\lambda}^{\infty}(z), \ z \in \mathcal{A}_{\lambda}(\infty).$$

Lemma 4.3.1. For $\lambda \in \mathbb{C}^*$, the Green functions satisfy:

$$G_{\lambda} \circ t_{\lambda}(z) = G_{\lambda}^{\infty}(z), \ z \in \mathcal{A}_{\lambda}(\infty),$$

$$G_{\lambda}^{\infty} \circ t_{\lambda}(z) = 2G_{\lambda}(z), \ z \in \mathcal{A}_{\lambda}(1).$$

Proof. First suppose $\lambda \in \mathbb{C}^* \setminus \mathcal{H}_0$. By Lemma 4.2.5, the only critical point of T_{λ} that lies in $\mathcal{A}_{\lambda}(1)$ is 1 itself, thus the Böttcher map $\phi_{\lambda} : \mathcal{A}_{\lambda}(1) \to \mathbb{D}$ defined by $\phi_{\lambda}(z) = \lim_{k \to \infty} (T_{\lambda}^k(z) - 1)^{2^{-k}}$ is a conformal isomorphism. Similarly, the only critical point of T_{λ} that lies in $\mathcal{A}_{\lambda}(\infty)$ is ∞ itself, the Böttcher map $\phi_{\lambda}^{\infty} : \mathcal{A}_{\lambda}(\infty) \to \overline{\mathbb{C}} \setminus \overline{\mathbb{D}}$ defined by $\phi_{\lambda}^{\infty}(z) = \lim_{k \to \infty} (T_{\lambda}^k(z))^{2^{-k}}$ is also a conformal isomorphism. Since $t_{\lambda} : \mathcal{A}_{\lambda}(\infty) \to \mathcal{A}_{\lambda}(1)$ is a proper map of degree one, the Böttcher maps satisfy:

$$\phi_{\lambda} \circ t_{\lambda}(z) = (\phi_{\lambda}^{\infty}(z))^{-1}, \ z \in \mathcal{A}_{\lambda}(\infty),$$

$$\phi_{\lambda}^{\infty} \circ t_{\lambda}(z) = (\phi_{\lambda}(z))^{-2}, \ z \in \mathcal{A}_{\lambda}(1).$$

Thus the Green functions satisfy:

$$\begin{aligned} G_{\lambda} \circ t_{\lambda}(z) &= -\log |\phi_{\lambda} \circ t_{\lambda}(z)| = \log |\phi_{\lambda}^{\infty}(z)| = G_{\lambda}^{\infty}(z), \ z \in \mathcal{A}_{\lambda}(\infty), \\ G_{\lambda}^{\infty} \circ t_{\lambda}(z) &= \log |\phi_{\lambda}^{\infty} \circ t_{\lambda}(z)| = -2\log |\phi_{\lambda}(z)| = 2G_{\lambda}(z), \ z \in \mathcal{A}_{\lambda}(1). \end{aligned}$$

Now suppose $\lambda \in \mathcal{H}_0$, the Böttcher maps ϕ_{λ} and ϕ_{λ}^{∞} can be defined in neighborhoods of 1 and ∞ respectively, say $U_{\lambda}(1)$ and $U_{\lambda}(\infty)$. We may assume that $U_{\lambda}(1)$ and $U_{\lambda}(\infty)$ are small enough such that $t_{\lambda} : U_{\lambda}(\infty) \to U_{\lambda}(1)$ is a conformal isomorphism. Thus we have

$$\phi_{\lambda} \circ t_{\lambda}(z) = (\phi_{\lambda}^{\infty}(z))^{-1}, \ G_{\lambda} \circ t_{\lambda}(z) = G_{\lambda}^{\infty}(z), \ z \in U_{\lambda}(\infty)$$

Let $GO_{\lambda}(1) = \bigcup_{n\geq 0} T_{\lambda}^{-n}(1)$ and $GO_{\lambda}(\infty) = \bigcup_{n\geq 0} T_{\lambda}^{-n}(\infty)$ be the grand orbits of 1 and ∞ respectively. It is easy to check that $t_{\lambda}^{-1}(GO_{\lambda}(1)) =$ $GO_{\lambda}(\infty), t_{\lambda}^{-1}(GO_{\lambda}(\infty)) = GO_{\lambda}(1)$. Note that G_{λ} is harmonic in $\mathcal{A}_{\lambda}(1) \setminus GO_{\lambda}(1)$ and G_{λ}^{∞} is harmonic in $\mathcal{A}_{\lambda}(\infty) \setminus GO_{\lambda}(\infty)$. The function $G_{\lambda} \circ t_{\lambda}$ is harmonic in $t_{\lambda}^{-1}(\mathcal{A}_{\lambda}(1)\backslash GO_{\lambda}(1)) = \mathcal{A}_{\lambda}(\infty)\backslash GO_{\lambda}(\infty)$. Comparing $G_{\lambda} \circ t_{\lambda}$ and G_{λ}^{∞} , we see that both $G_{\lambda} \circ t_{\lambda}$ and G_{λ}^{∞} are equal to ∞ in $GO_{\lambda}(\infty)$, harmonic in $\mathcal{A}_{\lambda}(\infty)\backslash GO_{\lambda}(\infty)$. Since the two coincide in $U_{\lambda}(\infty)$, by identity theorem of harmonic functions, we have $G_{\lambda} \circ t_{\lambda}(z) = G_{\lambda}^{\infty}(z), \ z \in \mathcal{A}_{\lambda}(\infty)$. It follows that

$$G_{\lambda}^{\infty} \circ t_{\lambda}(z) = G_{\lambda} \circ T_{\lambda}(z) = 2G_{\lambda}(z), \ z \in \mathcal{A}_{\lambda}(1).$$

Lemma 4.3.2. Let $\lambda_1, \lambda_2 \in \mathcal{H}_0$ with $\arg \phi_{\lambda_1}(0) = \arg \phi_{\lambda_2}(0)$, then there is a quasiconformal map h with $0, 1, \infty$ fixed such that $h \circ t_{\lambda_1} = t_{\lambda_2} \circ h$. Moreover, the Beltrami coefficient μ of h satisfies

$$\|\mu\|_{\infty} = \left| \frac{G_{\lambda_1}(0) - G_{\lambda_2}(0)}{G_{\lambda_1}(0) + G_{\lambda_2}(0)} \right|.$$

Proof. For $\lambda \in \mathcal{H}_0$ and $n \in \mathbb{Z}$, let $E_{\lambda}(n)$ be the component of $\{z \in \mathcal{A}_{\lambda}(1); G_{\lambda}(z) > 2^{-n}G_{\lambda}(0)\}$ that contains 1 and $E_{\lambda}^{\infty}(n)$ be the component of $\{z \in \mathcal{A}_{\lambda}(\infty); G_{\lambda}^{\infty}(z) > 2^{-n}G_{\lambda}^{\infty}(1-\lambda)\}$ that contains ∞ . By Lemma 4.3.1, we can verify by induction that:

- For $n \in \mathbb{Z}$, $E_{\lambda}(n) \subset E_{\lambda}(n+1)$, $E_{\lambda}^{\infty}(n) \subset E_{\lambda}^{\infty}(n+1)$.
- $\bigcup_{n>0} E_{\lambda}(n) = \mathcal{A}_{\lambda}(1), \ \bigcup_{n>0} E_{\lambda}^{\infty}(n) = \mathcal{A}_{\lambda}(\infty).$
- $E_{\lambda}(n)$ and $E_{\lambda}^{\infty}(n)$ are simply connected for all $n \in \mathbb{Z}$.

• $t_{\lambda}: E_{\lambda}^{\infty}(n) \to E_{\lambda}(n)$ is a proper map of degree one (if $n \leq 0$) or two (if $n \geq 1$) while $t_{\lambda}: E_{\lambda}(n+1) \to E_{\lambda}^{\infty}(n)$ is a proper map of degree two for all $n \in \mathbb{Z}$.

Given $\lambda_1, \lambda_2 \in \mathcal{H}_0$ with $\arg \phi_{\lambda_1}(0) = \arg \phi_{\lambda_2}(0)$, we define a quasiconformal map $\delta : \mathbb{D}_{|\phi_{\lambda_1}(0)|} \to \mathbb{D}_{|\phi_{\lambda_2}(0)|}$ by $\delta(re^{i\theta}) = r^{\alpha}e^{i\theta}$, where α satisfies $|\phi_{\lambda_1}(0)|^{\alpha} = |\phi_{\lambda_2}(0)|$. In complex coordinate, $\delta(z) = z^{(\alpha+1)/2} \bar{z}^{(\alpha-1)/2}$. The Beltrami coefficient μ_{δ} of δ satisfies

$$\mu_{\delta}(z) = \frac{\alpha - 1}{\alpha + 1} \frac{z}{\bar{z}}, \ \|\mu_{\delta}\|_{\infty} = \left|\frac{\alpha - 1}{\alpha + 1}\right| = \left|\frac{G_{\lambda_1}(0) - G_{\lambda_2}(0)}{G_{\lambda_1}(0) + G_{\lambda_2}(0)}\right|$$

It is easy to check that $\delta(z^2) = \delta(z)^2$ for $z \in \mathbb{D}_{|\phi_{\lambda_1}(0)|}$.

We first construct three quasiconformal maps $h_0: E_{\lambda_1}(0) \to E_{\lambda_2}(0), h_0^{\infty}: E_{\lambda_1}^{\infty}(0) \to E_{\lambda_2}^{\infty}(0)$ and $h_1: E_{\lambda_1}(1) \to E_{\lambda_2}(1)$ such that $h_1|_{E_{\lambda_1}(0)} = h_0$ and the following diagram is commutative.

$$\begin{array}{c|c} E_{\lambda_1}(1) \xrightarrow{t_{\lambda_1}} E_{\lambda_1}^{\infty}(0) \xrightarrow{t_{\lambda_1}} E_{\lambda_1}(0) \xrightarrow{\phi_{\lambda_1}} \mathbb{D}_{|\phi_{\lambda_1}(0)|} \\ h_1 & & h_0^{\infty} \\ h_1 & & h_0^{\infty} \\ E_{\lambda_2}(1) \xrightarrow{t_{\lambda_2}} E_{\lambda_2}^{\infty}(0) \xrightarrow{t_{\lambda_2}} E_{\lambda_2}(0) \xrightarrow{\phi_{\lambda_2}} \mathbb{D}_{|\phi_{\lambda_2}(0)|} \end{array}$$

Note that $\phi_{\lambda_i} : E_{\lambda_i}(0) \to \mathbb{D}_{|\phi_{\lambda_i}(0)|}$ is a conformal isomorphism for $i \in \{1, 2\}$, we can define a quasiconformal map $h_0 : E_{\lambda_1}(0) \to E_{\lambda_2}(0)$ as follows:

$$h_0(z) = \phi_{\lambda_2}^{-1} \circ \delta \circ \phi_{\lambda_1}(z), \ z \in E_{\lambda_1}(0).$$

By construction, $h_0(1) = 1$ and $h_0 \circ T_{\lambda_1}(z) = T_{\lambda_2} \circ h_0(z)$ for $z \in E_{\lambda_1}(0)$. Moreover, the map h_0 can be extended to a homeomorphism from $\overline{E_{\lambda_1}(0)}$ to $\overline{E_{\lambda_2}(0)}$ with the boundary point 0 fixed.

Since $t_{\lambda_i} : E_{\lambda_i}^{\infty}(0) \to E_{\lambda_i}(0)$ is a proper map of degree one for $i \in \{1, 2\}$, the map $h_0^{\infty} : E_{\lambda_1}^{\infty}(0) \to E_{\lambda_2}^{\infty}(0)$ with ∞ fixed can be defined by

$$h_0^{\infty}(z) = t_{\lambda_2}^{-1} \circ h_0 \circ t_{\lambda_1}(z), \ z \in E_{\lambda_1}^{\infty}(0)$$

Now we define h_1 . Since $t_{\lambda_i} : E_{\lambda_i}(1) \setminus \{1\} \to E_{\lambda_i}^{\infty}(0) \setminus \{\infty\}$ is a covering map of degree two, we can get a lifting of h_0^{∞} , say $h_1 : E_{\lambda_1}(1) \setminus \{1\} \to E_{\lambda_2}(1) \setminus \{1\}$, such that $h_1(0) = 0$ and $h_0^{\infty} \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ h_1(z), z \in E_{\lambda_1}(1) \setminus \{1\}$. By continuity, we can define $h_1(1) = 1$. Now we show $h_1|_{E_{\lambda_1}(0)} = h_0$. First note that both $h_1|_{E_{\lambda_1}(0)}$ and h_0 are liftings of h_0 via branch covering maps T_{λ_1} and T_{λ_2} . That is, the following diagram is commutative.

$$\begin{array}{c|c} E_{\lambda_1}(0) & \xrightarrow{F} & E_{\lambda_2}(0) \\ T_{\lambda_1} & & & \downarrow^{T_{\lambda_2}} \\ E_{\lambda_1}(-1) & \xrightarrow{h_0} & E_{\lambda_2}(-1) \end{array}$$

where $F \in \{h_1|_{E_{\lambda_1}(0)}, h_0\}$. Since $T_{\lambda_i} : E_{\lambda_i}(0) \setminus \{1\} \to E_{\lambda_i}(-1) \setminus \{1\}$ is a covering map of degree two and both h_1 and h_0 fix the boundary point 0 on $\partial E_{\lambda_1}(0)$, we conclude $h_1|_{E_{\lambda_1}(0)} = h_0$ by uniqueness of lifting and continuity.

Suppose for some $n \ge 0$, we already get quasiconformal maps $h_n : E_{\lambda_1}(n) \to E_{\lambda_2}(n), h_n^{\infty} : E_{\lambda_1}^{\infty}(n) \to E_{\lambda_2}^{\infty}(n)$ and $h_{n+1} : E_{\lambda_1}(n+1) \to E_{\lambda_2}(n+1)$ such that $h_{n+1}|_{E_{\lambda_1}(n)} = h_n$ and the right part of the following diagram is commutative.

$$\begin{array}{c|c} E_{\lambda_1}(n+2) \xrightarrow{t_{\lambda_1}} E_{\lambda_1}^{\infty}(n+1) \xrightarrow{t_{\lambda_1}} E_{\lambda_1}(n+1) \xrightarrow{t_{\lambda_1}} E_{\lambda_1}^{\infty}(n) \xrightarrow{t_{\lambda_1}} E_{\lambda_1}(n) \\ \hline \\ h_{n+2} & & & \\ h_{n+2} & & & \\ h_{n+1} & & & \\ h_{n} & & & \\ E_{\lambda_2}(n+2) \xrightarrow{t_{\lambda_2}} E_{\lambda_2}^{\infty}(n+1) \xrightarrow{t_{\lambda_2}} E_{\lambda_2}(n+1) \xrightarrow{t_{\lambda_2}} E_{\lambda_2}(n) \xrightarrow{t_{\lambda_2}} E_{\lambda_2}(n) \end{array}$$

We want to get extensions of h_n^{∞} and h_{n+1} , denoted by h_{n+1}^{∞} and h_{n+2} respectively, such that the left part of the diagram is commutative.

We first construct h_{n+1}^{∞} . Since $t_{\lambda_i} : E_{\lambda_i}^{\infty}(n+1) \setminus \{1-\lambda_i\} \to E_{\lambda_i}(n+1) \setminus \{0\}$ is a covering map of degree two for $i \in \{1, 2\}$, there is a lifting of h_{n+1} , say
$$\begin{split} h_{n+1}^{\infty} &: E_{\lambda_1}^{\infty}(n+1) \setminus \{1-\lambda_1\} \to E_{\lambda_2}^{\infty}(n+1) \setminus \{1-\lambda_2\}, \text{ such that } h_{n+1}^{\infty}(\infty) = \infty \\ \text{and } h_{n+1} \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ h_{n+1}^{\infty}(z) \text{ for } z \in E_{\lambda_1}^{\infty}(n+1) \setminus \{1-\lambda_1\}. \text{ By continuity,} \\ \text{we can define } h_{n+1}^{\infty}(1-\lambda_1) = 1-\lambda_2. \text{ Now we show } h_{n+1}^{\infty}|_{E_{\lambda_1}^{\infty}(n)} = h_n^{\infty}. \text{ By assumption } h_{n+1}|_{E_{\lambda_1}(n)} = h_n, \text{ we have} \end{split}$$

$$h_n \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ F(z), \ z \in E^{\infty}_{\lambda_1}(n), \ F \in \{h^{\infty}_{n+1}|_{E^{\infty}_{\lambda_1}(n)}, h^{\infty}_n\}.$$

Since $t_{\lambda_i}: E_{\lambda_i}^{\infty}(n) \setminus \{1 - \lambda_i\} \to E_{\lambda_i}(n) \setminus \{0\}$ is a covering map of degree one (if n = 0) or two (if $n \ge 1$) for $i \in \{1, 2\}$ and $h_{n+1}^{\infty}(\infty) = h_n^{\infty}(\infty) = \infty$, we conclude $h_{n+1}^{\infty}|_{E_{\lambda_i}^{\infty}(n)} = h_n^{\infty}$ by uniqueness of lifting and continuity.

We then construct h_{n+2} . Since $t_{\lambda_i} : E_{\lambda_i}(n+2) \setminus \{1\} \to E_{\lambda_i}^{\infty}(n+1) \setminus \{\infty\}$ is a covering map of degree two for $i \in \{1, 2\}$, h_{n+1}^{∞} can be lifted to h_{n+2} such that $h_{n+2}(0) = 0$. We have just proved that $h_{n+1}^{\infty}|_{E_{\lambda_1}(n)} = h_n^{\infty}$, thus both $h_{n+2}|_{E_{\lambda_1}(n+1)}$ and h_{n+1} satisfy

$$h_n^{\infty} \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ F(z), \ z \in E_{\lambda_1}(n+1), \ F \in \{h_{n+2}|_{E_{\lambda_1}(n+1)}, h_{n+1}\}.$$

Again by uniqueness of lifting and continuity, we have $h_{n+2}|_{E_{\lambda_1}(n+1)} = h_{n+1}$.

By induction, we finally get two sequences of quasiconformal maps $\{h_n : E_{\lambda_1}(n) \to E_{\lambda_2}(n); n \ge 0\}$ and $\{h_n^{\infty} : E_{\lambda_1}^{\infty}(n) \to E_{\lambda_2}^{\infty}(n); n \ge 0\}$ such that h_{n+1} and h_{n+1}^{∞} are extensions of h_n and h_n^{∞} respectively for $n \ge 0$.

Up to now, we can define two quasiconformal maps $\beta : \mathcal{A}_{\lambda_1}(1) \to \mathcal{A}_{\lambda_2}(1)$ and $\gamma : \mathcal{A}_{\lambda_1}(\infty) \to \mathcal{A}_{\lambda_2}(\infty)$ such that $\beta|_{E_{\lambda_1}(n)} = h_n, \gamma|_{E_{\lambda_1}^{\infty}(n)} = h_n^{\infty}$ for all $n \geq 0$. It's easy to check

- $\gamma \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ \beta(z), \ z \in \mathcal{A}_{\lambda_1}(1),$
- $\beta \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ \gamma(z), \ z \in \mathcal{A}_{\lambda_1}(\infty),$

• The Beltrami coefficients of β and γ , say μ_{β} and μ_{γ} , satisfy $\|\mu_{\beta}\|_{\infty} = \|\mu_{\gamma}\|_{\infty} = \|\mu_{\delta}\|_{\infty}$.

By Lemma 4.2.2, the Julia sets $J(t_{\lambda_1})$ and $J(t_{\lambda_2})$ are quasicircles, thus we can get extensions of β and γ , denoted by $\bar{\beta}$ and $\bar{\gamma}$ respectively, such that $\bar{\beta}$: $\mathcal{A}_{\lambda_1}(1) \cup J(t_{\lambda_1}) \to \mathcal{A}_{\lambda_2}(1) \cup J(t_{\lambda_2})$ and $\bar{\gamma} : \mathcal{A}_{\lambda_1}(\infty) \cup J(t_{\lambda_1}) \to \mathcal{A}_{\lambda_2}(\infty) \cup J(t_{\lambda_2})$ are homeomorphisms (See [Ahl]). By continuity, we have

$$\bar{\gamma} \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ \bar{\beta}(z), \ \bar{\beta} \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ \bar{\gamma}(z), \ z \in J(t_{\lambda_1}).$$
(4.1)

In the following, we show $\bar{\beta}|_{J(t_{\lambda_1})} = \bar{\gamma}|_{J(t_{\lambda_1})}$. Let $\operatorname{Fix}(R)$ be the set of all fixed points for a rational map R. First note that for $\lambda \in \mathcal{H}_0$, the set $\operatorname{Fix}(t_{\lambda})$ consists of three repelling fixed points and $\operatorname{Fix}(T_{\lambda}) = \operatorname{Fix}(t_{\lambda}) \cup \{1, \infty\}$. The maps $\bar{\beta}$ and $\bar{\gamma}$ satisfy

$$\tau \circ T_{\lambda_1}(z) = T_{\lambda_2} \circ \tau(z), \ z \in J(t_{\lambda_1}), \ \tau \in \{\bar{\beta}, \bar{\gamma}\}.$$

Thus both $\bar{\beta}$ and $\bar{\gamma}$ map $\operatorname{Fix}(T_{\lambda_1}) \cap J(t_{\lambda_1}) = \operatorname{Fix}(t_{\lambda_1})$ onto $\operatorname{Fix}(T_{\lambda_2}) \cap J(t_{\lambda_2}) = \operatorname{Fix}(t_{\lambda_2})$. By (4.1), we have $\bar{\beta}|_{\operatorname{Fix}(t_{\lambda_1})} = \bar{\gamma}|_{\operatorname{Fix}(t_{\lambda_1})}$. Let $e(z) = \bar{\beta}^{-1} \circ \bar{\gamma}(z)$

for $z \in J(t_{\lambda_1})$. The map $e : J(t_{\lambda_1}) \to J(t_{\lambda_1})$ is an orientation preserving homeomorphism with three fixed points and satisfies

$$e \circ t_{\lambda_1} \circ e(z) = t_{\lambda_1}(z), \ z \in J(t_{\lambda_1}).$$

$$(4.2)$$

We show e is in fact the identity map by induction. For $p \in \text{Fix}(t_{\lambda_1})$, e(p) = p. Suppose $t_{\lambda_1}^{-1}(p) = \{p,q\}$, by (4.2), we have $t_{\lambda_1}^{-1}(p) = \{p,q\} = \{e(p), e(q)\}$, thus e(q) = q. Assume that for some $n \geq 1$, $e|_{t_{\lambda_1}^{-n}(p)} = \text{id}$. For any $q \in t_{\lambda_1}^{-(n+1)}(p) \setminus t_{\lambda_1}^{-n}(p)$, there is $p_n \in t_{\lambda_1}^{-n}(p)$, such that $q \in t_{\lambda_1}^{-1}(p_n) = \{q,q'\}$. By (4.2), we have $\{q,q'\} = \{e(q), e(q')\}$. Since e is orientation preserving, the triples $\{q,q',p_n\}$ and $\{e(q), e(q'), e(p_n) = p_n\}$ have the same cyclic order, thus e(q) = q, e(q') = q'. In this way, we have $e|_{t_{\lambda_1}^{-(n+1)}(p)} = \text{id}$. By induction, $e|_{\bigcup_{n\geq 0} t_{\lambda_1}^{-n}(p)} = \text{id}$. Since $J(t_{\lambda_1}) = \overline{\bigcup_{n\geq 0} t_{\lambda_1}^{-n}(p)}$, we conclude that e is the identity map by continuity. This means $\bar{\beta}$ and $\bar{\gamma}$ coincide on the Julia set $J(t_{\lambda_1})$.

Now we define

$$h(z) = \begin{cases} \bar{\beta}(z), & z \in \mathcal{A}_{\lambda_1}(1) \bigcup J(t_{\lambda_1}), \\ \bar{\gamma}(z), & z \in \mathcal{A}_{\lambda_1}(\infty). \end{cases}$$

The map $h: \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a homeomorphism with 0, 1 and ∞ fixed such that

$$h \circ t_{\lambda_1}(z) = t_{\lambda_2} \circ h(z), \ z \in \mathbb{C}.$$

By construction, $h|_{F(t_{\lambda_1})} : F(t_{\lambda_1}) \to F(t_{\lambda_2})$ is a quasiconformal map. Since the Julia set $J(t_{\lambda_1})$ is a quasicircle which is quasiconformally removable, the map $h : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is actually a quasiconformal map. Moreover, the Beltrami coefficient μ of h satisfies

$$|\mu(z)| = \|\mu\|_{\infty} = \left|\frac{G_{\lambda_1}(0) - G_{\lambda_2}(0)}{G_{\lambda_1}(0) + G_{\lambda_2}(0)}\right|, \text{ a.e. } z \in \mathbb{C}.$$

Lemma 4.3.3. For $\lambda_1, \lambda_2 \in \mathbb{C}^*$, suppose there is a quasiconformal map h such that $h \circ t_{\lambda_1} = t_{\lambda_2} \circ h$, then either $\lambda_1 = \lambda_2 \in \partial \mathcal{M}$ or λ_1 and λ_2 lie in the same component of $\mathbb{C} \setminus \partial \mathcal{M}$. In the latter case, suppose λ_1 and λ_2 lie in the component \mathcal{U} of $\mathbb{C} \setminus \partial \mathcal{M}$, then

$$\|\mu\|_{\infty} \ge \frac{\exp(d_{\mathcal{U}}(\lambda_1, \lambda_2)) - 1}{\exp(d_{\mathcal{U}}(\lambda_1, \lambda_2)) + 1},$$

where μ is the Beltrami coefficient of h.

Proof. First note that h fixes 0, 1 and ∞ . Since t_{λ_1} and t_{λ_2} are quasiconformally conjugate via h, the Beltrami coefficient μ of h satisfies $t^*_{\lambda_1}(\mu) = \mu$.

If $\|\mu\|_{\infty} = 0$, then h is the identity map and $\lambda_1 = \lambda_2$. The conclusion follows.

In the following, we assume $\|\mu\|_{\infty} > 0$. For $c \in \mathbb{D}$, let f_c solve the Beltrami equation

$$\frac{\partial f_c}{\partial \bar{z}} / \frac{\partial f_c}{\partial z} = c\mu / \|\mu\|_{\infty}$$

with $0, 1, \infty$ fixed. f_c is holomorphic with respect to $c \in \mathbb{D}$ and $f_0 = \mathrm{id}$.

The map $R_c = f_c \circ t_{\lambda_1} \circ f_c^{-1}$ is a rational map since it preserves the standard complex structure. It is of degree two and satisfies the following properties:

- R_c is holomorphic with respect to $c \in \mathbb{D}$ and $R_0 = t_{\lambda_1}$;
- R_c permutes 1 and ∞ . Moreover, 1 is a pole of R_c of order two;
- R_c has a zero of order two.

It is easy to check that

$$R_{c}(z) = f_{c} \circ t_{\lambda_{1}} \circ f_{c}^{-1}(z) = \left(\frac{z - 1 + \lambda(c)}{z - 1}\right)^{2} = t_{\lambda(c)}(z),$$

where $c \mapsto \lambda(c)$ is a holomorphic map from \mathbb{D} to \mathbb{C} and $\lambda(0) = \lambda_1$. Since

$$t_{\lambda(c)} \circ f_c(1-\lambda_1) = f_c \circ t_{\lambda_1}(1-\lambda_1) = f_c(0) = 0,$$

we have $1 - \lambda(c) = f_c(1 - \lambda_1)$. Thus λ has the expression $\lambda(c) = 1 - f_c(1 - \lambda_1)$. It is obvious that $\lambda(\|\mu\|_{\infty}) = \lambda_2$ since $f_{\|\mu\|_{\infty}} = h$.

First suppose $\lambda_1 \in \partial \mathcal{M}$. Note that $t_{\lambda(c)}$ has the same dynamical property as t_{λ_1} for all $c \in \mathbb{D}$, the map $\lambda : \mathbb{D} \to \mathbb{C}$ can not take values outside of the nonescape locus \mathcal{M} . By open map theorem, λ is a constant map. In particular, $\lambda_1 = \lambda_2$.

Now suppose $\lambda_1 \in \mathbb{C} \setminus \partial \mathcal{M}$. In this case, the image $\lambda(\mathbb{D})$ under the map $\lambda : \mathbb{D} \to \mathbb{C}$ has no intersection with $\partial \mathcal{M}$ by the previous argument, thus $\lambda(\mathbb{D})$ must be contained in some component \mathcal{U} of $\mathbb{C} \setminus \partial \mathcal{M}$ (\mathcal{U} is a hyperbolic Riemann surface since $\partial \mathcal{M} \cap \mathbb{R}$ contains at least three points: 0, 3 and 32/27. See [QL]). Since the map $\lambda : \mathbb{D} \to \mathcal{U}$ is holomorphic, by Schwarz lemma,

$$d_{\mathcal{U}}(\lambda_1, \lambda_2) = d_{\mathcal{U}}(\lambda(0), \lambda(\|\mu\|_{\infty})) \le d_{\mathbb{D}}(0, \|\mu\|_{\infty}) = \log \frac{1 + \|\mu\|_{\infty}}{1 - \|\mu\|_{\infty}}.$$

Thus we have

$$\|\mu\|_{\infty} \ge \frac{\exp(d_{\mathcal{U}}(\lambda_1, \lambda_2)) - 1}{\exp(d_{\mathcal{U}}(\lambda_1, \lambda_2)) + 1}.$$

Lemma 4.3.4. \mathcal{H}_0 is connected and $\mathcal{H}_0 \cup \{\infty\}$ contains a neighborhood of ∞ .

Proof. First we make a coordinate change for the family $\{T_{\lambda}; \lambda \in \mathcal{H}_0\}$. Let $\lambda = \nu^{-3}$ and $\varphi_{\nu}(z) = \nu^2(z-1)$. We can conjugate T_{λ} to a new map $S_{\nu} = \varphi_{\nu} \circ T_{\lambda} \circ \varphi_{\nu}^{-1}$. Calculation shows

$$S_{\nu}(\zeta) = \frac{\zeta^2 (2\nu + 4\nu^2 \zeta + \zeta^2)}{(1 + 2\nu\zeta)^2}.$$

To prove $\mathcal{H}_0 \cup \{\infty\}$ contains a neighborhood of ∞ is equivalent to prove $0 \in \mathcal{A}_{\lambda}(1)$ when $|\lambda|$ is large. It's also equivalent to prove

$$-\nu^2 = \varphi_{\nu}(0) \in \varphi_{\nu}(\mathcal{A}_{\lambda}(1)) = \mathcal{A}_{\nu}(0)$$

when $|\nu|$ is small, where $\mathcal{A}_{\nu}(0)$ is the immediate basin of attraction for the superattracting fixed point $\zeta = 0$ of the map S_{ν} . By continuity, there is a small positive number δ such that when $|\nu| < \delta$, we have $S_{\nu}(\mathbb{D}_{1/2}) \subset \mathbb{D}_{1/2}$. This means $\mathbb{D}_{1/2}$ lies in the Fatou set $F(S_{\nu})$. Thus $-\nu^2 \in \mathbb{D}_{1/2} \subset \mathcal{A}_{\nu}(0)$ when $|\nu|$ is small.

To conclude, we show \mathcal{H}_0 is connected. The map $\Phi : \mathcal{H}_0 \cup \{\infty\} \to \mathbb{D}$ defined by $\Phi(\lambda) = \phi_{\lambda}(0)$ for $\lambda \in \mathcal{H}_0$ and $\Phi(\infty) = 0$ is holomorphic and locally injective in a neighborhood \mathcal{V}_{∞} of ∞ (See the proof of Proposition 4.3.1). Thus the image $\Phi(\mathcal{H}_0 \cup \{\infty\})$ contains a neighborhood of 0. For any $\lambda \in \mathcal{H}_0$, there is $\lambda_0 \in \mathcal{V}_{\infty} \setminus \{\infty\}$ such that $\arg \phi_{\lambda_0}(0) = \arg \phi_{\lambda}(0)$. By Lemma 4.3.2, t_{λ} and t_{λ_0} are quasiconformally conjugate. By Lemma 4.3.3, λ and λ_0 lie in the same component of \mathcal{H}_0 . Thus \mathcal{H}_0 is connected and it is the unbounded component of $\mathbb{C} \setminus \partial \mathcal{M}$.

Proposition 4.3.1. The map $\Phi : \mathcal{H}_0 \to \mathbb{D}^*$ defined by $\Phi(\lambda) = \phi_{\lambda}(0)$ is a conformal isomorphism, where ϕ_{λ} is the Böttcher map for T_{λ} defined near the fixed point 1.

Proof. We develop two methods to prove the proposition. The first is to prove that the map Φ is a proper map of degree one while the second is to prove the map Φ preserves the Poincaré metrics. Lemma 4.3.2 and Lemma 4.3.3 play a crucial role in both proofs.

First note that the map $\Phi: \mathcal{H}_0 \to \mathbb{D}^*$ has Laurent expansion

$$\Phi(\lambda) = -2\lambda^{-1} + O(\lambda^{-2})$$

when $|\lambda|$ is large and ∞ is a removable singularity for Φ . Thus we can define $\Phi(\infty) = 0$ such that $\Phi : \mathcal{H}_0 \cup \{\infty\} \to \mathbb{D}$ is holomorphic and locally injective near ∞ . Moreover, there exist a neighborhood \mathcal{V}_{∞} of ∞ and $\varepsilon \in (0, 1)$ such that $\Phi : \mathcal{V}_{\infty} \to \mathbb{D}_{\varepsilon}$ is biholomorphic. We may assume $\partial \mathcal{V}_{\infty}$ is an analytic simple curve by choosing ε small enough.

Proof 1. Given any $\lambda \in \mathcal{H}_0 \setminus \mathcal{V}_\infty$, there is a unique $\lambda_0 \in \partial \mathcal{V}_\infty$ such that $\arg \phi_{\lambda_0}(0) = \arg \phi_{\lambda}(0)$. It follows from Lemma 4.3.2 and Lemma 4.3.3 that

$$\frac{\exp(d_{\mathcal{H}_0}(\lambda,\lambda_0)) - 1}{\exp(d_{\mathcal{H}_0}(\lambda,\lambda_0)) + 1} \le \left| \frac{G_{\lambda}(0) - G_{\lambda_0}(0)}{G_{\lambda}(0) + G_{\lambda_0}(0)} \right| = \left| \frac{G_{\lambda}(0) - \log(1/\varepsilon)}{G_{\lambda}(0) + \log(1/\varepsilon)} \right|$$

From this inequality and by continuity of the function $\lambda \mapsto G_{\lambda}(0) = -\log |\Phi(\lambda)|$, we conclude $G_{\lambda}(0) \leq \log(1/\varepsilon)$ for all $\lambda \in \mathcal{H}_0 \setminus \mathcal{V}_{\infty}$. Thus we have

$$\frac{\exp(d_{\mathcal{H}_0}(\lambda,\partial\mathcal{V}_{\infty}))-1}{\exp(d_{\mathcal{H}_0}(\lambda,\partial\mathcal{V}_{\infty}))+1} \le \frac{\exp(d_{\mathcal{H}_0}(\lambda,\lambda_0))-1}{\exp(d_{\mathcal{H}_0}(\lambda,\lambda_0))+1} \le \frac{\log(1/\varepsilon) - G_{\lambda}(0)}{\log(1/\varepsilon) + G_{\lambda}(0)},$$

where $d_{\mathcal{H}_0}(\lambda, \partial \mathcal{V}_{\infty}) = \inf_{\zeta \in \partial \mathcal{V}_{\infty}} d_{\mathcal{H}_0}(\lambda, \zeta)$. This inequality indicates

$$G_{\lambda}(0) \leq \log(1/\varepsilon) \exp(-d_{\mathcal{H}_0}(\lambda, \partial \mathcal{V}_{\infty})), \ \lambda \in \mathcal{H}_0 \setminus \mathcal{V}_{\infty}.$$

From this we know that when $\lambda \to \lambda^* \in \mathbb{C} \cap \partial \mathcal{H}_0$, we have $d_{\mathcal{H}_0}(\lambda, \partial \mathcal{V}_\infty) \to \infty$, $G_{\lambda}(0) \to 0$ and $|\Phi(\lambda)| \to 1$. This implies that $\Phi : \mathcal{H}_0 \cup \{\infty\} \to \mathbb{D}$ is a proper map. Since $\Phi^{-1}(0) = \{\infty\}$ and Φ is locally injective near ∞ , Φ is a conformal isomorphism.

Proof 2. Given $\lambda_0 \in \mathcal{V}_{\infty} \setminus \{\infty\}$, there is an analytic arc γ in \mathcal{V}_{∞} passing through λ_0 such that $\arg \phi_{\lambda_0}(0) = \arg \phi_{\lambda}(0)$ for all $\lambda \in \gamma$. By Lemma 4.3.2 and Lemma 4.3.3, we have

$$\frac{\exp(d_{\mathcal{H}_0}(\lambda,\lambda_0)) - 1}{\exp(d_{\mathcal{H}_0}(\lambda,\lambda_0)) + 1} \le \left| \frac{G_{\lambda}(0) - G_{\lambda_0}(0)}{G_{\lambda}(0) + G_{\lambda_0}(0)} \right| = \frac{\left|\log \Phi(\lambda) - \log \Phi(\lambda_0)\right|}{\left|\log |\Phi(\lambda)| + \log |\Phi(\lambda_0)|\right|}, \ \lambda \in \gamma(4.3)$$

Taking a limit $\lambda \to \lambda_0$ along γ , we have

$$\lim_{\lambda \to \lambda_0} \frac{1}{|\lambda - \lambda_0|} \frac{\exp(d_{\mathcal{H}_0}(\lambda, \lambda_0)) - 1}{\exp(d_{\mathcal{H}_0}(\lambda, \lambda_0)) + 1} = \frac{\rho_{\mathcal{H}_0}(\lambda_0)}{2},$$
$$\lim_{\lambda \to \lambda_0} \frac{1}{|\lambda - \lambda_0|} \frac{|\log \Phi(\lambda) - \log \Phi(\lambda_0)|}{|\log |\Phi(\lambda)| + \log |\Phi(\lambda_0)||} = \frac{|\Phi'(\lambda_0)|}{2|\Phi(\lambda_0)|\log(1/|\Phi(\lambda_0)|)},$$

where $ds_{\mathcal{H}_0} = \rho_{\mathcal{H}_0}(\lambda) |d\lambda|$ is the Poincaré metric on \mathcal{H}_0 . By (4.3), we have

$$ds_{\mathcal{H}_0}(\lambda_0) = \rho_{\mathcal{H}_0}(\lambda_0) |d\lambda| \le \frac{|\Phi'(\lambda_0)||d\lambda|}{|\Phi(\lambda_0)|\log(1/|\Phi(\lambda_0)|)} = \Phi^*(ds_{\mathbb{D}^*})(\lambda_0).$$

On the other hand, by Schwarz Lemma, $\Phi^*(ds_{\mathbb{D}^*})(\lambda_0) \leq ds_{\mathcal{H}_0}(\lambda_0)$. Thus we have $\Phi^*(ds_{\mathbb{D}^*})(\lambda_0) = ds_{\mathcal{H}_0}(\lambda_0)$. This indicates the map $\Phi : \mathcal{H}_0 \to \mathbb{D}^*$ is a covering map. Note that \mathcal{H}_0 is not simply connected since ∞ is a cusp, \mathcal{H}_0 is conformally equivalent to \mathbb{D}^* and Φ is a proper map (See [F]). Since $\Phi : \mathcal{H}_0 \to \mathbb{D}^*$ has Laurent expansion $\Phi(\lambda) = -2\lambda^{-1} + O(\lambda^{-2})$ near ∞ , Φ is a conformal isomorphism. \Box **Corollary 4.3.1.** The non-escape locus \mathcal{M} has logarithmic capacity equal to 2.

Proof. We know from Proposition 4.3.1 that the map $\Phi : \mathcal{H}_0 \to \mathbb{D}^*$ is a conformal isomorphism, thus the Green function for \mathcal{H}_0 is $G_{\lambda}(0) = -\log |\Phi(\lambda)|$, whose asymptotic behavior at ∞ is of the form

$$G_{\lambda}(0) = \log |\lambda| + \gamma + o(1),$$

where $\gamma = -\log 2$ is the Robin constant. Since $\mathbb{C} \cap \partial \mathcal{H}_0$ is the outer boundary of \mathcal{M} , the capacity of \mathcal{M} is equal to $e^{-\gamma} = 2$.

Proposition 4.3.2. Let \mathcal{H} be a component of \mathcal{H}_n for $n \geq 1$, the map $\Phi_{\mathcal{H}}$: $\mathcal{H} \to \mathbb{D}$ defined by $\Phi_{\mathcal{H}}(\lambda) = \phi_{\lambda}(t^n_{\lambda}(0))$ is a conformal isomorphism, where ϕ_{λ} is the Böttcher map for T_{λ} defined throughout $\mathcal{A}_{\lambda}(1)$.

Proof. The proof will be based on the following claim:

Claim: There is a holomorphic map $\lambda : \mathbb{D} \to \mathcal{H}$ such that $\Phi_{\mathcal{H}}(\lambda(\zeta)) = \zeta$ for all $\zeta \in \mathbb{D}$.

Once the claim is proved, we see that $\Phi_{\mathcal{H}}$ is surjective and admits a global inverse map λ . Thus $\Phi_{\mathcal{H}} : \mathcal{H} \to \mathbb{D}$ is in fact a conformal isomorphism. Now we will prove the claim via quasiconformal surgery and holomorphic motion theorem.

Given any $\lambda_0 \in \mathcal{H}$, let W_{λ_0} be the component of $t_{\lambda_0}^{-1}(\mathcal{A}_{\lambda_0}(1))$ other than $\mathcal{A}_{\lambda_0}(\infty)$. By definition of \mathcal{H} , $t_{\lambda_0}^{n-1}(0) \in W_{\lambda_0}$. The Böttcher map ϕ_{λ_0} can be defined in the whole basin $\mathcal{A}_{\lambda_0}(1)$ since there is no critical point in $\mathcal{A}_{\lambda_0}(1)$ other than 1 (By Lemma 4.2.5). Let $\zeta_0 = \phi_{\lambda_0}(t_{\lambda_0}^n(0))$. For $\kappa > 0$, let $D(\zeta_0, \kappa) = \{\zeta \in \mathbb{D}; d_{\mathbb{D}}(\zeta, \zeta_0) < \kappa\}$ be the hyperbolic disk centered at ζ_0 with radius $\kappa, W_{\lambda_0,\kappa} = (t_{\lambda_0}|_{W_{\lambda_0}})^{-1} \circ \phi_{\lambda_0}^{-1}(D(\zeta_0,\kappa))$ be the relatively compact subset of W_{λ_0} . For any $\zeta \in D(\zeta_0, \kappa)$, we will define a map $\delta_{\zeta} : W_{\lambda_0,\kappa} \to \phi_{\lambda_0}^{-1}(D(\zeta_0,\kappa))$ satisfying the following properties:

- $\delta_{\zeta_0}(z) = t_{\lambda_0}(z)$ for all $z \in W_{\lambda_0,\kappa}$;
- $\delta_{\zeta}(t_{\lambda_0}^{n-1}(0)) = \phi_{\lambda_0}^{-1}(\zeta);$
- $\delta_{\zeta}: W_{\lambda_0,\kappa} \to \phi_{\lambda_0}^{-1}(D(\zeta_0,\kappa))$ is a quasiconformal map for any fixed ζ ;
- δ_{ζ} is holomorphic with respect to $\zeta \in D(\zeta_0, \kappa)$ for any fixed $z \in W_{\lambda_0, \kappa}$.

Let $E = \partial D(\zeta_0, \kappa) \cup \{\zeta_0\}$. To construct such a map δ_{ζ} , we first define a map $h: D(\zeta_0, \kappa) \times E \to \mathbb{D}$ as follows:

$$h(\zeta, z) = \begin{cases} z, & z \in \partial D(\zeta_0, \kappa), \\ \zeta, & z = \zeta_0. \end{cases}$$

It is easy to check that

• $h(\zeta_0, z) = z, \ z \in E,$

- for every fixed $\zeta \in D(\zeta_0, \kappa), \ z \mapsto h(\zeta, z)$ is injective on E,
- for every fixed $z \in E$, $\zeta \mapsto h(\zeta, z)$ is holomorphic in $D(\zeta_0, \kappa)$.

Thus $h : D(\zeta_0, \kappa) \times E \to \mathbb{D}$ is a holomorphic motion parameterized by $D(\zeta_0, \kappa)$ with base point ζ_0 . By Slodkowski's theorem (See Theorem 2.5.1 or [Slo]), there is a holomorphic motion $H : D(\zeta_0, \kappa) \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ extending h. Moreover, for any fixed $\zeta \in D(\zeta_0, \kappa), H(\zeta, \cdot) : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a quasiconformal map with dilatation $K(H(\zeta, \cdot)) \leq \exp(d_{D(\zeta_0, \kappa)}(\zeta_0, \zeta))$. In particular, $H(\zeta_0, \cdot)$ is the identity map. Let

$$\delta_{\zeta}(z) = \phi_{\lambda_0}^{-1} \circ H(\zeta, \phi_{\lambda_0} \circ t_{\lambda_0}(z))$$

for $z \in W_{\lambda_0,\kappa}$. It is easy to check that δ_{ζ} satisfies the required properties.

Now we define a quasiregular map

$$L_{\zeta}(z) = \begin{cases} \delta_{\zeta}(z), & z \in W_{\lambda_0,\kappa}, \\ t_{\lambda_0}(z), & z \in \overline{\mathbb{C}} \setminus W_{\lambda_0,\kappa} \end{cases}$$

Let σ be the standard complex structure, we can construct a complex structure σ_{ζ} invariant under L_{ζ} as follows:

$$\sigma_{\zeta} = \begin{cases} (t_{\lambda_0}^m)^* (\delta_{\zeta}^* \sigma), & \text{in } t_{\lambda_0}^{-m} (W_{\lambda_0,\kappa}) \text{ for } m \ge 0, \\ \sigma, & \text{in } \bar{\mathbb{C}} \setminus \bigcup_{m \ge 0} t_{\lambda_0}^{-m} (W_{\lambda_0,\kappa}). \end{cases}$$

Let μ_{ζ} be the Beltrami coefficient for the complex structure σ_{ζ} . Since L_{ζ} is equal to t_{λ_0} outside of $W_{\lambda_0,\kappa}$, it is easy to check that

$$\|\mu_{\zeta}\|_{\infty} \leq \frac{K(H(\zeta, \cdot)) - 1}{K(H(\zeta, \cdot)) + 1} \leq \frac{\exp(d_{D(\zeta_0, \kappa)}(\zeta_0, \zeta)) - 1}{\exp(d_{D(\zeta_0, \kappa)}(\zeta_0, \zeta)) + 1} < 1$$

for all $\zeta \in D(\zeta_0, \kappa)$. Moreover, μ_{ζ} is holomorphic with respect to ζ in the distribution sense by Slodkowski's theorem. By Ahlfors-Bers Theorem, there is a quasiconformal map $f_{\zeta} : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ with 0,1 and ∞ fixed such that $f_{\zeta}^*(\sigma) = \sigma_{\zeta}$. f_{ζ} is holomorphic with respect to ζ and $f_{\zeta_0} = \text{id}$.

The map $R_{\zeta} = f_{\zeta} \circ L_{\zeta} \circ f_{\zeta}^{-1}$ is a rational map since it preserves the standard complex structure σ . It is of degree two and satisfies the following properties:

- R_{ζ} is holomorphic with respect to $\zeta \in D(\zeta_0, \kappa)$ and $R_{\zeta_0} = t_{\lambda_0}$;
- R_{ζ} permutes 1 and ∞ . Moreover, 1 is a pole of R_{ζ} of order two;
- R_{ζ} has a zero of order two.

From these information, we conclude

$$R_{\zeta}(z) = f_{\zeta} \circ L_{\zeta} \circ f_{\zeta}^{-1}(z) = \left(\frac{z-1+\lambda_{\kappa}(\zeta)}{z-1}\right)^2 = t_{\lambda_{\kappa}(\zeta)}(z) \tag{4.4}$$

where $\zeta \mapsto \lambda_{\kappa}(\zeta)$ is holomorphic for $\zeta \in D(\zeta_0, \kappa)$ and $\lambda_{\kappa}(\zeta_0) = \lambda_0$. From (4.4), we get

$$\lambda_{\kappa}(\zeta) = 1 - f_{\zeta}(1 - \lambda_0), \ \zeta \in D(\zeta_0, \kappa).$$

This relation indicates that the map λ_{κ} is determined by a slice of holomorphic motion.

It is easy to check that the map $f_{\zeta} : \mathcal{A}_{\lambda_0}(1) \to \mathcal{A}_{\lambda_{\kappa}(\zeta)}(1)$ is a conformal isomorphism and the following diagram is commutative.

$$\begin{array}{c|c} \mathcal{A}_{\lambda_0}(1) \xrightarrow{T_{\lambda_0}} \mathcal{A}_{\lambda_0}(1) \\ f_{\zeta} & & \downarrow f_{\zeta} \\ \mathcal{A}_{\lambda_{\kappa}(\zeta)}(1) \xrightarrow{T_{\lambda_{\kappa}(\zeta)}} \mathcal{A}_{\lambda_{\kappa}(\zeta)}(1) \end{array}$$

Thus the Böttcher maps for T_{λ_0} and $T_{\lambda_{\kappa}(\zeta)}$ satisfy the following relation

$$\phi_{\lambda_0}(z) = \phi_{\lambda_{\kappa}(\zeta)} \circ f_{\zeta}(z), \quad z \in \mathcal{A}_{\lambda_0}(1).$$

For any $\zeta \in D(\zeta_0, \kappa)$, by definition of L_{ζ} , we see that $t^n_{\lambda_{\kappa}(\zeta)}(0) \in \mathcal{A}_{\lambda_{\kappa}(\zeta)}(1)$ and $t^{n-1}_{\lambda_{\kappa}(\zeta)}(0) \notin \mathcal{A}_{\lambda_{\kappa}(\zeta)}(\infty)$. Thus $\lambda_{\kappa}(D(\zeta_0, \kappa)) \subset \mathcal{H}_n$. In fact, $\lambda_{\kappa}(D(\zeta_0, \kappa))$ is contained in \mathcal{H} since $\lambda_{\kappa}(D(\zeta_0, \kappa))$ is connected and $\lambda_{\kappa}(\zeta_0) = \lambda_0 \in \mathcal{H} \cap \lambda_{\kappa}(D(\zeta_0, \kappa))$. For $\zeta \in D(\zeta_0, \kappa)$, we have

$$\Phi_{\mathcal{H}}(\lambda_{\kappa}(\zeta)) = \phi_{\lambda_{\kappa}(\zeta)}(t^{n}_{\lambda_{\kappa}(\zeta)}(0)) = \phi_{\lambda_{\kappa}(\zeta)} \circ f_{\zeta} \circ L^{n}_{\zeta} \circ f^{-1}_{\zeta}(0)
= \phi_{\lambda_{\kappa}(\zeta)} \circ f_{\zeta} \circ \delta_{\zeta} \circ t^{n-1}_{\lambda_{0}}(0) = \phi_{\lambda_{0}} \circ \delta_{\zeta} \circ t^{n-1}_{\lambda_{0}}(0) = \zeta.$$

Note that for $\kappa_1 > \kappa_2 > 0$, both λ_{κ_1} and λ_{κ_2} are local inverse of $\Phi_{\mathcal{H}}$ such that $\lambda_{\kappa_1}(\zeta_0) = \lambda_{\kappa_2}(\zeta_0) = \lambda_0$, thus we have $\lambda_{\kappa_1}|_{D(\zeta_0,\kappa_2)} = \lambda_{\kappa_2}$. Since $\bigcup_{\kappa>0} D(\zeta_0,\kappa) = \mathbb{D}$, there is a holomorphic map $\lambda : \mathbb{D} \to \mathcal{H}$ such that $\lambda|_{D(\zeta_0,\kappa)} = \lambda_{\kappa}$ for all $\kappa > 0$ and $\Phi_{\mathcal{H}}(\lambda(\zeta)) = \zeta$ for all $\zeta \in \mathbb{D}$. This ends the proof of the claim.

Proof of Theorem 4.1.1. First note that \mathcal{M} is compact. To prove \mathcal{M} is connected is equivalent to prove every connected component of $\mathbb{C} \setminus \mathcal{M}$ is simply connected. By Lemma 4.2.4, a connected component of $\mathbb{C} \setminus \mathcal{M}$ is either $\mathcal{H}_0 \cup \{\infty\}$ or a capture domain of level $n \geq 1$. These components are simply connected by Proposition 4.3.1 and Proposition 4.3.2.

An concluding remark: extremal property of Beltrami coefficient

At the end of this section, we give a remark about Lemma 4.3.2. Given $\lambda_1, \lambda_2 \in \mathcal{H}_0$ with $\arg \phi_{\lambda_1}(0) = \arg \phi_{\lambda_2}(0)$, let $QC(\lambda_1, \lambda_2)$ be the set of all quasiconformal maps $\varphi : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that $\varphi \circ t_{\lambda_1} = t_{\lambda_2} \circ \varphi$. We will show in the following that the map h constructed in Lemma 4.3.2 has the following extremal property:

$$\|\mu\|_{\infty} = \inf\{\|\mu_{\varphi}\|_{\infty}; \ \varphi \in QC(\lambda_1, \lambda_2)\},\$$

where μ and μ_{φ} are the Beltrami coefficients of h and $\varphi \in QC(\lambda_1, \lambda_2)$ respectively.

Indeed, from Proposition 4.3.1 we know that the map $\Phi : \mathcal{H}_0 \to \mathbb{D}^*$ is a conformal isomorphism. We may assume $|\Phi(\lambda_1)| \leq |\Phi(\lambda_2)|$, thus we have

$$d_{\mathcal{H}_0}(\lambda_1, \lambda_2) = d_{\mathbb{D}^*}(\Phi(\lambda_1), \Phi(\lambda_2)) = \int_{|\Phi(\lambda_1)|}^{|\Phi(\lambda_2)|} \frac{dr}{r \log(1/r)} = \log \frac{G_{\lambda_1}(0)}{G_{\lambda_2}(0)}.$$

By Lemma 4.3.3, for any $\varphi \in QC(\lambda_1, \lambda_2)$,

$$\|\mu_{\varphi}\|_{\infty} \geq \frac{\exp(d_{\mathcal{H}_{0}}(\lambda_{1},\lambda_{2})) - 1}{\exp(d_{\mathcal{H}_{0}}(\lambda_{1},\lambda_{2})) + 1} = \left|\frac{G_{\lambda_{1}}(0) - G_{\lambda_{2}}(0)}{G_{\lambda_{1}}(0) + G_{\lambda_{2}}(0)}\right| = \|\mu\|_{\infty}.$$

Here is a more general result on extremal quasiconformal conjugacy: Let \mathcal{H} be a capture domain of level $n \geq 0$, $\lambda_1, \lambda_2 \in \mathcal{H}$ be two parameters such that each t_{λ_i} is not postcritically finite. We still use $QC(\lambda_1, \lambda_2)$ to denote the set of all quasiconformal maps $\varphi : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that $\varphi \circ t_{\lambda_1} = t_{\lambda_2} \circ \varphi$. Then there exists $\phi \in QC(\lambda_1, \lambda_2)$ such that

$$\|\mu_{\phi}\| = \inf\{\|\mu_{\varphi}\|; \varphi \in \mathcal{Q}(\lambda_1, \lambda_2)\} = \left|\frac{\operatorname{Log}\phi_{\lambda_1}(t_{\lambda_1}^n(0)) - \operatorname{Log}\phi_{\lambda_2}(t_{\lambda_2}^n(0))}{\operatorname{Log}\phi_{\lambda_1}(t_{\lambda_1}^n(0)) + \overline{\operatorname{Log}\phi_{\lambda_2}(t_{\lambda_2}^n(0))}}\right|$$

where the Log is chosen such that $|\arg \phi_{\lambda_1}(t^n_{\lambda_1}(0)) - \arg \phi_{\lambda_2}(t^n_{\lambda_2}(0))| \leq \pi$.

4.4 Proof of Theorem 4.1.2

It is known from the previous section that when $\lambda \in \mathcal{H}_0$, $J(t_{\lambda})$ is a quasicircle. There are two natural questions:

- 1. What's the asymptotic behavior of $J(t_{\lambda})$ when $|\lambda| \to \infty$?
- 2. How to estimate the Hausdorff dimension of $J(t_{\lambda})$ for all $\lambda \in \mathcal{H}_0$?

It is observed by Hu and Lin (See [HL]) that $J(t_{\lambda})$ becomes larger and more circular as the real parameter $\lambda \to \infty$. In 1995, Osbaldestin shows that the Hausdorff dimension $HD(J(t_{\lambda}))$ of $J(t_{\lambda})$ has the following expansion (see [O]):

$$HD(J(t_{\lambda})) = 1 + \frac{|\lambda|^{-2/3}}{4\log 2} + \mathcal{O}(|\lambda|^{-1})$$

when the real parameter $\lambda \to \infty$. This expansion also holds for $\lambda \in \mathbb{C}$ when $|\lambda|$ is large.

Theorem 4.1.2 provides an answer to the second question. In the following, we prove Theorem 4.1.2. It's equivalent to prove the following inequality:

$$HD(J(t_{\lambda})) \le 1 + |\Phi(\lambda)|^{2/3}, \quad \lambda \in \mathcal{H}_0$$

where $\Phi : \mathcal{H}_0 \to \mathbb{D}^*$ is the map that constructed in Proposition 4.3.1. The proof of Theorem 4.1.2 will be based on Smirnov's theorem ([Smi], See also Theorem 2.4.3) and Slodkowski's theorem ([Slo], See also Theorem 2.5.1). **Proof of Theorem 4.1.2.** Let ℓ be a quasicircle, the dilatation $K[\ell]$ of ℓ is defined by

 $K[\ell] = \inf\{K(\varphi); \ \varphi: \overline{\mathbb{C}} \to \overline{\mathbb{C}} \text{ is a quasiconformal map such that } \varphi(\mathbb{S}^1) = \ell\}.$

By Smirnov's theorem, we have

$$HD(J(t_{\lambda})) \le 1 + \left(\frac{K[J(t_{\lambda})] - 1}{K[J(t_{\lambda})] + 1}\right)^{2}, \ \lambda \in \mathcal{H}_{0}$$

In the following, we prove

$$K[J(t_{\lambda})] \leq \frac{1 + |\Phi(\lambda)|^{1/3}}{1 - |\Phi(\lambda)|^{1/3}}, \ \lambda \in \mathcal{H}_0.$$

Recall that under the coordinate change in Lemma 4.3.4, the family $\{T_{\lambda}; \lambda \in \mathcal{H}_0\}$ becomes to $\{S_{\nu}; \nu \in \mathcal{V}_0^* = \mathcal{V}_0 \setminus \{0\}\}$, where \mathcal{V}_0 is a neighborhood of 0 such that the map $\nu \mapsto \lambda = \nu^{-3}$ is a proper map of degree three from \mathcal{V}_0^* to \mathcal{H}_0 . It is easy to check that under coordinate change, $K[J(t_{\lambda})] = K[J(S_{\nu})], HD(J(t_{\lambda})) = HD(J(S_{\nu}))$. Let $\varphi : \mathcal{V}_0 \to \mathbb{D}$ be a Riemann mapping with $\varphi(0) = 0$. The map Φ and φ satisfy

$$\Phi(\lambda) = \Phi(\nu^{-3}) = e^{i\theta} \varphi(\nu)^3, \ \nu \in \mathcal{V}_0,$$

where θ is a real constant.

Given a pair of compact sets (X, Y), recall that the Hausdorff distance $\sigma_H(X, Y)$ between X and Y is defined by

$$\sigma_H(X,Y) = \max\{\max_{x \in X} \sigma(x,Y), \ \max_{y \in Y} \sigma(X,y)\},\$$

where $\sigma(\cdot, \cdot)$ denotes the spherical distance.

We claim $\sigma_H(J(S_{\nu}), \mathbb{S}^1) \to 0$ as $\nu \to 0$. Indeed, using basic analysis, we can verify that for any $\varepsilon \in (0, 1)$, there is a small positive number δ such that when $|\nu| < \delta$, we have $S_{\nu}(\mathbb{D}_{1-\varepsilon}) \subset \mathbb{D}_{1-\varepsilon}$, $S_{\nu}(\mathbb{C} \setminus \overline{\mathbb{D}}_{1+\varepsilon}) \subset \mathbb{C} \setminus \overline{\mathbb{D}}_{1+\varepsilon}$. Thus both $\mathbb{D}_{1-\varepsilon}$ and $\mathbb{C} \setminus \overline{\mathbb{D}}_{1+\varepsilon}$ lie in the Fatou set $F(S_{\nu})$. It turns out that $\sigma_H(J(S_{\nu}), \mathbb{S}^1) \leq 2\varepsilon$. This means the Julia set $J(S_{\nu})$ moves continuously at $\nu = 0$ in Hausdorff topology.

It is obvious that the Julia set $J(S_{\nu})$ moves continuously on \mathcal{V}_0^* in Hausdorff topology. Thus by adding an new map $S_0(\zeta) = \zeta^4$ to the family $\{S_{\nu}; \nu \in \mathcal{V}_0^*\}$, we see that the Julia set $J(S_{\nu})$ moves continuously on \mathcal{V}_0 in Hausdorff topology. By characterizations of stability (see [McM1], Theorem 4.2), the Julia set

 $J(S_{\nu})$ moves holomorphically on \mathcal{V}_0 . So there is a holomorphic motion h: $\mathcal{V}_0 \times \mathbb{S}^1 \to \overline{\mathbb{C}}$ parameterized by \mathcal{V}_0 with base point 0 such that $h(0, \cdot) = \mathrm{id}$ and $h(\nu, \mathbb{S}^1) = J(S_{\nu})$ for all $\nu \in \mathcal{V}_0$. By Slodkowski's theorem (see [Slo]), there is a holomorphic motion $H : \mathcal{V}_0 \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ extending h. Moreover for any fixed $\nu \in \mathcal{V}_0, H(\nu, \cdot) : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a quasiconformal map with dilatation $K(H(\nu, \cdot)) \leq \exp(d_{\mathcal{V}_0}(0, \nu))$. Since $H(\nu, \mathbb{S}^1) = J(S_{\nu})$, we have

$$\begin{split} K[J(t_{\lambda})] &= K[J(S_{\nu})] \leq K(H(\nu, \cdot)) \leq \exp(d_{\mathcal{V}_{0}}(0, \nu)) \\ &= \exp(d_{\mathbb{D}}(0, \varphi(\nu))) = \frac{1 + |\varphi(\nu)|}{1 - |\varphi(\nu)|} = \frac{1 + |\Phi(\lambda)|^{1/3}}{1 - |\Phi(\lambda)|^{1/3}}. \end{split}$$

A concluding remark We remark that the exponent 2/3 in Theorem 4.1.2 is sharp in the following sense:

$$\max\{t; HD(J(t_{\lambda})) \le 1 + |\Phi(\lambda)|^t, \forall \lambda \in \mathcal{H}_0\} = \frac{2}{3}$$

Moreover,

$$\lim_{\lambda \to \infty} \frac{HD(J(t_{\lambda})) - 1}{|\Phi(\lambda)|^{2/3}} = \frac{1}{2^{8/3} \log 2}.$$

This follows from Osbaldestin's result on the asymptotic behavior of $HD(J(t_{\lambda}))$ near ∞ .

5.1 Introduction

The local connectivity of Julia sets for rational maps is a central problem in complex dynamical systems. It is well studied for classical type of rational maps, for example: hyperbolic and semihyperbolic maps, geometrically finite maps, see [CJY],[M2],[TY]. In polynomial case, it is also known a lot, see [DH2],[GS],[Kiwi],[Ly],[M2]. For quadratic polynomials, Yoccoz proved that the Julia set is locally connected provided that all periodic points are repelling and the map is not infinitely renormalizable, see [Hu],[M2]. Douady exhibited striking example of infinitely renormalizable quadratic polynomial with non locally connected Julia set, see [M2]. For general polynomial with connected Julia set and without irrationally neutral cycles, Kiwi shows in [Kiwi] that the local connectivity of Julia set is equivalent to the non existence of wandering continua.

The powerful tool to study the local connectivity of Julia sets for polynomials is the so-called 'Branner-Hubbard-Yoccoz puzzle' techniques, which is introduced by Branner-Hubbard and Yoccoz, [BH]. It has a natural way of construction, which is induced by finite periodic external rays together with an equipotential curve.

However, for general rational maps, things are different. The construction of Yoccoz puzzle becomes quite involved, even impossible. Up to now, the only known rational maps which admit Yoccoz puzzle structures are the cubic Newton maps, whose Yoccoz puzzles are constructed by Roesch. In [Ro1], by Yoccoz puzzle techniques, Roesch shows striking differences between rational maps and polynomials. The method also leads to the local connectivity of Julia sets except some specific cases.

In this article, we exhibit Yoccoz puzzle structure for another family of rational maps, known as McMullen maps, of the form

$$f_{\lambda}: z \mapsto z^n + \lambda/z^n, \ \lambda \in \mathbb{C}^* = \mathbb{C} \setminus \{0\}, \ n \ge 3.$$

Dynamics of this family have been studied by Devaney and his group, see [D1],[D2],[DK],[DLU].

The difference of Yoccoz puzzle between cubic Newton maps and McMullen maps is as follows: For cubic Newton maps, the ingredient of the Yoccoz puzzle is an converging ray that intersects the Julia set in a countably many points while for McMullen maps, the element to construct Yoccoz puzzle is a Jordan curve (will be called 'cut ray') that intersects the Julia set in a Cantor set of points. This kind of Jordan curve is induced by some particular angle and can be viewed as an extention of the corresponding external ray (see Section 5.3).

We denote by B_{λ} the immediate attractive basin of ∞ . The topology of ∂B_{λ} is of special interest. Based on Yoccoz puzzle techniques and combinatorial and topological analysis, we prove:

Theorem 5.1.1. (Cantor or Jordan) For any $n \geq 3$ and any complex parameter λ , if the Julia set $J(f_{\lambda})$ is not a Cantor set, then ∂B_{λ} is a Jordan curve.

This affirmatively answers a question posed by Devaney at the Snowbird Conference on the 25th Birthday of the Mandelbrot set, see [DK]. For the higher regularity of ∂B_{λ} , we show that ∂B_{λ} is a quasicircle except two special cases.

Theorem 5.1.2. Suppose the Julia set $J(f_{\lambda})$ is not a Cantor set, then ∂B_{λ} is a quasicircle if it contains neither parabolic point nor recurrent critical point.

Here, a recurrent critical point c on the Julia set of a rational map f is a critical point such that $c \in \omega(c)$, where $\omega(c)$ is the ω -limit set of c, defined as $\{z \in \overline{\mathbb{C}}; \text{ there exist } n_k \to \infty \text{ such that } z = \lim f^{n_k}(c)\}$. It follows from Proposition 5.7.5 that if ∂B_{λ} contains a parabolic point, then ∂B_{λ} is not a quasicircle by Leau-Fatou-Flower Theorem, see [M2]. The question whether ∂B_{λ} is a quasicircle when ∂B_{λ} contains a recurrent critical point is still unknown.

For the topology of the Julia set, we show

Theorem 5.1.3. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then $J(f_{\lambda})$ is locally connected in either of the following cases:

- 1. The critical orbit does not accumulate on the boundary ∂B_{λ} .
- 2. The map f_{λ} is neither renormalizable nor *-renormalizable.
- 3. The parameter λ is real and positive.

Here, the definitions of *renormalization* and *-*renormalization* can be found in Section 5.5. Theorem 5.1.3 implies that the Julia set is locally connected except some special cases. In fact, it's stronger than the following statement:

Theorem 5.1.4. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then $J(f_{\lambda})$ is locally connected if the critical orbit does not accumulate on the boundary ∂B_{λ} .

Theorem 5.1.4 is an analogue of Roesch's Theorem [Ro1]:

Theorem 5.1.5. (Roesch) A genuine cubic Newton map, without Siegel disks, has a locally connected Julia set provided that the orbit of the non-fixed critical point does not accumulate on the boundary of any invariant basin of attraction.

This chapter is organized as follows:

In Section 5.2, we present some basic results on McMullen maps.

In Section 5.3, we construct the 'cut rays', a kind of Jordan curves that cut the Julia set into two different parts. We first construct a Cantor set of angles on the unit circle which is used to generate 'cut rays'. Then we discuss the construction of 'cut rays' based on the work of Devaney.

In Section 5.4, basic knowledge of Yoccoz puzzles, graphs and tableaux are presented. The aim of this section is to find a Yoccoz puzzle with a non-degenerate critical annulus (See Section 5.4.2). A natural construction of 'modified puzzle piece' is discussed (See Section 5.4.3).

In Section 5.5, we discuss the renormalizations of McMullen maps from the viewpoint of puzzle piece.

In Section 5.6, we present a criterion of local connectivity. We introduce a '**BD** condition' on the boundary of immediate basin of attraction. Such condition can be considered as 'local semi-hyperbolicity'. We show that '**BD** condition' implies good topology.

In Section 5.7, we study the local connectivity of ∂B_{λ} in all possible cases and show that ∂B_{λ} enjoys higher regularity except two special cases.

In Section 5.8, we study the local connectivity of the Julia set $J(f_{\lambda})$ based on the 'Characterization of Local Connectivity' and the 'Shrinking Lemma'.

5.2 Preliminaries and Notations

In this section, we present some basic results and notations for the family of rational maps:

$$f_{\lambda}(z) = z^n + \lambda/z^n$$

where $\lambda \in \mathbb{C}^*$ and $n \geq 3$. This kind of map is known as 'McMullen map' since it is first studied by McMullen. McMullen proved that when $|\lambda|$ is small enough, the Julia set of $z \mapsto z^2 + \lambda/z^3$ is Cantor set of circles, see [McM3]. For any $\lambda \in \mathbb{C}^*$, the map f_{λ} has a superattracting fixed point at ∞ . The immediate basin of ∞ is denoted by B_{λ} , and the component of $f_{\lambda}^{-1}(B_{\lambda})$ that contains 0 is denoted by T_{λ} . The set of all critical points of f_{λ} is $\{0, \infty\} \cup C_{\lambda}$, where $C_{\lambda} = \{\sqrt[2n]{\lambda}\omega; \omega^{2n} = 1\}$. Besides ∞ , there are only two critical values for $f_{\lambda}: v_{\lambda}^+ = 2\sqrt{\lambda}$ and $v_{\lambda}^- = -2\sqrt{\lambda}$. In fact, there is only one critical orbit (up to a sign). Let $P(f_{\lambda}) = \overline{\bigcup_{n \geq 1} f_{\lambda}^k(C_{\lambda})} \cup \{\infty\}$ be the postcritical set.

The Böttcher map ϕ_{λ} for f_{λ} is defined in a neighborhood of ∞ by $\phi_{\lambda}(z) = \lim_{k \to \infty} (f_{\lambda}^{k}(z))^{n^{-k}}$. The Böttcher map is unique if we require $\phi'_{\lambda}(\infty) = 1$. It is known that the Böttcher map ϕ_{λ} can be extended to a domain $\text{Dom}(\phi_{\lambda}) \subset B_{\lambda}$ such that $\phi_{\lambda} : \text{Dom}(\phi_{\lambda}) \to \{z \in \overline{\mathbb{C}} : |z| > R\}$ is a conformal isomorphism for some smallest number $R \geq 1$. In particular, if B_{λ} contains no critical point other than ∞ , then $\text{Dom}(\phi_{\lambda}) = B_{\lambda}$; if B_{λ} contains a critical point $c \in \{0\} \cup C_{\lambda}$, then by 'The Escape Trichotomy' (Theorem 5.2.1), the Julia set $J(f_{\lambda})$ is a Cantor set.

The Green function $G_{\lambda}: B_{\lambda} \to (0, \infty]$ is defined by

$$G_{\lambda}(z) = \lim_{k \to \infty} n^{-k} \log |f_{\lambda}^{k}(z)|.$$

By definition, $G_{\lambda}(f_{\lambda}(z)) = nG_{\lambda}(z)$ for $z \in B_{\lambda}$ and $G_{\lambda}(z) = \log |\phi_{\lambda}(z)|$ for $z \in \text{Dom}(\phi_{\lambda})$. The Green function G_{λ} can be extended to $A_{\lambda} = \bigcup_{k \ge 0} f_{\lambda}^{-k}(B_{\lambda})$ by defining

$$G_{\lambda}(z) = n^{-k} G_{\lambda}(f_{\lambda}^{k}(z)) \text{ for } z \in f_{\lambda}^{-k}(B_{\lambda}).$$

In the following, for a set E in $\overline{\mathbb{C}}$ and $a \in \mathbb{C}$, let $aE = \{az; z \in E\}$, $a + E = \{a + z; z \in E\}$, $E^* = \{\overline{z}; z \in E\}$, \overline{E} be the closure of E and int(E) be the interior of E.

Lemma 5.2.1. (Symmetry of the Dynamical Plane) Let ω satisfy $\omega^{2n} = 1$, then

1.
$$\omega J(f_{\lambda}) = J(f_{\lambda}).$$

2. $G_{\lambda}(\omega z) = G_{\lambda}(z) \text{ for } z \in A_{\lambda}.$
3. $\omega \text{Dom}(\phi_{\lambda}) = \text{Dom}(\phi_{\lambda}), \text{ and } \phi_{\lambda}(\omega z) = \omega \phi_{\lambda}(z) \text{ for } z \in \text{Dom}(\phi_{\lambda}).$

Proof. For 1, since $A_{\lambda} = \{z \in \overline{\mathbb{C}}; f_{\lambda}^{k}(z) \text{ tends to infinity as } k \to \infty\}$ and $f_{\lambda}^{k}(\omega z) = \pm f_{\lambda}^{k}(z)$ for $k \ge 1$, $f_{\lambda}^{k}(\omega z)$ tends to infinity if and only if $f_{\lambda}^{k}(z)$ tends to infinity as $k \to \infty$. Thus $\omega A_{\lambda} = A_{\lambda}$. The conclusion follows from the fact that $J(f_{\lambda}) = \partial A_{\lambda}$.

2. By the definition of G_{λ} .

3. Since $\text{Dom}(\phi_{\lambda})$ is the connected component of $\{z \in B_{\lambda}; G_{\lambda}(z) > \log R\}$ that contains ∞ , we conclude that $\omega \text{Dom}(\phi_{\lambda}) = \text{Dom}(\phi_{\lambda})$. Note that $\phi_{\lambda}(\omega z)$ and $\omega \phi_{\lambda}(z)$ are two Riemann mappings from $\text{Dom}(\phi_{\lambda})$ onto $\{z \in \overline{\mathbb{C}}; |z| > R\}$ with the same derivative at ∞ , we have $\phi_{\lambda}(\omega z) = \omega \phi_{\lambda}(z)$ by the uniqueness of Riemann mapping theorem. \Box The Mandelbrot set for this family is defined by

$$M = \{\lambda \in \mathbb{C}^*; f_{\lambda}^k(v_{\lambda}^+) \text{ does not tend to infinity as } k \to \infty \}.$$

Lemma 5.2.2. (Symmetry of the Parameter Plane) The Mandelbrot set *M* satisfies:

- 1. $M^* = M$.
- 2. $\nu M = M$ with $\nu^{n-1} = 1$.
- 3. For any line $\ell \in \{\varepsilon \mathbb{R}; \varepsilon^{2n-2} = 1\}$, M is symmetric about ℓ .

Proof. 1. Since $\overline{f_{\lambda}(\bar{z})} = f_{\bar{\lambda}}(z)$, the critical orbit of f_{λ} and the critical orbit of $f_{\bar{\lambda}}$ are symmetric under the map $z \to \bar{z}$. They either both remain bounded or both tend to infinity. Thus $M^* = M$.

2. Let $\nu = e^{2\pi i/(n-1)}$ and $\varphi(z) = e^{\pi i/(n-1)}z$. For $k \ge 1$,

$$\varphi^{-1} \circ f_{\nu\lambda}^k \circ \varphi(z) = \begin{cases} (-1)^k f_{\lambda}^k(z), & n \text{ odd,} \\ f_{\lambda}^k(z), & n \text{ even} \end{cases}$$

Thus the critical orbit of f_{λ} tends to infinity if and only if the critical orbit of $f_{\nu\lambda}$ tends to infinity. Equivalently, $\lambda \in M$ if and only if $\nu\lambda \in M$.

3. It follows from 1 and 2.

From Lemma 5.2.2, f_{λ} and $f_{\lambda e^{2\pi i/(n-1)}}$ have the same dynamical property and their Julia sets are identical up to a rotation. Thus the fundamental domain of the parameter plane is $\{\lambda \in \mathbb{C}^*; \arg \lambda \in [0, \frac{2\pi}{n-1})\}$.

The following theorem due to Devaney, Look and Uminsky gives a classification of Julia sets with different topological type, see [DLU].

Theorem 5.2.1. (Devaney-Look-Uminsky) We have the following 'Escape Trichotomy':

1. If $v_{\lambda}^+ \in B_{\lambda}$, then $J(f_{\lambda})$ is a Cantor set.

2. If $v_{\lambda}^+ \in T_{\lambda} \neq B_{\lambda}$, then $J(f_{\lambda})$ is a Cantor set of circles.

3. If $f_{\lambda}^{k}(v_{\lambda}^{+}) \in T_{\lambda} \neq B_{\lambda}$ for some $k \geq 1$, then $J(f_{\lambda})$ is a Sierpiński curve, which is locally connected.

In all other cases, the critical orbits remain bounded and the Julia set $J(f_{\lambda})$ is connected.

For $n \geq 3$, it is known that the unbounded component of $\mathbb{C}^* - M$ consists of the parameters for which the Julia set is a Cantor set, this region is called *Cantor set locus*, see Figure 5.1. The component of $\mathbb{C}^* - M$ that contains a punctured neighborhood of 0 is the region where the Julia set $J(f_{\lambda})$ is a Cantor set of circles; this is the *McMullen domain*, as it is McMullen who first discovered this type of Julia set. The complement of these two regions is



Figure 5.1: Parameter plane for McMullen maps when n = 3.

the *connected locus*. The small copies of quadratic Mandelbrot set correspond to the renormalizable parameters while the 'holes' in the connected locus are always called *Sierpiński holes* according to Devaney. These regions correspond to the parameters for which the Julia set is a Sierpiński curve.

We will see later that when the critical orbit tends to ∞ , the boundary ∂B_{λ} is a quasicircle if it is connected. So this case is already well studied. For this reason, throughout the paper, all discussions are based on the following:

Hypothesis: The critical orbit remain bounded, or equivalently, $C_{\lambda} \cap A_{\lambda} = \emptyset$.

At the end of this section, we give some notations. We restrict our attention to the parameters $\lambda \in \mathcal{H} = \{\lambda \in \mathbb{C}^*; \arg \lambda \in (0, \frac{2\pi}{n-1})\}$ because of the symmetry of the parameter plane. The real positive parameters will be considered separately in Section 5.7.3.

Let $c_0 = c_0(\lambda) = \sqrt[2n]{\lambda}$ be the critical point that lies on \mathbb{R}^+ when $\lambda \in \mathbb{R}^+$ and varies analytically as λ ranges over \mathcal{H} . Let $c_k = c_0 e^{k\pi i/n}$ for $1 \le k \le 2n-1$. The critical points c_k with k even are mapped to $v_{\lambda}^+ = 2\sqrt{\lambda}$ while the critical points c_k with k odd are mapped to $v_{\lambda}^- = -2\sqrt{\lambda}$.

Let $\ell_k = c_k \mathbb{R}^+$ be the straight line connecting the origin with ∞ and passing through c_k for $0 \le k \le 2n - 1$. We call ℓ_k a *critical ray*. The closed sector bounded by ℓ_k and ℓ_{k+1} is denoted by S_k for $0 \le k \le n$. Define $S_{-k} = -S_k$ for $1 \le k \le n - 1$. So the sectors are arranged in counterclockwise order about the origin as $S_0, S_1, \dots, S_n, S_{-1}, \dots, S_{-(n-1)}$.

The critical value v_{λ}^+ always lies in S_0 since $\arg c_0 < \arg v_{\lambda}^+ < \arg c_1$ for all



Figure 5.2: Sectors in the dynamical plane when n = 3.

 $\lambda \in \mathcal{H}$. Correspondingly, the critical value v_{λ}^{-} lies in S_n . It's easy to check that the image of ℓ_k under f_{λ} is a straight ray connecting one of the critical values to ∞ , this ray is called *critical value ray*. As a consequence, f_{λ} maps the interior of each sectors of $\{S_{\pm 1}, \dots, S_{\pm (n-1)}\}$ univalently onto a region Υ_{λ} which can be identified as the complex sphere \mathbb{C} minus two critical value rays.

Let \mathcal{P} denote the set of all components of $\bigcup_{k\geq 0} f_{\lambda}^{-k}(B_{\lambda})$. For $U \in \mathcal{P}$ and v > 0, let $\mathbf{e}(U, v) = \{z \in U; G_{\lambda}(z) = v\}$ be the equipotential curve. The annulus bounded by $\mathbf{e}(B_{\lambda}, v)$ and $\mathbf{e}(T_{\lambda}, v)$ is denoted by \mathbf{A}_{v} . We may choose v large enough such that $\partial \mathbf{A}_{v}$ intersects with every critical ray at exactly two points. The bounded component and unbounded component of $\overline{\mathbb{C}} \setminus \mathbf{e}(B_{\lambda}, v)$ are denoted by $\mathbf{V}(v)$ and $\mathbf{U}(v)$, respectively.

Now, we define radial rays of U for every $U \in \mathcal{P} \setminus \{B_{\lambda}\}$. Under Hypothesis, we see that there is a unique Riemann mapping $\phi_{T_{\lambda}} : T_{\lambda} \to \mathbb{D}$ such that

$$\phi_{T_{\lambda}}(z)^{-n} = \phi_{\lambda}(f_{\lambda}(z)), \ z \in T_{\lambda}; \ \phi'_{T_{\lambda}}(0) = 1/\sqrt[n]{\lambda}$$

The radial ray $R_{T_{\lambda}}(\theta)$ of angle θ is defined as $\phi_{T_{\lambda}}^{-1}((0,1)e^{2\pi i\theta})$. For $U \in \mathcal{P} \setminus \{B_{\lambda}, T_{\lambda}\}$, there is a smallest integer $k \geq 1$ such that $f_{\lambda}^{k} : U \to T_{\lambda}$ is a conformal map. The radial ray $R_{U}(\theta)$ is defined as the pull back of $R_{T_{\lambda}}(\theta)$ under f_{λ}^{k} .

Let $\mathbb{I} = \{0, n, \pm 1, \dots, \pm (n-1)\}$ be an index set. $\mathbf{I}_k = \overline{\mathbf{A}_v} \cap S_k$ for $k \in \mathbb{I}$ and $\mathbf{I} = \bigcup_{k \in \mathbb{I} \setminus \{0,n\}} \mathbf{I}_k$. The set of all points whose orbits remain in \mathbf{I} under all iterations of f_{λ} is denoted by Λ_{λ} . Obviously $\Lambda_{\lambda} = \bigcap_{k \geq 0} f_{\lambda}^{-k}(\mathbf{I})$ and Λ_{λ} is a subset of the Julia set $J(f_{\lambda})$.

For any $k \in \mathbb{I} \setminus \{0, n\}$, $f_{\lambda} : \operatorname{int}(S_k) \to \Upsilon_{\lambda}$ is a conformal map, and the inverse is denoted by $h_k : \Upsilon_{\lambda} \to \operatorname{int}(S_k)$.

Given a point $z \in \Lambda_{\lambda}$, suppose $f_{\lambda}^{k}(z) \in S_{s_{k}}$ for $k \geq 0$, define the itinerary of z by $\mathbf{s}_{\lambda}(z) = (s_{0}, s_{1}, s_{2}, \cdots)$. The itinerary is always well defined in the set Λ_{λ} since if some iteration $f_{\lambda}^{k}(z)$ lies on the boundary of two adjacent sectors, then the next iteration $f_{\lambda}^{k+1}(z)$ will lie inside $S_{0} \cup S_{n}$.

Let $\Sigma = \{\mathbf{s} = (s_0, s_1, s_2, \cdots); s_k \in \mathbb{I} \setminus \{0, n\}$ for every $k \ge 0\}$ be the space of one-sided sequences of the symbols $\pm 1, \ldots, \pm (n-1)$. For $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma$, the shift map $\sigma : \Sigma \to \Sigma$ is defined by $\sigma(\mathbf{s}) = (s_1, s_2, \cdots)$. We denote (s_0, s_1, s_2, \cdots) by $(\overline{s_0, \cdots, s_{p-1}})$ if $s_{k+p} = s_k$ for $k \ge 0$.

It's obvious that $\mathbf{s}_{\lambda}(f_{\lambda}(z)) = \sigma(\mathbf{s}_{\lambda}(z))$ for $z \in \Lambda_{\lambda}$.

]

Lemma 5.2.3. The set Λ_{λ} is a Cantor set and the itinerary map $\mathbf{s}_{\lambda} : \Lambda_{\lambda} \to \Sigma$ is bijective.

Proof. First note that for any $\lambda \in \mathcal{H}$, **I** is a compact subset of Υ_{λ} . With respect to the hyperbolic metric of Υ_{λ} and by Schwarz Lemma, there is a number $\delta \in (0, 1)$ such that for any $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma$ and any $m \ge 0$,

Hyper.diam
$$\left(\bigcap_{0 \le k \le m} f_{\lambda}^{-k}(\mathbf{I}_{s_k})\right) \le$$
 Hyper.diam $(\mathbf{I}) \cdot \delta^m$

Thus $\bigcap_{k\geq 0} f_{\lambda}^{-k}(\mathbf{I}_{s_k})$ consists of a single point, say $z_{\mathbf{s}}$. This implies that Λ_{λ} is a Cantor set and the map $\mathbf{s}_{\lambda} : \Lambda_{\lambda} \to \Sigma$ defined by $\mathbf{s}_{\lambda}(z_{\mathbf{s}}) = \mathbf{s}$ is bijective. \Box

5.3 Cut Rays in the Dynamical Plane

In this section, we will construct the 'cut rays', a kind of Jordan curves that cut the Julia set into two different parts. The construction is due to Devaney [D2]. We give some more properties which will be used in our paper.

We first construct a Cantor set of angles on the unit circle which is used to generate 'cut rays' in [D2]. These angles can be considered as a combinatorial invariant when the parameter λ ranges over \mathcal{H} .

To begin with, we identify the unit circle $\mathbb{S} = \mathbb{R}/\mathbb{Z}$ with (0, 1]. We say three angles satisfy $t_1 \leq t_2 \leq t_3$ on \mathbb{S} if t_1, t_2, t_3 are in the counterclockwise order.

5.3.1 A Cantor set on the unit circle

In the following, we construct a subset Θ of (0,1]. The set Θ is a Cantor set and is used to generate 'cut rays' in the next section.

First, define a map $\tau : (0,1] \to (0,1]$ by $\tau(\theta) = n\theta \mod 1$. Let $\Theta_k = (\frac{k}{2n}, \frac{k+1}{2n}]$ for $0 \le k \le n$ and $\Theta_{-k} = \Theta_k + \frac{1}{2}$ for $1 \le k \le n-1$. Obviously, $(0,1] = \bigcup_{k \in \mathbb{I}} \Theta_k$.

Define a map $\chi : \mathbb{I} \to \mathbb{N}$ by

$$\chi(k) = \begin{cases} k, & \text{if } 0 \le k \le n, \\ n - k, & \text{if } -(n - 1) \le k \le -1. \end{cases}$$

For $k \in \mathbb{I}$, we have

$$\tau(\Theta_k) \supset \begin{cases} \bigcup_{j=1}^{n-1} \Theta_j, & \text{if } \chi(k) \text{ is even,} \\ \bigcup_{j=1}^{n-1} \Theta_{-j}, & \text{if } \chi(k) \text{ is odd.} \end{cases}$$

For $\theta \in (0, 1]$, suppose $\tau^k(\theta) \in \Theta_{s_k}$ for $k \ge 0$, define the itinerary $\mathbf{s}(\theta)$ of θ by $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$.

Let Θ be the set of all angles $\theta \in (0,1]$ whose orbit remains in $\mathcal{E} = \bigcup_{k=1}^{n-1} (\Theta_k \cup \Theta_{-k})$ under all iterations of τ . The set Θ can be written as $\Theta = \bigcap_{k\geq 0} \tau^{-k}(\mathcal{E}) = \bigcap_{k\geq 0} \tau^{-k}(\overline{\mathcal{E}})$. One can easily verify that Θ is a Cantor set.

The image of Θ under the itinerary map is denoted by $\Sigma_0 = {\mathbf{s}(\theta); \theta \in \Theta}$. One can easily verify that Σ_0 is a subspace of Σ that consists of all elements $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma$ such that for $k \ge 0$, if $\chi(s_k)$ is even, then $s_{k+1} \in {1, \dots, n-1}$; if $\chi(s_k)$ is odd, then $s_{k+1} \in {-1, \dots, -(n-1)}$.

The itinerary map $\mathbf{s} : \Theta \to \Sigma_0$ is bijective since for any $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma_0$, the intersection $\bigcap_{k \ge 0} \tau^{-k}(\Theta_{s_k})$ consists of a single point. In the following, we first construct an inverse map for \mathbf{s} (Lemma 5.3.1).

Let $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma$, define a map $\kappa : \Sigma \to (0, 1]$ by

$$\kappa(\mathbf{s}) = \frac{1}{2} \left(\frac{\chi(s_0)}{n} + \sum_{k \ge 1} \frac{|s_k|}{n^{k+1}} \right).$$

Lemma 5.3.1. $\kappa(\Sigma) = \Theta$ and $\kappa(\mathbf{s}(\theta)) = \theta$ for all $\theta \in \Theta$.

Proof. First, we show $\kappa(\mathbf{s}(\theta)) = \theta$ for $\theta \in \Theta$. Let $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$ and $\hat{\theta} = \kappa(\mathbf{s}(\theta))$. It suffices to show $\mathbf{s}(\hat{\theta}) = \mathbf{s}(\theta)$ since $\mathbf{s} : \Theta \to \Sigma_0$ is bijective. It follows that $\hat{\theta} \in \Theta_{s_0}$ since

$$\frac{\chi(s_0)}{2n} < \hat{\theta} \le \frac{1}{2} \left(\frac{\chi(s_0)}{n} + \sum_{k \ge 1} \frac{n-1}{n^{k+1}} \right) = \frac{\chi(s_0)}{2n} + \frac{1}{2n}.$$

For $k \geq 1$,

$$\tau^{k}(\hat{\theta}) = \begin{cases} \frac{1}{2}(\chi(s_{0}) + |s_{1}| + \dots + |s_{k-1}|) + \frac{1}{2}\sum_{j \ge k} \frac{|s_{j}|}{n^{j-k+1}}, & \text{if } n \text{ is odd,} \\ \frac{|s_{k-1}|}{2} + \frac{1}{2}\sum_{j \ge k} \frac{|s_{j}|}{n^{j-k+1}}, & \text{if } n \text{ is even.} \end{cases}$$

Since $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots) \in \Sigma_0$, we have for $j \ge 1$,

$$\frac{|s_j|}{2} = \begin{cases} \frac{1}{2}(\chi(s_j) - \chi(s_{j-1})) \mod 1, & \text{if } n \text{ is odd,} \\ \frac{1}{2}\chi(s_j) \mod 1, & \text{if } n \text{ is even} \end{cases}$$

and

$$\frac{\chi(s_{j-1})}{2} + \frac{|s_j|}{2n} = \frac{\chi(s_j)}{2n} \mod 1.$$

Thus we have

$$\tau^k(\hat{\theta}) = \frac{\chi(s_{k-1})}{2} + \frac{1}{2} \sum_{j \ge k} \frac{|s_j|}{n^{j-k+1}} = \frac{\chi(s_k)}{2n} + \frac{1}{2} \sum_{j \ge k+1} \frac{|s_j|}{n^{j-k+1}}.$$

This means $\tau^k(\hat{\theta}) \in \Theta_{s_k}$ for $k \ge 1$. So θ and $\hat{\theta}$ have the same itinerary.

In the following, we show $\kappa(\Sigma) = \Theta$. First, by the previous argument, $\Theta = \kappa(\Sigma_0) \subset \kappa(\Sigma)$. Conversely, for any $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma$, there is a unique sequence of symbols $\varepsilon_1, \varepsilon_2, \cdots \in \{\pm 1\}$, such that $\mathbf{s}^* = (s_0, \varepsilon_1 s_1, \varepsilon_2 s_2, \cdots) \in \Sigma_0$. Thus $\kappa(\mathbf{s}) = \kappa(\mathbf{s}^*) \in \Theta$.

Remark 5.3.1. For any $\mathbf{s} = (s_0, s_1, s_2, \cdots) \in \Sigma$, one can verify that

$$\kappa^{-1}(\kappa(\mathbf{s})) = \{(s_0, \pm s_1, \pm s_2, \cdots)\}.$$

Lemma 5.3.2. The set Θ satisfies:

- 1. $\tau(\Theta) = \Theta$.
- 2. $\Theta + \frac{1}{2} = \Theta$.
- 3. Periodic angles are dense in Θ .

Proof. 1. It is obvious that $\tau(\Theta) \subset \Theta$. τ is surjective since $\tau^{-1}(\theta) \cap \mathcal{E} \neq \emptyset$ for all $\theta \in \Theta$.

2. First note that $\mathcal{E} + \frac{1}{2} = \mathcal{E} \mod 1$. For $k \ge 1$, since $\tau^k(\theta + \frac{1}{2}) = \tau^k(\theta)$ when *n* is even and $\tau^k(\theta + \frac{1}{2}) = \tau^k(\theta) + \frac{1}{2}$ when *n* is odd, we have $\tau^k(\theta + \frac{1}{2}) \in \mathcal{E}$ if and only if $\tau^k(\theta) \in \mathcal{E}$. Thus $\theta \in \Theta$ if and only if $\theta + \frac{1}{2} \in \Theta$.

3. Let $\theta \in \Theta$ with itinerary $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. For any $k \geq 1$, either $(\overline{s_0, \cdots, s_k}) \in \Sigma_0$, or there is a symbol $s_{k+1}^* \in \{\pm 1, \cdots, \pm (n-1)\}$ such that $(\overline{s_0, \cdots, s_k, s_{k+1}^*}) \in \Sigma_0$. If $(\overline{s_0, \cdots, s_k}) \in \Sigma_0$, let $\theta_k = \kappa((\overline{s_0, \cdots, s_k}))$. Else, let $\theta_k = \kappa((\overline{s_0, \cdots, s_k, s_{k+1}^*}))$. It's obvious that θ_k is periodic. By Lemma 5.3.1, $\theta_k \in \Theta$ and

$$|\theta - \theta_k| \le C(n)n^{-k} (\to 0 \text{ as } k \to \infty),$$

where C(n) is a constant, depending only on n. This implies that periodic angles are dense in Θ .

Remark 5.3.2. The Hausdorff dimension of Θ is $\frac{\log(n-1)}{\log n}$.

For $\lambda \in \mathcal{H}$ and $k \in \mathbb{I}$, let $\Theta_k^{\lambda} = \Theta_k + \frac{\arg c_0(\lambda)}{2\pi} = \Theta_k + \frac{\arg \lambda}{4n\pi} \mod 1$. Recall that for $\lambda \in \mathcal{H}$, $\arg \lambda \in (0, \frac{2\pi}{n-1})$. It's easy to check that

$$\tau(\Theta_k^{\lambda}) \supset \begin{cases} \bigcup_{j=1}^{n-1} \Theta_j^{\lambda}, & \text{if } \chi(k) \text{ is even,} \\ \bigcup_{j=1}^{n-1} \Theta_{-j}^{\lambda}, & \text{if } \chi(k) \text{ is odd.} \end{cases}$$

As before, we define Θ^{λ} as the set of all angles in (0, 1] whose orbits remain in $\mathcal{E}^{\lambda} = \bigcup_{k=1}^{n-1} (\Theta_{k}^{\lambda} \cup \Theta_{-k}^{\lambda})$ under all iterations of τ . Thus $\Theta^{\lambda} = \bigcap_{k \geq 0} \tau^{-k}(\mathcal{E}^{\lambda})$. For $\theta \in (0, 1]$, suppose $\tau^{k}(\theta) \in \Theta_{s_{k}}^{\lambda}$ for $k \geq 0$, define the itinerary of θ by $\mathbf{s}^{\lambda}(\theta) = (s_{0}, s_{1}, s_{2}, \cdots)$. It's easy to show that the itinerary map $\mathbf{s}^{\lambda} : \Theta^{\lambda} \to \Sigma_{0}$ is bijective.

Lemma 5.3.3. (Combinatorial invariant) $\Theta^{\lambda} = \Theta$ and for any $\theta \in \Theta$, $\mathbf{s}^{\lambda}(\theta) = \mathbf{s}(\theta)$.

Proof. It suffices to show that if $\mathbf{s}^{\lambda}(\alpha) = \mathbf{s}(\beta)$ for $\alpha \in \Theta^{\lambda}$ and $\beta \in \Theta$, then $\alpha = \beta$.

First note that $\Theta_k^{\lambda} \cap \Theta_k \neq \emptyset$ for any $k \in \mathbb{I}$. Suppose $\mathbf{s}^{\lambda}(\alpha) = \mathbf{s}(\beta) = (s_0, s_1, s_2, \cdots)$, and let $A_m = \bigcap_{0 \leq k \leq m} \tau^{-k}(\Theta_{s_k}^{\lambda} \cap \Theta_{s_k})$ for $m \geq 0$. By induction argument, we see that A_m is a connected interval of the form $(a_m, b_m]$ with $a_{m+1} > a_m, b_{m+1} < b_m$ and $n(b_{m+1} - a_{m+1}) = b_m - a_m$ for $m \geq 0$. Thus $A_{m+1} \subset \overline{A_{m+1}} \subset A_m$ and $\bigcap_{k \geq 0} A_m = \bigcap_{k \geq 0} \overline{A_m}$ consists of a single point, say θ . On the other hand,

$$\{\theta\} = \bigcap_{k \ge 0} A_m = \left(\bigcap_{k \ge 0} \tau^{-k}(\Theta_{s_k}^{\lambda})\right) \bigcap \left(\bigcap_{k \ge 0} \tau^{-k}(\Theta_{s_k})\right) = \{\alpha\} \cap \{\beta\}.$$

Thus we have $\alpha = \beta = \theta$.

5.3.2 Cut rays

In this section, for any $\lambda \in \mathcal{H}$ and any $\theta \in \Theta$, we will construct a Jordan curve, say $\Omega_{\lambda}^{\theta}$, that cuts the dynamical plane of f_{λ} into two parts. The curve will meet the Julia set $J(f_{\lambda})$ in a Cantor set of points. This kind of Jordan curve $\Omega_{\lambda}^{\theta}$ will be called a cut ray of angle θ . In the following, we devote to construct such rays, but with a slightly different presentation from Devaney's in [D2].

Recall that the itinerary map $\mathbf{s}_{\lambda} : \Lambda_{\lambda} \to \Sigma$ is bijective from a Cantor set onto a symbolic space. We first extend the definition of \mathbf{s}_{λ} to a larger set. Let $\mathbf{E}_{\lambda} = \bigcap_{k \geq 0} f_{\lambda}^{-k} (\bigcup_{j \in \mathbb{I} \setminus \{0,n\}} S_j)$ be the set of all points in the dynamical plane whose orbits remain in $\bigcup_{j \in \mathbb{I} \setminus \{0,n\}} S_j$ under all iterations of f_{λ} . The map $\mathbf{s}_{\lambda} : \mathbf{\Lambda}_{\lambda} \to \Sigma$ can be extended to $\mathbf{s}_{\lambda} : \mathbf{E}_{\lambda} \to \Sigma$ as follows: For any $z \in \mathbf{E}_{\lambda}$, suppose $f_{\lambda}^{k}(z) \in S_{s_{k}}$ for $k \geq 0$, then the itinerary of z is defined by $\mathbf{s}_{\lambda}(z) = (s_{0}, s_{1}, s_{2}, \cdots)$. One can see that the map $\mathbf{s}_{\lambda} : \mathbf{E}_{\lambda} \to \Sigma$ is not welldefined for the points that are eventually mapped to ∞ . For example, under this definition, the itinerary of 0 or ∞ can be defined as any element of Σ . Even though there is some confusion of definition on the set $\mathbf{E}_{\lambda} \cap \bigcup_{k \geq 0} f_{\lambda}^{-k}(\infty)$, it's allowed to define the itinerary of $p \in \mathbf{E}_{\lambda} \cap \bigcup_{k \geq 0} f_{\lambda}^{-k}(\infty)$ as any element $(s_{0}, s_{1}, s_{2}, \cdots) \in \Sigma$ provided that $f_{\lambda}^{k}(p) \in S_{s_{k}}$ for any $k \geq 0$.

Given an angle $\theta \in \Theta$ with itinerary $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$, it's easy to check that when n is odd, $\mathbf{s}(\theta + 1/2) = (-s_0, -s_1, -s_2, \cdots) = -\mathbf{s}(\theta)$; when n is even, $\mathbf{s}(\theta + 1/2) = (-s_0, s_1, s_2, \cdots)$. We consider the set of all points in \mathbf{E}_{λ} whose itinerary is of the form $(s_0, \pm s_1, \pm s_2, \cdots)$. This set is denoted by $\omega_{\lambda}^{\theta}$. By definition, $\omega_{\lambda}^{\theta}$ contains 0 and ∞ . Moreover,

$$\omega_{\lambda}^{\theta} = \{ z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = (s_0, \pm s_1, \pm s_2, \cdots) \} = \{ z \in \mathbf{E}_{\lambda}; \kappa(\mathbf{s}_{\lambda}(z)) = \theta \}.$$

According to Devaney, the set $\omega_{\lambda}^{\theta}$ is called a full ray of angle θ . Let $\Omega^{\theta}_{\lambda} = \omega^{\theta}_{\lambda} \cup \omega^{\theta+1/2}_{\lambda}$, we call the set $\Omega^{\theta}_{\lambda}$ a cut ray of angle θ (or $\theta + 1/2$). It's obvious that

$$\Omega_{\lambda}^{\theta} = \{ z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = (\pm s_0, \pm s_1, \pm s_2, \cdots) \} = \bigcap_{k \ge 0} f_{\lambda}^{-k}(S_{s_k} \cup S_{-s_k}).$$

We first give an intuitionistic description of the cut ray $\Omega^{\theta}_{\lambda}$. For $m \geq 0$, let

$$\Omega^{\theta}_{\lambda,m} = \bigcap_{0 \le k \le m} f_{\lambda}^{-k} (S_{s_k} \cup S_{-s_k}).$$

Note that the set $\Omega^{\theta}_{\lambda,0}$ is a union of two closed sectors S_{s_0} and S_{-s_0} . $\Omega^{\theta}_{\lambda,1}$ is a string of four closed disks that lie inside $\Omega^{\theta}_{\lambda,0}$. Inductively, $\Omega^{\theta}_{\lambda,m}$ is a string of 2^{m+1} closed disks that are contained in $\Omega^{\theta}_{\lambda,m-1}$, and each of these disks meets exactly two others at the preimages of ∞ . Hence $\Omega^{\theta}_{\lambda,m}$ is a connected and compact set. One can show that $\Omega^{\theta}_{\lambda,m}$ converges to $\Omega^{\theta}_{\lambda} = \bigcap_{k \geq 0} \Omega^{\theta}_{\lambda,k}$ in Hausdorff topology as $m \to \infty$ (This is because a shrinking sequence of compact sets always converges in Hausdorff topology). Roughly speaking, the set $\Omega^{\theta}_{\lambda,m}$ becomes thinner when m becomes larger and $\Omega^{\theta}_{\lambda,m}$ finally shrinks to $\Omega^{\theta}_{\lambda}$. So it is believed that $\Omega^{\theta}_{\lambda}$ is a Jordan curve(A rigorous proof of this fact will be given in Proposition 5.3.3).

By construction, the cut ray satisfies:

- $\Omega^{\theta}_{\lambda} = -\Omega^{\theta}_{\lambda}.$
- Ω_λ^θ \ {0,∞} is contained in the interior of S_{s0} ∪ S_{-s0}.
 f_λ : Ω_λ^θ → Ω_λ^{τ(θ)} is a two-to-one map.

Lemma 5.3.4. Let $\lambda \in \mathcal{H}$, then there is a constant v > 0 such that for any $\theta \in \Theta$,

$$\overline{R_{\lambda}(\theta)} \cap \mathbf{U}(v) = \{ z \in \mathbf{E}_{\lambda} \cap \mathbf{U}(v); \mathbf{s}_{\lambda}(z) = \mathbf{s}(\theta) \}.$$

Proof. The proof is based on $\phi'_{\lambda}(\infty) = 1$ and Lemma 5.3.3. We omit the detail since it's easy.

Proposition 5.3.1. For any $\lambda \in \mathcal{H}$ and any $\theta \in \Theta$, the external ray $R_{\lambda}(\theta)$ lands at ∂B_{λ} and $R_{\lambda}(\theta) = \{z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = \mathbf{s}(\theta)\}.$

Proof. Suppose $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. Let $\ell_{\lambda}(v, \theta) = \{z \in R_{\lambda}(\theta); v \leq G_{\lambda}(z) \leq nv\}$ be the portion of $R_{\lambda}(\theta)$ that lies between two equipotential curves $\mathbf{e}(B_{\lambda}, v)$ and $\mathbf{e}(B_{\lambda}, nv)$. By Lemma 5.3.4, we may assume v large enough such that for any $\beta \in \Theta$, $\overline{R_{\lambda}(\beta)} \cap \mathbf{U}(v) = \{z \in \mathbf{E}_{\lambda} \cap \mathbf{U}(v); \mathbf{s}_{\lambda}(z) = \mathbf{s}(\beta)\}$. By pulling back $\ell_{\lambda}(v, \tau(\theta))$ by f_{λ}^{-1} to S_{s_0} , we can extend the portion of $\overline{R_{\lambda}(\theta)}$, say $\gamma_0 = \overline{R_{\lambda}(\theta)} \cap \mathbf{U}(v)$, to a longer one $\gamma_1 = h_{s_0}(\ell_{\lambda}(v, \tau(\theta))) \cup \gamma_0$. Obviously, $\gamma_1 \subset S_{s_0} \cap \overline{R_{\lambda}(\theta)}$. Continuing inductively, suppose we have already constructed a portion γ_k of $\overline{R_{\lambda}(\theta)}$, then we add a segment $h_{s_0} \circ \cdots \circ h_{s_k}(\ell_{\lambda}(v, \tau^{k+1}(\theta)))$ to γ_k and get $\gamma_{k+1} = \gamma_k \cup h_{s_0} \circ \cdots \circ h_{s_k}(\ell_{\lambda}(v, \tau^{k+1}(\theta)))$. By construction, one can check that $h_{s_0} \circ \cdots \circ h_{s_k}(\ell_{\lambda}(v, \tau^{k+1}(\theta))) \subset S_{s_0} \cap \overline{R_{\lambda}(\theta)}$ and for any $z \in h_{s_0} \circ \cdots \circ h_{s_k}(\ell_{\lambda}(v, \tau^{k+1}(\theta)))$, $\mathbf{s}_{\lambda}(z) = (s_0, s_1, s_2, \cdots)$. It turns out that $R_{\lambda}(\theta) \subset \{z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = \mathbf{s}(\theta)\}$ and

$$R_{\lambda}(\theta) \setminus \gamma_0 = \bigcup_{k \ge 0} h_{s_0} \circ \cdots \circ h_{s_k}(\ell_{\lambda}(v, \tau^{k+1}(\theta))).$$

In the following, we show the external ray $R_{\lambda}(\theta)$ lands at ∂B_{λ} . Since $h_k : \Upsilon_{\lambda} \to \Upsilon_{\lambda}$ contracts the hyperbolic metric ρ_{λ} of Υ_{λ} for any $k \in \mathbb{I} \setminus \{0, n\}$, there is a constant $\delta \in (0, 1)$ such that

$$\rho_{\lambda}(h_k(x), h_k(y)) \leq \delta \rho_{\lambda}(x, y), \quad \forall x, y \in \overline{\mathbf{V}(nv)} \cap \left(\cup_{j \in \mathbb{I} \setminus \{0, n\}} S_j \right), \forall k \in \mathbb{I} \setminus \{0, n\}.$$

Thus with respect to the hyperbolic metric of Υ_{λ} , we have

Hyper.length
$$(h_{s_0} \circ \cdots \circ h_{s_k}(\ell_{\lambda}(v, \tau^{k+1}(\theta)))) = \mathcal{O}(\delta^k).$$

This implies that $R_{\lambda}(\theta) \setminus \gamma_0$ has finite hyperbolic length in Υ_{λ} , thus the external ray $R_{\lambda}(\theta)$ lands at ∂B_{λ} . Let $p_{\lambda}(\theta)$ be the landing point. It's easy to check that $\mathbf{s}_{\lambda}(p_{\lambda}(\theta)) = \mathbf{s}(\theta)$ and $p_{\lambda}(\theta) \in \partial B_{\lambda} \cap \Lambda_{\lambda}$.

To finish, we show $R_{\lambda}(\theta) \supset \{z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = \mathbf{s}(\theta)\}$. For any $x \in \{z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = \mathbf{s}(\theta)\} \setminus \{\infty\}$, we consider the orbit of x.

If the orbit of x remains bounded, then by Lemma 5.2.3, we have $x \in \Lambda_{\lambda}$. Since $\mathbf{s}_{\lambda}|_{\Lambda_{\lambda}} : \underline{\Lambda_{\lambda}} \to \Sigma$ is bijective and $\mathbf{s}_{\lambda}(x) = \mathbf{s}_{\lambda}(p_{\lambda}(\theta)) = \mathbf{s}(\theta)$, we conclude $x = p_{\lambda}(\theta) \in \overline{R_{\lambda}(\theta)}$.

If the orbit of x tends to ∞ , then by Lemma 5.3.4, there is an integer $M \ge 1$ such that $f_{\lambda}^{M}(z) \in R_{\lambda}(\tau^{M}(\theta))$. Note that for any $k \ge 1$, $h_{s_{k-1}}(R_{\lambda}(\tau^{k}(\theta))) =$ $R_{\lambda}(\tau^{k-1}(\theta))$ and $h_{s_{k-1}}(f_{\lambda}^{k}(z)) = f_{\lambda}^{k-1}(z)$, we have $z \in R_{\lambda}(\theta)$. Thus $\overline{R_{\lambda}(\theta)} \supset$ $\{z \in \mathbf{E}_{\lambda}; \mathbf{s}_{\lambda}(z) = \mathbf{s}(\theta)\}$.

Proposition 5.3.2. For any $\lambda \in \mathcal{H}$ and any $\theta \in \Theta$ with itinerary $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$, the cut ray $\Omega^{\theta}_{\lambda}$ satisfies

1. $\Omega^{\theta}_{\lambda}$ meets the Julia set $J(f_{\lambda})$ in a Cantor set of points. More precisely, $\Omega^{\theta}_{\lambda} \cap J(f_{\lambda}) = (\kappa \circ \mathbf{s}_{\lambda}|_{\Lambda_{\lambda}})^{-1}(\{\theta, \theta + \frac{1}{2}\}).$



Figure 5.3: Combinatorial structure of a full ray $\omega_{\lambda}^{\theta}$ with $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$.

2. $\Omega^{\theta}_{\lambda}$ meets the Fatou set $F(f_{\lambda})$ in a countable union of external rays and radial rays, together with the preimages of ∞ that lie in the closure of these rays. More precisely,

$$\begin{split} \Omega_{\lambda}^{\theta} \cap B_{\lambda} &= R_{\lambda}(\theta) \cup R_{\lambda}(\theta + \frac{1}{2}) \cup \{\infty\} \\ \Omega_{\lambda}^{\theta} \cap T_{\lambda} &= \begin{cases} h_{-s_{0}}(R_{\lambda}(\tau(\theta))) \cup h_{s_{0}}(R_{\lambda}(\tau(\theta) + \frac{1}{2})) \cup \{0\}, & \text{if } n \text{ is odd,} \\ h_{s_{0}}(R_{\lambda}(\tau(\theta) + \frac{1}{2})) \cup h_{-s_{0}}(R_{\lambda}(\tau(\theta) + \frac{1}{2})) \cup \{0\}, & \text{if } n \text{ is even.} \end{cases} \end{split}$$

For any $U \in \mathcal{P} \setminus \{B_{\lambda}, T_{\lambda}\}$ with $U \cap \Omega_{\lambda}^{\theta} \neq \emptyset$, U is of the form $h_{b_0} \circ \cdots \circ h_{b_{k-1}}(T_{\lambda})$, where $k \ge 1$ and $(b_0, \dots, b_{k-1}) \in \{(\pm s_0, \dots, \pm s_{k-1})\}$. Moreover

$$\Omega_{\lambda}^{\theta} \cap U = h_{b_0} \circ \dots \circ h_{b_{k-1}} (\Omega_{\lambda}^{\tau^{\kappa}(\theta)} \cap T_{\lambda})$$

$$= \begin{cases} h_{b_0} \circ \dots \circ h_{b_{k-1}} \left(h_{-s_k}(R_{\lambda}(\tau^{k+1}(\theta))) \cup h_{s_k}(R_{\lambda}(\tau^{k+1}(\theta) + \frac{1}{2})) \cup \{0\} \right), & \text{if } n \text{ is odd,} \\ h_{b_0} \circ \dots \circ h_{b_{k-1}} \left(h_{-s_k}(R_{\lambda}(\tau^{k+1}(\theta) + \frac{1}{2})) \cup h_{s_k}(R_{\lambda}(\tau^{k+1}(\theta) + \frac{1}{2})) \cup \{0\} \right), & \text{if } n \text{ is even} \end{cases}$$

See Figure 5.3 for the combinatorial structure of a part of a cut ray.

Proof. 1. For $z \in \Omega_{\lambda}^{\theta}$, first note that $z \in \Omega_{\lambda}^{\theta} \cap J(f_{\lambda})$ if and only if the orbit of z remains bounded, if and only if $z \in \Lambda_{\lambda}$ and $\mathbf{s}_{\lambda}(z) \in \{(\pm s_0, \pm s_1, \pm s_2, \cdots)\} =$ $\kappa^{-1}(\{\theta, \theta + \frac{1}{2}\})$. Thus we have $\Omega_{\lambda}^{\theta} \cap J(f_{\lambda}) = (\kappa \circ \mathbf{s}_{\lambda}|_{\Lambda_{\lambda}})^{-1}(\{\theta, \theta + \frac{1}{2}\})$. 2. Let U be a Fatou component such that $U \cap \Omega_{\lambda}^{\theta} \neq \emptyset$. Then by 1, U is

eventually mapped onto B_{λ} .

Case 1: $U = B_{\lambda}$. By Proposition 5.3.1, $\Omega_{\lambda}^{\theta} \cap B_{\lambda} \supset R_{\lambda}(\theta) \cup R_{\lambda}(\theta + \frac{1}{2}) \cup \{\infty\}$. On the other hand, for any $z \in (\Omega_{\lambda}^{\theta} \setminus B_{\lambda}) \setminus \{\infty\}$, there is a integer $M \ge 1$ such that $f_{\lambda}^{M}(z) \in \mathbf{U}(v)$, where v is a positive constant chosen by Lemma 5.3.4. Since $\mathbf{s}_{\lambda}(f_{\lambda}^{M}(z)) \in \{(\pm s_{M}, \pm s_{M+1}, \pm s_{M+2}, \cdots)\}$, we conclude that the itinerary of $f_{\lambda}^{M}(z)$ must be identical as an itinerary of some angle $\beta \in \Theta$. Thus

$$\mathbf{s}_{\lambda}(f_{\lambda}^{M}(z)) = \begin{cases} (s_{M}, s_{M+1}, \cdots) \text{ or } (-s_{M}, -s_{M+1}, \cdots), & \text{ if } n \text{ is odd,} \\ (s_{M}, s_{M+1}, \cdots), & \text{ if } n \text{ is even.} \end{cases}$$

Case 1.1: *n* is odd. By Proposition 5.3.1, $f_{\lambda}^{M}(z) \in R_{\lambda}(\tau^{M}(\theta)) \cup R_{\lambda}(\tau^{M}(\theta) + \frac{1}{2})$. Note that $f_{\lambda}^{-1}(R_{\lambda}(\tau^{M}(\theta))) \cap (S_{s_{M-1}} \cup S_{-s_{M-1}}) \cap B_{\lambda} = R_{\lambda}(\tau^{M-1}(\theta)), f_{\lambda}^{-1}(R_{\lambda}(\tau^{M}(\theta) + \frac{1}{2})) \cap (S_{s_{M-1}} \cup S_{-s_{M-1}}) \cap B_{\lambda} = R_{\lambda}(\tau^{M-1}(\theta) + \frac{1}{2})$. We conclude $f_{\lambda}^{M-1}(z) \in R_{\lambda}(\tau^{M-1}(\theta)) \cup R_{\lambda}(\tau^{M-1}(\theta) + \frac{1}{2})$. It turns out that $z \in R_{\lambda}(\theta) \cup R_{\lambda}(\theta + \frac{1}{2})$ by induction. So in this case, $\Omega_{\lambda}^{\theta} \cap B_{\lambda} = R_{\lambda}(\theta) \cup R_{\lambda}(\theta + \frac{1}{2}) \cup \{\infty\}$.

Case 1.2: *n* is even. By Proposition 5.3.1, $f_{\lambda}^{M}(z) \in R_{\lambda}(\tau^{M}(\theta))$. Since $f_{\lambda}^{-1}(R_{\lambda}(\tau^{M}(\theta))) \cap (S_{s_{M-1}} \cup S_{-s_{M-1}}) \cap B_{\lambda} = R_{\lambda}(\tau^{M-1}(\theta)) \cup R_{\lambda}(\tau^{M-1}(\theta) + \frac{1}{2})$, we have $f_{\lambda}^{M-1}(z) \in R_{\lambda}(\tau^{M-1}(\theta)) \cup R_{\lambda}(\tau^{M-1}(\theta) + \frac{1}{2})$. If M = 1, then $z \in R_{\lambda}(\theta) \cup R_{\lambda}(\theta + \frac{1}{2})$ and the proof is done. If M > 1, then we claim $f_{\lambda}^{M-1}(z) \in R_{\lambda}(\tau^{M-1}(\theta))$. This is because $f_{\lambda}^{-1}(R_{\lambda}(\tau^{M-1}(\theta) + \frac{1}{2})) \cap (S_{s_{M-2}} \cup S_{-s_{M-2}}) \cap B_{\lambda} = \emptyset$. Again by induction, we have $z \in R_{\lambda}(\theta) \cup R_{\lambda}(\theta + \frac{1}{2})$ in this case.

Case 2: $U = T_{\lambda}$. In this case, if *n* is odd, then $f_{\lambda}(\Omega_{\lambda}^{\theta} \cap T_{\lambda} \cap S_{s_0}) = \Omega_{\lambda}^{\tau(\theta)} \cap B_{\lambda} \cap S_{-s_1} = R_{\lambda}(\tau(\theta) + \frac{1}{2}) \cup \{\infty\}$ and $f_{\lambda}(\Omega_{\lambda}^{\theta} \cap T_{\lambda} \cap S_{-s_0}) = \Omega_{\lambda}^{\tau(\theta)} \cap B_{\lambda} \cap S_{s_1} = R_{\lambda}(\tau(\theta)) \cup \{\infty\}$. So $\Omega_{\lambda}^{\theta} \cap T_{\lambda} = h_{-s_0}(R_{\lambda}(\tau(\theta))) \cup h_{s_0}(R_{\lambda}(\tau(\theta) + \frac{1}{2})) \cup \{0\}$; if *n* is even, then $f_{\lambda}(\Omega_{\lambda}^{\theta} \cap T_{\lambda} \cap S_{s_0}) = f_{\lambda}(\Omega_{\lambda}^{\theta} \cap T_{\lambda} \cap S_{-s_0}) = \Omega_{\lambda}^{\tau(\theta)} \cap B_{\lambda} \cap S_{-s_1} = R_{\lambda}(\tau(\theta) + \frac{1}{2}) \cup \{\infty\}$. So $\Omega_{\lambda}^{\theta} \cap T_{\lambda} = h_{s_0}(R_{\lambda}(\tau(\theta)) + \frac{1}{2}) \cup h_{-s_0}(R_{\lambda}(\tau(\theta) + \frac{1}{2})) \cup \{0\}$.

Case 3: $U \in \mathcal{P} \setminus \{B_{\lambda}, T_{\lambda}\}$. In this case, there is a smallest integer $k \geq 1$ such that $f_{\lambda}^{k}(U) = T_{\lambda}$. Since $f_{\lambda}^{k}: U \to T_{\lambda}$ is a conformal map and for any $0 \leq j \leq k-1, f_{\lambda}^{j}(U)$ lies inside some sector $S_{k_{j}}$, we conclude U must take the form $h_{b_{0}} \circ \cdots \circ h_{b_{k-1}}(T_{\lambda})$ for some $(b_{0}, \cdots, b_{k-1}) \in \{(\pm s_{0}, \cdots, \pm s_{k-1})\}$. By pulling back $f_{\lambda}^{k}(U \cap \Omega_{\lambda}^{\theta}) = \Omega_{\lambda}^{\tau^{k}(\theta)} \cap T_{\lambda}$ via f_{λ}^{k} , we have $\Omega_{\lambda}^{\theta} \cap U = h_{b_{0}} \circ \cdots \circ$ $h_{b_{k-1}}(\Omega_{\lambda}^{\tau^{k}(\theta)} \cap T_{\lambda})$. The conclusion follows by case 2.

Proposition 5.3.3. For any $\lambda \in \mathcal{H}$ and any $\theta \in \Theta$, the cut ray $\Omega_{\lambda}^{\theta}$ is a Jordan curve.

Proof. Suppose $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. For $k \ge 0$, define

$$\widehat{\Omega}_{\lambda,0}^{\tau^{k}(\theta)} = \Omega_{\lambda}^{\tau^{k}(\theta)} \cup \mathbf{I}_{s_{k}} \cup \mathbf{I}_{-s_{k}}, \quad \widehat{\Omega}_{\lambda,k}^{\theta} = \bigcap_{0 \le j \le k} f_{\lambda}^{-j}(\widehat{\Omega}_{\lambda,0}^{\tau^{j}(\theta)}).$$



Figure 5.4: Cut rays with angles 1/4, 1/3, 1/2 when n = 3.

The set $\widehat{\Omega}^{\theta}_{\lambda,k}$ is connected and compact and it contains $\Omega^{\theta}_{\lambda}$. It's easy to check that $\widehat{\Omega}^{\theta}_{\lambda,k} \supset \widehat{\Omega}^{\theta}_{\lambda,k+1}$ and $\bigcap_{k\geq 0} \widehat{\Omega}^{\theta}_{\lambda,k} = \Omega^{\theta}_{\lambda}$. Let D^{+}_{k} be the component of $\overline{\mathbb{C}} \setminus \widehat{\Omega}^{\theta}_{\lambda,k}$ that contains v^{+}_{λ} and D^{-}_{k} be the component of $\overline{\mathbb{C}} \setminus \widehat{\Omega}^{\theta}_{\lambda,k}$ that contains v^{-}_{λ} . Let $D^{+}_{\infty} = \bigcup_{k\geq 0} D^{+}_{k}$ and $D^{-}_{\infty} = \bigcup_{k\geq 0} D^{-}_{k}$, then $D^{+}_{\infty} \cup D^{-}_{\infty} \cup \Omega^{\theta}_{\lambda} = \overline{\mathbb{C}}$.

We first construct a Cantor set on $\mathbb{S} = \mathbb{R}/\mathbb{Z}$. Let $E_1 = (5/24, 13/24), E_2 = (17/24, 25/24)$ be two open intervals on \mathbb{S} and ζ be the map $t \mapsto 3t \mod \mathbb{Z}$. By definition, $\zeta(E_i) \supset \overline{E_1 \cup E_2}$. Let $T_k = \bigcap_{0 \le j \le k} \zeta^{-j}(E_1 \cup E_2)$. Then $T_k \supset T_{k+1}$ and T_k has 2^{k+1} components. The intersection $\bigcap_{k \ge 0} T_k$ is denoted by T_∞ . Since $T_\infty = \bigcap_{k \ge 0} \zeta^{-k}(E_1 \cup E_2) = \bigcap_{k \ge 0} \zeta^{-k}(\overline{E_1} \cup \overline{E_2})$, we conclude that T_∞ is a Cantor set.

Now, we define two sequences of Jordan curves $\{\gamma_k^+ : \mathbb{S} \to \partial D_k^+\}$, $\{\gamma_k^- : \mathbb{S} \to \partial D_k^-\}$ in the following way: for $k \ge 0$,

1. $\gamma_{k+1}^+|_{\mathbb{S}\setminus T_k} = \gamma_k^+|_{\mathbb{S}\setminus T_k} = \gamma_k^-|_{\mathbb{S}\setminus T_k} = \gamma_{k+1}^-|_{\mathbb{S}\setminus T_k}.$ 2. $\gamma_k^+(\mathbb{S}\setminus T_k) = \Omega_\lambda^\theta \cap \partial D_k^+ = \Omega_\lambda^\theta \cap \partial D_k^- = \gamma_k^-(\mathbb{S}\setminus T_k).$ 3. $\gamma_k^+(T_k) = \partial D_k^+ \setminus \Omega_\lambda^\theta, \ \gamma_k^-(T_k) = \partial D_k^- \setminus \Omega_\lambda^\theta.$

In the following, we show that each sequence of maps $\{\gamma_k^+ : \mathbb{S} \to \partial D_k^+\}$, $\{\gamma_k^- : \mathbb{S} \to \partial D_k^-\}$ converges in the spherical metric. By construction, $\gamma_{k+1}^+|_{\mathbb{S}\setminus T_k} = \gamma_k^+|_{\mathbb{S}\setminus T_k}$ and for any component W of T_k , $\gamma_{k+1}^+(W)$ and $\gamma_k^+(W)$ are contained in the same component of $\bigcap_{0 \leq j \leq k} f_{\lambda}^{-j}(\mathbf{I}_{s_j} \cup \mathbf{I}_{-s_j})$. Since the spherical metric and the hyperbolic metric are comparable in any compact subset
of Υ_{λ} , we conclude by Lemma 5.2.3 that

$$\max_{t \in \mathbb{S}} \operatorname{dist}_{\overline{\mathbb{C}}} \left(\gamma_{k+1}^+(t), \gamma_k^+(t) \right) = \mathcal{O}(\delta^k),$$

where dist_{$\overline{\mathbb{C}}$} is the spherical metric and $\delta \in (0,1)$ is a constant. Thus the sequence $\{\gamma_k^+\}$ has a limit map $\gamma_{\infty}^+ : \mathbb{S} \to \partial D_{\infty}^+$ which is continuous and surjective. Similarly, the sequence $\{\gamma_k^-\}$ also has a limit map $\gamma_{\infty}^- : \mathbb{S} \to \partial D_{\infty}^-$, continuous and surjective. The limit maps γ_{∞}^+ and γ_{∞}^- satisfy $\gamma_{\infty}^+|_{\mathbb{S}\setminus T_{\infty}} = \gamma_{\infty}^-|_{\mathbb{S}\setminus T_{\infty}}$. By continuity, γ_{∞}^+ and γ_{∞}^- are identical on \mathbb{S} . This implies that $\partial D_{\infty}^+ = \partial D_{\infty}^- = \Omega_{\lambda}^{\theta}$ and $\Omega_{\lambda}^{\theta}$ is locally connected.

To finish, we show $\Omega^{\theta}_{\lambda}$ is Jordan curve. Let $\Phi : \mathbb{D} \to D^{+}_{\infty}$ be a Riemann mapping. Since ∂D^{+}_{∞} is locally connected, Φ has an extension from $\overline{\mathbb{D}}$ to $\overline{D^{+}_{\infty}}$. If two distinct radical segments $\Phi((0,1)e^{2\pi i\theta_1})$ and $\Phi((0,1)e^{2\pi i\theta_2})$ converge to the same point p, then the Jordan curve $\Phi((0,1)e^{2\pi i\theta_1}) \cup \Phi((0,1)e^{2\pi i\theta_2}) \cup$ $\{\Phi(0), p\}$ separates a section of the boundary ∂D^{+}_{∞} from D^{-}_{∞} . But this is a contradiction since D^{+}_{∞} and D^{-}_{∞} share a common boundary. \Box

Proposition 5.3.4. For $\lambda \in \mathcal{H}$ and $\theta \in \Theta$, all periodic points on $\Omega_{\lambda}^{\theta} \cap J(f_{\lambda})$ are repelling.

Proof. Suppose $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. Let $z \in \Omega_{\lambda}^{\theta} \cap J(f_{\lambda})$ be a periodic point, with period p. Then the itinerary of z is of the form $(\overline{a_0, a_1, \cdots, a_{p-1}})$, where $a_j \in \{\pm s_j\}$ for $0 \leq j \leq p-1$. Let $a_k = a_k \mod p$ for $k \geq 0$ and $\mathbf{I}_{a_0 \cdots a_s} = \bigcap_{0 \leq k \leq s} f_{\lambda}^{-k}(\mathbf{I}_{a_k})$. By Lemma 5.2.3, the hyperbolic diameter of $\mathbf{I}_{a_0 \cdots a_s}$ is $O(\delta^s)$ when s is large. So we can choose N large enough such that f_{λ}^p : $\operatorname{int}(\mathbf{I}_{a_0 \cdots a_N}) \to \operatorname{int}(\mathbf{I}_{a_p \cdots a_N}) = \operatorname{int}(\mathbf{I}_{a_0 \cdots a_{N-p}})$ is a conformal map. Since $z \in$ $\operatorname{int}(\mathbf{I}_{a_0 \cdots a_N}) \subset \mathbf{I}_{a_0 \cdots a_N} \subset \operatorname{int}(\mathbf{I}_{a_0 \cdots a_{N-p}})$, we conclude $|(f_{\lambda}^p)'(z)| > 1$ by Schwarz Lemma. Thus z is a repelling periodic point.

Proposition 5.3.2 tells us the combinatorial structure of the cut ray $\Omega_{\lambda}^{\theta}$. The following proposition shows that the iterated preimages of $\Omega_{\lambda}^{\theta}$ have the same combinatorial structure as $\Omega_{\lambda}^{\theta}$ provided that $\Omega_{\lambda}^{\theta}$ doesn't meet the critical orbit.

Proposition 5.3.5. For $\lambda \in \mathcal{H}$ and $\theta \in \Theta$, suppose the cut ray $\Omega_{\lambda}^{\theta}$ doesn't meet the critical orbit, then for any $\alpha \in \bigcup_{k\geq 0} \tau^{-k}(\theta)$, there is a unique ray $\omega_{\lambda}^{\alpha}$ such that

1. $\omega_{\lambda}^{\alpha}$ is a continuous curve connecting 0 with ∞ . 2. $\omega_{\lambda}^{\alpha+1/2} = -\omega_{\lambda}^{\alpha}$. 3. $f_{\lambda}(\omega_{\lambda}^{\alpha}) = \omega_{\lambda}^{\tau(\alpha)} \cup \omega_{\lambda}^{\tau(\alpha)+1/2}$. 4. $\omega_{\lambda}^{\alpha} \cap B_{\lambda} = R_{\lambda}(\alpha) \cup \{\infty\}$.

For this reason, we still call $\omega_{\lambda}^{\alpha}$ a full ray of angle α , and $\Omega_{\lambda}^{\alpha} = \omega_{\lambda}^{\alpha} \cup \omega_{\lambda}^{\alpha+1/2}$ a cut ray of angle α (or $\alpha + \frac{1}{2}$). *Proof.* The proof is based on induction argument. Suppose $\alpha \in \bigcup_{k\geq 0} \tau^{-k}(\theta)$ is an angle such that the full ray $\omega_{\lambda}^{\alpha}$ and the cut ray $\Omega_{\lambda}^{\alpha}$ satisfy 1,2,3,4. Then for $\beta \in \tau^{-1}(\alpha)$, we define ω_{λ}^{β} by lifting $\Omega_{\lambda}^{\alpha}$ in the following way:

$$f_{\lambda}(\omega_{\lambda}^{\beta}) = \Omega_{\lambda}^{\alpha}, \ \ \omega_{\lambda}^{\beta} \cap B_{\lambda} = R_{\lambda}(\beta) \cup \{\infty\}.$$

This ray ω_{λ}^{β} is unique since we require $\omega_{\lambda}^{\beta} \cap B_{\lambda} = R_{\lambda}(\beta) \cup \{\infty\}$. Also by uniqueness of lifting maps, we conclude $\omega_{\lambda}^{\beta+\frac{1}{2}} = -\omega_{\lambda}^{\beta}$ by the fact $R_{\lambda}(\beta+\frac{1}{2}) = -R_{\lambda}(\beta)$ and $\Omega_{\lambda}^{\alpha} = -\Omega_{\lambda}^{\alpha}$.

In the following, we show that ω_{λ}^{β} connects ∞ and 0. If not, then ω_{λ}^{β} must be a curve connecting ∞ with itself, hence a Jordan curve. This implies that ω_{λ}^{β} doesn't meet with 0. Since $\Omega_{\lambda}^{\alpha} = -\Omega_{\lambda}^{\alpha}$, all curves in the set $\mathcal{C} = \{e^{k\pi i/n}\omega_{\lambda}^{\beta}, H_{\lambda}(e^{k\pi i/n}\omega_{\lambda}^{\beta}); 0 \leq k < 2n\}$ are preimages of $\Omega_{\lambda}^{\alpha}$, where $H_{\lambda}(z) = \sqrt[n]{\lambda}/z$. Since $\Omega_{\lambda}^{\alpha}$ doesn't meet the critical orbit, we conclude that for any $\gamma_{1}, \gamma_{2} \in \mathcal{C}$ with $\gamma_{1} \neq \gamma_{2}, \gamma_{1}$ and γ_{2} are disjoint outside $\{0, \infty\}$. This means $\#\mathcal{C} = 4n$. But this is a contradiction since the degree of f_{λ} is 2n.

Recall that for any $\theta \in \Theta$ with itinerary $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$, the cut ray $\Omega^{\theta}_{\lambda}$ contains at least two points: 0 and ∞ , and $\Omega^{\theta}_{\lambda} \setminus \{0, \infty\}$ is contained in the interior of $S_{s_0} \cup S_{-s_0}$. Now given two angles $\alpha, \beta \in \Theta$ with $\Omega^{\alpha}_{\lambda} \neq \Omega^{\beta}_{\lambda}$, suppose $\mathbf{s}(\alpha) = (s_0^{\alpha}, s_1^{\alpha}, s_2^{\alpha}, \cdots)$, $\mathbf{s}(\beta) = (s_0^{\beta}, s_1^{\beta}, s_2^{\beta}, \cdots)$. Let $\mathbf{J}(\alpha, \beta)$ be the first integer $k \geq 0$ such that $|s_k^{\alpha}| \neq |s_k^{\beta}|$. Note that the intersection $\Omega^{\alpha}_{\lambda} \cap \Omega^{\beta}_{\lambda}$ consists of at least two points 0 and ∞ . If furthermore $\mathbf{J}(\alpha, \beta) = 0$, then $\Omega^{\alpha}_{\lambda} \cap \Omega^{\beta}_{\lambda} = \{0, \infty\}$. The following proposition tells us the number of intersection points in general case.

Proposition 5.3.6. Let $\alpha, \beta \in \Theta$ with $\Omega_{\lambda}^{\alpha} \neq \Omega_{\lambda}^{\beta}$, then the intersection $\Omega_{\lambda}^{\alpha} \cap \Omega_{\lambda}^{\beta}$ consists of $2^{\mathbf{J}(\alpha,\beta)+1}$ points.

Proof. We consider the orbit of $\Omega_{\lambda}^{\alpha} \cap \Omega_{\lambda}^{\beta}$ under f_{λ} :

$$\Omega_{\lambda}^{\alpha} \cap \Omega_{\lambda}^{\beta} \to \Omega_{\lambda}^{\tau(\alpha)} \cap \Omega_{\lambda}^{\tau(\beta)} \to \dots \to \Omega_{\lambda}^{\tau^{\mathbf{J}(\alpha,\beta)}(\alpha)} \cap \Omega_{\lambda}^{\tau^{\mathbf{J}(\alpha,\beta)}(\beta)}$$

Note that for any $0 \leq k \leq \mathbf{J}(\alpha, \beta) - 1$, $f_{\lambda} : \Omega_{\lambda}^{\tau^{k}(\alpha)} \cap \Omega_{\lambda}^{\tau^{k+1}(\alpha)} \to \Omega_{\lambda}^{\tau^{k+1}(\alpha)} \cap \Omega_{\lambda}^{\tau^{k+1}(\beta)}$ is a two-to-one map, thus we have

$$\#(\Omega_{\lambda}^{\alpha} \cap \Omega_{\lambda}^{\beta}) = 2\#(\Omega_{\lambda}^{\tau(\alpha)} \cap \Omega_{\lambda}^{\tau(\beta)}) = \dots = 2^{\mathbf{J}(\alpha,\beta)} \#(\Omega_{\lambda}^{\tau^{\mathbf{J}(\alpha,\beta)}(\alpha)} \cap \Omega_{\lambda}^{\tau^{\mathbf{J}(\alpha,\beta)}(\beta)}) = 2^{\mathbf{J}(\alpha,\beta)+1}$$

Remark 5.3.3. From the proof of Proposition 5.3.6, we know that any two different cut rays $\Omega_{\lambda}^{\alpha}$ and Ω_{λ}^{β} intersect at the preimages of ∞ . More precisely, $\Omega_{\lambda}^{\alpha} \cap \Omega_{\lambda}^{\beta} \subset \bigcup_{0 \le k \le \mathbf{J}(\alpha,\beta)+1} f_{\lambda}^{-k}(\infty)$, and for $2 \le k \le \mathbf{J}(\alpha,\beta)+1$, the intersection $\Omega_{\lambda}^{\alpha} \cap \Omega_{\lambda}^{\beta} \cap f_{\lambda}^{-(k-1)}(0)$ consists of 2^{k-1} points.

5.4 Puzzles, Graphs and Tableaux

5.4.1 The Yoccoz Puzzle

Let $X_{\lambda} = \mathbb{C} \setminus \{z \in B_{\lambda}; G_{\lambda}(z) \ge 1\} = \mathbf{V}(1)$. Given N periodic angles $\theta_1, \dots, \theta_N$ that lie in different periodic cycles of Θ , let

$$g_{\lambda}(\theta_1,\cdots,\theta_N) = \bigcup_{k\geq 0} \left(\Omega_{\lambda}^{\tau^k(\theta_1)}\cup\cdots\cup\Omega_{\lambda}^{\tau^k(\theta_N)}\right).$$

Obviously, $g_{\lambda}(\theta_1, \dots, \theta_N)$ is f_{λ} -invariant. The graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ generated by $\theta_1, \dots, \theta_N$ is defined as following:

$$\mathbf{G}_{\lambda}(\theta_{1},\cdots,\theta_{N})=\partial X_{\lambda}\cup\Big(X_{\lambda}\cap g_{\lambda}(\theta_{1},\cdots,\theta_{N})\Big).$$

The Yoccoz Puzzle induced by the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is constructed in the following way: The Yoccoz Puzzle of depth zero consists of all connected components of $X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$, and each component is called a puzzle piece of depth zero. The Yoccoz Puzzle of greater depth can be constructed by induction: If $P_d^{(1)}, \dots, P_d^{(m)}$ are the puzzle pieces of depth d, then the connected components of the set $f_{\lambda}^{-1}(P_d^{(i)})$ are the puzzle pieces $P_{d+1}^{(j)}$ of depth d+1. One can verify that the puzzle pieces of depth d consists of all connected components of $f_{\lambda}^{-d}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N))$ and each puzzle piece is a disk.

To make the puzzle well-defined, we should avoid the situation that the critical orbits touch the set $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. If the critical orbits touch the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$, we say the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is touchable. In this case, since there are infinitely many periodic angles in Θ , we can change the *N*-tuple $(\theta_1, \dots, \theta_N)$ to another *N*-tuple $(\theta'_1, \dots, \theta'_N)$ to make the graph not touchable. So in the following discussion, we always assume the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is not touchable.

Let J_0 be the set of all points on the Julia set $J(f_{\lambda})$ whose orbits eventually meet the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. Then $J_0 = \bigcup_{k \ge 0} f_{\lambda}^{-k}(\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N) \cap J(f_{\lambda}))$. For any $z \in \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$, there is a unique sequence of puzzle pieces $P_0(z) \supset$ $P_1(z) \supset P_2(z) \supset \cdots$ which contain z. By Proposition 5.3.4, if f_{λ} has a nonrepelling cycle in \mathbb{C} , say $\mathcal{C} = \{z, f_{\lambda}(z), \dots, f_{\lambda}^p(z) = z\}$, then this cycle must avoid the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. This implies that $\mathcal{C} \subset \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$. Thus for any $d \ge 0$ and any $x \in \mathcal{C}$, the puzzle piece $P_d(x)$ is well defined.

Lemma 5.4.1. Suppose the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is not touchable, then for any $z \in \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$, the puzzle pieces satisfy:

$$-P_0(z) = P_0(-z); \quad \omega P_d(z) = P_d(\omega z), \ \omega^{2n} = 1, \ d \ge 1.$$



Figure 5.5: A graph with Yoccoz puzzle to depth one $(n = 3 \text{ and } \mathbf{G}_{\lambda} = \mathbf{G}_{\lambda}(1/2))$.

Proof. By definition of the graph $\mathbf{G}_{\lambda}(\theta_1, \cdots, \theta_N)$ and the symmetry of the Green function $G_{\lambda} : A_{\lambda} \to (0, +\infty]$ (See Lemma 5.2.1), we have $X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_1, \cdots, \theta_N) = -X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_1, \cdots, \theta_N)$. Thus $-P_0(z) = P_0(-z)$. Suppose that for some $d \geq 0$,

$$f_{\lambda}^{-d}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_{1}, \cdots, \theta_{N})) = -f_{\lambda}^{-d}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_{1}, \cdots, \theta_{N})).$$

Since $f_{\lambda}(\omega z) = \pm f_{\lambda}(z)$ and $G_{\lambda}(\omega z) = G_{\lambda}(z)$, we have $f_{\lambda}(z) \in f_{\lambda}^{-d}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N))$ if and only if $f_{\lambda}(\omega z) \in f_{\lambda}^{-d}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N))$. Thus

$$f_{\lambda}^{-(d+1)}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_{1}, \cdots, \theta_{N})) = \omega f_{\lambda}^{-(d+1)}(X_{\lambda} \setminus \mathbf{G}_{\lambda}(\theta_{1}, \cdots, \theta_{N})).$$

The conclusion follows by induction.

Lemma 5.4.2. Suppose the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is not touchable, then for any $d \geq 0$ and any puzzle piece P_d of depth d, the intersection $\overline{P}_d \cap J(f_{\lambda})$ is connected.

Proof. It's equivalent to prove that every connected component of $\overline{\mathbb{C}} \setminus (\overline{P}_d \cap J(f_\lambda))$ is simply connected. Since the Julia set $J(f_\lambda)$ is connected, every component of $\overline{\mathbb{C}} \setminus (\overline{P}_d \cap J(f_\lambda))$ that lies inside P_d is simply connected. So we just need consider the components of $\overline{\mathbb{C}} \setminus (\overline{P}_d \cap J(f_\lambda))$ that intersect with ∂P_d . Note that the puzzle piece P_d is bounded by finitely many cut rays, say $\Omega_{\lambda}^{\beta_1}, \dots, \Omega_{\lambda}^{\beta_s}$, together with finitely many equipotential curves $\mathbf{e}(U_1, v), \dots, \mathbf{e}(U_t, v)$. By the structure of cut rays (Proposition 5.3.2), there is exactly one component of

 $\overline{\mathbb{C}} \setminus (\overline{P}_d \cap J(f_\lambda))$ that intersects with the boundary ∂P_d . This component is the union of $\overline{\mathbb{C}} \setminus \overline{P}_d$ and countably many Fatou components that intersect with the cut rays $\Omega_{\lambda}^{\beta_1}, \cdots, \Omega_{\lambda}^{\beta_s}$. Thus it's also simply connected. \Box

5.4.2 Admissible graphs

Given a point $z \in \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$, the difference set $A_d(z) = P_d(z) \setminus P_{d+1}(z)$ is an annulus, either degenerate or of positive modulus. Here, d is called the depth of $A_d(z)$. For $d \geq 1$ and $c \in C_{\lambda}$, the annulus $A_d(z)$ is called off-critical, c-critical or c-semi-critical if $P_d(z)$ contains no critical points, $P_{d+1}(z)$ contains the critical point c or $A_d(z)$ contains the critical point c, respectively.

Since the critical annuli play a crucial rule in our discussion, we will devote ourself to finding a graph such that with respect to the Yoccoz puzzle induced by such a graph, the critical annulus $A_d(c)$ is non-degenerate for some $d \ge$ 1. By Lemma 5.4.1, if some critical annulus $A_d(c)$ of depth $d \ge 1$ is nondegenerate, then all critical annuli of the same depth are non-degenerate. The graph that satisfies this property is of special favourite.

Definition 5.4.1. We say the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is admissible if with respect to the Yoccoz puzzle induced by $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$, there exists a nondegenerate critical annulus $A_d(c)$ for some critical point $c \in C_{\lambda}$ and some depth $d \geq 1$. Else, we say the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is non-admissible.

The following remark tells us that a graph may be non-admissible in some cases.

Remark 5.4.1. There exist non-admissible graphs. For example, for any $n \geq 3$, suppose f_{λ} is 1-renormalizable at c_0 (See Section 5.5 for definition), then the graph $\mathbf{G}_{\lambda}(1)$ is non-admissible since $A_d(c_0)$ is degenerate for all depths $d \geq 1$, see Figure 5.5.

However, even if there are non-admissible graphs, we can always find an admissible graph by an elaborate choice. The aim of this section is to prove the existence of admissible graphs for $n \geq 3$.

Proposition 5.4.1. For any $n \geq 3$ and any $\lambda \in \mathcal{H}$, if f_{λ} is not critically finite, then there always exists an admissible graph.

The proof is divided into three lemmas: Lemma 5.4.3, Lemma 5.4.4 and Lemma 5.4.5. In fact, these lemmas enable us to prove much more: when $n \geq 5$, there exist infinitely many admissible graphs without the assumption of the critical finiteness of f_{λ} (See Lemma 5.4.5); when n = 4, there exists at least one admissible graph without the assumption of the critical finiteness of f_{λ} (See Lemma 5.4.4); when n = 3, there exists at least one admissible graph except some particular critically finite cases (See Lemma 5.4.3). **Lemma 5.4.3.** When n = 3, there exists an admissible graph except when the critical orbit of f_{λ} eventually lands at a repelling cycle of period one or two. More precisely,

1. If neither $\mathbf{G}_{\lambda}(1/4)$ nor $\mathbf{G}_{\lambda}(1/2)$ is touchable, then at least one of the graphs $\mathbf{G}_{\lambda}(1/4)$, $\mathbf{G}_{\lambda}(1/2)$, $\mathbf{G}_{\lambda}(1/4, 1/2)$ is admissible.

2. If $\mathbf{G}_{\lambda}(1/2)$ is touchable, then either $\mathbf{G}_{\lambda}(1/4)$ is admissible or the critical orbit of f_{λ} eventually lands at a repelling cycle of period two.

3. If $\mathbf{G}_{\lambda}(1/4)$ is touchable, then either $\mathbf{G}_{\lambda}(1/2)$ is admissible or the critical orbit of f_{λ} eventually lands at a repelling fixed point.

Proof. First note that

$$f_{\lambda}^{-1}(\Omega_{\lambda}^{1/4}) = \Omega_{\lambda}^{1/12} \cup \Omega_{\lambda}^{1/4} \cup \Omega_{\lambda}^{5/12}, \ f_{\lambda}^{-1}(\Omega_{\lambda}^{1/2}) = \Omega_{\lambda}^{1/6} \cup \Omega_{\lambda}^{1/3} \cup \Omega_{\lambda}^{1/2}.$$

1. In this case, the full rays $\omega_{\lambda}^{1/12}$ and $\omega_{\lambda}^{1/6}$ decompose S_0 into four domains: D_1, D_2, D_3 and D_4 , see Figure 6. If neither $\mathbf{G}_{\lambda}(1/4)$ nor $\mathbf{G}_{\lambda}(1/2)$ is touchable, then the graphs $\mathbf{G}_{\lambda}(1/4)$, $\mathbf{G}_{\lambda}(1/2)$, $\mathbf{G}_{\lambda}(1/4, 1/2)$ are all well-defined. Now, we consider the location of the critical value v_{λ}^+ , there are four possibilities:

Case 1: $v_{\lambda}^{+} \in D_1$. In this case, the annulus $A_0(v_{\lambda}^{+}) = P_0(v_{\lambda}^{+}) \setminus \overline{P_1(v_{\lambda}^{+})}$ is non-degenerate with respect to the Yoccoz puzzle induced by either of the graphs $\mathbf{G}_{\lambda}(1/4)$, $\mathbf{G}_{\lambda}(1/2)$ and $\mathbf{G}_{\lambda}(1/4, 1/2)$. It turns out that the critical annulus $A_1(c)$ is non-degenerate for all $c \in C_{\lambda}$. Thus, in this case, all the graphs $\mathbf{G}_{\lambda}(1/4)$, $\mathbf{G}_{\lambda}(1/2)$, $\mathbf{G}_{\lambda}(1/4, 1/2)$ are admissible.

Case 2: $v_{\lambda}^{+} \in D_{2}$. The annulus $A_{0}(v_{\lambda}^{+}) = P_{0}(v_{\lambda}^{+}) \setminus P_{1}(v_{\lambda}^{+})$ is nondegenerate with respect to the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(1/4)$. So all critical annuli $A_{1}(c)$ are non-degenerate. Thus the graph $\mathbf{G}_{\lambda}(1/4)$ is admissible.

Case 3: $v_{\lambda}^+ \in D_3$. The annulus $A_0(v_{\lambda}^+)$ is non-degenerate with respect to the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(1/4, 1/2)$. So all critical annuli $A_1(c)$ are non-degenerate and the graph $\mathbf{G}_{\lambda}(1/4, 1/2)$ is admissible.

Case 4: $v_{\lambda}^+ \in D_4$. Similar argument as above, we conclude the graph $\mathbf{G}_{\lambda}(1/2)$ is admissible.

2. In this case, the graph $\mathbf{G}_{\lambda}(1/4)$ is necessarily non-touchable. First note that the cut ray $\Omega_{\lambda}^{5/12}$ decomposes $\Omega_{\lambda}^{1/2}$ into four parts: $\Omega_{\lambda}^{1/2}(2,2)$, $\Omega_{\lambda}^{1/2}(2,-2)$, $\Omega_{\lambda}^{1/2}(-2,2)$ and $\Omega_{\lambda}^{1/2}(-2,-2)$, where

$$\Omega_{\lambda}^{1/2}(\varepsilon_0,\varepsilon_1) = \{ z \in \Omega_{\lambda}^{1/2}; \mathbf{s}_{\lambda}(z) = (\varepsilon_0,\varepsilon_1,\pm 2,\pm 2,\cdots) \}, \quad (\varepsilon_0,\varepsilon_1) \in \{(\pm 2,\pm 2)\}.$$

Moreover, for any $z \in (\Omega_{\lambda}^{1/2}(2,2) \cup \Omega_{\lambda}^{1/2}(-2,-2)) \cap J(f_{\lambda})$, the annulus $A_0(z)$ is non-degenerate with respect to the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(1/4)$.



Figure 5.6: Candidates for admissible graph when n = 3.

Since $\mathbf{G}_{\lambda}(1/2)$ is touchable, there is an integer $p \geq 1$ and a critical point $c \in C_{\lambda}$ such that $f_{\lambda}^{p}(c) \in \Omega_{\lambda}^{1/2}$. Consider the itinerary of $f_{\lambda}^{p}(c)$, say $\mathbf{s}_{\lambda}(f_{\lambda}^{p}(c)) = (s_{0}, s_{1}, s_{2}, \cdots)$. There are two possibilities:

Case 1. There is an integer $n \ge 0$ such that $(s_n, s_{n+1}) = (2, 2)$ or (-2, -2). In this case, $f_{\lambda}^{n+p}(c) \in (\Omega_{\lambda}^{1/2}(2, 2) \cup \Omega_{\lambda}^{1/2}(-2, -2)) \cap J(f_{\lambda})$, thus the annulus $A_0(f_{\lambda}^{n+p}(c))$ is non-degenerate. It turns out that the critical annulus $A_{n+p}(c)$ is non-degenerate. So the graph $\mathbf{G}_{\lambda}(1/4)$ is admissible.

Case 2. For any integer $n \ge 0$, $(s_n, s_{n+1}) = (2, -2)$ or (-2, 2). In this case, either $\mathbf{s}_{\lambda}(f_{\lambda}^p(c)) = (2, -2, 2, -2, \cdots) = (\overline{2, -2})$ or $\mathbf{s}_{\lambda}(f_{\lambda}^p(c)) = (-2, 2, -2, 2, \cdots) = (\overline{-2, 2})$. By Proposition 5.3.4, $f_{\lambda}^p(c)$ lies in a repelling cycle of period two.

3. The proof is similar as 2. In this case, the graph $\mathbf{G}_{\lambda}(1/2)$ is necessarily non-touchable. First note that the cut ray $\Omega_{\lambda}^{1/3}$ decomposes $\Omega_{\lambda}^{1/4}$ into four parts: $\Omega_{\lambda}^{1/4}(1,-1)$, $\Omega_{\lambda}^{1/4}(1,1)$, $\Omega_{\lambda}^{1/4}(-1,-1)$ and $\Omega_{\lambda}^{1/4}(-1,1)$, where

$$\Omega_{\lambda}^{1/4}(\varepsilon_0,\varepsilon_1) = \{ z \in \Omega_{\lambda}^{1/4}; \mathbf{s}_{\lambda}(z) = (\varepsilon_0,\varepsilon_1,\pm 1,\pm 1,\cdots) \}, \quad (\varepsilon_0,\varepsilon_1) \in \{(\pm 1,\pm 1)\}.$$

Moreover, for any $z \in (\Omega_{\lambda}^{1/4}(1,-1) \cup \Omega_{\lambda}^{1/4}(-1,1)) \cap J(f_{\lambda})$, the annulus $A_0(z)$ is non-degenerate with respect to the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(1/2)$.

Since $\mathbf{G}_{\lambda}(1/4)$ is touchable, there is an integer $p \geq 1$ and a critical point $c \in C_{\lambda}$ such that $f_{\lambda}^{p}(c) \in \Omega_{\lambda}^{1/4}$. Consider the itinerary of $f_{\lambda}^{p}(c)$, say $\mathbf{s}_{\lambda}(f_{\lambda}^{p}(c)) = (s_{0}, s_{1}, s_{2}, \cdots)$. There are two possibilities:



Figure 5.7: Candidates for admissible graph when n = 4.

Case 1. There is an integer $n \ge 0$ such that $(s_n, s_{n+1}) = (-1, 1)$ or (1, -1). In this case, $f_{\lambda}^{n+p}(c) \in (\Omega_{\lambda}^{1/4}(1, -1) \cup \Omega_{\lambda}^{1/4}(-1, 1)) \cap J(f_{\lambda})$, thus the annulus $A_0(f_{\lambda}^{n+p}(c))$ is non-degenerate. It turns out that the critical annulus $A_{n+p}(c)$ is non-degenerate. So the graph $\mathbf{G}_{\lambda}(1/2)$ is admissible.

Case 2. For any integer $n \ge 0$, $(s_n, s_{n+1}) = (1, 1)$ or (-1, -1). In this case, either $\mathbf{s}_{\lambda}(f_{\lambda}^p(c)) = (1, 1, \cdots) = (\overline{1})$ or $\mathbf{s}_{\lambda}(f_{\lambda}^p(c)) = (-1, -1, \cdots) = (\overline{-1})$. By Proposition 5.3.4, $f_{\lambda}^p(c)$ is a repelling fixed point.

Lemma 5.4.4. When n = 4, if $\mathbf{G}_{\lambda}(1/3)$ is not touchable, then $\mathbf{G}_{\lambda}(1/3)$ is admissible; if $\mathbf{G}_{\lambda}(1/3)$ is touchable, then $\mathbf{G}_{\lambda}(2/3, 1)$ is admissible.

Proof. First note that $\mathbf{s}(1/3) = (2, 2, \dots) = (\overline{2})$, $\mathbf{s}(2/3) = (-1, -1, \dots) = (\overline{-1})$ and $\mathbf{s}(1) = (-3, -3, \dots) = (\overline{-3})$. Thus $\Omega_{\lambda}^{1/3} \subset S_2 \cup S_{-2}, \Omega_{\lambda}^{2/3} \subset S_1 \cup S_{-1}$ and $\Omega_{\lambda}^1 \subset S_3 \cup S_{-3}$, see Figure 7. It's easy to verify

$$f_{\lambda}^{-1}(\Omega_{\lambda}^{1/3}) = \Omega_{\lambda}^{1/12} \cup \Omega_{\lambda}^{5/24} \cup \Omega_{\lambda}^{1/3} \cup \Omega_{\lambda}^{11/24}.$$

If the graph $\mathbf{G}_{\lambda}(1/3)$ is not touchable, then with respect to the Yoccoz puzzle induced by $\mathbf{G}_{\lambda}(1/3)$, the puzzle piece $P_1(v_{\lambda}^+)$ is a subset of the domain bounded by $\omega_{\lambda}^{5/24}$ and $\omega_{\lambda}^{23/24}$ together with the equipotential curves $\mathbf{e}(B_{\lambda}, 1/n)$ and $\mathbf{e}(T_{\lambda}, 1/n)$. Thus the annulus $A_0(v_{\lambda}^+)$ is non-degenerate. It turns out that all critical annuli $A_1(c)$ are non-degenerate. So the graph $\mathbf{G}_{\lambda}(1/3)$ is admissible. If the graph $\mathbf{G}_{\lambda}(1/3)$ is touchable, then there exist an integer $p \geq 1$ and a critical point $c \in C_{\lambda}$ such that $f_{\lambda}^p(c) \in \Omega_{\lambda}^{1/3}$. Note that the



Figure 5.8: Candidates for admissible graph when $n \geq 5$.

preimage of $\Omega_{\lambda}^{2/3}$ that lies in $S_2 \cup S_{-2}$ is $\Omega_{\lambda}^{7/24}$ and the preimage of Ω_{λ}^1 that lies in $S_2 \cup S_{-2}$ is $\Omega_{\lambda}^{3/8}$. In this case, with respect to the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(2/3, 1)$, the puzzle piece $P_1(f_{\lambda}^p(c))$ is bounded by $\Omega_{\lambda}^{7/24}$ and $\Omega_{\lambda}^{3/8}$, thus the annulus $A_0(f_{\lambda}^p(c))$ is non-degenerate. So all critical annuli $A_p(c)$ are non-degenerate and the graph $\mathbf{G}_{\lambda}(2/3, 1)$ is admissible.

Lemma 5.4.5. When $n \ge 5$, there are infinitely many periodic angles $\theta \in \Theta$ such that the graph $\mathbf{G}_{\lambda}(\theta)$ is admissible.

Proof. Let $\widehat{\Theta} = \bigcup_{j\geq 0} \tau^{-j} \left(\bigcup_{2\leq k\leq n-2} (\Theta_k \cup \Theta_{-k}) \right)$ be the set of all angles in Θ whose orbits remain in $\bigcup_{2\leq k\leq n-2} (\Theta_k \cup \Theta_{-k})$ under all iterations of τ and let $\widehat{\Theta}_{per}$ be the set of all periodic angles in $\widehat{\Theta}$. Similar argument as Lemma 5.3.2, we can show that $\widehat{\Theta}_{per}$ is a dense subset of $\widehat{\Theta}$. By Lemma 5.3.1, one can check that the set $\widehat{\Theta}_{per}$ can be written as

$$\widehat{\Theta}_{per} = \bigcup_{p \ge 1} \{ \kappa(\mathbf{s}); \mathbf{s} = (\overline{s_0, \cdots, s_{p-1}}) \in \Sigma_0 \text{ and } s_0, \cdots, s_{p-1} \in \{ \pm 2, \cdots, \pm (n-2) \} \}$$

and any angle $\theta \in \widehat{\Theta}_{per}$ is of the form

$$\theta = \frac{1}{2} \left(\frac{\chi(s_0)}{n} + \frac{|s_0|}{n(n^p - 1)} + \frac{n^p}{n^p - 1} \sum_{1 \le k < p} \frac{|s_k|}{n^{k+1}} \right).$$

We can choose an angle $\theta \in \widehat{\Theta}_{per}$ such that the critical orbit avoids the graph $\mathbf{G}_{\lambda}(\theta)$ (Note that there are infinitely many such choices of angle θ). Then with respect to the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(\theta)$, $\overline{P_1(v_{\lambda}^+)}$ is a proper subset of $P_0(v_{\lambda}^+)$, thus the graph $\mathbf{G}_{\lambda}(\theta)$ is admissible. See Figure 8.

In the rest of this section, we prove an important property of the cut rays that are used to generate admissible graphs. Let

$$\Theta_{ad} = \begin{cases} \{\frac{1}{4}, \frac{1}{2}\}, & n = 3, \\ \{\frac{1}{3}, \frac{2}{3}, 1\}, & n = 4, \\ \widehat{\Theta}_{per}, & n \ge 5. \end{cases}$$

Note that for any admissible graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ constructed by Lemma 5.4.3, Lemma 5.4.4 and Lemma 5.4.5, $\{\theta_1, \dots, \theta_N\} \subset \Theta_{ad}$. In the following, we will prove

Proposition 5.4.2. For any $\theta \in \Theta_{ad}$, the intersection $\Omega^{\theta}_{\lambda} \cap \partial B_{\lambda}$ consists of two points.

The proof is based on the following

Lemma 5.4.6. Suppose $\theta \in \Theta$ and θ satisfies one of the following conditions: C1. There are two sequences $\{\theta_k^+\}_{k\geq 1}, \{\theta_k^-\}_{k\geq 1} \subset \Theta$ such that for all $k \geq 1$, $\theta_k^- < \theta < \theta_k^+$ and $\mathbf{J}(\theta_k^+, \theta) = \mathbf{J}(\theta_k^-, \theta) \to \infty$ as $k \to \infty$.

C2. There is a sequence $\{\theta_k\}_{k\geq 1} \subset \Theta$ such that $\theta_1 < \theta_2 < \theta_3 < \ldots$ (or $\theta_1 > \theta_2 > \theta_3 > \ldots$) and $\mathbf{J}(\theta_k, \theta) = k$ for any $k \geq 1$.

Then the intersection $\Omega^{\theta}_{\lambda} \cap \partial B_{\lambda}$ consists of two points.

Proof. 1. Suppose θ satisfies C1 and $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. By Proposition 5.3.6, the cut ray $\Omega_{\lambda}^{\theta_k^+}$ (or $\Omega_{\lambda}^{\theta_k^-}$) intersects with $\Omega_{\lambda}^{\theta}$ at $2^{\mathbf{J}(\theta_k^+, \theta)+1}$ points, hence decomposes $\Omega_{\lambda}^{\theta}$ into $2^{\mathbf{J}(\theta_k^+, \theta)+1}$ parts:

$$\Omega^{\theta}_{\lambda}(\varepsilon_0,\varepsilon_1,\cdots,\varepsilon_{\mathbf{J}(\theta^+_k,\theta)}),(\varepsilon_0,\varepsilon_1,\cdots,\varepsilon_{\mathbf{J}(\theta^+_k,\theta)})=(\pm s_0,\pm s_1,\cdots,\pm s_{\mathbf{J}(\theta^+_k,\theta)}),$$

where

$$\Omega^{\theta}_{\lambda}(\varepsilon_0,\varepsilon_1,\cdots,\varepsilon_p) := \{ z \in \Omega^{\theta}_{\lambda}; \mathbf{s}_{\lambda}(z) = (\varepsilon_0,\varepsilon_1,\cdots,\varepsilon_p,\pm s_{p+1},\pm s_{p+2},\cdots) \}.$$

By the structure of the cut rays (Proposition 5.3.2), since the angle θ satisfies condition C1, we conclude that among these $2^{\mathbf{J}(\theta_k^+,\theta)+1}$ parts, only two intersect with $\overline{B_{\lambda}}$: $\Omega_{\lambda}^{\theta}(s_0, s_1, \cdots, s_{\mathbf{J}(\theta_k^+,\theta)})$ and $\Omega_{\lambda}^{\theta}(-s_0, (-1)^n s_1, \cdots, (-1)^n s_{\mathbf{J}(\theta_k^+,\theta)})$. Moreover, for any $k \geq 1$

$$\Omega_{\lambda}^{\theta} \cap \overline{B_{\lambda}} \subset \Omega_{\lambda}^{\theta}(s_0, s_1, \cdots, s_{\mathbf{J}(\theta_k^+, \theta)}) \cup \Omega_{\lambda}^{\theta}(-s_0, (-1)^n s_1, \cdots, (-1)^n s_{\mathbf{J}(\theta_k^+, \theta)}).$$

It turns out that

$$\Omega_{\lambda}^{\theta} \cap \overline{B_{\lambda}} \subset \bigcap_{k \ge 1} \left(\Omega_{\lambda}^{\theta}(s_0, s_1, \cdots, s_{\mathbf{J}(\theta_k^+, \theta)}) \cup \Omega_{\lambda}^{\theta}(-s_0, (-1)^n s_1, \cdots, (-1)^n s_{\mathbf{J}(\theta_k^+, \theta)}) \right)$$
$$= \frac{\{z \in \Omega_{\lambda}^{\theta}; \mathbf{s}_{\lambda}(z) = (s_0, s_1, s_2, \cdots) \text{ or } (-s_0, (-1)^n s_1, (-1)^n s_2, \cdots)\}$$
$$= \overline{R_{\lambda}(\theta)} \cup \overline{R_{\lambda}(\theta + 1/2)}.$$

By Proposition 5.3.2, the intersection $\Omega_{\lambda}^{\theta} \cap \partial B_{\lambda}$ consists of two points. These two points are the landing points of the external rays $R_{\lambda}(\theta)$ and $R_{\lambda}(\theta + 1/2)$.

2. Now we suppose θ satisfies C2 and $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. We only prove the case when n is odd. The argument applies equally well to the case when n is even. Let $\{\theta_k\}_{k\geq 1} \subset \Theta$ be the sequence such that $\theta_1 < \theta_2 < \theta_3 < \ldots$ and $\mathbf{J}(\theta_k, \theta) = k$ for any $k \geq 1$. The following facts are straightforward:

Fact 1. Let $z \in \Omega_{\lambda}^{\theta}$. If the itinerary $\mathbf{s}_{\lambda}(z)$ is of the form $(\varepsilon_0, \cdots, \varepsilon_k, s_{k+1}, s_{k+2}, \cdots)$ or $(\varepsilon_0, \cdots, \varepsilon_k, -s_{k+1}, -s_{k+2}, \cdots)$ for some $k \geq 0$, then z lies in the closure of some external ray or radial ray $R_U(\theta_U)$ for $U \in \mathcal{P}$. (By Proposition 5.3.2)

Fact 2. For any k > 1, $\overline{B_{\lambda}}$ has no intersection with any bounded component of $\overline{\mathbb{C}} \setminus \bigcup_{1 \le j \le k} \Omega_{\lambda}^{\theta_j}$. (By Proposition 5.3.1 and 5.3.2)

Fact 3. The sections of $\Omega_{\lambda}^{\theta}$ that intersect with the unbounded component of $\overline{\mathbb{C}} \setminus \bigcup_{1 \le j \le k} \Omega_{\lambda}^{\theta_j}$ are as follows:

$$\Omega^{\theta}_{\lambda}(s_0, \cdots, s_k), \ \Omega^{\theta}_{\lambda}(-s_0, \cdots, -s_k),$$
$$\Omega^{\theta}_{\lambda}(s_0, \cdots, s_j, -s_{j+1}, \cdots, -s_k), \ \Omega^{\theta}_{\lambda}(-s_0, \cdots, -s_j, s_{j+1}, \cdots, s_k), \ 0 \le j < k.$$

Let \mathcal{E}_k be the collection of these sections.

By Fact 2 and Fact 3, we have $\overline{B_{\lambda}} \cap \Omega_{\lambda}^{\theta} \subset \bigcup_{E \in \mathcal{E}_{k}} E$ for any k > 1. It follows that $\overline{B_{\lambda}} \cap \Omega_{\lambda}^{\theta} \subset \bigcap_{k>1} \bigcup_{E \in \mathcal{E}_{k}} E = \{z \in \Omega_{\lambda}^{\theta}, \mathbf{s}_{\lambda}(z) \text{ is of the form } \pm \mathbf{s}(\theta) \text{ or } \pm (s_{0}, s_{1}, \cdots, s_{k}, -s_{k+1}, -s_{k+2}, \cdots) \text{ for some } k \geq 0\}.$

By Fact 1, for any $z \in \overline{B_{\lambda}} \cap \Omega_{\lambda}^{\theta}$, either $z \in R_{\lambda}(\theta) \cup R_{\lambda}(\theta + 1/2)$ or there exist $U \in \mathcal{P} \setminus \{B_{\lambda}\}$ and an angle θ_U such that $z \in \overline{R_U(\theta_U)}$. In the following, we show that the latter is impossible. In fact, if $z \in \overline{B_{\lambda}} \cap \Omega_{\lambda}^{\theta} \cap \overline{R_U(\theta_U)}$, then $z \in \partial B_{\lambda} \cap \partial U$. Let $p \geq 0$ be the first integer such that $f_{\lambda}^p(U) = T_{\lambda}$.

After iterations, we see that $f_{\lambda}^{p}(z) \in \partial B_{\lambda} \cap \partial T_{\lambda}$ and $f_{\lambda}^{p}(z)$ is the landing point of the radial ray $R_{T_{\lambda}}(\theta_{T_{\lambda}}) = f_{\lambda}^{p}(R_{U}(\theta_{U}))$. On the other hand, $f_{\lambda}^{p+1}(z)$ is the landing point of the external ray $R_{\lambda}(\theta_{\lambda}) = f_{\lambda}^{p+1}(R_{U}(\theta_{U}))$. This implies $f_{\lambda}^{p}(z)$ is also a landing point of some external ray $R_{\lambda}(\beta)$, $\beta \in \tau^{-1}(\theta_{\lambda})$. Since both $R_{T_{\lambda}}(\theta_{T_{\lambda}})$ and $R_{\lambda}(\beta)$ land at $f_{\lambda}^{p}(z)$, and $f_{\lambda}(R_{T_{\lambda}}(\theta_{T_{\lambda}})) = f_{\lambda}(R_{\lambda}(\beta)) =$ $R_{\lambda}(\theta_{\lambda}), f_{\lambda}^{p}(z)$ is necessarily a critical point in C_{λ} .

But this is a contradiction since for any $\alpha \in \Theta$, the cut ray $\Omega_{\lambda}^{\alpha}$ avoids the critical set C_{λ} .

Now we are in the situation $\overline{B_{\lambda}} \cap \Omega_{\lambda}^{\theta} \subset \overline{R_{\lambda}(\theta)} \cup \overline{R_{\lambda}(\theta+1/2)}$ and the conclusion follows. \Box

Proof of Proposition 5.4.2. It suffices to verify that for any $\theta \in \Theta_{ad}$, θ satisfies either C1 or C2 by Lemma 5.4.6.

When n = 3, $\mathbf{s}(1/4) = (\overline{1, -1})$, $\mathbf{s}(1/2) = (\overline{2})$. Define two sequences of angles $\{\alpha_k\}_{k\geq 1}, \{\beta_k\}_{k\geq 1} \subset \Theta$ such that:

$$\mathbf{s}(\alpha_1) = (1, -2, -1, 1, -1, 1, \cdots), \ \mathbf{s}(\beta_1) = (2, 1, -1, 2, 2, 2, \cdots), \mathbf{s}(\alpha_2) = (1, -1, 2, 1, -1, 1, \cdots), \ \mathbf{s}(\beta_2) = (2, 2, 1, -1, 2, 2, \cdots), \mathbf{s}(\alpha_3) = (1, -1, 1, -2, -1, 1, \cdots), \ \mathbf{s}(\beta_3) = (2, 2, 2, 1, -1, 2, \cdots), \cdots$$

Then $\alpha_1 > \alpha_2 > \alpha_3 > \cdots$ and $\mathbf{J}(\alpha_k, 1/4) = k$ for any $k \ge 1$; $\beta_1 < \beta_2 < \beta_3 < \cdots$ and $\mathbf{J}(\beta_k, 1/2) = k$. Thus both 1/4 and 1/2 satisfy condition C2.

When n = 4, $\mathbf{s}(1/3) = (\overline{2})$, $\mathbf{s}(2/3) = (\overline{-1})$, $\mathbf{s}(1) = (\overline{-3})$. Define three sequences of angles $\{\alpha_k\}_{k\geq 1}, \{\beta_k\}_{k\geq 1}, \{\gamma_k\}_{k\geq 1} \subset \Theta$ such that:

$$\mathbf{s}(\alpha_1) = (2, 1, -2, 2, 2, \cdots), \ \mathbf{s}(\beta_1) = (-1, -3, -1, -1, \cdots), \ \mathbf{s}(\gamma_1) = (-3, -1, -3, -3, \cdots), \\ \mathbf{s}(\alpha_2) = (2, 2, 1, -2, 2, \cdots), \ \mathbf{s}(\beta_2) = (-1, -1, -3, -1, \cdots), \\ \mathbf{s}(\gamma_2) = (-3, -3, -1, -3, \cdots), \\ \mathbf{s}(\alpha_3) = (2, 2, 2, 1, -2, \cdots), \ \mathbf{s}(\beta_3) = (-1, -1, -1, -3, \cdots), \\ \mathbf{s}(\gamma_3) = (-3, -3, -3, -1, \cdots), \\ \cdots$$

Then $\alpha_1 < \alpha_2 < \alpha_3 < \cdots$ and $\mathbf{J}(\alpha_k, 1/3) = k$; $\beta_1 > \beta_2 > \beta_3 > \cdots$ and $\mathbf{J}(\beta_k, 2/3) = k$; $\gamma_1 < \gamma_2 < \gamma_3 < \cdots$ and $\mathbf{J}(\gamma_k, 1) = k$. Thus 1/3, 2/3, 1 all satisfy condition C2.

When $n \geq 5$, we can prove that for any $\theta \in \widehat{\Theta} \supset \widehat{\Theta}_{per}$, θ satisfies condition C1, as follows. Suppose $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$. For any $k \geq 1$, we choose $s_k^-, s_k^+ \in \{\pm 1, \pm (n-1)\}$ and $s_{k+1}^-, s_{k+1}^+ \in \mathbb{I} \setminus \{0, n\}$ such that

(1) $|s_k^-| < |s_k| < |s_k^+|,$

(2) $(s_0, \cdots, s_{k-1}, s_k^-, s_{k+1}^-, s_{k+2}, \cdots), (s_0, \cdots, s_{k-1}, s_k^+, s_{k+1}^+, s_{k+2}, \cdots) \in \Sigma_0.$ Let

$$\theta_k^+ = \kappa((s_0, \cdots, s_{k-1}, s_k^+, s_{k+1}^+, s_{k+2}, s_{k+3}, \cdots)), \theta_k^- = \kappa((s_0, \cdots, s_{k-1}, s_k^-, s_{k+1}^-, s_{k+2}, s_{k+3}, \cdots)).$$

It's easy to check that $\theta_k^- < \theta < \theta_k^+$ and $\mathbf{J}(\theta_k^+, \theta) = \mathbf{J}(\theta_k^-, \theta) = k \to \infty$ as $k \to \infty$.

5.4.3 Modified puzzle piece

Following the idea of 'thickened puzzle piece' in [M2] to study the quadratic Julia set, we construct the 'modified puzzle piece' for McMullen maps. The



Figure 5.9: An example of 'modified puzzle pieces', to depth one.

'modified puzzle piece' can be applied to study the local connectivity of $J(f_{\lambda})$ in the non-renormalizable case (See Lemma 5.7.1). It is also used to define renormalizations (See Remark 5.5.1).

Given an angle $\theta \in \Theta$ with itinerary $\mathbf{s}(\theta) = (s_0, s_1, s_2, \cdots)$, recall that the cut ray $\Omega_{\lambda}^{\theta}$ is identified as $\Omega_{\lambda}^{\theta} = \bigcap_{k \geq 0} f_{\lambda}^{-k}(S_{s_k} \cup S_{-s_k})$. As is known that $\Omega_{\lambda}^{\theta}$ can be approximated by the sequence of compact sets $\{\Omega_{\lambda,m}^{\theta} = \bigcap_{0 \leq k \leq m} f_{\lambda}^{-k}(S_{s_k} \cup S_{-s_k})\}_{m \geq 0}$ in Hausdorff topology. Now we consider the set $\mathbb{C} \setminus \Omega_{\lambda,m}^{\theta}$. The open set $\mathbb{C} \setminus \Omega_{\lambda,m}^{\theta}$ consists of two connected components and the boundary of each component is a Jordan curve. Denote these two boundary curves by $\gamma_{\lambda,m}^{1}(\theta)$ and $\gamma_{\lambda,m}^{2}(\theta)$. Let $V_m(\theta) = \gamma_{\lambda,m}^{1}(\theta) \cap \gamma_{\lambda,m}^{2}(\theta)$ be the intersection of these two curves. It is obvious that $V_m(\theta)$ consists of finite points and $V_m(\theta) = \Omega_{\lambda}^{\theta} \cap \left(\bigcup_{0 \leq k \leq m+1} f_{\lambda}^{-k}(\infty)\right)$. For any $v \in V_m(\theta)$, let D(v) be the connected component of $\{z \in A_{\lambda}; G_{\lambda}(z) > 1\}$ that contains v. Obviously, D(v) is a disk.

In the following, we construct 'modified puzzle piece'. For the Yoccoz puzzle induced by the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$, recall that each puzzle piece P_0 of depth zero is contained in a unique component of $\mathbb{C} \setminus g_{\lambda}(\theta_1, \dots, \theta_N)$. This component is simply connected and is denoted by Q_0 . We may choose *m* large enough such that for any $\alpha, \beta \in \{\tau^k(\theta_j); 1 \leq j \leq N, k \geq 0\}$ with $\Omega^{\alpha}_{\lambda} \neq \Omega^{\beta}_{\lambda}$,

$$\Omega^{\alpha}_{\lambda,m} \cap \Omega^{\beta}_{\lambda,m} = \Omega^{\alpha}_{\lambda} \cap \Omega^{\beta}_{\lambda}.$$

The disk Q_0 is bounded by some collection of cut rays, say $\{\Omega_{\lambda}^{\alpha}; \alpha \in \Lambda(Q_0)\}$, where $\Lambda(Q_0)$ is an index set induced by Q_0 . For any $\alpha \in \Lambda(Q_0)$, choose a curve $\gamma(\alpha) \in \{\gamma_{\lambda,m}^1(\alpha), \gamma_{\lambda,m}^2(\alpha)\}$ such that $\gamma(\alpha) \cap Q_0 = \emptyset$. Let \widehat{Q}_0 be the connected component of $\overline{\mathbb{C}} \setminus \bigcup_{\alpha \in \Lambda(Q_0)} \gamma(\alpha)$ that contains Q_0 and let $V(Q_0) = \bigcup_{\alpha \in \Lambda(Q_0)} (V_m(\alpha) \cap \partial Q_0)$. The modified puzzle piece \widehat{P}_0 of P_0 is defined as follows:

$$\widehat{P}_0 = \widehat{Q}_0 - \bigcup_{v \in V(Q_0)} \overline{D(v)}.$$

Roughly speaking, we can get \widehat{P}_0 from Q_0 by thickening Q_0 near $\partial Q_0 \setminus V(Q_0)$ and truncating Q_0 near the points in $V(Q_0)$. The puzzle piece P_0 is not contained in \widehat{P}_0 , that is the reason why we call \widehat{P}_0 the 'modified puzzle piece' of P_0 rather than the 'thickened puzzle piece' of P_0 .

The modified puzzle pieces of greater depth can be constructed by the usual inductive procedure: If $\widehat{P}_d^{(j)}$ is the modified puzzle piece of depth d, then each component of $f_{\lambda}^{-1}(\widehat{P}_d^{(j)})$ is the modified puzzle piece of depth d+1, see Figure 5.9.

The virtue of these modified puzzle pieces is: If a puzzle piece $P_d^{(j)}$ contains $P_{d+1}^{(k)}$, then the modified puzzle piece $\widehat{P}_d^{(j)}$ contains $\overline{\widehat{P}_{d+1}^{(k)}}$. This can be easily proved by induction. In other words, this construction replace all of our annuli by non-degenerate annuli.

For $z \in \mathbb{C} \setminus (A_{\lambda} \cup J_0)$, let $\widehat{P}_d(z)$ be the modified puzzle piece of $P_d(z)$. We will only make use of modified puzzle pieces which are small enough to satisfy the following added restriction: If $\widehat{P}_d(z)$ contains a critical point, then $P_d(z)$ must already contain this critical point. Note that if the graph $G_{\lambda}(\theta_1, \dots, \theta_N)$ is not touchable, then this requirement is easily satisfied for any bounded value of depth d by choosing m large enough, and this will suffice for applications.

By construction, the puzzle piece $P_d(z)$ and the modified puzzle piece $\hat{P}_d(z)$ satisfy the following relation:

$$\overline{P_d(z)} \subset \widehat{P}_d(z) \cup A_{\lambda}, \quad \bigcap_{d \ge 0} \overline{P_d(z)} \subset \bigcap_{d \ge 0} \widehat{P}_d(z).$$

The modified puzzle pieces also satisfy the following symmetry properties: For any $z \in \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$,

$$-\widehat{P}_0(z) = \widehat{P}_0(-z); \quad \omega \widehat{P}_d(z) = \widehat{P}_d(\omega z), \ \omega^{2n} = 1, \ d \ge 1.$$

5.4.4 Tableaux

In this section, we present some basic knowledge of tableaux based on Milnor's Lecture [M2]. The applications of tableaux analysis combined with puzzle techniques can be found in [BH], [Hu], [M2], [PQRTY], [QY], [R01] and [RY] and many other papers.

Recall that J_0 is the set of all points on $J(f_{\lambda})$ whose orbits eventually meet the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. For $x \in \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$, the tableau T(x) is defined as two dimensional array $(P_{d,l}(x))_{d,l \geq 0}$, where $P_{d,l}(x) = f_{\lambda}^l(P_{d+l}(x)) = P_d(f_{\lambda}^l(x))$. The position (d, l) is called critical if $P_{d,l}(x)$ contains a critical point in C_{λ} . If $P_{d,l}(x)$ contains a critical point $c \in C_{\lambda}$, the position (d, l) is called a *c*-position. We call T(x) a critical tableau if $x \in C_{\lambda}$.

For any $x \in \overline{\mathbb{C}} \setminus (A_{\lambda} \cup J_0)$, the tableau T(x) satisfies the following three rules:

(T1) For each column $l \ge 0$, either the position (d, l) is critical for all $d \ge 0$ or there is a unique integer $d_0 \ge 0$ such that the position (d, l) is critical for all $d < d_0$ and not critical for $d \ge d_0$.

(T2) If $P_{d,l}(x) = P_d(y)$ for some $y \in \overline{\mathbb{C}} \setminus (A_\lambda \cup J_0)$, then $P_{i,l+j}(x) = P_{i,j}(y)$ for $0 \le i+j \le d$.

(T3) Let T(c) be a critical tableau. Assume

(a) $P_{d+1-l,l}(c) = P_{d+1-l}(c')$ for some critical point $c' \in C_{\lambda}$, $0 \leq l < d$, and $P_{d-i,i}(c)$ contains no critical points for 0 < i < l,

(b) $P_{d,m}(x) = P_d(c)$ and $P_{d+1,m}(x) \neq P_{d+1}(c)$ for some m > 0, Then $P_{d+1-l,m+l}(x) \neq P_{d+1-l}(c')$.

Remark 5.4.2. The tableau rule (T3) is based on the fact that every puzzle piece of depth $d \ge 1$ contains at most one critical point in C_{λ} .

Definition 5.4.2. 1. The tableau T(x) is non-critical if there is an integer $d_0 \ge 0$ such that (d_0, j) is not critical for all j > 0. Otherwise, T(x) is called critical.

2. The tableau T(x) is called pre-periodic if there exist two integers $l \ge 0$ and $p \ge 1$ such that $P_{d,l+p}(x) = P_{d,l}(x)$ for all $d \ge 0$. In this case, if l = 0, T(x) is called periodic and the smallest integer $p \ge 1$ is called the period of T(x).

3. Let $\operatorname{Row}_c(d)$ be the d-th row of the critical tableau T(c), we say $\operatorname{Row}_c(d+l)$ with l > 0 is a child of $\operatorname{Row}_c(d)$ if there is a critical point $c' \in C_\lambda$ such that $A_d(f_\lambda^l(c)) = A_d(c')$ and $f_\lambda^l : A_{d+l}(c) \to A_d(c')$ is a degree two covering map.

4. (following 3) For $d \ge 1$, we say $\operatorname{Row}_c(d)$ is excellent if $A_d(f_{\lambda}^l(c))$ is not semi-critical for all $l \ge 0$.

Remark 5.4.3. By Lemma 5.4.1, and the fact $f_{\lambda}^{k}(\omega z) = \pm f_{\lambda}^{k}(z)$ for $k \geq 1, \omega^{2n} = 1$, we have

1. If (d,l) is a critical position for some critical tableau, then (d,l) is a critical position for every critical tableau.

2. If some critical tableau T(c) is critical, non-critical or pre-periodic, then every critical tableau is critical, non-critical or pre-periodic, respectively.

3. If $\operatorname{Row}_c(d)$ is excellent or has a child $\operatorname{Row}_c(d+l)$ for some critical point $c \in C_{\lambda}$, then for every $c' \in C_{\lambda}$, $\operatorname{Row}_{c'}(d)$ is excellent or has a child $\operatorname{Row}_{c'}(d+l)$, respectively.

Lemma 5.4.7. Suppose some critical tableau T(c) is critical but not preperiodic, then

1. For every $d \ge 1$, $\operatorname{Row}_c(d)$ has at least one child.

2. If $\operatorname{Row}_{c}(d)$ is excellent, then $\operatorname{Row}_{c}(d)$ has at least two children.

3. If $\operatorname{Row}_c(d)$ is excellent and $\operatorname{Row}_c(d+l)$ is its child, then $\operatorname{Row}_c(d+l)$ is also excellent.

4. If $\operatorname{Row}_c(d)$ has only one child, say $\operatorname{Row}_c(d+l)$, then $\operatorname{Row}_c(d+l)$ is excellent.

Proof. 1. By hypothesis, for every $d \ge 1$, we can find a smallest integer l > 0, such that the annulus $A_d(f_{\lambda}^l(c))$ is c'-critical for some $c' \in C_{\lambda}$. The map $f_{\lambda}^l : A_{d+l}(c) \to A_d(c')$ is a degree two covering map. This implies that $\operatorname{Row}_c(d+l)$ is a child of $\operatorname{Row}_c(d)$.

2. (following 1) There exists d' > d such that the annulus $A_{d'}(f_{\lambda}^{l}(c))$ is c'semi-critical. Since $\operatorname{Row}_{c}(d)$ is excellent, by tableau rule (T3), $A_{d'-t}(f_{\lambda}^{l+t}(c))$ is either off-critical or semi-critical for $0 < t \leq d' - d$. In particular, $A_{d}(f_{\lambda}^{l+d'-d}(c))$ is off-critical. Hence, we can find a smallest integer l' > l+d'-dsuch that the annulus $A_{d}(f_{\lambda}^{l'}(c))$ is critical, so $\operatorname{Row}_{c}(d+l')$ is another child of $\operatorname{Row}_{c}(d)$.

3. If $\operatorname{Row}_c(d+l)$ is not excellent, then there is a column $l' \geq l$ such that $A_{d+l}(f_{\lambda}^{l'}(c))$ is semi-critical. By tableau rule (T3), $A_d(f_{\lambda}^{l+l'}(c))$ is also semi-critical, which contradict the fact that $\operatorname{Row}_c(d)$ is excellent.

4. If $\operatorname{Row}_c(d+l)$ is not excellent, then as in (3), $A_d(f_{\lambda}^{l+l'}(c))$ is semi-critical for some $l' \geq l$. Suppose $l' \geq l$ is the smallest integer. We can find a smallest integer t > l' + l such that $A_d(f_{\lambda}^t(c))$ is c'-critical for some $c' \in C_{\lambda}$. Then $\operatorname{Row}_c(d+t)$ is also a child of $\operatorname{Row}_c(d)$, which is a contradiction. \Box

Lemma 5.4.8. Suppose the critical tableau T(c) is critical and pre-periodic.

1. If n is odd, then there exist exactly two critical points $\pm c' \in C_{\lambda}$ such that T(c') and T(-c') are periodic.

2. If n is even, then there is a unique critical point $\tilde{c} \in C_{\lambda}$ such that $T(\tilde{c})$ is periodic.

Proof. Since T(c) is critical, there exist a smallest integer $p \ge 1$ and a unique critical point $c' \in C_{\lambda}$ such that (d, p) is a c'-position for all $d \ge 0$.

1. *n* is odd. There are two possibilities, either $f_{\lambda}(c) = f_{\lambda}(c')$ or $f_{\lambda}(c) + f_{\lambda}(c') = 0$.

If $f_{\lambda}(c) = f_{\lambda}(c')$, then the critical tableaux T(c') and T(-c') are periodic with period p. In this case, there is an integer $d_0 \ge 0$ such that for any $d \ge d_0, 0 < l < p$, the position (d, l) is not critical. It's easy to check that for any $\tilde{c} \in C_{\lambda} \setminus \{\pm c'\}$, the tableau $T(\tilde{c})$ is strictly pre-periodic. In particular, if p = 1, then $P_d(c') = P_d(f_{\lambda}(c'))$ for all $d \ge 0$. This means that for any $d \ge 0$, c' and $f_{\lambda}(c')$ lie in the same puzzle piece of depth d. Thus we conclude $\{\pm c'\} = \{c_0, c_n\}.$

If $f_{\lambda}(c) + f_{\lambda}(c') = 0$, then the critical tableau T(c') and T(-c') are periodic with period 2p. Consider the tableau T(c'), there is an integer $d_0 \ge 0$, such that for any $d \ge d_0, 0 < l < p$, the position (d, l) is not critical and for any $d \ge 0$ the position (d, p) is (-c')-critical. It's easy to check that for any $\tilde{c} \in C_{\lambda} \setminus \{\pm c'\}$, the tableau $T(\tilde{c})$ is strictly pre-periodic. In particular, if p = 1, then $P_d(-c') = P_d(f_{\lambda}(c'))$ for all $d \ge 0$. This means that for any $d \ge 0$, -c' and $f_{\lambda}(c')$ lie in the same puzzle piece of depth d. Thus we conclude $\{\pm c'\} = \{c_1, c_{n+1}\}$.

2. *n* is even. In this case, by the fact that $f_{\lambda}^{k}(v_{\lambda}^{+}) = f_{\lambda}^{k}(v_{\lambda}^{-})$ for all $k \geq 1$, we conclude the tableau $T(f_{\lambda}(c'))$ is periodic with period *p* and the tableau $T(-f_{\lambda}(c'))$ is strictly pre-periodic. Thus there is a unique critical point $\tilde{c} \in f_{\lambda}^{-1}(f_{\lambda}(c'))$ such that $T(\tilde{c})$ is periodic. For this tableau, there is an integer $d_0 \geq 0$ such that for any $d \geq d_0, 0 < l < p$, the position (d, l) is not critical. It's easy to check that for any $c'' \in C_{\lambda} \setminus \{\tilde{c}\}$, the tableau T(c'') is strictly pre-periodic. In particular, if p = 1 and $T(v_{\lambda}^{+})$ is periodic, then $\tilde{c} = c_0$; if p = 1 and $T(v_{\lambda}^{-})$ is periodic, then $\tilde{c} = c_{n+1}$.

5.5 Renormalizations

In this section, we discuss the renormalizations of McMullen maps from the viewpoint of puzzle piece.

Definition 5.5.1. If there exist a critical point c of f_{λ} , an integer $p \ge 1$ and two disks U and V containing c such that

 $\varepsilon f_{\lambda}^p : U \to V$

is a quadratic like map whose Julia set is connected (here $\varepsilon \in \{\pm 1\}$ is a symbol), then we say f_{λ} is p-renormalizable at c if $\varepsilon = 1$ and f_{λ} is p-*-renormalizable at c if $\varepsilon = -1$. In the former case, the triple (f_{λ}^{p}, U, V) is called a p-renormalization of f_{λ} at c. In the latter case, the triple $(-f_{\lambda}^{p}, U, V)$ is called a p-*-renormalization of f_{λ} at c.

In the following, we use $K_c = \{z \in U; (\varepsilon f_{\lambda}^p)^k(z) \in U, \forall k \geq 0\} = \bigcap_{k\geq 0} (\varepsilon f_{\lambda}^p)^{-k}(U)$ to denote the small filled Julia set of the (*-)renormalization $(\varepsilon f_{\lambda}^p, U, V)$. By straightening theorem of Douady and Hubbard [DH3], if $(\varepsilon f_{\lambda}^p, U, V)$ is a *p*-(*-)renormalization of f_{λ} , then $\varepsilon f_{\lambda}^p$ is conjugate by a quasiconformal map σ to a unique quadratic polynomial $p_{\mu}(z) = z^2 + \mu$ in a neighborhood of filled Julia set K_c . Let β be the β -fixed point (i.e. landing point of the zero external ray) of p_{μ} , and β' be the other preimage of β . We call $\beta_c = \sigma^{-1}(\beta)$ the β -fixed point of the renormalization ($\varepsilon f_{\lambda}^p, U, V$). The other preimage of β_c under the renormalization is $\beta'_c = \sigma^{-1}(\beta')$.

In this section, we always assume that the graph $\mathbf{G}_{\lambda}(\theta_1, \cdots, \theta_N)$ is admissible.

5.5.1 Periodic critical tableau implies renormalization

Lemma 5.5.1. Suppose the critical tableau T(c) is pre-periodic.

1. If T(c) is non-critical, then f_{λ} is critically finite.

2. If T(c) is critical, then f_{λ} is either renormalizable or *-renormalizable.

Proof. Since T(c) is pre-periodic, there exist two integers $l \ge 0$ and $p \ge 1$ such that $P_d(f_{\lambda}^{l+p}(c)) = P_{d,l+p}(c) = P_{d,l}(c) = P_d(f_{\lambda}^l(c))$ for all $d \ge 0$.

1. T(c) is non-critical. In this case, the tableaux $T(f_{\lambda}^{l}(c))$ and $T(f_{\lambda}^{l+p}(c))$ are also non-critical. By Lemma 5.7.1 (Notice that the proof of Lemma 5.7.1 is independent of Lemma 5.5.1), $\{f_{\lambda}^{l+p}(c)\} = \bigcap_{d\geq 0} P_d(f_{\lambda}^{l+p}(c)) =$ $\bigcap_{d\geq 0} P_d(f_{\lambda}^{l}(c)) = \{f_{\lambda}^{l}(c)\}$. Therefore, $f_{\lambda}^{l+p}(c) = f_{\lambda}^{l}(c)$ and f_{λ} is critically finite.

2. T(c) is critical. If *n* is odd, then by Lemma 5.4.8, there are exactly two critical points $\pm c' \in C_{\lambda}$ such that T(c') and T(-c') are periodic. Suppose the period is *p*. Consider the tableau T(c'), there are two possibilities :

Case 1. There is an integer $d_0 \geq 0$ such that for any $d \geq d_0, 0 < l < p$, the position (d, l) is not critical. Then $f_{\lambda}^p : P_{d_0+p}(c') \rightarrow P_{d_0}(c')$ is a quadratic-like map and $\{f_{\lambda}^{kp}(c'); k \geq 0\} \subset P_{d_0+p}(c')$. Thus $(f_{\lambda}^p, P_{d_0+p}(c'), P_{d_0}(c'))$ is a *p*-renormalization of f_{λ} at c'. Since f_{λ} is an odd function, $(f_{\lambda}^p, P_{d_0+p}(-c'), P_{d_0}(-c'))$ is a *p*-renormalization of f_{λ} at -c'.

Case 2. p is even and there is an integer $d_0 \geq 0$ such that for any $d \geq d_0, 0 < l < p/2$, the position (d, l) is not critical and for any $d \geq 0$, the position (d, p/2) is (-c')-critical. Then $-f_{\lambda}^{p/2} : P_{d_0+p/2}(c') \to P_{d_0}(c')$ is a quadratic-like map with $\{(-1)^k f_{\lambda}^{kp/2}(c'); k \geq 0\} \subset P_{d_0+p/2}(c')$. Thus $(-f_{\lambda}^{p/2}, P_{d_0+p/2}(c'), P_{d_0}(c'))$ is a p/2-*-renormalization of f_{λ} at c'. It turns out that $(-f_{\lambda}^{p/2}, P_{d_0+p/2}(-c'), P_{d_0}(-c'))$ is a p/2-*-renormalization of f_{λ} at -c'.

If n is even, then by Lemma 5.4.8, there is a unique critical point $\tilde{c} \in C_{\lambda}$ such that $T(\tilde{c})$ is periodic. Suppose the period is p. Then there is an integer $d_0 \geq 0$ such that for any $d \geq d_0, 0 < l < p$, the position (d, l) is not critical. Then $f_{\lambda}^p : P_{d_0+p}(\tilde{c}) \to P_{d_0}(\tilde{c})$ is a quadratic-like map and $\{f_{\lambda}^{kp}(\tilde{c}); k \geq 0\} \subset P_{d_0+p}(\tilde{c})$. Thus $(f_{\lambda}^p, P_{d_0+p}(\tilde{c}), P_{d_0}(\tilde{c}))$ is a p-renormalization of f_{λ} at \tilde{c} . Since f_{λ} is an even function, $(-f_{\lambda}^p, P_{d_0+p}(-\tilde{c}), P_{d_0}(-\tilde{c}))$ is a p-*-renormalization of f_{λ} at $-\tilde{c}$. **Remark 5.5.1.** Lemma 5.5.1 also holds when the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is non-admissible. Indeed, in this case, we can use modified puzzle pieces to define renormalizations.

Proposition 5.5.1. Suppose f_{λ} has a non-repelling cycle in \mathbb{C} , then f_{λ} is either renormalizable or *-renormalizable. Under this situation, there are three possibilities:

1. If f_{λ} is renormalizable and n is odd, then f_{λ} has exactly two non-repelling cycles in \mathbb{C} .

2. If f_{λ} is *-renormalizable and n is odd, then f_{λ} has exactly one non-repelling cycle in \mathbb{C} .

3. If f_{λ} is renormalizable and n is even, then f_{λ} has exactly one non-repelling cycle in \mathbb{C} .

Proof. Let $\mathcal{C} = \{z_0, f_\lambda(z_0), \dots, f_\lambda^q(z_0) = z_0\}$ be the non-repelling cycle of f_λ in \mathbb{C} . By Proposition 5.4.1, we can find an admissible graph $\mathbf{G}_\lambda(\theta_1, \dots, \theta_N)$. By Proposition 5.3.4, the cycle \mathcal{C} avoid the graph $\mathbf{G}_\lambda(\theta_1, \dots, \theta_N)$. Thus for any $z \in \mathcal{C}$ and any integer $d \geq 0$, the puzzle piece $P_d(z)$ is well-defined.

We claim that there exist $z \in \mathcal{C}$ and a critical point $c \in C_{\lambda}$, such that $P_d(z) = P_d(c)$ for all $d \geq 0$. Otherwise, the tableau T(z) is non-critical for any $z \in \mathcal{C}$. So there is an integer $d_0 \geq 0$ such that the map $f_{\lambda}^q : P_{d_0+q}(z_0) \to P_{d_0}(z_0)$ is conformal. By Schwarz lemma, $|(f_{\lambda}^q)'(z_0)| > 1$, which is a contradiction.

In this way, we can find a critical point $c \in C_{\lambda}$ whose tableau T(c) is periodic. By Lemma 5.5.1, f_{λ} is either renormalizable or *-renormalizable.

To continue, suppose the period of T(c) is p, which is necessarily a divisor of q. By Lemma 5.5.1, there are three possibilities:

(P1). n is odd and $(f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ is a p-renormalization of f_{λ} at c. In this case, $(f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ is quasiconformally conjugate to a polynomial $z \mapsto z^{2} + \mu$. Since a quadratic polynomial has at most one non-repelling cycle (See [CG] or [Sh1]), it turns out that C is the only non-repelling cycle contained in $\bigcup_{0 \leq j < p} f_{\lambda}^{j}(K_{c})$. On the other hand, -C is the only non-repelling cycle contained tableaux in this case and $(\bigcup_{0 \leq j < p} f_{\lambda}^{j}(K_{c})) \cap (\bigcup_{0 \leq j < p} f_{\lambda}^{j}(-K_{c})) = \emptyset$, we conclude that f_{λ} has exactly two non-repelling cycles in \mathbb{C} .

(P2). n is odd and $(-f_{\lambda}^{p/2}, P_{d_0+p/2}(c), P_{d_0}(c))$ is a p/2-*- renormalization of f_{λ} at c. In this case, the cycle \mathcal{C} meets both K_c and $-K_c$. Similar argument as above, one see that \mathcal{C} is the only non-repelling cycle contained in $\bigcup_{0 \leq j < p} f_{\lambda}^j(K_c)$. Since the cycle $-\mathcal{C}$ is also contained in $\bigcup_{0 \leq j < p} f_{\lambda}^j(K_c)$, it turns out that $\mathcal{C} = -\mathcal{C}$.

(P3). *n* is even and $(f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ is a *p*-renormalization of f_{λ} at *c*. In this case, T(c) is the only periodic critical tableau. Similar argument as above, we see that \mathcal{C} is the only non-repelling cycle in \mathbb{C} . In the following, we discuss the case when f_{λ} has an indifferent cycle of multiplier $e^{2\pi i\theta}$. Douady [Dou1] conjectured that for any rational map, whenever it is linearizable (i.e. the map is conformally conjugate to an irrational rotation) near an indifferent fixed point of multiplier $e^{2\pi i\theta}$, then θ must be a Brjuno number. Here an irrational number θ of convergent p_k/q_k (rational approximations obtained by the continued fraction expansion) is a Brjuno number (denoted by \mathcal{B}) if

$$\sum_{k\geq 1} \frac{\log q_{k+1}}{q_k} < +\infty$$

It follows from Cremer, Siegel and Brjuno that if $\theta \in \mathcal{B}$, then every germ $f(z) = e^{2\pi i\theta}z + \mathcal{O}(z^2)$ is linearizable. Yoccoz [Yo] shows that if the quadratic polynomial $z \mapsto e^{2\pi i\theta}z + z^2$ is linearizable, then $\theta \in \mathcal{B}$. For general case, Geyer [Gey1] shows that for any $d \geq 2$, if $z \mapsto z^d + c$ has an indifferent cycle of multiplier $e^{2\pi i\theta}$ near which the map is linearizable, then $\theta \in \mathcal{B}$. Based on these results and Proposition 5.5.1, we establish immediately:

Proposition 5.5.2. Suppose f_{λ} has an indifferent cycle of multiplier $e^{2\pi i\theta}$, then f_{λ} is linearizable near the indifferent cycle if and only if $\theta \in \mathcal{B}$.

5.5.2 Properties of renormalizations

In this section, we assume that the critical tableau T(c) is periodic with period k. By Lemma 5.5.1, f_{λ} is either k-renormalizable at c or k/2-*-renormalizable at c. Let $(\varepsilon f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ be the corresponding renormalization, where

$$(\varepsilon, p) = \begin{cases} (1, k), & \text{if } f_{\lambda} \text{ is } k \text{-renormalizable at } c, \\ (-1, k/2), & \text{if } f_{\lambda} \text{ is } k/2 \text{-}*\text{-renormalizable at } c. \end{cases}$$

The small filled Julia set $K_c = \bigcap_{d \ge 0} \overline{P_d(c)} = \bigcap_{d \ge 0} P_d(c)$.

If $K_c \cap \partial B_\lambda \neq \emptyset$, we will show that there is a unique external ray in B_λ accumulating on K_c . Before the proof, we need a classic result for quadratic polynomials:

Lemma 5.5.2. Let $p_{\mu}(z) = z^2 + \mu$ be a quadratic polynomial with a connected filled Julia set K. If there is a curve $\delta \subset \mathbb{C} \setminus K$ converging to $x \in K$ and $p_{\mu}(\delta) \supset \delta$, then x is the β -fixed point of p_{μ} .

Here, a curve $\delta \subset \mathbb{C} \setminus K$ converges to $x \in K$ means that δ can be parameterized as $\delta : [0,1) \to \mathbb{C} \setminus K$ such that $\lim_{t\to 1} \delta(t)$ exists and $\lim_{t\to 1} \delta(t) = x \in K$. See [McM1] for a proof of Lemma 5.5.2. The conclusion also holds for quadratic like maps. **Lemma 5.5.3.** Suppose the critical tableau T(c) is k-periodic and $K_c \cap \partial B_{\lambda} \neq \emptyset$, then

1. The small filled Julia sets $K_c, f_{\lambda}(K_c), \dots, f_{\lambda}^{k-1}(K_c)$ are pairwise disjoint.

2. There is a unique external ray $R_{\lambda}(t)$ in B_{λ} accumulating on K_c . This external ray lands at $\beta_c \in K_c$ and the angle t is k-periodic.

Proof. 1. If $f_{\lambda}^{i}(K_{c}) \cap f_{\lambda}^{j}(K_{c}) \neq \emptyset$ for some $0 \leq i < j < k$, then $K_{c} \cap f_{\lambda}^{k+i-j}(K_{c}) \neq \emptyset$. Thus $P_{d,k+i-j}(c) = f_{\lambda}^{k+i-j}(P_{d+k+i-j}(c)) = P_{d}(c)$ for all $d \geq 0$. This implies that the critical tableau T(c) is (k+i-j)-periodic, which is a contradiction.

2. First note that $f_{\lambda}^{k}(P_{d+k}(c)) = P_{d}(c)$ for $d \geq 0$. Since $K_{c} \cap \partial B_{\lambda} \neq \emptyset$, $P_{mk}(c) \cap B_{\lambda}$ is nonempty and bounded by two external rays, say $R_{\lambda}(\theta_{m}^{-})$ and $R_{\lambda}(\theta_{m}^{+})$ with $\theta_{m}^{-} < \theta_{m}^{+}$. Let $Q(\theta_{m}^{-}, \theta_{m}^{+}) = \overline{P_{mk}(c) \cap B_{\lambda}}, m \geq 1$. Since $f_{\lambda}^{k}(Q(\theta_{m+1}^{-}, \theta_{m+1}^{+})) = Q(\theta_{m}^{-}, \theta_{m}^{+})$, we have

$$\theta_m^- \le \theta_{m+1}^- \le \dots \le \theta_{m+1}^+ \le \theta_m^+, \ \ \theta_m^+ - \theta_m^- = n^k(\theta_{m+1}^+ - \theta_{m+1}^-)$$

Thus there is a common limit $t = \lim \theta_m^+ = \lim \theta_m^-$. Since $\theta_m^- \le t \le \theta_m^+$ for any m, we have $n^k t \equiv t \pmod{\mathbb{Z}}$. Thus t is a periodic angle and the external ray $R_{\lambda}(t)$ lands at a point $z \in K_c \cap \partial B_{\lambda}$ (This is because rational external rays always land). Since $R_{\lambda}(n^j t)$ lands at $f_{\lambda}^j(z) \in f_{\lambda}^j(K_c) \cap \partial B_{\lambda}$ for $0 \le j < k$ and the small filled Julia sets $K_c, f_{\lambda}(K_c), \cdots, f_{\lambda}^{k-1}(K_c)$ are pairwise disjoint, we conclude that the angles $t, nt, \cdots, n^{k-1}t$ are different from each other. Thus t is k-periodic.

Suppose θ is another angle such that the external ray $R_{\lambda}(\theta)$ accumulating on K_c . Then $\theta_m^- \leq \theta \leq \theta_m^+$ for any m. Thus $\theta = \lim \theta_m^+ = \lim \theta_m^- = t$.

To finish, we show $z = \beta_c$. Since T(c) is k-periodic, f_{λ} is either krenormalizable or k/2-*-renormalizable. In the former case, $f_{\lambda}^k(R_{\lambda}(t)) = R_{\lambda}(t)$. Thus by Lemma 5.5.2, $z = \beta_c$. In the latter case, since $R_{\lambda}(t)$ is the unique external ray accumulating on K_c , we conclude $R_{\lambda}(t + 1/2) = -R_{\lambda}(t)$ is the unique external ray accumulating on $-K_c$. On the other hand, $f_{\lambda}^{k/2}(R_{\lambda}(t))$ is also an external ray accumulating on $-K_c$, we have $f_{\lambda}^{k/2}(R_{\lambda}(t)) = R_{\lambda}(t + 1/2) = -R_{\lambda}(t)$. In this case, $-f_{\lambda}^{k/2}(R_{\lambda}(t)) = R_{\lambda}(t)$. Again by Lemma 5.5.2, $z = \beta_c$.

5.6 A Criterion of Local Connectivity

In this section, we present a criterion to characterize the local connectivity of the immediate basin of attraction. This criterion together with Yoccoz puzzle techniques can be applied to study the local connectivity and higher regularity of the boundary ∂B_{λ} .

In the following discussion, let f be a rational map of degree at least two, C(f) be the critical set of f and $P(f) = \overline{\bigcup_{k\geq 1} f^k(C(f))}$ be the postcritical set. Suppose that f has an attracting periodic point z_0 and the immediate basin B of z_0 is simply connected. Let $B(z, \delta) = \{x \in \mathbb{C}; |x - z| < \delta\}$.

Definition 5.6.1. We say f satisfies **BD**(*i.e.* bounded degree) condition on ∂B if for any $u \in \partial B$, there is a number $\varepsilon_u > 0$ such that for any integer $m \ge 0$ and any component $U_m(u)$ of $f^{-m}(B(u, \varepsilon_u))$ intersecting with ∂B , $U_m(u)$ is simply connected and the degree deg $(f^m : U_m(u) \to B(u, \varepsilon_u))$ is bounded by some constant D, which is independent of u, m and $U_m(u)$.

For the definition, here is a remark: since $f^m : U_m(u) \to B(u, \varepsilon_u)$ is a proper map between two disks, we conclude by Maximum Principle that for any disk $W \subset B(u, \varepsilon_u)$ and any component V of $f^{-m}(W)$ that lies inside $U_m(u), V$ is also a disk.

The aim of this section is to prove the following:

Proposition 5.6.1. If f satisfies **BD** condition on ∂B , then

- 1. ∂B is locally connected.
- 2. If furthermore ∂B is a Jordan curve, then ∂B is a quasicircle.

The proof of Proposition 5.6.1 is based on Theorem 2.3.2.

Proof. By replacing f with f^k , we assume z_0 is a fixed point of f. By quasiconformal surgery, we assume z_0 is a superattracting fixed point with local degree $d = \deg(f : B \to B) \ge 2$. Thus B contains no critical points other that z_0 . By Möbius conjugation, we assume $z_0 = \infty$.

Since f satisfies **BD** condition on ∂B , there exists a constant $\delta > 0$ such that for any $u \in \partial B$, any integer $m \ge 0$ and any component $U_m(u)$ of $f^{-m}(B(u,\delta))$ that intersects with ∂B , $U_m(u)$ is simply connected and $\deg(f^m : U_m(u) \to B(u,\delta)) \le D$. In fact, we can choose δ as the Lebesgue number of the family $\mathcal{F} = \{B(u, \varepsilon_u); u \in \partial B\}$, which is an open covering of the boundary ∂B .

The proof consists of four steps.

Step 1. Let $V_m(z)$ be the component of $f^{-m}(B(z, \delta/2))$ contained in $U_m(z)$ and intersecting with ∂B , then

$$\lim_{m \to \infty} \sup_{z \in \partial B} \operatorname{diam}(V_m(z)) = 0.$$

For else, there is a constant $d_0 \ge 0$ and two sequences $\{z_k\} \subset \partial B$ and $\{\ell_k\}$ such that $\operatorname{diam}(V_{\ell_k}(z_k)) \ge d_0$. For every $k \ge 1$, choose a point $y_k \in$

 $f^{-\ell_k}(z_k) \cap V_{\ell_k}(z_k)$. By passing to a subsequence, we assume $y_k \to y_\infty \in \partial B$ and $z_k \to z_\infty \in \partial B$. By Theorem 2.3.2, there is a constant C(D) such that

$$\operatorname{Shape}(V_{\ell_k}(z_k), y_k) \leq C(D)\operatorname{Shape}(B(z_k, \delta/2), z_k) = C(D)$$

Since diam $(V_{\ell_k}(z_k)) \geq d_0$, $V_{\ell_k}(z_k)$ contains a round disk centered at y_k of definite size. So there is a constant $r_0 = r_0(d_0, D)$ such that $V_{\ell_k}(z_k) \supset B(y_{\infty}, r_0)$ for large k. This means $f^{\ell_k}(B(y_{\infty}, r_0)) \subset B(z_k, \delta/2) \subset B(z_{\infty}, \delta)$. But this contradicts the fact that $f^{\ell_k}(B(y_{\infty}, r_0)) \supset J(f)$ when k is large.

Step 2. There are two constants L > 0 and $\nu \in (0, 1)$, such that for any $z \in \partial B$ and any $k \ge 1$, diam $(V_k(z)) \le L\nu^k$.

By Step 1, there is an integer s > 0, such that diam $(V_s(z)) < \delta/4$ for all $z \in \partial B$. For each $x \in \partial B$ and each integer k > 0, let $V_{ks}(x)$ be a component of $f^{-ks}(B(x, \delta/2))$ intersecting with ∂B and $x_{ks} \in V_{ks}(x) \cap f^{-ks}(x)$. For $0 \le j \le k$, let $x_{js} = f^{(k-j)s}(x_{ks})$ and U_j be the component of $f^{-js}(B(x_{(k-j)s}, \delta/2))$ containing x_{ks} . Then

$$x_{ks} \in V_{ks}(x) = U_k \subset \cdots \subset U_0 = B(x_{ks}, \delta/2).$$

For every $1 \leq j < k$, $f^{js}: U_j \to B(x_{(k-j)s}, \delta/2)$ is a proper map of degree $\leq D$. Since $f^{js}(U_{j+1})$ is contained in $B(x_{(k-j)s}, \delta/4)$,

$$\operatorname{mod}(U_j \setminus \overline{U_{j+1}}) \ge \frac{1}{D} \operatorname{mod}(B(x_{(k-j)s}, \delta/2) \setminus \overline{f^{js}(U_{j+1})}) \ge \frac{\log 2}{2\pi D},$$
$$\operatorname{mod}(B(x_{ks}, \delta/2) \setminus \overline{V_{ks}(x)}) \ge \sum_{0 \le j < k} \operatorname{mod}(U_j \setminus \overline{U_{j+1}}) \ge \frac{k \log 2}{2\pi D}.$$

So there are two constants M > 0 and $\mu \in (0, 1)$ such that for any $x \in \partial B$, diam $(V_{ks}(x)) \leq M\mu^k$. This implies that there are two constants L > 0 and $\nu \in (0, 1)$ such that diam $(V_k(x)) \leq L\nu^k$ for all $k \geq 1$.

Step 3. There exists a sequence of Jordan curves $\{\gamma_k : \mathbb{S} \to B\}$ such that γ_k converges uniformly to a continuous and surjective map $\gamma_{\infty} : \mathbb{S} \to \partial B$, where $\mathbb{S} = \mathbb{R}/\mathbb{Z}$ is the unit circle. Hence, ∂B is locally connected.

Recall that the Böttcher map $\phi : B \to \overline{\mathbb{C}} \setminus \overline{\mathbb{D}}$ defined by $\phi(z) = \lim_{k \to \infty} (f_{\lambda}^{k}(z))^{d^{-k}}$ is a conformal isomorphism. It satisfies $\phi^{-1}(r^{d}e^{2\pi i dt}) = f(\phi^{-1}(re^{2\pi i t}))$ for $(r,t) \in (1,+\infty) \times \mathbb{S}$. Let $\ell(R,t) = \phi^{-1}([\sqrt[d]{R},R]e^{2\pi i t})$ for $(R,t) \in (1,2) \times \mathbb{S}$. By the boundary behavior of Poincaré metric, there is a constant C > 0 such that for $\operatorname{any}(R,t) \in (1,2) \times \mathbb{S}$,

$$\begin{aligned} \text{Eucl.length}(\ell(R,t)) &\leq C \text{Hyper.length}(\ell(R,t)) \cdot \text{H.dist}(\phi^{-1}(R\mathbb{S}), \partial B) \\ &\leq C(\log d) \text{H.dist}(\phi^{-1}(R\mathbb{S}), \partial B) \quad (\to 0 \text{ as } R \to 1), \end{aligned}$$

where Hyper.length is the hyperbolic length in B and H.dist is the Hausdorff distance in the sphere $\overline{\mathbb{C}}$. Thus we can choose R sufficiently close to 1 such that for any $t \in \mathbb{S}$, $\ell(R,t) \subset B(z,\delta/2)$ for some $z \in \partial B$. For $k \geq 0$, define a curve $\gamma_k : \mathbb{S} \to B$ by $\gamma_k(t) = \phi^{-1}(R^{1/d^k}e^{2\pi i t})$. Since $f^k(\gamma_{k+q}(t)) = \gamma_q(d^k t)$ for $q \geq 0$ and $\gamma_0(d^k t), \gamma_1(d^k t) \in \ell(R, d^k t) \subset B(z, \delta/2)$ for some $z \in \partial B$, we conclude that $\gamma_k(t)$ and $\gamma_{k+1}(t)$ lie in the same component of $f^{-k}(B(z, \delta/2))$ intersecting with ∂B . By Step 2,

$$\max_{t \in \mathbb{S}} |\gamma_{k+1}(t) - \gamma_k(t)| = \mathcal{O}(\nu^k).$$

So $\{\gamma_k : \mathbb{S} \to B\}$ is a Cauchy sequence, hence converges to a continuous map $\gamma_{\infty} : \mathbb{S} \to \partial B$. By construction, γ_{∞} is surjective.

Step 4. If furthermore ∂B is a Jordan curve, then ∂B is a quasi-circle.

Since ∂B is a Jordan curve, the Böttcher map $\phi : B \to \overline{\mathbb{C}} \setminus \overline{\mathbb{D}}$ can be extended to a homeomorphism $\phi : \overline{B} \to \overline{\mathbb{C}} \setminus \mathbb{D}$. Define a map $\psi : \mathbb{S} \to \partial B$ by $\psi(\zeta) = \phi^{-1}(\zeta)$ for $\zeta \in \mathbb{S}$. Then $f(\psi(\zeta)) = \psi(\zeta^d)$. Let $\varphi = \phi|_{\partial B}$ be the inverse of ψ . Both ψ and φ are uniformly continuous. Thus for any sufficiently small positive number ε , there are two small constants $a(\varepsilon), b(\varepsilon)$ such that

$$\forall \ (\zeta_1, \zeta_2) \in \mathbb{S} \times \mathbb{S}, |\zeta_1 - \zeta_2| < a(\varepsilon) \implies |\psi(\zeta_1) - \psi(\zeta_2)| < \varepsilon; \\ \forall \ (z_1, z_2) \in \partial B \times \partial B, |z_1 - z_2| < b(\varepsilon) \implies |\varphi(z_1) - \varphi(z_2)| < a(\varepsilon).$$

Given two points $z_1, z_2 \in \partial B$, $\partial B \setminus \{z_1, z_2\}$ consists of two components, say E_1 and E_2 . Let $L(z_1, z_2) \in \{\overline{E}_1, \overline{E}_2\}$ be the section of ∂B such that diam $(L(z_1, z_2)) = \min\{\text{diam}(E_1), \text{diam}(E_2)\}$. Thus for any positive number $\varepsilon \ll \text{diam}(\partial B)$, by uniform continuity, we have

$$|z_1 - z_2| < b(\varepsilon) \Rightarrow \operatorname{diam}(L(z_1, z_2)) < \varepsilon.$$
(5.1)

By Ahlfors' characterization of quasicircle [Ahl], to prove that ∂B is a quasicircle, it suffices to show that there is a constant C > 0 such that for any $z_1, z_2 \in \partial B$ with $z_1 \neq z_2$, $\Delta(L(z_1, z_2); z_1, z_2) \leq C$. In fact, if $|z_1 - z_2| \geq \varepsilon$ for some positive constant ε , then $\Delta(L(z_1, z_2); z_1, z_2) \leq \operatorname{diam}(\partial B)/\varepsilon$. So we just need consider the case when $|z_1 - z_2|$ is small. In the following, we assume $\delta \ll \operatorname{diam}(\partial B)$ and $|z_1 - z_2| \leq b(\delta/2)$, it turns out that $\operatorname{diam}(L(z_1, z_2)) < \delta/2$.

Since f is expanding on ∂B , there is an integer N > 0 such that $f^k(L(z_1, z_2)) = \partial B$ for all $k \ge N$. So we can find a smallest integer $\ell \ge 0$, such that

diam
$$(f^{\ell}(L(z_1, z_2))) < \delta/2, \text{ diam}(f^{\ell+1}(L(z_1, z_2))) \ge \delta/2.$$

On the other hand, there exist two points $w_1, w_2 \in f^{\ell}(L(z_1, z_2))$ such that

diam
$$(f^{\ell+1}(L(z_1, z_2))) = |f(w_1) - f(w_2)| \le \int_{[w_1, w_2]} |f'(z)| |dz|$$

 $\le M |w_1 - w_2| \le M \operatorname{diam}(f^{\ell}(L(z_1, z_2))),$

where $[w_1, w_2]$ is the straight segment connecting w_1 with w_2 and

 $M = \max\{|f'(z)|; \text{Eucl.dist}(z, \partial B) \le \delta/2\}.$

Thus we have

$$\frac{\delta}{2M} \le \text{diam}(f^{\ell}(L(z_1, z_2))) = \text{diam}(L(f^{\ell}(z_1), f^{\ell}(z_2))) < \frac{\delta}{2}$$

By (5.1), there is a constant $c(\delta, M) > 0$ such that $|f^{\ell}(z_1) - f^{\ell}(z_2)| \ge c(\delta, M)$.

Applying Theorem 2.3.2 to the following situation $(V_1, U_1) = (U_{\ell}(f^{\ell}(z_1)), V_{\ell}(f^{\ell}(z_1))), (V_2, U_2) = (B(f^{\ell}(z_1), \delta), B(f^{\ell}(z_1), \delta/2))$ and $g = f^{\ell}$, we conclude that there is a constant C(D) > 0 such that

$$\Delta(L(z_1, z_2); z_1, z_2) \le C(D)\Delta(f^{\ell}(L(z_1, z_2)); f^{\ell}(z_1), f^{\ell}(z_2)) \le \frac{C(D)\delta}{2c(\delta, M)}$$

Thus for any $x, y \in \partial B$ with $x \neq y$, the turning $\Delta(L(x, y); x, y)$ is bounded by

$$\max\left\{\frac{\operatorname{diam}(\partial B)}{b(\delta/2)}, \frac{C(D)\delta}{2c(\delta, M)}\right\}.$$

Remark 5.6.1. Using the same argument as [CJY], one can show further that if f satisfies **BD** condition on ∂B , then ∂B is a John domain.

The following proposition gives a criterion when f satisfies **BD** condition on ∂B .

Proposition 5.6.2. If $\#(P(f) \cap \partial B) < \infty$ and all periodic points in $P(f) \cap \partial B$ are repelling, then f satisfies **BD** condition on ∂B .

Proof. The proof is based on the following claim.

Claim: For any $u \in \partial B$, there is a constant $\varepsilon_u > 0$ such that for any $m \geq 0$ and any component $U_m(u)$ of $f^{-m}(B(u, \varepsilon_u))$ that intersects with ∂B , $U_m(u)$ contains at most one critical point of f^m .

The claim implies that $U_m(u)$ is simply connected by Riemann-Hurwitz Formula. Since the sequence $U_m(u) \to f(U_m(u)) \to \cdots \to f^{m-1}(U_m(u)) \to$

 $B(u, \varepsilon_u)$ meets every critical point of f at most once, we conclude that $\deg(f^m : U_m(u) \to B(u, \varepsilon_u))$ is bounded by $D = \prod_{c \in C(f)} \deg(f, c)$.

In the following, we prove the claim.

First note that every point in $P(f) \cap \partial B$ is pre-periodic, we can decompose ∂B into three disjoint sets: X, Y and Z, where $X = \partial B \setminus P(f), Z$ is the union of all repelling cycles in $P(f) \cap \partial B$ and $Y = (P(f) \cap \partial B) \setminus Z$.

For any $x \in X$, choose a small number $\varepsilon_x > 0$ such that $B(x, \varepsilon_x) \cap P(f) = \emptyset$. Then for any component $W_m(x)$ of $f^{-m}(B(x, \varepsilon_x))$ intersecting with ∂B , $f^m: W_m(x) \to B(x, \varepsilon_x)$ is a conformal map.

The set Y consists of all strictly pre-periodic points. Thus there is an integer $q \ge 1$ such that for any $y \in Y$, $f^{-q}(y) \cap P(f) \cap \partial B = \emptyset$. For a open set U in $\overline{\mathbb{C}}$ and a point $u \in U$, we use $\operatorname{Comp}_u(U)$ to denote the component of U that contains u. For every $y \in Y$, choose $\varepsilon_y > 0$ small enough such that for any $x \in f^{-q}(y) \cap \partial B \subset X$, $\operatorname{Comp}_x(f^{-q}(B(y, \varepsilon_y))) \subset B(x, \varepsilon_x)$ and $\operatorname{Comp}_x(f^{-q}(B(y, \varepsilon_y)))$ contains at most one critical point of f^q .

Finally, we deal with Z. For $z \in Z$, suppose z lies in a repelling cycle of period p. Choose $\varepsilon_z > 0$ such that

(1) $B(z, \varepsilon_z)$ is contained in the linearizable neighborhood of z and $\operatorname{Comp}_z(f^{-p}(B(z, \varepsilon_z)))$ is a subset of $B(z, \varepsilon_z)$,

(2) For every $u \in (f^{-p}(z) \cap \partial B) \setminus \{z\} \subset X \cup Y$, $\operatorname{Comp}_u(f^{-p}(B(z,\varepsilon_z)))$ contains at most one critical point of f^p and $\operatorname{Comp}_u(f^{-p}(B(z,\varepsilon_z))) \subset B(u,\varepsilon_u)$.

One can easily verify that the collection of neighborhoods $\{B(u, \varepsilon_u), u \in \partial B\}$ are just as required.

Corollary 5.6.1. If f is critically finite, then f satisfies **BD** condition on ∂B .

Proof. Since f is critically finite, every periodic point of f is either repelling or superattracting. This implies that $\#(P(f) \cap \partial B) < \infty$ and all periodic points in $P(f) \cap \partial B$ are repelling. Thus by Proposition 5.6.2, f satisfies **BD** condition on ∂B .

5.7 The boundary ∂B_{λ} is a Jordan curve

In this section, we will prove:

Theorem 5.7.1. For any $n \geq 3$ and any complex parameter λ , if the Julia set $J(f_{\lambda})$ is not a Cantor set, then ∂B_{λ} is a Jordan curve.

An immediate corollary of the theorem is the following:

Corollary 5.7.1. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then every Fatou component is a Jordan domain.

For the higher regularity of ∂B_{λ} , we show

Theorem 5.7.2. Suppose the Julia set $J(f_{\lambda})$ is not a Cantor set, then the boundary ∂B_{λ} is a quasicircle if it contains neither parabolic point nor recurrent critical point.

The strategy of the proof is as follows:

First, consider the McMullen maps f_{λ} with parameter $\lambda \in \mathcal{H}$. If f_{λ} is critically finite, then the Julia set is locally connected. Else, by Proposition 5.4.1, we can find an admissible graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. With respect to the Yoccoz puzzle induced by this graph, there are two possibilities of the critical tableaux:

Case 1: There is no periodic critical tableau. This case is discussed in section 5.7.1 and the local connectivity of $J(f_{\lambda})$ follows from Proposition 5.7.1. The idea of the proof is based on the combinatorial analysis for tableaux introduced by Branner and Hubbard (see [BH], [M2]), together with 'modified puzzle piece' techniques.

Case 2: There is a periodic critical tableau T(c). In this case, the map f_{λ} is either renormalizable or *-renormalizable. This case is discussed in section 5.7.2. The local connectivity of ∂B_{λ} follows from Proposition 5.7.2. The idea of the proof of Proposition 5.7.2 is to construct a closed curve separating ∂B_{λ} from the small filled Julia set K_c .

In section 5.7.3, we deal with the real parameters $\lambda \in \mathbb{R}^+$.

In section 5.7.4, we improve the regularity of the boundary ∂B_{λ} . We first include a proof of R.Devaney which claims that the local connectivity of ∂B_{λ} implies that ∂B_{λ} is a Jordan curve. Then we show ∂B_{λ} is a quasicircle except two specific cases.

In section 5.7.5, we present some corollaries.

5.7.1 No periodic critical tableau case

Recall that J_0 is the set of all points on the Julia set $J(f_{\lambda})$ whose orbits eventually meet the graph $\mathbf{G}_{\lambda}(\theta_1, \cdots, \theta_N)$.

Lemma 5.7.1. Let $z \in J(f_{\lambda}) \setminus J_0$. If T(z) is non-critical, then $End(z) := \bigcap_{d>0} \overline{P_d(z)} = \{z\}.$

Proof. It suffice to prove $\operatorname{End}(f_{\lambda}(z)) = \{f_{\lambda}(z)\}$. Since T(z) is non-critical, there is an integer $d_0 \geq 1$ such that for any j > 0, the position (d_0, j) is not critical. Let $\{\widehat{P}_{d_0-1}^{(i)}; 1 \leq i \leq M\}$ be the collection of all modified puzzle pieces of depth $d_0 - 1$, numbered so that $\widehat{P}_{d_0-1}^{(1)} = \widehat{P}_{d_0-1}(v_{\lambda}^+)$, $\widehat{P}_{d_0-1}^{(2)} = \widehat{P}_{d_0-1}(v_{\lambda}^-)$, recall that we use $\widehat{P}_d(w)$ to denote the modified puzzle piece of $P_d(w)$. Every modified puzzle piece of depth $\geq d_0$ is contained in a unique modified puzzle piece $\widehat{P}_{d_0-1}^{(i)}$ of depth d_0-1 . Let dist_i(x, y) be the Poincaré metric of $\widehat{P}_{d_0-1}^{(i)}$. For $2 < i \leq M$, there are exactly 2n branches of f_{λ}^{-1} on $\widehat{P}_{d_0-1}^{(i)}$, say $g_1^i, g_2^i, \cdots, g_{2n}^i$, and each g_k^i on $\widehat{P}_{d_0-1}^{(i)}$ is univalent and carries $\widehat{P}_{d_0}^{(\alpha)} \subset \widehat{P}_{d_0-1}^{(i)}$ onto a proper subset of some $\widehat{P}_{d_0-1}^{(j)}$. It follows that there is a uniform constant $0 < \nu < 1$, such that

 $\operatorname{dist}_j(g_k^i(x), g_k^i(y)) \le \nu \operatorname{dist}_i(x, y)$

for any $x, y \in \widehat{P}_{d_0}^{(\alpha)} \subset \subset \widehat{P}_{d_0-1}^{(i)}$ and any $2 < i \le M, 1 \le k \le 2n$.

Let D be the maximum of Poincaré diameters of the modified puzzle pieces of depth d_0 . For any integer h > 0, since the sequence

$$\widehat{P}_{d_0+h}(f_{\lambda}(z)) \to \widehat{P}_{d_0+h-1}(f_{\lambda}^2(z)) \to \cdots \widehat{P}_{d_0+1}(f_{\lambda}^h(z)) \to \widehat{P}_{d_0}(f_{\lambda}^{h+1}(z))$$

contains no critical point, it follows that

Hyper.diam
$$\left(\widehat{P}_{d_0+h}(f_{\lambda}(z))\right) \leq D\nu^h$$

with respect to the Poincaré metric of $\widehat{P}_{d_0-1}(f_{\lambda}(z))$. Thus we have $\underbrace{\bigcap_{d\geq 0} \widehat{P}_d(f_{\lambda}(z))}_{P_d(f_{\lambda}(z))} = \{f_{\lambda}(z)\}$. By the construction of modified puzzle piece, $\overline{P_d(f_{\lambda}(z))} \subset \widehat{P}_d(f_{\lambda}(z)) \cup A_{\lambda}$ for any $d \geq 0$, thus $\operatorname{End}(f_{\lambda}(z)) \subset \{f_{\lambda}(z)\} \cup A_{\lambda}$. Since $\operatorname{End}(f_{\lambda}(z))$ has no intersection with A_{λ} , $\operatorname{End}(f_{\lambda}(z)) = \{f_{\lambda}(z)\}$. \Box

Proposition 5.7.1. If T(c) is not periodic for any $c \in C_{\lambda}$, then the Julia set $J(f_{\lambda})$ is locally connected.

Proof. Note that T(c) is either critical or non-critical. First we prove $\text{End}(c) = \{c\}$ and $\text{End}(z) = \{z\}$ for any $z \in J(f_{\lambda}) \setminus J_0$. Then we deal with the points that lie in J_0 .

Case 1: T(c) is critical. Since the graph is admissible, we can find a non-degenerate annulus $A_{d_0}(c)$. Consider the descendants of $\operatorname{Row}_c(d_0)$. It's obvious that if $\operatorname{Row}_c(t)$ is a descendent in the k-th generation of $\operatorname{Row}_c(d_0)$, the annulus $A_t(c)$ is non-degenerate with modulus $\operatorname{mod}(A_{d_0}(c))/2^k$. If $\operatorname{Row}_c(d_0)$ has at least 2^k descendants in the k-th generation for each $k \geq 1$, then each of these contributes exactly $\operatorname{mod}(A_{d_0}(c))/2^k$ to the sum $\sum_d \operatorname{mod}(A_d(c))$. Hence $\sum_d \operatorname{mod}(A_d(c)) = \infty$, as required. On the other hand, if there are fewer descendants in some generation, then one of them, say $\operatorname{Row}_c(m)$ must be an only child, hence excellent by Lemma 5.4.7. Again by Lemma 5.4.7, we see that $\sum_d \operatorname{mod}(A_d(c)) = \infty$. Therefore in either case, $\operatorname{End}(c) = \{c\}$.

Now, consider a point $z \in J(f_{\lambda}) \setminus (J_0 \cup C_{\lambda})$. If T(z) is non-critical, then by Lemma 5.7.1, End $(z) = \{z\}$. If T(z) is critical, then for each $d \ge 1$, there is a smallest integer $l_d \ge 0$ such that both (d, l_d) and (d, l_d+1) are critical positions. It follows that $f_{\lambda}^{l_d} : A_{d+l_d}(z) \to A_d(c')$ is a conformal map for some $c' \in C_{\lambda}$. In this case $\sum_{d \mod A_d(z)} \ge \sum_{d \mod A_{d+l_d}(z)} = \sum_{d \mod A_d(c)} = \infty$, hence $\operatorname{End}(z) = \{z\}.$

Case 2: T(c) is non-critical. It follows from Lemma 5.7.1 that $\operatorname{End}(c) = \{c\}$. For $z \in J(f_{\lambda}) \setminus (J_0 \cup C_{\lambda})$, we assume T(z) is critical, for else $\operatorname{End}(z) = \{z\}$ by Lemma 5.7.1. Suppose $A_{d_0}(c)$ is a non-degenerate annulus, and $(d_0 + 1, l_1), (d_0 + 1, l_2), \cdots$ are all critical positions in the $(d_0 + 1)$ -th row of the tableau T(z). Since all critical tableaux are non-critical, there is a constant D such that $\operatorname{deg}(f_{\lambda}^{l_k}: P_{d_0+l_k}(z) \to P_{d_0,l_k}(z)) \leq D$ for all $k \geq 1$. Thus

$$\operatorname{mod}(A_{d_0+l_k}(z)) \ge D^{-1} \operatorname{mod}(A_{d_0}(c))$$

for all $k \ge 1$. Hence $\sum_d \operatorname{mod}(A_d(z)) \ge \sum_k \operatorname{mod}(A_{d_0+l_k}(z)) = \infty$ and $\operatorname{End}(z) = \{z\}.$

Points that lie in J_0 . For any $z \in J_0$, the orbit $z \mapsto f_{\lambda}(z) \mapsto f_{\lambda}^2(z) \mapsto \cdots$ eventually meets the graph $\mathbf{G}_{\lambda}(\theta_1, \cdots, \theta_N)$. So the Euclidean distance between the critical set C_{λ} and the orbit $\{f_{\lambda}^k(z)\}_{k\geq 0}$ is bounded below by some positive number $\varepsilon(z)$. In addition, for every d large enough, z lies in the common boundary of exactly two puzzle pieces of depth d. We denote these two puzzle pieces by $P'_d(z)$ and $P''_d(z)$. In the previous argument, we have already proved that $\operatorname{End}(c) = \{c\}$, this implies $\operatorname{Eucl.diam}(P_d(c)) \to 0$ as $d \to \infty$. Choose d_0 large enough such that

Eucl.diam
$$(P_{d_0}(c)) < \varepsilon(z) \leq \text{Eucl.dist}(C_{\lambda}, \{f_{\lambda}^k(z)\}_{k\geq 0}).$$

Then the orbit $z \mapsto f_{\lambda}(z) \mapsto f_{\lambda}^2(z) \mapsto \cdots$ avoids all the critical puzzle pieces of depth d_0 . Let $P_d^*(z) = \overline{P'_d(z) \cup P''_d(z)}$ for d large enough. Then the proof of Lemma 5.7.1 applies equally well to this situation and $\bigcap_d P_d^*(z) = \{z\}$ follows immediately.

Connectivity of neighborhoods. Let

$$P_d^*(z) = \begin{cases} \overline{P_d(z)}, & \text{if } z \in J(f_\lambda) \setminus J_0, \\ \overline{P_d'(z) \cup P_d''(z)}, & \text{if } z \in J_0 \text{ and } d \text{ is large.} \end{cases}$$

By Lemma 5.4.2, for every $z \in J(f_{\lambda})$ and every large integer d, the intersection $P_d^*(z) \cap J(f_{\lambda})$ is a connected and compact subset of $J(f_{\lambda})$. Thus $\{P_d^*(z) \cap J(f_{\lambda})\}$ forms a basis of connected neighborhoods of z. Since $\bigcap(P_d^*(z) \cap J(f_{\lambda})) = \{z\}$, the Julia set is locally connected at z. Note that zis arbitrarily chosen, $J(f_{\lambda})$ is locally connected.

5.7.2 Periodic critical tableau case

Suppose the critical tableau T(c) is k-periodic for some k > 0. By the proof of Lemma 5.5.1, f_{λ} is either k-renormalizable at c or k/2-*-renormalizable at c.

Let $(\varepsilon f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ be the renormalization, where d_{0} is a large integer and

$$(\varepsilon, p) = \begin{cases} (1, k), & \text{if } f_{\lambda} \text{ is } k \text{-renormalizable at } c, \\ (-1, k/2), & \text{if } f_{\lambda} \text{ is } k/2 \text{-}*\text{-renormalizable at } c. \end{cases}$$

The small filled Julia set of the renormalization $(\varepsilon f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ is denoted by K_{c} . Recall that β_{c} is the β -fixed point of the renormalization and β'_{c} is the other preimage of β_{c} under the map $\varepsilon f_{\lambda}^{p}|_{P_{d_{0}+p}(c)}$.

Assume now $K_c \cap \partial B_\lambda \neq \emptyset$, then by Lemma 5.5.3, $\beta_c \in K_c \cap \partial B_\lambda$ and there is a unique external ray, say $R_\lambda(\theta)$, landing at β_c . The angle θ is of the form $\frac{m}{2^k-1}$. It follows that $\beta'_c \in K_c \cap \partial T_\lambda$ and there is a unique radical ray $R_{T_\lambda}(\alpha_\theta)$ in T_λ landing at β'_c . The radical ray $R_{T_\lambda}(\alpha_\theta)$ satisfies $\varepsilon f^p_\lambda(R_{T_\lambda}(\alpha_\theta)) = R_\lambda(\theta)$. Let

$$K = K_c \cup \overline{R_{\lambda}(\theta)} \cup \overline{R_{T_{\lambda}}(\alpha_{\theta})} \cup (-K_c) \cup (-\overline{R_{\lambda}(\theta)}) \cup (-\overline{R_{T_{\lambda}}(\alpha_{\theta})}).$$

The set K is a connected and compact subset of $\overline{\mathbb{C}}$. Note that $-R_{T_{\lambda}}(\alpha_{\theta}) = R_{T_{\lambda}}(\alpha_{\theta} + 1/2)$. Let Δ_1 be the component of $\overline{\mathbb{C}} \setminus (K \cup \overline{B_{\lambda}})$ that intersects with $Q_{T_{\lambda}}(\alpha_{\theta}, \alpha_{\theta} + 1/2)$ and Δ_2 be the component of $\overline{\mathbb{C}} \setminus (K \cup \overline{B_{\lambda}})$ that intersects with $Q_{T_{\lambda}}(\alpha_{\theta} + 1/2, \alpha_{\theta})$, where we use $Q_{T_{\lambda}}(\theta_1, \theta_2)$ to denote the set $\overline{\{\phi_{T_{\lambda}}(re^{2\pi i t}); 0 < r < 1, \theta_1 \leq t \leq \theta_2\}}$. Since $K \cup \overline{B_{\lambda}}$ is connected and compact, both Δ_1 and Δ_2 are disks. Let Z_i be the component of $\overline{\mathbb{C}} \setminus K$ that contains Δ_i .

The aim of this section is to prove:

Proposition 5.7.2. Assume that $K_c \cap \partial B_\lambda \neq \emptyset$, then for $i \in \{1, 2\}$, there is a curve $\mathcal{L}_i \subset \Delta_i \cup \{0\}$ stemming from T_λ and converging to β_c . More precisely, \mathcal{L}_i can be parameterized as $\mathcal{L}_i : [0, +\infty) \to \Delta_i \cup \{0\}$ such that $\mathcal{L}_i(0) = 0, \mathcal{L}_i((0, +\infty)) \subset \Delta_i$ and $\lim_{t\to +\infty} \mathcal{L}_i(t) = \beta_c$.

Proof. Let $\Gamma = \bigcup_{j\geq 0} (\pm f_{\lambda}^{j}(K_{c} \cup \overline{R_{\lambda}(\theta)}))$. By Lemma 5.5.3, any two different elements in the set $\{\pm f_{\lambda}^{j}(K_{c} \cup \overline{R_{\lambda}(\theta)}); j \geq 0\}$ intersect only at the point ∞ . This implies $U = \overline{\mathbb{C}} \setminus \Gamma$ is a disk.

Step 1. There exists $G_i : U \to U \cap Z_i$, which is an inverse branch of $\varepsilon f_{\lambda}^p$, such that the sequence $\{G_i^k; k \ge 0\}$ converges locally and uniformly in U to a constant $z_i \in K_c$.

Since U has no intersection with the critical orbits, its preimage $f_{\lambda}^{-1}(U)$ has exactly 2n components, say V_1, \dots, V_{2n} . These components are arranged symmetrically about the origin under the rotation $z \mapsto e^{\pi i/n} z$. For every $1 \leq j \leq 2n, f_{\lambda} : V_j \to U$ is a conformal map. Moreover, $f_{\lambda}^{-1}(U) \subset \mathbb{C} \setminus K$.

For $1 \leq j \leq p-1$, let Ω_j be the component of $f_{\lambda}^{-1}(U)$ such that $\overline{\Omega}_j \cap f_{\lambda}^j(K_c) \neq \emptyset$ and the inverse of $f_{\lambda} : \Omega_j \to U$ is denoted by g_j . For j = 0, let Ω_0^i



Figure 5.10: Constructing two curves L_1 and L_2 that converge to β_c , here n = 3 and f_{λ} is 1-renormalizable at $c = c_0$.

be the component of $f_{\lambda}^{-1}(U)$ such that $\overline{\Omega_0^i} \cap K_c \neq \emptyset$ and $\Omega_0^i \subset Z_i$. The inverse of $f_{\lambda} : \Omega_0^i \to U$ is denoted by g_0^i for $i \in \{1, 2\}$.

Now, we define

$$G_i(z) = \begin{cases} g_0^i \circ g_1 \circ \cdots \circ g_{p-1}(\varepsilon z), & z \in U & \text{if } p \ge 2, \\ g_0^i(\varepsilon z), & z \in U & \text{if } p = 1. \end{cases}$$

Since $(\varepsilon f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ is a p-(*-)renormalization of f_{λ} at c, we have $G_{i}(P_{d_{0}}(c) \cap U) \subset P_{d_{0}+p}(c) \cap Z_{i}$. The map $G_{i} : U \to U$ is not surjective, thus by Denjoy-Wolff theorem(See [M1]), the sequence $\{G_{i}^{k}; k \geq 0\}$ converges locally and uniformly in U to a constant z_{i} . It follows from $G_{i}(P_{d_{0}}(c) \cap U) \subset P_{d_{0}+p}(c) \cap Z_{i}$ that $z_{i} \in K_{c}$.

Step 2. There exists a curve $C_i \subset U \cap (\Delta_i \cup \{0\})$ connecting 0 with $G_i(0)$ for $i \in \{1, 2\}$.

Since the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$ is admissible, the filled Julia set K_c is disjoint from the boundary of any puzzle piece. Thus for any $\alpha \in \{\tau^s(\theta_j); 1 \leq j \leq N, s \geq 0\}$, Γ is disjoint from the cut ray $\Omega^{\alpha}_{\lambda}$ outside ∞ . For any angle $\alpha \in \{\tau^s(\theta_j); 1 \leq j \leq N, s \geq 0\}$ and any map $g \in \{g_0^1, g_0^2, g_1, \dots, g_{p-1}\}$, by Proposition 5.4.2, only one curve of $g(\omega^{\alpha}_{\lambda} \setminus \{\infty\}), g(\omega^{\alpha+1/2}_{\lambda} \setminus \{\infty\})$ intersects with ∂B_{λ} and the other curve connects 0 with a preimage of 0.

Fix an angle $\alpha \in \{\tau^s(\theta_j); 1 \leq j \leq N, s \geq 0\}$, we define a curve family \mathcal{F} by

$$\mathcal{F} = \{ \varepsilon \omega_{\lambda}^{\alpha} \setminus \{ \infty \}; \ \varepsilon^{2n} = 1 \text{ and } \ \varepsilon \omega_{\lambda}^{\alpha} \subset \bigcup_{j \in \mathbb{I} \setminus \{0,n\}} S_j \}$$

We construct the curve C_i by inductive procedure as following:

First, choose a curve $\zeta_{p-1} \in \mathcal{F}$ such that $g_{p-1}(\zeta_{p-1}) \cap \partial B_{\lambda} = \emptyset$, and let $\gamma_{p-1} = g_{p-1}(\zeta_{p-1})$. Suppose that for some $2 \leq j \leq p-1$, we have already constructed the curves $\gamma_{p-1}, \dots, \gamma_j$. Then we choose $\zeta_{j-1} \in \mathcal{F}$ such that $g_{j-1}(\zeta_{j-1}) \cap \partial B_{\lambda} = \emptyset$ and $\zeta_{j-1} \cap \gamma_j = \emptyset$, and let $\gamma_{j-1} = g_{j-1}(\zeta_{j-1} \cup \gamma_j)$. In this way, we can construct a sequence of curves $\gamma_{p-1}, \gamma_{p-2}, \dots, \gamma_2, \gamma_1$ step by step and each curve has no intersection with ∂B_{λ} , connecting 0 with some iterated preimage of 0. By construction,

$$\gamma_1 = \bigcup_{1 \le j \le p-1} g_1 \circ \cdots \circ g_j(\zeta_j).$$

Now, we choose $\zeta_0^i \in \mathcal{F}$ such that $g_0^i(\zeta_0^i) \cap \partial B_\lambda = \emptyset$ and $\zeta_0^i \cap \gamma_1 = \emptyset$, and let

$$C_i = \begin{cases} g_0^i(\zeta_0^i \cup \gamma_1) \cup \{0\}, & \text{if } p \ge 2, \\ g_0^i(\zeta_0^i) \cup \{0\}, & \text{if } p = 1. \end{cases}$$

The curve C_i connects 0 with $G_i(0)$ and $C_i \subset U \cap (\Delta_i \cup \{0\})$, as required. **Step 3.** The union $\mathcal{L}_i = \bigcup_{j \geq 0} G_i^j(C_i)$ is the curve contained in $\Delta_i \cup \{0\}$ and converging to β_c .

By construction, $G_i(\mathcal{L}_i) \subset G_i(\mathcal{L}_i) \cup C_i = \mathcal{L}_i$ and $\mathcal{L}_i \setminus \{0\} \subset \Delta_i$.

To finish, we show \mathcal{L}_i converges to β_c . By step 1, the sequence $\{G_i^k; k \ge 0\}$ converges uniformly on any compact subset of U to a constant $z_i \in K_c$. Since C_i is a compact subset of U, the curve \mathcal{L}_i converges to $z_i \in K_c$ and $G_i(z_i) = z_i$. Since $\varepsilon f_{\lambda}^p(\mathcal{L}_i) \supset \mathcal{L}_i$, we conclude $z_i = \beta_c$ by Lemma 5.5.2.

Corollary 5.7.2. If T(c) is periodic for some $c \in C_{\lambda}$, then ∂B_{λ} is locally connected.

Proof. We may assume that f_{λ} is not geometrically finite, otherwise the Julia set is locally connected (see [**TY**]). Thus f_{λ} has no parabolic point.

If $K_c \cap \partial B_\lambda = \emptyset$, then for all $j \ge 0$, $f_\lambda^j(K_c) \cap \partial B_\lambda = \emptyset$. Since $P(f_\lambda)$ is a subset of $(\bigcup_{j\ge 0} f_\lambda^j(\pm f_\lambda(K_c))) \bigcup \{\infty\}$, we conclude $P(f_\lambda) \cap \partial B_\lambda = \emptyset$. By Proposition 5.6.1 and Proposition 5.6.2, ∂B_λ is locally connected.

If $K_c \cap \partial B_\lambda \neq \emptyset$, then by Proposition 5.7.2, the closed curve $\mathcal{L} = \mathcal{L}_1 \cup \mathcal{L}_2 \cup \{\beta_c\}$ separates $K_c \setminus \{\beta_c\}$ from $\partial B_\lambda \setminus \{\beta_c\}$. In this case, for all $j \geq 0$, $f_\lambda^j(K_c) \cap \partial B_\lambda = \{f_\lambda^j(\beta_c)\}$. Thus $\#(P(f_\lambda) \cap \partial B_\lambda) < \infty$ and all periodic points in $P(f_\lambda) \cap \partial B_\lambda$ are repelling. Again by Proposition 5.6.1 and Proposition 5.6.2, ∂B_λ is locally connected.

5.7.3 Real case

In this section, we will deal with real parameters. By the symmetry of the parameter plane, we just need consider $\lambda \in \mathbb{R}^+ = (0, +\infty)$. In this case, the

Julia set $J(f_{\lambda})$ is symmetric about the real axis. If $C_{\lambda} \subset A_{\lambda}$, by 'The Escape Trichotomy' (Theorem 5.2.1), the Julia set $J(f_{\lambda})$ is either a Cantor set, a Cantor set of circles or a Sierpinski curve. The local connectivity of ∂B_{λ} is already known in the latter two cases. In the following discussion, we assume $C_{\lambda} \cap A_{\lambda} = \emptyset$.

Lemma 5.7.2. Suppose $\lambda \in \mathbb{R}^+$ and $C_{\lambda} \cap A_{\lambda} = \emptyset$, then f_{λ} is 1-renormalizable at $c_0 = \sqrt[2n]{\lambda}$.

Proof. Let U be the interior of $(S_0 \cup S_{-(n-1)}) \setminus \{z \in B_\lambda \cup T_\lambda; G_\lambda(z) \ge 1\}$ and $V = \overline{\mathbb{C}} \setminus (\{z \in B_\lambda; G_\lambda(z) \ge n\} \cup [-\infty, v_\lambda^-])$. One can easily verify that $f_\lambda : U \to V$ is a quadratic-like map. Since $C_\lambda \cap A_\lambda = \emptyset$, the critical orbit $\{f_\lambda^k(c_0); k \ge 0\}$ is contained in $U \cap \mathbb{R}^+$. This implies that (f_λ, U, V) is a 1-renormalization of f_λ at c_0 .

Let $K_{c_0} = \bigcap_{k\geq 0} f_{\lambda}^{-k}(U)$ be the small filled Julia set of the renormalization (f_{λ}, U, V) , β_{c_0} be the β -fixed point and β'_{c_0} be the preimage of β_{c_0} . It's easy to check that K_{c_0} is symmetric about the real axis and $K_{c_0} \cap \mathbb{R}^+$ is a connected and closed interval.

Proposition 5.7.3. $K_{c_0} \cap \partial B_{\lambda} = \{\beta_{c_0}\}.$

Proof. The idea of the proof is to construct a Jordan curve C that separates $K_{c_0} \setminus \{\beta_{c_0}\}$ from $\partial B_{\lambda} \setminus \{\beta_{c_0}\}$, similar as the proof of Proposition 5.7.2.

We first show that β_{c_0} is the landing point of the zero external ray $R_{\lambda}(0)$. Note that rational external rays (i.e. external rays with a rational angle) always land. Let z_0 be the landing point of $R_{\lambda}(0)$. Obviously, $R_{\lambda}(0) \subset \mathbb{R}^+$ and z_0 is a fixed point of f_{λ} . This implies $z_0 \in U \cap \mathbb{R}^+$ and the orbit of z_0 does not escape from U, so $z_0 \in K_{c_0}$. Since $R_{\lambda}(0)$ is an f_{λ} -invariant ray that lands at z_0 , we conclude $z_0 = \beta_{c_0}$ by Lemma 5.5.2.

Let $K = K_{c_0} \cup [\beta_{c_0}, +\infty] \cup (-K_{c_0}) \cup [-\infty, -\beta_{c_0}]$. One can easily verify $f_{\lambda}^{-1}(K) = \bigcup_{\omega^{2n}=1} \omega(K_{c_0} \cup [0, +\infty])$. The set $Y = \overline{\mathbb{C}} \setminus K$ is a disk and its preimage $f_{\lambda}^{-1}(Y)$ consists of 2n components, which are symmetric about the origin under the rotation $z \mapsto e^{i\pi/n}z$. For each component X of $f_{\lambda}^{-1}(Y)$, $f_{\lambda}: X \to Y$ is a conformal map. Let X_0 be the component of $f_{\lambda}^{-1}(Y)$ that is contained in S_0 and g be the inverse map of $f_{\lambda}: X_0 \to Y$. By Denjoy-Wolff theorem, the sequence of maps $\{g^k; k \ge 0\}$ converges locally and uniformly in Y to a constant, say x. Since $g(Y \cap V) \subset X_0 \cap U$, we conclude $x \in K_{c_0}$.

Let Δ be the component of $\mathbb{C} \setminus (B_{\lambda} \cup K_{c_0} \cup (-K_{c_0}) \cup \mathbb{R})$ that intersects with T_{λ} and lies in the upper half plane.

Claim: There is a path $\mathcal{L} \subset \Delta \cup \{0\}$ stemming from T_{λ} and converging to β_{c_0} . More precisely, \mathcal{L} can be parameterized as $\mathcal{L} : [0, +\infty) \to \Delta \cup \{0\}$ such that $\mathcal{L}(0) = 0, \mathcal{L}((0, +\infty)) \subset \Delta$ and $\lim_{t\to +\infty} \mathcal{L}(t) = \beta_{c_0}$

Let $p_0 = \sqrt[2n]{-\lambda}$ be the preimage of 0 that lies in S_0 and $\gamma_0 = [0, p_0]$ be the segment connecting 0 with p_0 . Then $\gamma_0 \cap (K_{c_0} \cup \partial B_{\lambda}) = \emptyset$. Indeed, $\gamma_0 \cap K_{c_0} = \emptyset$ follows from the fact that $f_{\lambda}(\gamma_0) \cap K_{c_0} \subset i\mathbb{R} \cap K_{c_0} = \emptyset$. In the following, we show $\gamma_0 \cap \partial B_{\lambda} = \emptyset$. It suffices to show $B_{\lambda} \cap D = \emptyset$, where $D = \{z \in \mathbb{C}; |z| < \sqrt[2n]{\lambda}\}$. Otherwise, $B_{\lambda} \cap D \neq \emptyset$ implies $B_{\lambda} \cap \partial D \neq \emptyset$. Since $\varphi : z \mapsto \sqrt[n]{\lambda}/\overline{z}$ maps B_{λ} onto T_{λ} and the restriction $\varphi|_{\partial D}$ is the identity map, we have $B_{\lambda} \cap \partial D = \varphi(B_{\lambda} \cap \partial D) = T_{\lambda} \cap \partial D$. But this is a contradiction.

Note that g maps γ_0 outside D and $g(\gamma_0)$ connects p_0 with a preimage of p_0 that lies inside S_0 . Let $\mathcal{L} = \bigcup_{k\geq 0} g^k(\gamma_0)$. By construction, $\mathcal{L} \cap (K_{c_0} \cup \partial B_{\lambda}) = \emptyset$ and \mathcal{L} converges to $x \in K_{c_0}$. Since $f_{\lambda}(\mathcal{L}) = \mathcal{L} \cup f_{\lambda}(\gamma_0) \supset \mathcal{L}$, we conclude $x = \beta_{c_0}$ by Lemma 5.5.2.

Let $\mathcal{C} = \mathcal{L} \cup \mathcal{L}^* \cup \{\beta_{c_0}\}$, where $\mathcal{L}^* = \{\overline{z}; z \in \mathcal{L}\}$. \mathcal{C} is a Jordan curve separating $K_{c_0} \setminus \{\beta_{c_0}\}$ from $\partial B_{\lambda} \setminus \{\beta_{c_0}\}$. The conclusion follows. \Box

Remark 5.7.1. From the proof of Proposition 5.7.3, we conclude

$$\partial B_{\lambda} \cap \mathbb{R} = \{\pm \beta_{c_0}\}, \ K_{c_0} \cap \mathbb{R} = [\beta'_{c_0}, \beta_{c_0}], \ \partial T_{\lambda} \cap \mathbb{R} = \{\pm \beta'_{c_0}\}.$$

Corollary 5.7.3. Suppose $\lambda \in \mathbb{R}^+$ and $C_{\lambda} \cap A_{\lambda} = \emptyset$, then ∂B_{λ} is locally connected.

Proof. By Proposition 5.7.3, if n is odd, then $P(f_{\lambda}) \cap \partial B_{\lambda} \subset (-K_{c_0} \cup K_{c_0}) \cap \partial B_{\lambda} \subset \{\pm \beta_{c_0}\}$; if n is even, then $P(f_{\lambda}) \cap \partial B_{\lambda} \subset K_{c_0} \cap \partial B_{\lambda} \subset \{\beta_{c_0}\}$. If β_{c_0} is a parabolic point, then f_{λ} is geometrically finite, the local connectivity of ∂B_{λ} follows from [TY]. Else, by Proposition 5.6.1 and Proposition 5.6.2, ∂B_{λ} is also locally connected.

5.7.4 Local connectivity implies higher regularity

Up to now, we have already proved that ∂B_{λ} is locally connected if the Julia set is not a Cantor set. By the arguments of Devaney [D1], we prove the following proposition which will lead to Theorem 5.1.1.

Proposition 5.7.4. If ∂B_{λ} is locally connected, then ∂B_{λ} is a Jordan curve.

Proof. Let W_0 be the component of $\overline{\mathbb{C}} - \overline{B}_{\lambda}$ containing 0. It's obvious that $\partial W_0 \subset \partial B_{\lambda}, \ T_{\lambda} \subset W_0, \ \partial T_{\lambda} \subset \overline{W}_0$. By Lemma 5.2.1, $e^{i\pi/n}W_0 = W_0$.

Recall that $H_{\lambda}(z) = \sqrt[n]{\lambda}/z$, so $H_{\lambda}(\partial W_0) \subset H_{\lambda}(\partial B_{\lambda}) = \partial T_{\lambda} \subset \overline{W}_0$. Since ∂B_{λ} is locally connected, ∂W_0 is locally connected. It follows that $\overline{\mathbb{C}} - \overline{W}_0$ is connected and $H_{\lambda}(\overline{\mathbb{C}} - \overline{W}_0) \subset W_0$.

Now we show $f_{\lambda}^{-1}(0) \subset W_0$. If not, $f_{\lambda}^{-1}(0) \cap (\overline{\mathbb{C}} - \overline{W}_0) \neq \emptyset$. By the symmetry of $f_{\lambda}^{-1}(0)$ and $\overline{\mathbb{C}} - \overline{W}_0$, we have $f_{\lambda}^{-1}(0) \subset \overline{\mathbb{C}} - \overline{W}_0$. This will contradict the fact that $f_{\lambda}^{-1}(0) = H_{\lambda}(f_{\lambda}^{-1}(0)) \subset H_{\lambda}(\overline{\mathbb{C}} - \overline{W}_0) \subset W_0$.

Since any point on ∂W_0 can not be mapped into W_0 , we have $f_{\lambda}^{-1}(W_0) \subset W_0$ and $f_{\lambda}^{-1}(\overline{W}_0) \subset \overline{W}_0$. Take a point $z \in \partial W_0$, we have $\partial B_{\lambda} \subset J(f_{\lambda}) = \bigcup_{k>0} f_{\lambda}^{-k}(z) \subset \overline{W}_0$ and $\partial B_{\lambda} \subset \partial W_0$. Therefore $\partial W_0 = \partial B_{\lambda}$.

Now we show that ∂B_{λ} is a Jordan curve. If two different external rays, say $R_{\lambda}(t_1)$ and $R_{\lambda}(t_2)$, land at the same point $p \in \partial B_{\lambda}$, then $\overline{R_{\lambda}(t_1) \cup R_{\lambda}(t_2)}$ decomposes ∂B_{λ} into two parts. It turns out that $\partial W_0 \neq \partial B_{\lambda}$, which is a contradiction.

The aim of this section is to prove that ∂B_{λ} is a quasicircle in almost all cases. Formally, we have the following

Theorem 5.7.3. Suppose the Julia set $J(f_{\lambda})$ is not a Cantor set, then the boundary ∂B_{λ} is a quasicircle if it contains neither parabolic point nor recurrent critical point.

Proof. By Theorem 1 and Proposition 5.6.1, it suffices to show that f_{λ} satisfies **BD** condition on ∂B_{λ} . First we deal with three special cases:

Case 1. The critical orbit escapes to infinity.

Case 2. The parameter $\lambda \in \mathbb{R}^+$ and ∂B_{λ} contains no parabolic point.

Case 3. The map f_{λ} is critically finite.

In case 1, $P(f_{\lambda}) \cap \partial B_{\lambda} = \emptyset$. By Proposition 5.6.2, f_{λ} satisfies **BD** condition on ∂B_{λ} . For case 2, by Proposition 5.7.4, either $P(f_{\lambda}) \cap \partial B_{\lambda} = \emptyset$ or $P(f_{\lambda}) \cap \partial B_{\lambda} = \{\beta_c\}$ or $P(f_{\lambda}) \cap \partial B_{\lambda} = \{\pm \beta_c\}$. In either case, β_c is a repelling fixed point of f_{λ} . By Proposition 5.6.1, f_{λ} satisfies **BD** condition on ∂B_{λ} . For case 3, f_{λ} satisfies **BD** condition on ∂B_{λ} by Corollary 5.6.2.

In the remaining cases, we can use Yoccoz puzzle to study the higher regularity of ∂B_{λ} . There are two remaining cases:

Case 4. ∂B_{λ} contains no critical point.

Case 5. $C_{\lambda} \subset \partial B_{\lambda}$ and all critical points in C_{λ} are non-recurrent.

In either case, by Proposition 5.4.1, we can find an admissible graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. With respect to the Yoccoz puzzle induced by this graph, we consider the critical tableaux. For case 4, there are two possibilities:

Case 4.1. There is a periodic critical tableau T(c).

Case 4.2. There is no periodic critical tableau.

For case 4.1, we conclude from Proposition 5.7.2 that $\#(P(f_{\lambda}) \cap \partial B_{\lambda}) < \infty$. Since ∂B_{λ} contains no parabolic point, all periodic points in $P(f_{\lambda}) \cap \partial B_{\lambda}$ are repelling. Thus by Proposition 5.6.2, f_{λ} satisfies **BD** condition on ∂B_{λ} .

For case 4.2, we have already shown that $\operatorname{End}(c) = \bigcap_{d \ge 0} P_d(c) = \{c\}$ for $c \in C_{\lambda}$ in the proof of Proposition 5.7.1. Thus we can choose d_0 large enough such that

Eucl.diam $(P_{d_0}(c)) < \text{Eucl.dist}(c, \partial B_{\lambda}).$

For $d \geq d_0$, let U_d be the union of all puzzle pieces of depth d that intersect with ∂B_{λ} and V_d be the interior of $\overline{U_d}$. For every $u \in \partial B_{\lambda}$, there is a number $\varepsilon_u > 0$ such that $B(u, \varepsilon_u) \subset V_{d_0}$. For any $m \geq 0$ and any component $U_m(u)$ of $f_{\lambda}^{-m}(B(u, \varepsilon_u))$ intersecting with ∂B_{λ} , $U_m(u) \subset V_{d_0+m} \subset V_{d_0}$. By the choice of d_0 , the sequence $U_m(u) \to \cdots \to f_{\lambda}^{m-1}(U_m(u)) \to B(u, \varepsilon_u)$ meets no critical point of f_{λ} , thus $f_{\lambda}^m : U_m(u) \to B(u, \varepsilon_u)$ is a conformal map. So in this case, f_{λ} satisfies **BD** condition on ∂B_{λ} .

In the following, we deal with case 5. Again by Proposition 5.7.1, $\operatorname{End}(c) = \{c\}$ for $c \in C_{\lambda}$. Thus in this case, one can verify that ∂B_{λ} contains no recurrent critical point if and only if all critical tableaux are non-critical. By Lemma 5.5.1, f_{λ} is critically finite. It follows from Corollary 5.6.1 that f_{λ} satisfies **BD** condition on ∂B_{λ} .

5.7.5 Corollaries

In this section, we present some corollaries of Theorem 5.1.1.

Proposition 5.7.5. If ∂B_{λ} contains a parabolic cycle, then the multiplier of the cycle is 1 and the Julia set $J(f_{\lambda})$ contains a quasiconformal copy of quadratic Julia set of $z \mapsto z^2 + 1/4$.

Proof. Suppose $\mathcal{C} = \{z_0, f_{\lambda}(z_0), \cdots, f_{\lambda}^q(z_0) = z_0\}$ is a parabolic cycle on ∂B_{λ} . We will first consider the case $\lambda \in \mathbb{R}^+$, then deal with the case $\lambda \in \mathcal{H}$.

First suppose $\lambda \in \mathbb{R}^+$. By Lemma 5.7.2 and Proposition 5.7.3, f_{λ} is 1-renormalizable at c_0 and $P(f_{\lambda}) \cap \partial B_{\lambda} \subset (-K_{c_0} \cup K_{c_0}) \cap \partial B_{\lambda} = \{\pm \beta_{c_0}\}$. Since a parabolic point must attract a critical point, we conclude that β_{c_0} is a parabolic fixed point of f_{λ} . So (f_{λ}, U, V) is quasiconformally conjugate to a quadratic polynomial $z \mapsto z^2 + \mu$ whose β -fixed point is also a parabolic point, thus $\mu = 1/4$. The conclusion follows in this case.

In the following, we deal with the case $\lambda \in \mathcal{H}$. By Proposition 5.4.1, we can find an admissible graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. By Proposition 5.3.4, the parabolic cycle \mathcal{C} avoid the graph $\mathbf{G}_{\lambda}(\theta_1, \dots, \theta_N)$. With respect to the Yoccoz puzzle induced by this graph and by the similar argument as Corollary 5.5.1, we conclude that there is a critical point $c \in C_{\lambda}$ and a point $z \in \mathcal{C}$ such that $P_d(z) = P_d(c)$ for all $d \geq 0$. Thus the critical tableau T(c) is periodic. Suppose the period of T(c) is k. It is obvious that k is a divisor of q. By Lemma 5.5.1, when d_0 is large enough, the triple $(\varepsilon f_{\lambda}^p, P_{d_0+p}(c), P_{d_0}(c))$ is either a k-renormalization of f_{λ} at c (in this case, $(\varepsilon, p) = (-1, k/2)$). Moreover, the small filled Julia set $K_c = \operatorname{End}(c) = \bigcap_{d \geq 0} \overline{P_d(c)}$ and $z \in K_c \cap \partial B_{\lambda}$.

On the other hand, by Lemma 5.5.3, there is a unique external ray $R_{\lambda}(t)$ landing at β_c , which is the β -fixed point of the renormalization
$(\varepsilon f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$. Note that we have already proved that ∂B_{λ} is a Jordan curve, the intersection $\partial B_{\lambda} \cap \overline{P_{d}(c)}$ shrinks to a single point as $d \to \infty$. Thus we have $K_{c} \cap \partial B_{\lambda} = \{\beta_{c}\}$. By the previous argument, $\beta_{c} = z$.

By the straightening theorem of Douady and Hubbard, $(\varepsilon f_{\lambda}^{p}, P_{d_{0}+p}(c), P_{d_{0}}(c))$ is quasi-conformally conjugate to a quadratic polynomial $p_{\mu}(z) = z^{2} + \mu$ in a neighborhood of the small filled Julia set K_{c} . For this quadratic polynomial, the β -fixed point is also a parabolic point, thus $\mu = 1/4$. This means that the Julia set $J(f_{\lambda})$ contains a quasiconformal copy of quadratic Julia set of $z \mapsto z^{2} + 1/4$. Since the multiplier of the parabolic point of $z \mapsto z^{2} + 1/4$ if 1, it turns out that $(\varepsilon f_{\lambda}^{p})'(z) = 1$, $(f_{\lambda}^{k})'(z) = 1$ and $(f_{\lambda}^{q})'(z) = 1$.

Proposition 5.7.6. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then every Fatou component is a Jordan domain.

Proof. By Proposition 5.7.4 and the fact $H_{\lambda}(B_{\lambda}) = T_{\lambda}$, we conclude that both T_{λ} and B_{λ} are Jordan domains.

If the critical orbit tends to ∞ , then the Julia set is a Sierpinski curve which is locally connected, and all Fatou components are quasidisks (By Proposition 5.6.1).

If the critical orbit remains bounded, then for any $U \in \mathcal{P} \setminus \{T_{\lambda}, B_{\lambda}\}$, there is a smallest integer $k \geq 1$ such that $f_{\lambda}^{k} : U \to T_{\lambda}$ is a conformal map. Thus if two radial rays $R_{U}(\theta_{1})$ and $R_{U}(\theta_{2})$ land at the same point, then $R_{T_{\lambda}}(\theta_{1}) = f_{\lambda}^{k}(R_{U}(\theta_{1}))$ and $R_{T_{\lambda}}(\theta_{2}) = f_{\lambda}^{k}(R_{U}(\theta_{2}))$ also land at the same point. This implies that U is also a Jordan domain. If there are other Fatou components, then they are eventually mapped to a parabolic basin or an attracting basin. By Proposition 5.5.1, the map is either renormalizable or *-renormalizable. It is known that every bounded Fatou component of a quadratic polynomial(without Siegel disk) is a Jordan disk, it turns out that all Fatou components of f_{λ} are Jordan disks in this case. \Box

Proposition 5.7.7. If f_{λ} has a Cremer point, then the Cremer point cannot lie on the boundary of any Fatou component. In other words, all Cremer points are buried on the Julia set.

Proof. Suppose f_{λ} has a Cremer point z, then the Fatou set $F(f_{\lambda}) = \bigcup_{k\geq 0} f_{\lambda}^{-k}(B_{\lambda})$. If z lies on the boundary of some Fatou component, then after integrations, one sees that $z \in \partial B_{\lambda}$. By Theorem 5.1.1, there is a periodic external ray $R_{\lambda}(t)$ landing at z. But this is a contradiction since every periodic external ray can only land at a parabolic point or a repelling point (By Snail Lemma, see [M1]).

5.8 Local connectivity of the Julia set $J(f_{\lambda})$

In this section, we study the local connectivity of the Julia set $J(f_{\lambda})$. We will prove the following

Theorem 5.8.1. Suppose f_{λ} has no Siegel disk and the Julia set $J(f_{\lambda})$ is connected, then $J(f_{\lambda})$ is locally connected in either of the following cases:

- 1. The critical orbit does not accumulate on the boundary ∂B_{λ} .
- 2. f_{λ} is neither renormalizable nor *-renormalizable.
- 3. The parameter λ is real and positive.

The proof is based on the 'Characterization of Local Connectivity' (Proposition 5.8.1, See [Wh]) and 'Shrinking Lemma' (Proposition 5.8.2, See [TY] or [LM]), as follows

Proposition 5.8.1. A connected and compact set $X \subset \overline{\mathbb{C}}$ is locally connected if and only if it satisfies the following conditions:

1. Every component of $\overline{\mathbb{C}} \setminus X$ is locally connected.

2. For any $\varepsilon > 0$, there are at most finitely many components of $\overline{\mathbb{C}} \setminus X$ with spherical diameter greater than ε .

Proposition 5.8.2. Let $f : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ be a rational map and D be a topological disk whose closure \overline{D} has no intersection with the post-critical set P(f), then either \overline{D} is contained in a Siegel disk or a Herman ring, or for any $\varepsilon > 0$, there are at most finitely many iterated preimages of D with spherical diameter greater than ε .

Proof of Theorem 5.8.1.

1. If f_{λ} is geometrically finite, then $J(f_{\lambda})$ is locally connected (See [**TY**]). Else, the Fatou set $F(f_{\lambda}) = \bigcup_{k\geq 0} f_{\lambda}^{-k}(B_{\lambda})$. Since $\overline{B}_{\lambda} \cap P(f_{\lambda}) = \emptyset$, we conclude by Shrinking Lemma that for any $\varepsilon > 0$, there are at most finitely many iterated preimages of B_{λ} with spherical diameter greater than ε . By Proposition 5.8.1, $J(f_{\lambda})$ is locally connected.

2. If f_{λ} is neither renormalizable nor *-renormalizable, then the parameter $\lambda \in \mathcal{H}$ by Lemma 5.7.2. We may assume f_{λ} is not critically finite, for else the Julia set is locally connected. Thus by Proposition 5.4.1, we can find an admissible graph. By Lemma 5.5.1, none of the critical tableaux is periodic. The local connectivity of $J(f_{\lambda})$ follows from Proposition 5.7.1.

3. (The notations here are the same as in Section 5.7.3) We just need consider the case when f_{λ} is not geometrically finite. In this case, the Fatou set $F(f_{\lambda}) = \bigcup_{k\geq 0} f_{\lambda}^{-k}(B_{\lambda})$. Note that for any z > 0, $f_{\lambda}(z) \geq 2\sqrt{z^n \cdot \frac{\lambda}{z^n}} = 2\sqrt{\lambda} = v_{\lambda}^+$. Thus $\{f_{\lambda}^k(v_{\lambda}^+); k\geq 0\} \subset [v_{\lambda}^+, \beta_{c_0}]$.

If $v_{\lambda}^{+} = \beta_{c_0}'$, then one can easily verify that the triple (f_{λ}, U, V) is quasiconformally conjugate to the quadratic polynomial $z \mapsto z^2 - 2$, which is critically finite. So f_{λ} is also critically finite and the Julia set is locally connected.

If $v_{\lambda}^{+} > \beta_{c_{0}}'$, then $\overline{T}_{\lambda} \cap [v_{\lambda}^{+}, \beta_{c_{0}}] = \emptyset$ by Remark 5.7.1. Since $P(f_{\lambda}) \subset [-\beta_{c_{0}}, v_{\lambda}^{-}] \cup [v_{\lambda}^{+}, \beta_{c_{0}}] \cup \{\infty\}$, we have $\overline{T}_{\lambda} \cap P(f_{\lambda}) = \emptyset$. By Proposition 5.8.2, for any $\varepsilon > 0$, there are at most finitely many iterated preimages of T_{λ} with spherical diameter greater than ε . By Proposition 5.8.1, the Julia set is locally connected.

CHAPTER 6 Decomposition Theorem and Thurston-type Theorems

6.1 Introduction

Let $f: S^2 \to S^2$ be an orientation preserving branched covering of degree at leat two. We denote by $\deg(f, x)$ the local degree of f at $x \in S^2$. The critical set Ω_f of f is defined by

$$\Omega_f = \{ x \in S^2; \deg(f, x) > 1 \},\$$

and the postcritical set P_f of f is defined by

$$P_f = \overline{\bigcup_{n \ge 1} f^n(\Omega_f)}.$$

We say that f is postcritically finite (also called 'critically finite') if P_f is a finite set. Such a map is always called a Thurston map. For a Thurston map, we define a function $\nu_f : S^2 \to \mathbb{N} \cup \{\infty\}$ in the following way: For each $x \in S^2$, define $\nu_f(x)$ (may be ∞) as the least common multiple of the local degrees deg (f^n, y) for all n > 0 and all $y \in S^2$ such that $f^n(y) = x$. (Notice that $\nu_f(x) = 1$ if $x \notin P_f$). We call $\mathcal{O}_f = (S^2, \nu_f)$ the orbifold of f.

In 1980s, Thurston proved the following theorem:

Theorem 6.1.1. (Thurston) Let $f: S^2 \to S^2$ be a critically finite branched covering. Suppose that \mathcal{O}_f does not have signature (2, 2, 2, 2). Then f is combinatorially equivalent to a rational function R if and only if for any fstable multicurve Γ , we have $\lambda(\Gamma, f) < 1$. The rational function R is unique up to Möbius conjugation.

Here, the definitions of 'multicurve' and 'combinatorially equivalent' will be presented below for a larger category of branched coverings, that covers the postcritically finite cases. A detailed proof of Thurston's theorem is given by Douady and Hubbard [DH1].

Thurston's theorem has connections with a number of related areas such as Teichmuller theory, quasiconformal surgery, dynamics of several complex variables, transversality, group theory, algorithm, etc.

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 108 THEOREMS

There are many applications of Thurston's theorem. Here is an incomplete list: Geyer's sharp bounds on the number of harmonic polynomial roots [Gey2], Kiwi's characterization of polynomial laminations [Kiwi] (using previous work of Bielefield-Fisher-Hubbard [BFH] and Poirier [Poi]), Mikulich's classification of postcritically finite Newton maps, Milnor-Thurston's proof of monotonicity of entropy for unimodal maps [MT], McMullen's work on rational quotients [McM1], Pilgrim-Tan's cut-and-paste surgery along arcs ([PT1]), Rees' descriptions of parameter spaces [Rees2], Rees, Shishikura and Tan's studies on matings of polynomials ([Rees1],[ST], [Tan1], [Tan2]), ...

Over the years, there are several various attempts to generalize Thurston's theorem beyond postcritically finite rational maps. For example, David Brown [Bro], supported by the previous work of Hubbard and Schleicher [HS], has succeeded in extending the theory to the uni-critical polynomials with an infinite postcritical set (but always with a connected Julia set), and pushed it even further to the infinite degree case, namely the exponential maps. We would also like to mention a recent work of Hubbard-Schleicher-Shishikura [HSS]) extending Thurston's theorem to postcritically finite exponential maps. Cui-Tan[CT1] and Jiang-Zhang [JZ], independently, using different methods, extend Thurston's theorem to hyperbolic rational maps. Furthermore, Cui and Tan [CT2] extend Thurston's theorem to geometrically finite rational maps. Meanwhile, Zhang [Zh2] extends Thurston's theorem to a class of rational maps with Siegel disks.

In this work, we aim to extend Thurston's theorem to a large class of branched covering, namely 'non-parabolic' branched covering. Roughly speaking, a 'non-parabolic' branched covering is a proper branched covering for which each critical point either has finite orbit or is attracted to an attracting cycle, or is eventually mapped to the closure of some rotation domain. Before we are going on, we shall define these objects first.

We may identify S^2 with $\overline{\mathbb{C}}$.

Definition 6.1.1. (Rotation domain) We say $\langle U_0, \dots, U_{p-1} \rangle$ is a cycle of rotation domain of f if

1. All U_i are disks or annuli, with disjoint closures and Jordan curve boundaries.

2. f should induce conformal isomorphisms

$$U_0 \xrightarrow{\cong} U_1 \xrightarrow{\cong} \cdots \xrightarrow{\cong} U_{p-1} \xrightarrow{\cong} U_p = U_0$$

and the return map $f^p: U_0 \to U_0$ is conformally conjugate to an irrational rotation.

3. Each boundary cycle of ∂U_j contains at least one critical point of f.

One may compare this definition with the definitions of Siegel disks and Herman rings for rational maps ([M1]). Let P'_f be the accumulation set of P_f .

Definition 6.1.2. (Attracting cycle) We say $\langle z_0, \dots, z_{p-1} \rangle$ is an attracting cycle of f if $\langle z_0, \dots, z_{p-1} \rangle$ is contained in P'_f , and f is holomorphic in a neighborhood of this cycle with multiplier $|(f^p)'(z_0)| < 1$.

We remark that: A periodic cycle near which the map f is holomorphic and attracting (i.e. the multiplier λ satisfies $|\lambda| < 1$) is not necessarily contained in P'_f . This kind of 'attracting' cycle may be artificial. This is one of the differences between branched coverings and rational maps. Another important and essential difference is, for an attracting cycle $\langle z_0, \dots, z_{p-1} \rangle \subset P'_f$, the immediate attracting basin $\mathcal{A}_0 = \bigcup_{0 \le i < p} \mathcal{A}_0(z_i)$ of $\langle z_0, \dots, z_{p-1} \rangle$ does not necessarily contain a critical point, where $\mathcal{A}_0(z_i)$ is the component of $\{z \in \overline{\mathbb{C}}; f^{pk}(z) \to z_i \text{ as } k \to \infty\}$ that contains z_i . This case usually implies the existence of Thurston obstructions.

Definition 6.1.3. (Non-parabolic map) We say that f is a non-parabolic map if each critical point of f either has finite orbit or is attracted to an attracting cycle, or is eventually mapped to the closure of some rotation domain.

Given a non-parabolic map f, let $n_{RD}(f)$ be the number of rotation disk cycles, $n_{RA}(f)$ be the number of rotation annulus cycles and $n_A(f)$ be the number of attracting cycles. By definition, we see that

$$n_{RD}(f) + 2n_{RA}(f) + n_A(f) \le 2\deg(f) - 2.$$

Let f be a non-parabolic map,

we call $f \begin{cases} a \text{ Herman map, if } n_{RA}(f) > 0, n_{RD}(f) \ge 0 \text{ and } n_A(f) = 0, \\ a \text{ Siegel map, if } n_{RA}(f) = 0, n_{RD}(f) > 0 \text{ and } n_A(f) = 0. \end{cases}$

It's obvious that a Thurston map is a non-parabolic map with $n_{RA}(f) = n_{RD}(f) = n_A(f) = 0$.

Definition 6.1.4. (Marked set) Let f be a non-parabolic map and R_f be the union of all rotation domains of f. A marked set P is a compact set that satisfies the following:

1. $f(P) \subset P$. 2. $P \supset P_f \cup \overline{R_f}$ and $P - (P_f \cup \overline{R_f})$ is a finite set.

In the chapter, we always use a pair (f, P), a branched covering together with a marked set, to denote a non-parabolic map.

Definition 6.1.5. (C-equivalence) Two non-parabolic maps (f, P) and (g, Q) are called combinatorially equivalent or 'c-equivalent' for short (resp. q.c-equivalent), if there is a pair (ϕ, ψ) of homeomorphisms (resp. quasiconformal maps) of $\overline{\mathbb{C}}$ such that

1. $\phi \circ f = g \circ \psi$ and $\phi(P) = Q$.

2. ϕ and ψ are holomorphic in $R_f \cup N$, where R_f is the union of all rotation domains of f (if any) and N is a neighborhood of all attracting cycles (if any). If P'_f contains no attracting cycle, we set $N = \emptyset$.

3. ϕ and ψ are isotopic rel $P \cup N$. That is, there is a continuous map H: $[0,1] \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that for any $t \in [0,1]$, $H(t, \cdot) : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a homeomorphism (resp. quasiconformal map), $H(0, \cdot) = \phi$, $H(1, \cdot) = \psi$ and $H(t, z) = \phi(z)$ for any $t \in [0,1]$ and any $z \in P \cup N$.

In this case, we say (f, P) is c-equivalent (resp. q.c-equivalent) to (g, Q) via (ϕ, ψ) . Notice that a necessary condition for q.c-equivalence is that f is a quairegular map.

Multicurve and Thurston obstruction

Let (f, P) be a non-parabolic map. A Jordan curve γ in $\overline{\mathbb{C}} \setminus P$ is called *null-homotopic* (resp. *peripheral*) in $\overline{\mathbb{C}} \setminus P$ if one of its complementary components contains no (resp. one) point of P, and called *non-peripheral* in $\overline{\mathbb{C}} \setminus P$ if each component of $\overline{\mathbb{C}} \setminus \gamma$ contains at least two points of P.

We say that $\Gamma = \{\gamma_1, \dots, \gamma_n\}$ is a *multicurve* in $\overline{\mathbb{C}} \setminus P$ if each γ_i is a nonperipheral Jordan curve in $\overline{\mathbb{C}} \setminus P$, and they are mutually disjoint and no two homotopic in $\overline{\mathbb{C}} \setminus P$. Its (f, P)-transition matrix $W_{\Gamma} = (a_{ij})$ is defined by

$$a_{ij} = \sum_{\alpha \sim \gamma_i} \frac{1}{\deg(f : \alpha \to \gamma_j)},$$

where the summation is taken over all the components α of $f^{-1}(\gamma_j)$ which are homotopic to γ_i in $\overline{\mathbb{C}} \setminus P$.

A multicurve Γ in $\overline{\mathbb{C}} \setminus P$ is called (f, P)-stable if every component of $f^{-1}(\gamma)$ for $\gamma \in \Gamma$ is either null-homotopic, or peripheral, or homotopic in $\overline{\mathbb{C}} \setminus P$ to a curve $\delta \in \Gamma$.

We say that a multicurve Γ is a *Thurston obstruction* of (f, P) if Γ is (f, P)-stable and the leading eigenvalue $\lambda(\Gamma, f)$ of its transition matrix W_{Γ} satisfies $\lambda(\Gamma, f) \geq 1$.

For convention, an empty set $\Gamma = \emptyset$ is always considered as a (f, P)-stable multicurve with $\lambda(\Gamma, f) = 0$.

The main theorem of this chapter is:

Theorem 6.1.2. (Decomposition Theorem) Let (f, P) be a non-parabolic map, then there exist a (f, P)-stable multicurve Γ and a collection of Siegel

maps or Thurston maps, say $\{(h_k, P_k), k \in \Lambda\}$, where Λ is a finite index set, such that

1. (Combinatorial part) (f, P) has no Thurston obstructions if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) has no Thurston obstructions.

2. (Surgery part) (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is q.c-equivalent to a rational map.

3. (Analytic part) (f, P) is q.c-equivalent to a unique rational map up to Möbius conjugation if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) q.c-equivalent to a unique rational map up to Möbius conjugation.

From the viewpoint of 'decomposition', this theorem means that every nonparabolic branched covering can be decomposed along a stable multicurve into finitely many Siegel maps or Thurston maps, such that the combinatorics and rational realizations of these resulting maps essentially dominate the original one. These resulting maps can be viewed as the renormalizations of the original map.

From the viewpoint of 'reduction', the theorem implies that Thurstontype Theorems for every non-parabolic branched covering can be reduced to Thurston-type Theorems for finitely many Siegel type branched coverings. In particular, Thurston-type Theorems for rational maps with Herman rings can be reduced to Thurston-type Theorems for rational maps with Siegel disks.

Remark 6.1.1. 1. The number of Siegel maps is bounded by $n_{RD}(f) + 2n_{RA}(f)$.

2. In the surgery part of Theorem 6.1.2, we require that (f, P) is a quasiregular branched covering, and the pair of c-equivalences are quasiconformal. This is simply because we want to apply Measurable Riemann Mapping Theorem in our proof. In fact, this part can be restated as '(f, P) is c-equivalent to a rational map if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is c-equivalent to a rational map' and the proof goes through without any difficulty. The only modification of the proof is to replace the Measurable Riemann Mapping Theorem by the Uniformization Theorem.

3. The condition $\lambda(\Gamma, f) < 1$ implies that if (h_k, P_k) is a Thurston map for some $k \in \Lambda$, then the signature of the orbifold of (h_k, P_k) is not (2, 2, 2, 2). Thus by Thurston's Theorem, (h_k, P_k) has no Thurston obstructions if and only if (h_k, P_k) is c-equivalent to a rational map.

4. Theorem 6.1.2 consists of four cases:

1). $\Gamma = \emptyset$, $\Lambda = \emptyset$. In this case, (f, P) is c-equivalent to a unique rational map up to Möbius conjugation. Hyperbolic polynomials with Cantor Julia sets provide such examples.

2). $\Gamma \neq \emptyset$, $\Lambda = \emptyset$. In this case, (f, P) has no Thurston obstructions (or 'is c-equivalent to a rational map') if and only if $\lambda(\Gamma, f) < 1$. Such examples of rational maps can be found in the family of McMullen maps: $f_{n,\lambda}(z) = z^n + \lambda/z^n$ with $n \ge 3$ and λ suitably chosen such that $J(f_{n,\lambda})$ is a Cantor set of circles.

3). $\Gamma = \emptyset$, $\Lambda \neq \emptyset$. In this case, (f, P) is a Herman map. See Example 6.2.1.

4). $\Gamma \neq \emptyset$, $\Lambda \neq \emptyset$. This is the general case.

The 'Decomposition Theorem' provides a mechanism to produce Thurston type Theorems for non-parabolic maps. Thus it has many applications. For example, it can reduce the Thurston-type Theorem for hyperbolic maps to the so-called 'Marked Thurston Theorem' (this is the idea of Cui-Tan's work [CT1]), which is slightly stronger than Thurston's original theorem, as follows:

Theorem 6.1.3. (Marked Thurston Theorem) Let (f, P) be a Thurston map. Suppose that \mathcal{O}_f does not have signature (2, 2, 2, 2). Then (f, P) is c-equivalent to a rational function (R, Q) if and only if for any (f, P)-stable multicurve Γ , we have $\lambda(\Gamma, f) < 1$. The rational function (R, Q) is unique up to Möbius conjugation.

The detailed proof the 'Marked Thurston Theorem' can be found in [BCT], using the same idea as Douady-Hubbard's original one.

As another application, the 'Decomposition Theorem' enables us to give a characterization of a class of rational maps with Herman rings based on Zhang's work [Zh2] and the 'Marked Thurston Theorem', as follows:

Theorem 6.1.4. (Characterization of rational maps with Herman rings) Let (f, P) be a non-parabolic map, with only one rotation annulus cycle which is of period one and has rotation number of bounded type, and without rotation disk. Then (f, P) is c-equivalent to a rational map (R, Q) if and only if (f, P) has no Thurston obstructions. Moreover, the Lebesgue measure of the Julia set J(R) is zero, and (R, Q) is unique up to Möbius conjugation.

There is no reason to believe that the absence of Thurston obstruction is always equivalent to rational realization for postcritically infinite branched covering, even if the equivalence is true for hyperbolic case ([CT1], [JZ]), some Siegel cases [Zh2] and Herman cases (Theorem 6.1.4). The mating of two quadratic Siegel polynomials $f_{\theta}(z) = z^2 + c_{\theta}$ and $f_{-\theta}(z) = z^2 + c_{-\theta}$, where $c_{\alpha} = \frac{e^{2\pi i \alpha}}{2} (1 - \frac{e^{2\pi i \alpha}}{2})$, provides a non-parabolic map $g = f_{\theta} \sqcup f_{-\theta}$ for which the equivalence is false. As a supplement to the Decomposition Theorem, following the same idea as Shishikura's construction [Sh1] of rational maps with prescribed numbers of non-repelling cycles and Herman rings, we can construct many such examples by surgery. **Theorem 6.1.5.** Given nonnegative integers n_A , n_{RD} , n_{RA} , d satisfying

 $n_A + n_{RD} + 2n_{RA} \le 2d - 2, \ 1 \le n_{RA} \le d - 2, \ n_{RD} + n_{RA} \ge 2.$

There exists a non-parabolic map (f, P) of degree d, such that

1. $n_A(f) = n_A, n_{RD}(f) = n_{RD}, n_{RA}(f) = n_{RA}$, and the rotation number of each rotation cycle is of bounded type.

2. (f, P) has no Thurston obstructions.

3. (f, P) is not c-equivalent to a rational map.

This chapter is organized as follows:

From Section 6.2 to Section 6.4, we prove the 'combinatorial part' and 'surgery part' of Theorem 6.1.2 for Herman maps. More precisely, in Section 6.2, we will decompose a Herman map into finitely many Siegel maps and Thurston maps based on Shishikura's 'Herman ring-Siegel disk' surgery. In Section 6.3, we show the equivalence of absence of Thurston obstructions between the original map and the resulting maps. In Section 6.4, we show the equivalence of rational realizations between the original map and the resulting maps.

From Section 6.5 to Section 6.7, we prove that a non-parabolic map with $n_A(f) > 0$ can be decomposed along a stable multicurve into finitely many Herman maps, Siegel maps and Thurston maps whose combinatorics and rational realizations essentially dominate the original one. The proof is based on Cui-Tan's repelling system theory. The decomposition procedure, 'combinatorial part' and 'surgery part' are discussed in Section 6.5, Section 6.6 and Section 6.7, respectively.

In Section 6.8, we prove the 'combinatorial part' and 'surgery part' of Theorem 6.1.2.

In Section 6.9, we discuss the renormalizations of rational maps and prove the 'analytic part' of Theorem 6.1.2.

In Section 6.10, we give many applications of Theorem 6.1.2. These include characterizations of hyperbolic rational maps and a class of rational maps with Siegel disks. As another application, we prove a Thurston-type theorem for a class of rational maps with Herman rings.

In Section 6.11, as a supplement to the Decomposition Theorem, we show that for postcritically infinite non-parabolic maps, no Thurston obstruction does not always imply rational realization. We construct many such examples by surgery.

Definitions and Notations:

1. Given a collection of Jordan curves \mathcal{C} (not necessarily a multicurve) in $\overline{\mathbb{C}} - P$. For any integer $k \geq 0$, we denote by $f^{-k}(\mathcal{C})$ the collection of all components δ of $f^{-k}(\gamma)$ for $\gamma \in \mathcal{C}$. Set $\cup \mathcal{C} := \bigcup_{\gamma \in \mathcal{C}} \gamma$. 2. Let $A = (a_{ij})$ be a $n \times n$ real matrix. The Banach norm ||A|| of A is defined to be either $\sum |a_{ij}|$ or $(\sum |a_{ij}|^2)^{1/2}$ according to different situations. The spectral radius $\operatorname{sp}(A)$ of A is defined by $\operatorname{sp}(A) := \lim \sqrt[n]{||A^n||}$.

3. Given two multicurves Σ_1 and Σ_2 in $\overline{\mathbb{C}} - P$. We say that Σ_1 is *ho-motopically contained* in Σ_2 , denoted by $\Sigma_1 \prec \Sigma_2$, if each curve $\alpha \in \Sigma_1$ is homotopic in $\overline{\mathbb{C}} - P$ to some curve $\beta \in \Sigma_2$. We say that Σ_1 is identical to Σ_2 up to homotopy, if $\Sigma_1 \prec \Sigma_2$ and $\Sigma_2 \prec \Sigma_1$.

4. Let D and Ω be two planar domains and $f: D \to \Omega$ be a quasiregular map, the Beltrami coefficient μ_f of f is defined by $\mu_f = \frac{\partial f}{\partial z} / \frac{\partial f}{\partial z}$.

5. We use #E to denote the cardinality of the set E. The characteristic function $\chi_E : E \to \{0, 1\}$ is defined by

$$\chi_E(z) = \begin{cases} 1, & \text{if } z \in E, \\ 0, & \text{if } z \notin E. \end{cases}$$

6.2 Decompositions of Herman maps

In the following three sections we will prove the following theorem:

Theorem 6.2.1. (Herman=multicurve+Siegel+Thurston) Let (f, P) be a Herman map, then there exist a (f, P)-stable multicurve Γ and a collection of Siegel maps or Thurston maps, say $\{(h_k, P_k), k \in \Lambda\}$, where Λ is a finite index set, such that

1. (f, P) has no Thurston obstructions if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) has no Thurston obstructions.

2. (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\Gamma, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is q.c-equivalent to a rational map.

In this section, we will decompose a Herman map into finitely many Siegel maps and Thurston maps along a collection of f-periodic Jordan curves and their suitably chosen preimages. The method we use here is called 'Herman-Siegel' surgery which is pioneered by Shishikura [Sh1].

Let (f, P) be a Herman map, \mathcal{A} be the collection of all rotation annuli of f and $\cup \mathcal{A} := \bigcup_{A \in \mathcal{A}} A$ be the union of all these annuli. For each $A \in \mathcal{A}$, we choose an analytic curve $\gamma_A \subset A$ such that $\gamma_A \cap f(P - \cup \mathcal{A}) = \emptyset$ (This implies that γ_A avoids the postcritical points and the images of other marked points) and $f(\gamma_A) = \gamma_{f(A)}$. It's obvious that if $f^p(A) = A$, then $f^p(\gamma_A) = \gamma_A$.

Let $\Gamma_0 = \{\gamma_A; A \in \mathcal{A}\}$, we first show that Γ_0 can generate a unique (f, P)-stable multicurve up to homotopy.

Lemma 6.2.1. Given a choice of Γ_0 , there is a (f, P)-stable multicurve Γ such that:

• (Invariant) For any $\gamma \in \Gamma$, we have $f(\gamma) \in \Gamma \cup \Gamma_0$.

• (Maximal) Γ represents all homotopy classes of non-peripheral curves of $\bigcup_{k>1} f^{-k}(\Gamma_0) - \Gamma_0$ in $\overline{\mathbb{C}} - P$.

Moreover, the multicurve Γ is unique up to homotopy.

Proof. First, there is a multicurve Γ_1 in $\overline{\mathbb{C}} - P$ such that $\Gamma_1 \subset f^{-1}(\Gamma_0) - \Gamma_0$ and Γ_1 represents all homotopy classes of non-peripheral curves of $f^{-1}(\Gamma_0) - \Gamma_0$.

Such Γ_1 is not uniquely chosen. But any two such multicurves are identical up to homotopy, thus they have the same number of curves.

- For $n \geq 2$, we define Γ_n inductively in the following way:
- $\Gamma_n \subset f^{-1}(\Gamma_{n-1}).$

• $\Gamma_1 \cup \cdots \cup \Gamma_n$ is a multicurve in $\overline{\mathbb{C}} - P$.

• $\Gamma_1 \cup \cdots \cup \Gamma_n$ represents all homotopy classes of non-peripheral curves of $f^{-n}(\Gamma_0) - \Gamma_0$.

Since any two different curves in $\bigcup_{k\geq 1} f^{-k}(\Gamma_0) - \Gamma_0$ are disjoint and P has finitely many components, we conclude that $\bigcup_{k\geq 1} f^{-k}(\Gamma_0) - \Gamma_0$ has finitely many homotopy classes of non-peripheral curves in $\overline{\mathbb{C}} - P$. It turns out that $\#(\Gamma_1 \cup \cdots \cup \Gamma_n)$ is uniformly bounded above by some constant C(P). Thus there is an integer $N \geq 0$ such that $\Gamma_N \neq \emptyset$ and $\Gamma_{N+1} = \Gamma_{N+2} = \cdots = \emptyset$. (It can happen that N = 0, see Example 6.2.1.)

We set $\Gamma = \emptyset$ if N = 0 and $\Gamma = \bigcup_{1 \le j \le N} \Gamma_j$ if $N \ge 1$. By the choice of N, Γ is a (f, P)-stable multicurve. By construction, for any $\gamma \in \Gamma$, we have $f(\gamma) \in \Gamma \cup \Gamma_0$. The homotopy classes of Γ is uniquely determined by those of non-peripheral curves of $\bigcup_{k \ge 1} f^{-k}(\Gamma_0) - \Gamma_0$. So Γ is unique up to homotopy. \Box

Here we give an example to show that Γ can be an empty set.

Example 6.2.1. ($\Gamma = \emptyset$) The example is borrowed from Shishikura's paper [Sh1]. Let

$$f(z) = \frac{e^{i\alpha}}{z} \left(\frac{z-r}{1-rz}\right)^2,$$

where $\alpha \in \mathbb{R}$ and 0 < r < 1/5. We may assume that α is properly chosen such that f has a fixed Herman ring H containing the unit circle \mathbb{S} , with bounded type rotation number (Remark: in this case, each boundary component of His a quasicircle containing a critical point of f). There are two other critical points: r and 1/r, which are eventually mapped to a repelling cycle of period two, and $f(r) = f^3(r) = 0$, $f^2(r) = f(1/r) = \infty$. We choose $\Gamma_0 = \{\mathbb{S}\}$. Let $P = \overline{H} \cup P_f = \overline{H} \cup \{0, \infty\}$. Since each component of $\overline{\mathbb{C}} - \overline{H}$ is a disk containing exactly one point in the marked set P, the set Γ is necessarily empty. Let $\Sigma = \Gamma_0 \cup \Gamma$. In the following, we will use Σ to decompose the complex sphere $\overline{\mathbb{C}}$ into finitely many pieces. We define

$$\mathcal{S} = \{ \overline{U}; \ U \text{ is a connected component of } \overline{\mathbb{C}} - \cup \Sigma \},$$
$$\mathcal{E} = \{ \overline{V}; \ V \text{ is a connected component of } \overline{\mathbb{C}} - \cup f^{-1}(\Sigma) \}.$$

Each element of S (resp. \mathcal{E}) is called an S-piece (resp. \mathcal{E} -piece). Given an S-piece S (resp. \mathcal{E} -piece E), let $\partial(S)$ (resp. $\partial(E)$) be the collection of all boundary curves of S (resp. E). One should notice that ∂S and $\partial(S)$ are different notations, they satisfy $\partial S = \bigcup \partial(S)$.

The following facts are easy to verify:

- Every \mathcal{E} -piece E is contained in a unique \mathcal{S} -piece and $f(E) \in \mathcal{S}$.
- For every S-piece S, we have $\#(S \cap P) + \#\partial(S) \ge 3$.
- For each curve $\gamma \in \Sigma$, there exist exactly two S-pieces, say S_{γ}^+ and S_{γ}^- , that share γ as a common boundary component.



Figure 6.1: Four examples: E_i (shadow region) is parallel to S_i . p_i, q_i are marked points in P. Here, S_1 is an annulus with one marked point, S_2 has three boundary curves and contains no marked point, both S_3 and S_4 are disks with two marked points.

Let S be an S-pieces, T is a connected and closed subset of S, we say T is *parallel* to S if $\partial T \cap P = \emptyset$ and each component of $S \setminus T$ is either

- an annulus contained in S P, or
- a disk that contains at most one point of *P*.

Notice that if T is parallel to S and A is an annular component of $S \setminus T$, then one boundary curve of A is on S and $\#(T \cap P) + \#\partial(T) \ge \#(S \cap P) + \#\partial(S)$.

Here is an important property of the S-pieces:

Lemma 6.2.2. For every S-piece S, there is a unique \mathcal{E} -piece, say E_S , parallel to S.

Proof. The proof is based on the 'maximal' property of the (f, P)-stable multicurve Γ . We omit the details.

We define a map

$$f_*: \begin{cases} \mathcal{S} \to \mathcal{S}, \\ S \mapsto f(E_S). \end{cases}$$

Since there are finitely many S-pieces, every S-piece is eventually periodic under the map f_* .

For each curve $\gamma \in \partial(S)$, there is a unique boundary curve $\beta_{\gamma} \in \partial(E_S)$ such that either $\beta_{\gamma} = \gamma$, or β_{γ} and γ bound an annulus in S - P. We define three sets $\partial_0(S), \partial_1(S), \partial_2(S)$ as follows:

$$\begin{aligned} \partial_0(S) &= \{\gamma \in \partial(S); \gamma \in \Gamma_0\}, \\ \partial_1(S) &= \{\gamma \in \partial(S); \gamma \neq \beta_\gamma\}, \\ \partial_2(S) &= \{\gamma \in \partial(S); \gamma = \beta_\gamma\} - \Gamma_0. \end{aligned}$$

Lemma 6.2.3. If $\partial_0(S) \neq \emptyset$, then we have:

- 1. For any $\gamma \in \partial_0(S)$, $\gamma = \beta_{\gamma}$.
- 2. S is f_* -periodic.
- 3. $\#\partial_0(S) = \#\partial_0(f_*(S)).$

Proof. 1. Notice that every component of $S - E_S$ is either a disk containing at most one point in P, or an annulus in $\overline{\mathbb{C}} - P$. It follows that if $\gamma \in \partial_0(S)$, then $\gamma \subset P$ and $\gamma = \beta_{\gamma}$.

2. Take $\gamma \in \partial_0(S)$. Then there is a rotation annulus $A_\gamma \in \mathcal{A}$ containing γ . Then from 1 we see that $S \cap A_\gamma = E_S \cap A_\gamma$. This implies $f(S \cap A_\gamma) = f_*(S) \cap f(A_\gamma)$. Let $k \geq 1$ be the period of A_γ . Then we have $S \cap A_\gamma = f^k(S \cap A_\gamma) = f^k_*(S) \cap f^k(A_\gamma) = f^k_*(S) \cap A_\gamma$. Thus $f^k_*(S) = S$. So S is f_* -periodic, and the period of S is a divisor of k.

3. It follows from 1 that if $\gamma \in \partial_0(S)$, then $f(\gamma) \in \partial_0(f_*(S))$. So $\#\partial_0(S) \leq \#\partial_0(f_*(S)) \leq \cdots$. Since S is f_* -periodic (by 2), we have $\#\partial_0(S) = \#\partial_0(f_*(S))$.

It follows from Lemma 6.2.3 that $\partial_i(S), i \in \{0, 1, 2\}$ are mutually disjoint and $\partial(S) = \partial_0(S) \sqcup \partial_1(S) \sqcup \partial_2(S)$.

Remark 6.2.1. Suppose $\partial_0(S) \neq \emptyset$. For each $\gamma \in \partial_0(S)$, let $per(\gamma)$ be the period of γ . From Lemma 6.2.3 we see that the f_* -period of S is a devisor of $gcd\{per(\gamma); \gamma \in \partial_0(S)\}$. In particular, if $gcd\{per(\gamma); \gamma \in \partial_0(S)\} = 1$, then $f_*(S) = S$ and for every $\gamma \in \partial_0(S)$ and every integer $k \geq 0$, we have $f^k(\gamma) \in \partial_0(S)$.

For example, suppose that (f, P) has two cycles of rotation annuli whose periods are different prime numbers, say p and q. If $\partial_0(S) \neq \emptyset$, then $\#\partial_0(S)$ takes only four possible values: 1, p, q and p + q.

6.2.1 Marked disk extension

For each S-piece S, we denote by $\overline{\mathbb{C}}(S)$ the Riemann sphere containing S. We always consider that different S-pieces are embedded into different copies of Riemann spheres.

In the following, we will extend $f|_{E_S}$ to a quasiregular branched covering $H_S: \overline{\mathbb{C}}(S) \to \overline{\mathbb{C}}(f_*(S))$ such that $\deg(H_S) = \deg(f|_{E_S})$. To do this, we need to define the map $H_S: \overline{\mathbb{C}}(S) - E_S \to \overline{\mathbb{C}}(f_*(S)) - f_*(S)$ such that $H_S|_{\partial E_S} = f|_{\partial E_S}$. We will define H_S component by componet.

Notice that each component of $\overline{\mathbb{C}}(S) - E_S$ is a disk. Let U be such a component with boundary curve γ .

We first deal with the case when $\gamma \in \partial_0(S)$. In this case, there is a rotation annulus A_{γ} containing γ . Let $k \geq 1$ be the period of A_{γ} . Let $\phi_0: S \cap A_{\gamma} \to \mathbb{A}_R := \{z \in \mathbb{C}; 1 < |z| < R\}$ be the conformal map such that $\phi_0 f^k \phi_0^{-1}(z) = e^{2\pi i \theta} z$ for $z \in \mathbb{A}_R$. For $1 \leq j \leq k - 1$, we define a conformal map from $f^j(S \cap A_{\gamma})$ onto \mathbb{A}_R by $\phi_j = \phi_0 f^{k-j}|_{f^j(S \cap A_{\gamma})}$. Then we have the following commutative diagram



Let $\mathbb{D}_R = \{z \in \mathbb{C}; |z| < R\}$. For $0 \leq j < k$, we consider the disk Δ_j obtained by gluing $f^j(S \cap A_\gamma)$ and \mathbb{D}_R via the map ϕ_j . The disk Δ_j inherits a natural complex structure from \mathbb{D}_R since ϕ_j is holomorphic.



Figure 6.2: Marked disk extension. Here $\partial S = \gamma_1 \cup \gamma_2 \cup \gamma_3 \cup \gamma_4$, $\partial f_*(S) = \gamma_5 \cup \gamma_6 \cup \gamma_7$. Marked points are labeled by '•'.

The map $H_{f^j_*(S)}: \Delta_j \to \Delta_{j+1}$ defined by

$$H_{f^{j}_{*}(S)}(z) = \begin{cases} f(z), & z \in f^{j}(S \cap A_{\gamma}), \ 0 \le j < k, \\ e^{2\pi i \theta} z, & z \in \mathbb{D}, \ j = 0, \\ z, & z \in \mathbb{D}, \ 1 \le j < k. \end{cases}$$

is a holomorphic extension of $f|_{E_{f_*^j(S)}}$ along the boundary curve $f^j(\gamma) \in \partial_0(f_*^j(S))$. We call $(\Delta_j, 0)$ a holomorphic marked disk of $H_{f_*^j(S)}$. This construction allows us to define the extensions of $f|_{E_S}, \dots, f|_{E_{f_*^{l-1}(S)}}$ (where l is the f_* -period of S) along the curves in $\partial_0(S) \cup \dots \cup \partial_0(f_*^{l-1}(S))$ at the same time.

Now, we consider the case when $\gamma = \partial U \notin \partial_0(S)$. Notice that either U is a disk in S containing at most one point of P, or it contains a unique component V of $\overline{\mathbb{C}}(S) - S$. In the former case, if U contains a marked point $p \in P$, we get a marked disk (U, p); if $U \cap P = \emptyset$, we don't mark any point in U. In the latter case, we mark a point $p \in V$ and get two marked disks (U, p) and (V, p).

Now we extend $f|_{E_S}$ to U in the following fashion:

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 120 THEOREMS

We require that H_S maps U onto (W,q) with $\deg(H_S|_U) = \deg(f|_{\partial U})$, where (W,q) is the unique marked disk of $\overline{\mathbb{C}}(f_*(S)) - f_*(S)$ whose boundary curve is $f(\partial U)$. If U contains a marked point p, we require further H(p) = qand the local degree of H_S at p is equal to $\deg(f|_{\partial U})$. Else, we require that qis the only possible critical value (This implies that U contains at most one ramification point of H_S).

In this way, for each \mathcal{S} -piece S, we can get an extension $H_S : \overline{\mathbb{C}}(S) \to \overline{\mathbb{C}}(f_*(S))$ of $f|_{E_S}$. Let D(S) be the union of all holomorphic marked disks of H_S . Notice that if $\partial_0(S) = \emptyset$, then $D(S) = \emptyset$. Set

 $P(S) = (P \cap S) \cup \{\text{all marked points in } \overline{\mathbb{C}}(S) - S\} \cup \overline{D(S)}.$

We call $(\overline{\mathbb{C}}(S), P(S))$ a marked sphere of $\overline{\mathbb{C}}(S)$. By the construction of H_S , we see that $H_S(P(S)) \subset P(f_*(S))$.

Notice that every S-piece is eventually periodic under the map f_* . Let n be the number of all f_* -cycles of S-pieces. These cycles are listed as follows:

$$S_{\nu} \mapsto f_*(S_{\nu}) \mapsto \dots \mapsto f_*^{p_{\nu}-1}(S_{\nu}) \mapsto f_*^{p_{\nu}}(S_{\nu}) = S_{\nu}, \quad 1 \le \nu \le n,$$

where S_{ν} is a representative of the ν -th cycle and p_{ν} is the period of S_{ν} . Set

$$h_{\nu} = H_{f_*^{p_{\nu}-1}(S_{\nu})} \circ \dots \circ H_{f_*(S_{\nu})} \circ H_{S_{\nu}}, \ P_{\nu} = P(S_{\nu}), \ 1 \le \nu \le n.$$

Then $h_{\nu}: \overline{\mathbb{C}}(S_{\nu}) \to \overline{\mathbb{C}}(S_{\nu})$ is a branched covering with $h_{\nu}(P_{\nu}) \subset P_{\nu}$.

These resulting maps $(h_1, P_1), \dots, (h_n, P_n)$ can be considered as the renormalizations of the original map (f, P). There are three types of them:

• $\partial_0(S_\nu) \neq \emptyset$ or S_ν contains at least one rotation disk of (f, P). In this case, (h_ν, P_ν) has at least one cycle of rotation disks, so (h_ν, P_ν) is a Siegel map. Moreover, a curve $\gamma \in \partial_0(S_\nu)$ contained in a rotation annulus of f with period p and rotation number θ becomes a periodic curve contained in a rotation disk of h_ν , with period p/p_ν and rotation number θ . One may verify that the number of these resulting Siegel maps is at least two, and bounded above by $2n_{RA}(f) + n_{RD}(f)$.

• $\partial_0(S_{\nu}) = \emptyset$, S_{ν} contains no rotation disk of (f, P) and $\deg(h_{\nu}) > 1$. In this case, P_{ν} is a finite set and (h_{ν}, P_{ν}) is a Thurston map.

• $\partial_0(S_\nu) = \emptyset$, S_ν contains no rotation disk of (f, P) and $\deg(h_\nu) = 1$. In this case, (h_ν, P_ν) is a homeomorphism of $\overline{\mathbb{C}}(S_\nu)$ and $h_\nu(P_\nu) = P_\nu$. So every point of P_ν is periodic. Moreover, for any $S \in \{S_\nu, f_*(S_\nu), \cdots, f_*^{p_\nu - 1}(S_\nu)\}$, we have $\#\partial(E_S) = \#\partial(S)$.

Let Λ be the index set consisting of all $\nu \in \{1, \dots, n\}$ such that $\deg(h_{\nu}) > 1$. 1. That is, for each $\nu \in \Lambda$, (h_{ν}, P_{ν}) is either a Siegel map or a Thurston map. Let $\Lambda^* = \{1, \dots, n\} - \Lambda$. We use the following notation to record the decomposition procedure:

$$\operatorname{Dec}(f, P) = \left(\bigoplus_{\nu \in \Lambda \cup \Lambda^*} (h_{\nu}, P_{\nu})\right)_{\Gamma}.$$

Lemma 6.2.4. If $\lambda(\Gamma, f) < 1$, then

1. For any $1 \leq \nu \leq n$, every point in $(\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}$ is eventually mapped to either the center of some rotation disk or a periodic critical point of (h_{ν}, P_{ν}) .

2. $\Lambda^* = \emptyset$.

3. If (h_{ν}, P_{ν}) is a Thurston map, then the signature of the orbifold of (h_{ν}, P_{ν}) is not (2, 2, 2, 2).

Proof. Since $h_{\nu}((\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}) \subset (\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}$ and $(\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}$ is a finite set, every point in $(\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}$ is eventually periodic under the map (h_{ν}, P_{ν}) . Let z_0 be a periodic point in $(\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}$ with period k. Suppose that z_0 is not the center of rotation disk, and let β be the boundary curve of S_{ν} that encloses z_0 . Then there is a unique component of $h_{\nu}^{-k}(\beta)$, denoted by α , such that $\alpha \subset S_{\nu}$ and α is homotopic to β in $\overline{\mathbb{C}}(S_{\nu}) - P_{\nu}$. Thus

$$\deg(h_{\nu}^{k}, z_{0}) = \deg(h_{\nu}^{k} : \alpha \to \beta) = \deg(f^{kp_{\nu}} : \alpha \to \beta) \ge \lambda(\Gamma, f)^{-kp_{\nu}} > 1.$$

This implies that z_0 lies in a critical cycle and $\deg(h_{\nu}) > 1$. It follows that $\Lambda^* = \emptyset$ and there is no (2, 2, 2, 2)-type Thurston map among $(h_{\nu}, P_{\nu}), \nu \in \Lambda$.

A multicurve $\Gamma = \{\gamma_1, \dots, \gamma_k\}$ is called a *Levy cycle* of (f, P), if for each $1 \leq i \leq k, f^{-1}(\gamma_i)$ has a component α_{i-1} homotopic to γ_{i-1} in $\overline{\mathbb{C}} - P$ (set $\gamma_0 = \gamma_k$) and deg $(f : \alpha_{i-1} \to \gamma_i) = 1$.

Corollary 6.2.1. If $\Lambda^* \neq \emptyset$, then $\lambda(\Gamma, f) \geq 1$ and Γ contains a Levy cycle of (f, P).

Proof. If $\Lambda^* \neq \emptyset$, then $\lambda(\Gamma, f) \geq 1$ follows from Lemma 6.2.4. Take some $\nu \in \Lambda^*$, the boundary multicurve $\partial(S_{\nu})$ contains a submulticurve $\{\gamma_1, \dots, \gamma_k\}$ labeled in the way that for any $i \in [1, k]$, $h_{\nu}^{-1}(\gamma_i)$ is homotopic to γ_{i-1} ($\gamma_0 := \gamma_k$) in $\overline{\mathbb{C}} - P$. In particular, $\alpha := h_{\nu}^{-k}(\gamma_k)$ is homotopic to γ_k in $\overline{\mathbb{C}} - P$. So there exist two integers $0 \leq m < n \leq kp_{\nu}$ such that $f^m(\alpha)$ is homotopic to $f^n(\alpha)$ in $\overline{\mathbb{C}} - P$ and $\{f^k(\alpha); m < k \leq n\}$ is a multicurve in $\overline{\mathbb{C}} - P$. Since Γ is (f, P)-stable, there is a submulticurve $\Gamma_* \subset \Gamma$ identical to $\{f^k(\alpha); m < k \leq n\}$ up to homotopy. One may verify that Γ_* is a Levy cycle of (f, P).

To end this section, we give a concrete example to illustrate how the decomposition procedure works:

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 122 THEOREMS

Example 6.2.2. We still consider the rational map f defined as in Example 6.2.1. Recall that f has a fixed Herman ring containing the unit circle \mathbb{S} . Suppose $f^{-1}(\mathbb{S}) = \mathbb{S} \cup \gamma_1 \cup \gamma_2$, where γ_1 (resp. γ_2) is a Jordan curve in the connected component D_0 (resp. D_{∞}) of $\overline{\mathbb{C}} - \overline{H}$ containing 0 (resp. ∞). One may deduce that γ_1 encloses 0 and γ_2 encloses ∞ . Let $D_0^{\gamma_1}$ (resp. $D_{\infty}^{\gamma_2}$) be the disk neighborhood of 0 (resp. ∞) whose boundary curve is γ_1 (resp. γ_2). Notice that $r \in D_0 - D_0^{\gamma_1}$ and $1/r \in D_{\infty} - D_{\infty}^{\gamma_2}$.

One may check both $f^{-1}(\gamma_1)$ and $f^{-1}(\gamma_2)$ consist of two components. We denote $f^{-1}(\gamma_1) = \gamma_4 \cup \gamma_5$, where $\gamma_4 \subset D_{\infty}^{\gamma_2}$ and γ_4 encloses ∞ , $\gamma_5 \subset D_0 - D_0^{\gamma_1}$ and γ_5 encloses the critical point r. Correspondingly, We denote $f^{-1}(\gamma_2) = \gamma_3 \cup \gamma_6$, where $\gamma_3 \subset D_0^{\gamma_1}$ and γ_3 encloses 0, $\gamma_6 \subset D_{\infty} - D_{\infty}^{\gamma_2}$ and γ_6 encloses the critical point 1/r.

Let $D_0^{\gamma_3}$ (resp. $D_{\infty}^{\gamma_4}$) be the disk neighborhood of 0 (resp. ∞) whose boundary curve is γ_3 (resp. γ_4). We choose two points p and q, with $p \in D_0^{\gamma_3}$ and $q \in D_{\infty}^{\gamma_4}$. We modify the mappings $f|_{D_0^{\gamma_3}} : D_0^{\gamma_3} \to D_{\infty}^{\gamma_2}$ and $f|_{D_{\infty}^{\gamma_4}} : D_{\infty}^{\gamma_4} \to D_0^{\gamma_1}$, by $\varphi : D_0^{\gamma_3} \to D_{\infty}^{\gamma_2}$ and $\psi : D_{\infty}^{\gamma_4} \to D_0^{\gamma_1}$, respectively, such that $\varphi|_{\gamma_3} = f|_{\gamma_3}, \psi|_{\gamma_4} = f|_{\gamma_4}, \varphi(0) = \infty, \varphi(p) = q, \psi(\infty) = p, \psi(q) = 0$. Then we define a new branched covering

$$h(z) = \begin{cases} \varphi(z), & z \in D_0^{\gamma_3}, \\ \psi(z), & z \in D_\infty^{\gamma_4}, \\ f(z), & z \in \overline{\mathbb{C}} - D_0^{\gamma_3} \cup D_\infty^{\gamma_4}. \end{cases}$$

Then the postcritical set P_h of h is $\partial H \cup \{0, p, q, \infty\}$ and (h, P) is a Herman map, where $P = P_h \cup H$. The curves $\gamma_i, i \in [1, 4]$ are all non-peripheral in $\overline{\mathbb{C}} - P$.

According to the decomposition procedure, $\Gamma = \{\gamma_1, \gamma_2\}$ and $\Sigma = \{S, \gamma_1, \gamma_2\}$. There are four S-pieces: S_1, S_2, S_3, S_4 , and $h_*(S_1) = S_1, h_*(S_2) = S_2, h_*(S_3) = S_4, h_*(S_4) = S_3$. So there are three h_* -periodic cycles, we denote the resulting map associating with $S_i, i = 1, 2, 3$ by (h_i, P_i) .

One may check the resulting maps (h_1, P_1) and (h_2, P_2) are both degree two Siegel maps, c-equivalent to $z \mapsto e^{2\pi i \theta} (z-1)^2/z$. Moreover, (h_3, P_3) is a homeomorphism, satisfying $h_3(0) = p, h_3(p) = 0$.

The (h, P)-transition matrix of Γ is

$$\left(\begin{array}{cc} 0 & 1 \\ 1 & 0 \end{array}\right).$$

Thus Γ is a Levy cycle. It follows that (h, P) is not c-equivalent to a rational map (To see this, one may use 'modulus argument' or apply Theorem 6.4.2).

6.3 Combinatorial part

The aim of this section is to prove the following:

Theorem 6.3.1. Let (f, P) be a Herman map, and

$$\operatorname{Dec}(f, P) = \left(\bigoplus_{\nu \in \Lambda \cup \Lambda^*} (h_{\nu}, P_{\nu})\right)_{\Gamma}$$

Then (f, P) has no Thurston obstructions if and only if $\lambda(\Gamma, f) < 1$ and for each $\nu \in \Lambda$, (h_{ν}, P_{ν}) has no Thurston obstructions.

Notice that if (f, P) has no Thurston obstructions or $\lambda(\Gamma, f) < 1$, then $\Lambda^* = \emptyset$ (See Lemma 6.2.4).

The proof of the 'sufficiency' of Theorem 6.3.1 is based on the decomposition of (f, P)-stable multicurves. We will show that every (f, P)-stable multicurve contains an 'essential' submulticurve (see Lemma 6.3.1), and every such essential submulticurve can be decomposed into a ' Γ -part' multicurve together with a (h_{ν}, P_{ν}) -stable multicurve for each $\nu \in [1, n]$. Moreover, the leading eigenvalues of their transition matrices satisfy the so-called 'reduction identity' (Theorem 6.3.2).

To prove the 'necessity' of Theorem 6.3.1, we will show that every (h_{ν}, P_{ν}) stable multicurve Σ can generate a (f, P)-stable multicurve C with $\lambda(\Sigma, h_{\nu}) \leq \lambda(C, f)^{p_{\nu}}$.

Lemma 6.3.1. (Essential submulticurve) Let Σ_0 be a (f, P)-stable multicurve, then there is a (f, P)-stable multicurve Σ , such that

- 1. Σ is homotopically contained in Σ_0 .
- 2. Each curve of Σ is contained in the interior of some S-piece.
- 3. $\lambda(\Sigma, f) = \lambda(\Sigma_0, f)$.

Proof. For $n \geq 1$, we define a multicurve Σ_n inductively in the following way: $\Sigma_n \subset f^{-1}(\Sigma_{n-1})$ and Σ_n represents all homotopy classes of non-peripheral curves of $f^{-1}(\Sigma_{n-1})$. Since Σ_0 is a (f, P)-stable multicurve, all Σ_n are (f, P)stable, and Σ_n is homotopically contained in Σ_{n-1} . Let W_n be the (f, P)transition matrix of Σ_n for $n \geq 0$, then

$$W_n = \left(\begin{array}{cc} W_{n+1} & * \\ O & O \end{array}\right).$$

Thus $\lambda(\Sigma_0, f) = \lambda(\Sigma_1, f) = \lambda(\Sigma_2, f) = \cdots$. By the construction of Γ , there is an integer $N \ge 0$ such that $\Gamma \subset f^{-N}(\Gamma_0)$, where Γ_0 is the choice of a collection of f-periodic curves in the rotation annuli (see the previous section). Since $\cup \Gamma_0$ has no intersection with $\cup \Sigma_0$, we conclude that $f^{-n}(\cup \Gamma_0)$

has no intersection with $f^{-n}(\cup \Sigma_0)$ for all $n \ge 1$. Thus when $n \ge N$, we have $\cup \Gamma \subset f^{-n}(\cup \Gamma_0)$ and each curve of Σ_n is contained in the interior of some \mathcal{S} -piece. The proof is completed if we set $\Sigma = \Sigma_n$ for some $n \ge N$. \Box

Theorem 6.3.2. (Decomposition of stable multicurve) Let C be a (f, P)-stable multicurve. Suppose that each curve of C is contained in the interior of some S-piece. Let

$$\mathcal{C}_{\Gamma} = \{ \gamma \in \mathcal{C}; \gamma \text{ is homotopic to a curve of } \Gamma \},\\ \Sigma_{\nu} = \{ \gamma \in \mathcal{C} - \mathcal{C}_{\Gamma}; \gamma \text{ is contained in } S_{\nu} \}, \ \nu \in \Lambda \cup \Lambda^* = [1, n].$$

Then C_{Γ} is a (f, P)-stable multicurve, Σ_{ν} is a (h_{ν}, P_{ν}) -stable multicurve for each $\nu \in [1, n]$, and we have the following reduction identity:

$$\lambda(\mathcal{C}, f) = \max\left\{\lambda(\mathcal{C}_{\Gamma}, f), \sqrt[p_1]{\lambda(\Sigma_1, h_1)}, \cdots, \sqrt[p_n]{\lambda(\Sigma_n, h_n)}\right\}.$$

Remark 6.3.1. In Theorem 6.3.2, the multicurve Σ_{ν} can be viewed as a multicurve of (h_{ν}, P_{ν}) , this is because under the inclusion map $\iota_{\nu} : S_{\nu} \hookrightarrow \overline{\mathbb{C}}(S_{\nu})$, the set $\iota_{\nu}(\Sigma_{\nu}) := \{\iota_{\nu}(\gamma); \gamma \in \Sigma_{\nu}\}$ is a multicurve in $\overline{\mathbb{C}}(S_{\nu}) - P_{\nu}$. We still use Σ_{ν} to denote the multicurve $\iota_{\nu}(\Sigma_{\nu})$ if there is no confusion.

One may show directly that if $\Lambda^* \neq \emptyset$, then for any $\nu \in \Lambda^*$,

$$\lambda(\Sigma_{\nu}, h_{\nu}) = \begin{cases} 1, & \text{if } \Sigma_{\nu} \neq \emptyset, \\ 0, & \text{if } \Sigma_{\nu} = \emptyset. \end{cases}$$

This observation can simplify the reduction identity.

Proof. The fact that C_{Γ} is (f, P)-stable is easy to verify since both Γ and C are (f, P)-stable. Let $\Sigma_{\nu}^{k} = \{\gamma \in C - C_{\Gamma}; \gamma \text{ is contained in } f_{*}^{k}(S_{\nu})\}$ for $0 \leq k \leq p_{\nu}$. It's obvious that $\Sigma_{\nu}^{0} = \Sigma_{\nu}^{p_{\nu}} = \Sigma_{\nu}$. Since C is (f, P)-stable, each non-peripheral component of $f^{-1}(\gamma)$ for $\gamma \in \Sigma_{\nu}^{k+1} (0 \leq k < p_{\nu})$ is homotopic in $\overline{\mathbb{C}} - P$ to either a curve $\alpha \in C_{\Gamma}$, or a curve $\beta \in \Sigma_{\nu}^{k}$, or a curve δ contained in a non-periodic S-piece.

By the definition of the marked set $P(f_*^k(S_\nu))$, one can verify that the set Σ_{ν}^k is a multicurve in $\overline{\mathbb{C}}(f_*^k(S_\nu)) - P(f_*^k(S_\nu))$. Moreover, each curve $\gamma \in \mathcal{C}_{\Gamma}$ contained in $f_*^k(S_\nu)$ and homotopic (in $\overline{\mathbb{C}} - P$) to a boundary curve of $f_*^k(S_\nu)$ is peripheral in $\overline{\mathbb{C}}(f_*^k(S_\nu)) - P(f_*^k(S_\nu))$. Thus for $0 \leq k < p_\nu$, each non-peripheral component of $H_{f_*^k(S_\nu)}^{-1}(\gamma)$ for $\gamma \in \Sigma_{\nu}^{k+1}$ is homotopic to a curve $\delta \in \Sigma_{\nu}^k$ in $\overline{\mathbb{C}}(f_*^k(S_\nu)) - P(f_*^k(S_\nu))$. It follows that each non-peripheral component of $h_{\nu}^{-1}(\gamma)$ for $\gamma \in \Sigma_{\nu}$ is homotopic to a curve $\delta \in \Sigma_{\nu}$ in $\overline{\mathbb{C}}(S_\nu) - P_{\nu}$. This means Σ_{ν} is a (h_{ν}, P_{ν}) -stable multicurve.

In the following, we will prove the 'reduction identity'. Let $W_{\mathcal{C}_{\Gamma}}$ be the (f, P)-transition matrix of \mathcal{C}_{Γ} . We define $\mathcal{C}_s := \{\gamma \in \mathcal{C} - \mathcal{C}\}$

 $C_{\Gamma}; \gamma$ is contained in a strictly preperiodic S-piece} with (f, P)-transition matrix W_s . Let $C_{\nu} = \Sigma_{\nu}^0 \cup \cdots \cup \Sigma_{\nu}^{p_{\nu}-1}$ with (f, P)-transition matrix W_{ν} . Then the (f, P)-transition matrix $W_{\mathcal{C}}$ of \mathcal{C} has the following block decomposition:

$$W_{\mathcal{C}} = \begin{pmatrix} W_{\mathcal{C}_{\Gamma}} & * & * & \cdots & * \\ O & W_{s} & * & \cdots & * \\ O & O & W_{1} & \cdots & * \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ O & O & O & \cdots & W_{n} \end{pmatrix}$$

It follows that $\lambda(\mathcal{C}, f) = \max \{\lambda(\mathcal{C}_{\Gamma}, f), \lambda(\mathcal{C}_{s}, f), \lambda(\mathcal{C}_{1}, f), \dots, \lambda(\mathcal{C}_{n}, f)\}$. By the definition of \mathcal{C}_{s} , there is an integer M > 0 such that for any $\gamma \in \mathcal{C}_{s}$ and any component α of $f^{-M}(\gamma)$, α is either null-homotopic, or peripheral, or homotopic to a curve $\delta \in \mathcal{C}_{\Gamma}$ in $\overline{\mathbb{C}} - P$. This implies $W_{s}^{M} = 0$ and $\lambda(\mathcal{C}_{s}, f) = 0$. So we have

$$\lambda(\mathcal{C}, f) = \max \left\{ \lambda(\mathcal{C}_{\Gamma}, f), \lambda(\mathcal{C}_{1}, f), \cdots, \lambda(\mathcal{C}_{n}, f) \right\}.$$

Notice that the (f, P)-transition matrix W_{ν} of \mathcal{C}_{ν} has the form

$$W_{\nu} = \begin{pmatrix} O & B_0 & O & \cdots & O \\ O & O & B_1 & \cdots & O \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ O & O & O & \cdots & B_{p_{\nu}-2} \\ B_{p_{\nu}-1} & O & O & \cdots & O \end{pmatrix}$$

where B_j is a $n_j \times n_{j+1}$ matrix, n_j is equal to the number of curves in Σ_{ν}^j for $0 \le j \le p_{\nu} - 1$. By a direct calculation,

$$W_{\nu}^{p_{\nu}} = \begin{pmatrix} B_0 B_1 \cdots B_{p_{\nu}-1} & O & \cdots & O \\ O & B_1 B_2 \cdots B_0 & \cdots & O \\ \vdots & \vdots & \ddots & \vdots \\ O & O & \cdots & B_{p_{\nu}-1} B_0 \cdots B_{p_{\nu}-2} \end{pmatrix}.$$

For a square matrix $A = (a_{ij})$, we use the norm $||A|| = \sqrt{\sum |a_{ij}|^2}$. Then for any $k \ge 1$, we have

$$\|(W_{\nu}^{p_{\nu}})^{k}\|^{2} = \|(B_{0}B_{1}\cdots B_{p_{\nu}-1})^{k}\|^{2} + \dots + \|(B_{p_{\nu}-1}B_{0}\cdots B_{p_{\nu}-2})^{k}\|^{2}.$$

It follows from Lemma 6.3.2 that

$$\operatorname{sp}(W_{\nu})^{p_{\nu}} = \operatorname{sp}(B_0 B_1 \cdots B_{p_{\nu}-1}) = \cdots = \operatorname{sp}(B_{p_{\nu}-1} B_0 \cdots B_{p_{\nu}-2}).$$

On the other hand, one can verify that the (h_{ν}, P_{ν}) -transition matrix of Σ_{ν} is $B_0 B_1 \cdots B_{p_{\nu}-1}$. It follows from Perron-Frobenius Theorem that

$$\lambda(\Sigma_{\nu}, h_{\nu}) = \operatorname{sp}(B_0 B_1 \cdots B_{p_{\nu}-1}) = \operatorname{sp}(W_{\nu})^{p_{\nu}} = \lambda(\mathcal{C}_{\nu}, f)^{p_{\nu}}.$$

Finally we have

$$\lambda(\mathcal{C}, f) = \max \left\{ \lambda(\mathcal{C}_{\Gamma}, f), \sqrt[p_1]{\lambda(\Sigma_1, h_1)}, \cdots, \sqrt[p_n]{\lambda(\Sigma_n, h_n)} \right\}.$$

Lemma 6.3.2. Let B_{ν} be a $n_{\nu} \times n_{\nu+1}$ real matrix for $1 \leq \nu \leq k$, $n_{k+1} = n_1$, then

$$\operatorname{sp}(B_1B_2\cdots B_k) = \operatorname{sp}(B_2B_3\cdots B_1)\cdots = \operatorname{sp}(B_kB_1\cdots B_{k-1}).$$

Proof. For a square matrix $A = (a_{ij})$, we define a norm $||A|| := \sqrt{\sum |a_{ij}|^2}$. The basic property of this norm is $||AB|| \le ||A|| ||B||$. First we assume $n_1 = \cdots = n_k$, then for any $1 \le \nu \le k$,

$$sp(B_{1}B_{2}\cdots B_{k}) = \lim_{n \to \infty} \sqrt[n]{\|(B_{1}B_{2}\cdots B_{k})^{n}\|} \\ = \lim_{n \to \infty} \sqrt[n]{\|(B_{1}\cdots B_{\nu-1})(B_{\nu}B_{\nu+1}\cdots B_{\nu-1})^{n-1}(B_{\nu}\cdots B_{k})\|} \\ \leq \lim_{n \to \infty} \sqrt[n]{\|B_{1}\cdots B_{\nu-1}\|\|(B_{\nu}B_{\nu+1}\cdots B_{\nu-1})^{n-1}\|\|B_{\nu}\cdots B_{k}\|} \\ = sp(B_{\nu}B_{\nu+1}\cdots B_{\nu-1}).$$

The same argument leads to the other direction of the inequality. In the following, we deal with the general case. Choose $n \ge \max\{n_1, \dots, n_k\}$, for any $1 \le \nu \le k$, we define a $n \times n$ matrix \widehat{B}_{ν} by

$$\widehat{B}_{\nu} = \begin{pmatrix} B_{\nu} & O_{n_{\nu} \times (n - n_{\nu + 1})} \\ O_{(n - n_{\nu}) \times n_{\nu + 1}} & O_{(n - n_{\nu}) \times (n - n_{\nu + 1})} \end{pmatrix},$$

where we use $O_{p \times q}$ to denote the $p \times q$ zero matrix. Then by the above argument, $\operatorname{sp}(\widehat{B}_1 \widehat{B}_2 \cdots \widehat{B}_k) = \operatorname{sp}(\widehat{B}_{\nu} \widehat{B}_{\nu+1} \cdots \widehat{B}_{\nu-1})$. On the other hand,

$$\widehat{B}_{\nu}\widehat{B}_{\nu+1}\cdots\widehat{B}_{\nu-1} = \begin{pmatrix} B_{\nu}B_{\nu+1}\cdots B_{\nu-1} & O_{n_{\nu}\times(n-n_{\nu})} \\ O_{(n-n_{\nu})\times n_{\nu}} & O_{(n-n_{\nu})\times(n-n_{\nu})} \end{pmatrix}$$

•

This implies that $||(B_1B_2\cdots B_k)^n|| = ||(\widehat{B}_1\widehat{B}_2\cdots \widehat{B}_k)^n||$ for all $n \ge 1$. So

$$\operatorname{sp}(B_1 B_2 \cdots B_k) = \operatorname{sp}(\widehat{B}_1 \widehat{B}_2 \cdots \widehat{B}_k) = \operatorname{sp}(\widehat{B}_\nu \widehat{B}_{\nu+1} \cdots \widehat{B}_{\nu-1}) = \operatorname{sp}(B_\nu B_{\nu+1} \cdots B_{\nu-1}).$$

Proof of Theorem 6.3.1.

Sufficiency. Let \mathcal{C} be a (f, P)-stable multicurve in $\overline{\mathbb{C}} - P$. The multicurves $\mathcal{C}_{\Gamma}, \Sigma_1, \dots, \Sigma_n$ are the subsets of \mathcal{C} defined as in Theorem 6.3.2. We may assume that each curve $\gamma \in \mathcal{C}$ is contained in the interior of some \mathcal{S} -piece by Lemma 6.3.1. If $\lambda(\Gamma, f) < 1$ (notice that this implies $\Lambda^* = \emptyset$ by Lemma 6.2.4) and (h_{ν}, P_{ν}) has no Thurston obstructions for each $\nu \in \Lambda$, then by Theorem 6.3.2, we have

$$\begin{aligned} \lambda(\mathcal{C}, f) &= \max\left\{\lambda(\mathcal{C}_{\Gamma}, f), \sqrt[p_1]{\lambda(\Sigma_1, h_1)}, \cdots, \sqrt[p_n]{\lambda(\Sigma_n, h_n)}\right\} \\ &\leq \max\left\{\lambda(\Gamma, f), \sqrt[p_1]{\lambda(\Sigma_1, h_1)}, \cdots, \sqrt[p_n]{\lambda(\Sigma_n, h_n)}\right\} < 1. \end{aligned}$$

This means (f, P) has no Thurston obstructions.

Necessity. Suppose that (f, P) has no Thurston obstructions. Then $\lambda(\Gamma, f) < 1$ and $\Lambda^* = \emptyset$. Let Σ be a (h_{ν}, P_{ν}) -stable multicurve in $\overline{\mathbb{C}}(S_{\nu}) - P_{\nu}$. Up to homotopy, we may assume that each curve $\gamma \in \Sigma$ is contained in the interior of S_{ν} , so Σ can be considered as a multicurve in $\overline{\mathbb{C}} - P$. In the following, we will use Σ to generate a (f, P)-stable multicurve \mathcal{C} .

For $k \geq 0$, let $\Lambda_k \subset f^{-k}(\Sigma)$ be a multicurve in $\overline{\mathbb{C}} - P$, representing all homotopy classes of non-peripheral curves in $f^{-k}(\Sigma)$. We claim that

For any $\alpha \in \Lambda_i, \beta \in \Lambda_j$ with $0 \leq i < j$, if α is not homotopic to β in $\overline{\mathbb{C}} - P$, then α and β are homotopically disjoint.

In fact, the claim is obviously true in either of the following cases:

1. The curves α and β are contained in two different S-pieces.

2. Either α or β is homotopic a curve in Γ .

So in the following discussion, we assume that α and β are contained in the same S-piece S, and neither is homotopic to a boundary curve of S. We assume further that they intersect homotopically. In this case, one may check that both $f^i(\alpha)$ and $f^i(\beta)$ are contained in $f^i_*(S) = S_{\nu}$, but neither of $f^i(\alpha)$ and $f^i(\beta)$ is homotopic to a boundary curve of S_{ν} . So $f^i(\beta)$ is contained in the unique component of $f^{i-j}(S_{\nu})$ that is parallel to S_{ν} . This implies $i \equiv j \mod p_{\nu}$. Since $f^j(\beta) \in \Sigma$ and Σ is (h_{ν}, P_{ν}) -stable, we have that $f^i(\beta)$ is homotopic in $\overline{\mathbb{C}} - P$ to either a curve of Σ or a curve of Γ . But neither is possible due to our assumption. This ends the proof of the claim.

For $k \geq 0$, we define a collection of Jordan curves C_k such that $\Sigma \subset C_k \subset \Lambda_0 \cup \cdots \cup \Lambda_k$ and C_k represents all homotopy classes of non-peripheral curves in $\Lambda_0 \cup \cdots \cup \Lambda_k$. It follows from the above claim that we can consider C_k as a multicurve in $\overline{\mathbb{C}} - P$ up to homotopy. Notice that C_k is homotopically contained in C_{k+1} , we have $\#C_k \leq \#C_{k+1}$. Since P has finitely many components, $\#C_k$ is uniformly bounded above for all k. So there is an integer $N \geq 0$, such that $\#C_n = \#C_N$ for all $n \geq N$. Let $\mathcal{C} = \mathcal{C}_N$, then \mathcal{C} is a (f, P)-stable multicurve by the choice of N. Let $\mathcal{C}_{\Gamma} = \{\gamma \in \mathcal{C}; \gamma \text{ is homotopic to a curve in } \Gamma\}$, one may verify that $\Sigma = \{\gamma \in \mathcal{C} - \mathcal{C}_{\Gamma}; \gamma \text{ is contained in } S_{\nu}\}$. By Theorem 6.3.2,

$$\lambda(\Sigma, h_{\nu}) \le \lambda(\mathcal{C}, f)^{p_{\nu}} < 1.$$

Thus (h_{ν}, P_{ν}) has no Thurston obstructions.

6.4 Surgery part: Gluing holomorphic models

The aim of this section is to prove the following:

Theorem 6.4.1. Let (f, P) be a Herman map, and

$$\operatorname{Dec}(f, P) = \left(\bigoplus_{\nu \in \Lambda \cup \Lambda^*} (h_{\nu}, P_{\nu})\right)_{\Gamma}.$$

Then (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\Gamma, f) < 1$ and for each $\nu \in \Lambda$, (h_{ν}, P_{ν}) is q.c-equivalent to a rational map.

The proof of Theorem 6.4.1 is based on the quasiconformal surgery.

In Section 6.4.1, we prove the necessity of Theorem 6.4.1. The idea is as follows: we use the rational realization of (f, P), say (R, Q), to generate the partial holomorphic models of $(h_{\nu}, P_{\nu}), \nu \in \Lambda$. The partial holomorphic model of (h_{ν}, P_{ν}) takes the form $R^{p_{\nu}}|_{E_{\nu}}$, where E_{ν} is a multi-connected domain in the Riemann sphere $\overline{\mathbb{C}}$. The holomorphic map $R^{p_{\nu}}|_{E_{\nu}}$ can be extended to a Siegel map or Thurston map, say (g_{ν}, Q_{ν}) , q.c-equivalent to (h_{ν}, P_{ν}) . The map (g_{ν}, Q_{ν}) can be made holomorphic outside a neighborhood of the boundary ∂E_{ν} . In the final step, we apply quasiconformal surgery to make the map (g_{ν}, Q_{ν}) globally holomorphic and get a rational realization of (h_{ν}, P_{ν}) .

In Section 6.4.2, we prove the sufficiency of Theorem 6.4.1 assuming $\Gamma = \emptyset$. This part is the inverse procedure of Section 6.4.1. We use the rational realizations of $(h_{\nu}, P_{\nu}), \nu \in \Lambda$ to generate the partial holomorphic models for (f, P). These partial holomorphic models can be glued along $\Sigma = \Gamma_0$ in a suitable fashion into a branched covering (g, Q), holomorphic in most part of $\overline{\mathbb{C}}$ and q.c-equivalent to (f, P). Finally, we apply quasiconformal surgery to make the map (g, Q) globally holomorphic.

In Section 6.4.3, we prove the sufficiency of Theorem 6.4.1 in the more general case $\Gamma \neq \emptyset$. The idea of reconstruction of the rational realization of (f, P) via gluing the rational realizations of $(h_{\nu}, P_{\nu}), \nu \in \Lambda$ is essentially the same as that in Section 6.4.2. But this setion provides very interesting and technical flavor because of the algebraic condition $\lambda(\Gamma, f) < 1$. In most

part of this section, we deal with this condition and shows that this algebraic condition is equivalent to the Grötzsch inequality in the homorphic setting. Thus it enables us to glue the partial homorphic models of (f, P) along Σ in a suitable fashion into a branched covering (g, Q), holomorphic in most part of $\overline{\mathbb{C}}$ and q.c-equivalent to (f, P). The last step is similar to the previous sections, it is a quasiconformal surgery procedure.

6.4.1 Rational realizations can descend

Theorem 6.4.2. (Marked McMullen Theorem) Let R be a rational map, M be a closed set containing the postcritical set P_R and $R(M) \subset M$. Let Γ be a multicurve in $\overline{\mathbb{C}} - M$. Then $\lambda(\Gamma, R) \leq 1$. If $\lambda(\Gamma, R) = 1$, then either R is postcritically finite whose orbifold has signature (2, 2, 2, 2); or R is postcritically infinite, and Γ includes a curve contained in a periodic Siegel disk or Herman ring.

We remark that the definition of the multicurve in $\overline{\mathbb{C}} - M$ is similar to the definition of the multicurve in $\overline{\mathbb{C}} - P$. The 'Marked McMullen Theorem' is slightly stronger than McMullen's original result [McM1], but the proof goes through without any problem.

Proof of the necessity of Theorem 6.4.1 Suppose that (f, P) is q.cequivalent to a rational map (R, Q) via a pair of quasiconformal maps (ϕ_0, ϕ_1) . Then the (f, P)-stable multicurve Γ in $\overline{\mathbb{C}} - P$ induces a (R, Q)-stable multicurve $\phi_0(\Gamma) := \{\phi_0(\gamma); \gamma \in \Gamma\}$ in $\overline{\mathbb{C}} - Q$. Since the marked set Q contains all possible Siegel disks and Herman rings of R, it follows from Theorem 6.4.2 that $\lambda(\Gamma, f) = \lambda(\phi_0(\Gamma), R) < 1$.

Notice that $\lambda(\Gamma, f) < 1$ implies $\Lambda^* = \emptyset$ by Lemma 6.2.4. In the following, we will show that for each $\nu \in \Lambda$, (h_{ν}, P_{ν}) is q.c-equivalent to a rational map.

Let $H_0: [0,1] \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ be an isotopy between ϕ_0 and ϕ_1 rel P. That is, $H_0: [0,1] \times \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a continuous map such that $H_0(0,\cdot) = \phi_0, H_0(1,\cdot) = \phi_1$ and $H_0(t,z) = \phi_0(z)$ for all $(t,z) \in [0,1] \times P$. Moreover, for any $t \in [0,1]$, $H_0(t,\cdot): \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ is a quasiconformal map. Then there is a unique lift of H_0 , say H_1 , such that $H_0(t, f(z)) = R(H_1(t,z))$ for all $(t,z) \in [0,1] \times \overline{\mathbb{C}}$, with basepoint $H_1(0,\cdot) = \phi_1$. Set $\phi_2 = H_1(1,\cdot)$. Inductively, for any $k \ge 1$, let H_{k+1} be the unique lift of H_k such that $H_k(t, f(z)) = R(H_{k+1}(t,z))$ for all $(t,z) \in [0,1] \times \overline{\mathbb{C}}$, and $H_{k+1}(0,\cdot) = \phi_{k+1}$. Set $\phi_{k+2} = H_{k+1}(1,\cdot)$. In this way, we can get a sequence of quasiconformal maps $\phi_0, \phi_1, \phi_2, \cdots$, such that the following diagram commutes.

$$\cdots \xrightarrow{f} (\overline{\mathbb{C}}, P) \xrightarrow{\phi_3} \phi_2 \Big| \qquad \phi_1 \Big| \qquad \phi_0 \Big|$$
$$\cdots \xrightarrow{R} (\overline{\mathbb{C}}, Q) \xrightarrow{R} (\overline{\mathbb{C}}, Q) \xrightarrow{R} (\overline{\mathbb{C}}, Q) \xrightarrow{R} (\overline{\mathbb{C}}, Q) \xrightarrow{R} (\overline{\mathbb{C}}, Q)$$

One can verify that for any $k \ge 0$, ϕ_{k+1} is isotopic to ϕ_k rel $f^{-k}(P)$.

Fix some $\nu \in \Lambda$, let D_{ν} be the union of all rotation disks of (h_{ν}, P_{ν}) intersecting ∂S_{ν} . We set $D_{\nu} = \emptyset$ if $\partial_0(S_{\nu}) = \emptyset$.

Choose a large integer $\ell > 0$ such that $\cup \Gamma \subset f^{-\ell+p_{\nu}}(P)$. Then we extend $\phi_{\ell}|_{S_{\nu}}$ to a quasiconformal map $\Phi : \overline{\mathbb{C}}(S_{\nu}) \to \overline{\mathbb{C}}$. We require that Φ is holomorphic in D_{ν} if $D_{\nu} \neq \emptyset$.

Notice that there is a unique component E_{ν} of $f^{-p_{\nu}}(S_{\nu})$ parallel to S_{ν} . By the choice of ℓ , $\partial E_{\nu} \subset f^{-p_{\nu}}(\cup \Gamma) \subset f^{-\ell}(P)$, so $\phi_{\ell+p_{\nu}}$ and ϕ_{ℓ} are isotopic rel $f^{-\ell}(P)$. In particular, $\phi_{\ell+p_{\nu}}|_{\partial E_{\nu}} = \phi_{\ell}|_{\partial E_{\nu}} = \Phi|_{\partial E_{\nu}}$.

Denote the components of $\overline{\mathbb{C}}(S_{\nu}) - (E_{\nu} \cup D_{\nu})$ by $\{U_j; j \in I\}$, where Iis a finite index set. Each U_j is a disk, containing at most one point in P_{ν} . For any $j \in I$, let $V_j \Subset U_j$ be a disk such that $V_j \cap P_{\nu} = U_j \cap P_{\nu}$ and $U_j \setminus V_j \subset f^{-\ell}(P) \setminus P$. By Measurable Riemann Mapping Theorem, there is a quasiconformal homeomorphism $\Psi_j : V_j \to \Phi(V_j)$ whose Beltrami coefficient satisfies $\mu_{\Psi_j}(z) = \mu_{\Phi \circ h_{\nu}}(z)$ for $z \in V_j$. If U_j contains a point $p \in P_{\nu}$, we further require that $\Psi_j(p) = \Phi(p)$.

We can construct a quasiconformal map $\Psi : \overline{\mathbb{C}}(S_{\nu}) \to \overline{\mathbb{C}}$ by

$$\Psi(z) = \begin{cases} \Phi(z), & z \in D_{\nu}, \\ \phi_{\ell+p_{\nu}}(z), & z \in E_{\nu}, \\ \Psi_{j}(z), & z \in V_{j}, j \in I, \\ q.c \ interpolation, & z \in U_{j} \setminus V_{j}, j \in I. \end{cases}$$

One may verify that Φ is homotopic to Ψ rel P_{ν} . Thus (h_{ν}, P_{ν}) is cequivalent to $(g_{\nu}, Q_{\nu}) := (\Phi \circ h_{\nu} \circ \Psi^{-1}, \Phi(P_{\nu}))$ via (Φ, Ψ) . Moreover, (g_{ν}, Q_{ν}) is holomorphic outside $\Psi(\bigcup_{j \in I} (U_j \setminus \overline{V_j}))$.

In the following, we will construct a (g_{ν}, Q_{ν}) -invariant complex structure. For each $j \in I$, we may assume that the annulus $U_j \setminus \overline{V_j}$ is thin enough such that for k > 1 large enough, $g_{\nu}^k(\Psi(U_j \setminus \overline{V_j}))$ is contained either in a rotation disk of g_{ν} , or in a neighborhood of a periodic critical point near which g_{ν} is holomorphic. Let $k_j \geq 1$ be the first integer such that g_{ν} is holomorphic in $g_{\nu}^{k_j}(\Psi(U_j \setminus \overline{V_j}))$. Define a complex structure in $\Psi(U_j \setminus \overline{V_j})$ by pulling back the standard complex structure in $g_{\nu}^{k_j}(\Psi(U_j \setminus \overline{V_j}))$ via $g_{\nu}^{k_j}$. Then we define a complex structure in $g_{\nu}^{-k}(\Psi(\cup_{j \in I}(U_j \setminus \overline{V_j})))$ by pulling back the complex structure in $\Psi(\bigcup_{j\in I}(U_j\setminus \overline{V_j}))$ via g_{ν}^k for all $k \geq 0$ and define the standard complex structure elsewhere. In this way, we get a (g_{ν}, Q_{ν}) -invariant complex structure σ . The Beltrami coefficient μ of σ satisfies $\|\mu\|_{\infty} < 1$ since (g_{ν}, Q_{ν}) is holomorphic outside $\Psi(\bigcup_{j\in I}(U_j\setminus \overline{V_j}))$.

By Measurable Riemann Mapping Theorem, there is a quasiconformal map $\zeta : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ whose Beltrami coefficient is μ . Let $f_{\nu} = \zeta \circ g_{\nu} \circ \zeta^{-1}$, then f_{ν} is a rational map and (h_{ν}, P_{ν}) is q.c-equivalent to $(f_{\nu}, \zeta \circ \Phi(P_{\nu}))$ via $(\zeta \circ \Phi, \zeta \circ \Psi)$. See the following commutative diagram.

6.4.2 Promotion of rational realizations when $\Gamma = \emptyset$

Proof of the sufficiency of Theorem 6.4.1, assuming $\Gamma = \emptyset$

Since $\Gamma = \emptyset$, for each S-piece S, we have $\partial(S) = \partial_0(S) \subset \Gamma_0$, where Γ_0 is the collection of (f, P)-periodic curves defined in Section 6.2. It follows from Lemma 6.2.3 that S is f_* -periodic. So each S-piece is f_* -periodic, and S can be written as $\{f_*^j(S_\nu); 0 \leq j < p_\nu, \nu \in \Lambda\}$. Moreover, any two S-pieces contained in the same f_* -cycle have the same number of boundary curves.

Suppose that (h_{ν}, P_{ν}) is *c*-equivalent to a rational map (R_{ν}, Q_{ν}) via a pair of quasiconformal maps (Φ_{ν}, Ψ_{ν}) for $\nu \in \Lambda = [1, n]$.

Step 1: Getting partial holomorphic models. For each S-piece S, there exist a pair of quasiconformal maps $(\Phi_S, \Psi_S) : \overline{\mathbb{C}}(S) \to \overline{\mathbb{C}}$ and a rational map R_S such that the following diagram commutes:



It suffices to show that for each f_* -cycle $\langle S_{\nu}, \cdots, f_*^{p_{\nu}-1}(S_{\nu}) \rangle$, there exist a sequence of quasiconformal maps $\Psi_{S_{\nu}}, \Phi_{f_*^k(S_{\nu})}, 0 \leq k < p_{\nu}$ and a sequence of

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 132 THEOREMS

rational maps $R_{f_*^k(S_\nu)}, 0 \leq k < p_\nu$ such that the following diagram commutes

$$\overline{\mathbb{C}}(S_{\nu}) \xrightarrow{H_{S_{\nu}}} \overline{\mathbb{C}}(f_{*}(S_{\nu})) \xrightarrow{H_{f_{*}(S_{\nu})}} \overline{\mathbb{C}}(f_{*}^{2}(S_{\nu})) \longrightarrow \cdots \longrightarrow \overline{\mathbb{C}}(f_{*}^{p_{\nu}-1}(S_{\nu})) \xrightarrow{H_{f_{*}^{p_{\nu}-1}(S_{\nu})}} \overline{\mathbb{C}}(S_{\nu}) \xrightarrow{\Phi_{f_{*}^{p_{\nu}-1}(S_{\nu})}} \overline{\mathbb{C}}(S_{\nu}) \xrightarrow{\Phi_{f_{*}^{p_{\nu}-1}(S_{\nu})}} \xrightarrow{\Phi_{F_{\nu}}} \overline{\mathbb{C}} \xrightarrow{\Phi_{F_{\nu}}} \xrightarrow{\Phi_{F_{\nu}}} \overline{\mathbb{C}} \xrightarrow{\Phi_{F_{\nu}}} \xrightarrow{\mathbb{C}} \xrightarrow{\Phi_{F_{\nu}}} \overline{\mathbb{C}} \xrightarrow{\Phi_{F_{\nu}}} \overline{\mathbb{C}} \xrightarrow{\Phi_{F_{\nu}}} \xrightarrow{\mathbb{C}} \overline{\mathbb{C}} \xrightarrow{R_{f_{*}^{p_{\nu}-1}(S_{\nu})}} \xrightarrow{\mathbb{C}} \xrightarrow{\mathbb{$$

The constructions of the two sequences of maps are as follows: First, we set $\Phi_{S_{\nu}} = \Phi_{\nu}$ and $\Psi_{S_{\nu}} = \Psi_{\nu}$. By Measurable Riemann Mapping Theorem, there is a quasiconformal map $\Phi_{f_*^{p_{\nu}-1}(S_{\nu})} : \overline{\mathbb{C}}(f_*^{p_{\nu}-1}(S_{\nu})) \to \overline{\mathbb{C}}$ such that $\Phi_{f_*^{p_{\nu}-1}(S_{\nu})}^*(\sigma_0) = (\Phi_{S_{\nu}} \circ H_{f_*^{p_{\nu}-1}(S_{\nu})})^*(\sigma_0)$, where σ_0 is the standard complex structure. Then $R_{f_*^{p_{\nu}-1}(S_{\nu})} = \Phi_{S_{\nu}} \circ H_{f_*^{p_{\nu}-1}(S_{\nu})} \circ \Phi_{f_*^{p_{\nu}-1}(S_{\nu})}^{-1}$ is a rational map. Inductively, for $i = p_{\nu} - 2, \cdots, 1$, we can get a quasiconformal map $\Phi_{f_*^i(S_{\nu})} :$

Inductively, for $i = p_{\nu} - 2, \cdots, 1$, we can get a quasiconformal map $\Phi_{f_*^i(S_{\nu})} : \overline{\mathbb{C}}(f_*^i(S_{\nu})) \to \overline{\mathbb{C}}$ such that $R_{f_*^i(S_{\nu})} = \Phi_{f_*^{i+1}(S_{\nu})} \circ H_{f_*^i(S_{\nu})} \circ \Phi_{f_*^i(S_{\nu})}^{-1}$ is a rational map.

Finally, we set $R_{S_{\nu}} = \Phi_{f_*(S_{\nu})} \circ H_{S_{\nu}} \circ \Psi_{S_{\nu}}^{-1}$. Then $R_{\nu} = R_{f_*^{p_{\nu}-1}(S_{\nu})} \circ \cdots \circ R_{f_*(S_{\nu})} \circ R_{S_{\nu}}$. Hence $R_{S_{\nu}}$ is also a rational map.

Set $\Psi_{f^i_*(S_{\nu})} = \Phi_{f^i_*(S_{\nu})}$ for $1 \leq i < p_{\nu}$. Then the pair of quasiconformal maps $(\Phi_{f^i_*(S_{\nu})}, \Psi_{f^i_*(S_{\nu})})$ and the rational map $R_{f^i_*(S_{\nu})}(0 \leq i < p_{\nu})$ are as required.

Step 2: Gluing holomorphic models. For each S-piece S, recall that E_S is the unique \mathcal{E} -piece parallel to S. Since $\Gamma = \emptyset$, each boundary curve of S is also a boundary curve of E_S . So each component of $S - E_S$ is a disk, containing at most one point in P. Let $\{U_k; k \in I_S\}$ be the collection of all components of $S \setminus E_S$, where I_S is the finite index set induced by S. For any $k \in I_S$, let $V_k \Subset U_k$ be a disk such that $V_k \cap P = U_k \cap P$ and $U_k \setminus V_k \subset f^{-1}(P) \setminus P$. By the Measurable Riemann Mapping Theorem, there is a quasiconformal homeomorphism $\phi_k : V_k \to \Psi_S(V_k)$ whose Beltrami coefficient satisfies

$$\mu_{\phi_k}(z) = \sum_{\mathcal{E} \ni E \subset U_k} \chi_E(z) \mu_{\Phi_{f(E)} \circ f}(z), \ z \in V_k.$$

Here the summation is taken over all the \mathcal{E} -pieces contained in U_k . If V_k contains a point $p \in P$, we further require that $\phi_k(p) = \Phi_S(p)$.

Now we define a quasiconformal homeomorphism $\psi_S: S \to \Phi_S(S)$ by

$$\psi_S(z) = \begin{cases} \Psi_S(z), & z \in E_S, \\ \phi_k(z), & z \in V_k, k \in I_S, \\ q.c \ interpolation, & z \in U_k \setminus V_k, k \in I_S. \end{cases}$$

Define a quasiconformal map $\Theta : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ by $\Theta|_S = \psi_S^{-1} \circ \Phi_S$ for all $S \in \mathcal{S}$. The map Θ is isotopic to the identity map rel P. Let $\Phi : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ be a quasiconformal map such that

$$\mu_{\Phi}(z) = \sum_{S \in \mathcal{S}} \chi_{S}(z) \mu_{\Phi_{S}}(z), \ z \in \overline{\mathbb{C}}.$$

Let $\Psi = \Phi \circ \Theta^{-1}$. The pair of quasiconformal maps (Φ, Ψ) can be considered as the gluing of $(\Phi_S|_S, \Psi_S|_S)_{S \in S}$. In this way, (f, P) is q.c-equivalent to the Herman map $(g, Q) := (\Phi \circ f \circ \Psi^{-1}, \Phi(P))$ via (Φ, Ψ) .

Step 3: Applying quasiconformal surgery. We first show that the Herman map (g, Q) is holomorphic in most parts of $\overline{\mathbb{C}}$. In fact, it is holomorphic outside $X := \Psi(\bigcup_{S \in \mathcal{S}} \bigcup_{k \in I_S} (U_k \setminus V_k))$. To see this, we fix some \mathcal{S} -piece S. The restriction $g|_{\Psi(E_S)}$ can be decomposed into

$$g|_{\Psi(E_S)} = (\Phi \circ \Phi_{f(E_S)}^{-1}) \circ (\Phi_{f(E_S)} \circ f \circ \Psi_S^{-1})|_{\Psi_S(E_S)} \circ (\Phi_S \circ \Phi^{-1})|_{\Psi(E_S)}.$$

For any $k \in I_S$, any \mathcal{E} -piece $E \subset U_k$, the restriction $g|_{\Psi(V_k \cap E)}$ can be decomposed into

$$g|_{\Psi(V_k \cap E)} = (\Phi \circ \Phi_{f(E)}^{-1}) \circ (\Phi_{f(E)} \circ f \circ \phi_k^{-1})|_{\phi_k(V_k \cap E)} \circ (\Phi_S \circ \Phi^{-1})|_{\Psi(V_k \cap E)}$$

In either case, each factor of the decompositions of g is holomorphic in its domain of definition. So $g|_S$ is holomorphic outside $\Psi(\bigcup_{k \in I_S} (U_k \setminus V_k))$. It follows that (g, Q) is holomorphic outside X.

Let R_A be the union of all rotation annuli of g. Then one can check that $X \subset g^{-1}(R_A) \setminus R_A$. Let σ_0 be the standard complex structure in $\overline{\mathbb{C}}$. Define a g-invariant complex structure σ by

$$\sigma = \begin{cases} (g^k)^*(\sigma_0), & \text{in } g^{-k}(R_A) \setminus g^{-k+1}(R_A), \ k \ge 1, \\ \sigma_0, & \text{in } \overline{\mathbb{C}} - \bigcup_{k \ge 1} (g^{-k}(R_A) \setminus g^{-k+1}(R_A)). \end{cases}$$

Since g is holomorphic outside X, the Beltrami coefficient μ of σ satisfies $\|\mu\|_{\infty} < 1$. By Measurable Riemann Mapping Theorem, there is a quasiconformal map $\zeta : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that $\zeta^*(\sigma_0) = \sigma$. Let $R = \zeta \circ g \circ \zeta^{-1}$, then R is a rational map and (f, P) is c-equivalent to $(R, \zeta \circ \Phi(P))$ via $(\zeta \circ \Phi, \zeta \circ \Psi)$. \Box

6.4.3 Promotion of rational realizations when $\Gamma \neq \emptyset$

This is the technical part. We assume in this section that $\Gamma \neq \emptyset$, $\lambda(\Gamma, f) < 1$ and for each $\nu \in \Lambda = [1, n]$, the map (h_{ν}, P_{ν}) is q.c-equivalent to a rational map, we will show that (f, P) is q.c-equivalent to a rational map.

To begin with, we recall a result on non-negative matrix. Let W be a non-negative square matrix (i.e. each entry is a nonnegative real number).

It's known from Perron-Frobenius Theorem that the spectral radius of W is an eigenvalue of W, named the *leading eigenvalue*. Let $v = (v_1, \dots, v_n)^t \in \mathbb{R}^n$ be a vector, we say v > 0 if for each $i, v_i > 0$. The following Lemma can be found in [CT1], Lemma A.1.

Lemma 6.4.1. Let W be a non-negative square matrix with leading eigenvalue λ . Then $\lambda < 1$ iff there is a vector v > 0 such that Wv < v.

With the help of Lemma 6.4.1, we turn to our discussion. First, $\lambda(\Gamma, f) < 1$ implies Wv < v, where W is the (f, P)-transition matrix of Γ and $v \in \mathbb{R}^{\Gamma}$ is a positive vector. That is, there is a positive function $v : \Gamma \to \mathbb{R}^+$ such that for any $\gamma \in \Gamma$,

$$(Wv)(\gamma) = \sum_{\beta \in \Gamma} \sum_{\alpha \sim \gamma} \frac{v(\beta)}{\deg(f : \alpha \to \beta)} < v(\gamma),$$

where the second summation is taken over all components α of $f^{-1}(\beta)$ homotopic to γ in $\overline{\mathbb{C}} - P$.



Figure 6.3: Orientation and labeling

Recall that for each curve $\gamma \in \Sigma$, there exist exactly two S-pieces, say S_{γ}^+ and S_{γ}^- , such that $S_{\gamma}^+ \cap S_{\gamma}^- = \gamma$. For each curve $\gamma \in \Sigma$, we can associate an orientation such that f preserves the orientation. We may assume that the notations S_{γ}^+ and S_{γ}^- are chosen such that S_{γ}^+ lies on the left side of γ and S_{γ}^- lies on the right side of γ .

Here, we borrow some notations from Lemma 6.2.1. Recall that Γ_0 is the collection of the (f, P)-periodic curves that generates Γ , and $\Gamma_n = \{\gamma \in \Gamma; n \text{ is the first integer such that } f^n(\gamma) \in \Gamma_0\}$ for $n \ge 1$. One may verify that if $\delta \in f^{-1}(\Gamma)$ is homotopic to a curve $\gamma \in \Gamma$ in $\overline{\mathbb{C}} - P$, then δ is necessarily contained in $S_{\gamma}^+ \cup S_{\gamma}^-$. Moreover, if $\gamma \in \Gamma_1$, then $\delta \neq \gamma$; if $\gamma \in \Gamma_k$ for some $k \geq 2$, it can happen that $\delta = \gamma$.

In the following, for each curve $\gamma \in \Gamma = \bigcup_{n \geq 1} \Gamma_n$, we will associate two positive numbers $\rho(S_{\gamma}^+, \gamma)$ and $\rho(S_{\gamma}^-, \gamma)$ inductively.

For $\gamma \in \Gamma_1$, we choose two positive numbers $\rho(S^+_{\gamma}, \gamma)$ and $\rho(S^-_{\gamma}, \gamma)$ such that

$$\begin{split} \rho(S_{\gamma}^{+},\gamma)+\rho(S_{\gamma}^{-},\gamma)&=1,\\ \sum_{\beta\in\Gamma}\sum_{\alpha\sim\gamma,\alpha\subset S_{\gamma}^{\omega}}\frac{v(\beta)}{\deg(f:\alpha\to\beta)} < v(\gamma)\rho(S_{\gamma}^{\omega},\gamma), \ \omega\in\{\pm\}. \end{split}$$

Suppose that for each curve $\alpha \in \Gamma_1 \cup \cdots \cup \Gamma_k$, we have already chosen two numbers $\rho(S^+_{\alpha}, \alpha)$ and $\rho(S^-_{\alpha}, \alpha)$. Then for $\gamma \in \Gamma_{k+1}$ (notice that $f(\gamma) \in \Gamma_k$), we can find two positive numbers $\rho(S^+_{\gamma}, \gamma)$ and $\rho(S^-_{\gamma}, \gamma)$ such that:

$$\rho(S_{\gamma}^+, \gamma) + \rho(S_{\gamma}^-, \gamma) = 1,$$

$$\frac{v(f(\gamma))}{\deg(f|_{\gamma})}\rho(S^{\omega}_{f(\gamma)}, f(\gamma)) + \sum_{\beta \in \Gamma} \sum_{\alpha \sim \gamma, \alpha \subset S^{\omega}_{\gamma} \setminus \gamma} \frac{v(\beta)}{\deg(f: \alpha \to \beta)} < v(\gamma)\rho(S^{\omega}_{\gamma}, \gamma), \ \omega \in \{\pm\}$$

In fact, we can take

$$\rho(S_{\gamma}^{\omega},\gamma) = \frac{\frac{v(f(\gamma))}{\deg(f|_{\gamma})}\rho(S_{f(\gamma)}^{\omega},f(\gamma)) + \sum_{\beta\in\Gamma}\sum_{\alpha\sim\gamma,\alpha\subset S_{\gamma}^{\omega}\setminus\gamma}\frac{v(\beta)}{\deg(f:\alpha\to\beta)}}{\sum_{\beta\in\Gamma}\sum_{\alpha\sim\gamma}\frac{v(\beta)}{\deg(f:\alpha\to\beta)}}, \ \omega\in\{\pm\}.$$

Potentials

Let \mathbb{D} be the unit disk. A marked disk is a pair (Δ, a) with Δ an open hyperbolic disk in $\overline{\mathbb{C}}$ and $a \in \Delta$ a marked point. An equipotential γ of the marked disk (Δ, a) is a Jordan curve that is mapped to a round circle with center zero under a conformal isomorphism $\phi : \Delta \to \mathbb{D}$ with $\phi(a) = 0$. The potential $\varpi(\gamma)$ of γ is defined to be $\operatorname{mod}(A(\partial \Delta, \gamma))$, the modulus of the annulus between $\partial \Delta$ and γ . One may check that these definitions are independent of the choice of the Riemann mapping ϕ .

Suppose that (f, P) is either a Thurston rational map or a Siegel rational map, with a non-empty Fatou set. Recall that P is a marked set containing the poscritical set P_f . Then each periodic Fatou component is either a superattracting domain or a Siegel disk. If f has a superattracting Fatou component D, then every Fatou component Δ which is eventually mapped onto D can be marked by the unique eventually periodic point $a \in \Delta$. We

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 136 THEOREMS

call (Δ, a) a I-type marked disk of f. Notice that every equipotential in a superattracting Fatou component corresponds to a round circle in Böttcher coordinates. If f has a Siegel disk D, then it is known that the boundary ∂D is contained in the postcritical set P_f . Let z_0 be the center of the Siegel disk D, the intersection $P \cap (D - \{z_0\})$ is either empty or consists of finitely many (f, P)-periodic Jordan curves. Let $D_0 \subset D$ be the component of $\overline{\mathbb{C}} - (P \setminus \{z_0\})$ containing z_0 . For any $k \geq 0$ and any component Δ of $f^{-k}(D_0)$, one can verify that Δ is a disk and there is a unique point $a \in \Delta \cap f^{-k}(z_0)$. We call (Δ, a) a II-type marked disk of (f, P).

A positive function

For each curve $\gamma \in \Sigma$, we associate a very thin annular neighborhood A^{γ} of γ . The annulas A^{γ} is chosen as follows: If $\gamma \in \Gamma_0$, then A^{γ} is a proper subset of the rotation annulus containing γ such that $f(A^{\gamma}) = A^{f(\gamma)}$ and $\overline{A^{\gamma}} \cap f(P - \cup \mathcal{A}) = \emptyset$. If $\gamma \in \Gamma_k$ for some $k \geq 1$, then A^{γ} is the component of $f^{-k}(A^{f^k(\gamma)})$ containing γ .

We define

 $\mathcal{S}^{\star} = \{U; \ U \text{ is a connected component of } \overline{\mathbb{C}} - \cup_{\gamma \in \Sigma} A^{\gamma} \},\$ $\mathcal{E}^{\star} = \{V; \ V \text{ is a connected component of } \overline{\mathbb{C}} - f^{-1}(\cup_{\gamma \in \Sigma} A^{\gamma}) \}.$

Each element of S^* (resp. E^*) is called an \mathcal{S}^* -piece (resp. \mathcal{E}^* -piece). We will use S^* (resp. E^*) to denote an \mathcal{S}^* -piece (resp. \mathcal{E}^* -piece). Notice that for each \mathcal{S} -piece S (resp. \mathcal{E} -piece E), there is a unique \mathcal{S}^* -piece (resp. \mathcal{E}^* -piece) contained in S (resp. \mathcal{E}), we denote this piece by S^* (resp. E^*). Similarly as in Section 6.2, we can define E_{S^*} , the unique \mathcal{E}^* -piece contained in S^* and parallel to S^* . The map $f_* : \mathcal{S}^* \to \mathcal{S}^*$ is defined by $f_*(S^*) = f(E_{S^*})$. Moreover, the notations $\partial(E^*), \partial(S^*), \partial_0(S^*), \partial_1(S^*), \partial_2(S^*), \overline{\mathbb{C}}(S^*)$ and the marked disk extension $H_{S^*} : \overline{\mathbb{C}}(S^*) \to \overline{\mathbb{C}}(f_*(S^*))$ are defined similarly. Let

$$h_{\nu}^{\star} = H_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})} \circ \dots \circ H_{f_{*}(S_{\nu}^{\star})} \circ H_{S_{\nu}^{\star}}, \ P_{\nu}^{\star} = P(S_{\nu}^{\star}), \ 1 \le \nu \le n.$$

One can see that this modification doesn't change the combinatorics and rational realizations of the maps $(h_{\nu}, P_{\nu}), 1 \leq \nu \leq n$. That is to say

• (h_{ν}, P_{ν}) has no Thurston obstructions if and only if $(h_{\nu}^{\star}, P_{\nu}^{\star})$ has no Thurston obstructions.

• (h_{ν}, P_{ν}) is c-equivalent to a rational map if and only if $(h_{\nu}^{\star}, P_{\nu}^{\star})$ is c-equivalent to a rational map.

The virtue of this modification is that we can construct deformations in a neighborhood of each curve $\gamma \in \Sigma$. This will be seen in the last step of the proof of Theorem 6.4.1 when we apply the quasiconformal surgery to glue all holomorphic models together to obtain a rational realization of (f, P).



Figure 6.4: A S-piece S with boundary $\partial S = \gamma_1 \cup \gamma_2 \cup \gamma_3 \cup \gamma_4$. S contains a S^* -piece S^* , whose boundary curves are $\beta_1, \beta_2, \beta_3$ and α_4 .

For each curve $\gamma \in \Sigma$, let $\alpha_{\gamma}, \beta_{\gamma}$ be the two boundary curves of A^{γ} . Define

$$\Sigma^{\star} = \{\alpha_{\gamma}, \beta_{\gamma}; \gamma \in \Sigma\}, \quad \Gamma^{\star} = \{\alpha_{\gamma}, \beta_{\gamma}; \gamma \in \Gamma\}, \quad \Gamma_{k}^{\star} = \{\alpha_{\gamma}, \beta_{\gamma}; \gamma \in \Gamma_{k}\}, k \ge 0.$$

We define a map $\pi: \Sigma^* \to \Sigma$ by $\pi(\alpha) = \gamma$ if α is a boundary curve of A^{γ} . It's obvious that for each curve $\gamma \in \Sigma$, $\pi^{-1}(\gamma) = \{\alpha_{\gamma}, \beta_{\gamma}\}$. For each curve $\gamma \in \Sigma^*$, let S_{γ} (resp. S_{γ}^*) be the unique \mathcal{S} -piece (resp. \mathcal{S}^* -piece) that contains γ .

Now we define a positive function $\sigma_t : \Sigma^* \to \mathbb{R}^+$, where t is a positive parameter, as follows:

First suppose $\gamma \in \Gamma^*$. In this case, $\gamma \in \partial_1(S^*_{\gamma}) \cup \partial_2(S^*_{\gamma})$. If $\gamma \in \partial_1(S^*_{\gamma})$, we define

$$\sigma_t(\gamma) = t \cdot \rho(S_\gamma, \pi(\gamma)) \cdot v(\pi(\gamma)).$$

If $\gamma \in \partial_2(S^*_{\gamma})$, we define

$$\sigma_t(\gamma) = \begin{cases} \frac{\sigma_t(f(\gamma))}{\deg(f|_{\gamma})}, & \text{if } S_{\gamma}^{\star} \in \mathcal{S}^{\star} - \{S_1^{\star}, \cdots, S_n^{\star}\}, \\ t \cdot \rho(S_{\gamma}, \pi(\gamma)) \cdot v(\pi(\gamma)), & \text{if } S_{\gamma}^{\star} \in \{S_1^{\star}, \cdots, S_n^{\star}\}. \end{cases}$$

Now we consider $\gamma \in \Gamma_0^*$. In this case, there is an integer $p \ge 0$ such that $f^p(\gamma) \in \partial_0(S_k^*)$ for some $k \in \{1, \dots, n\}$. Then $f^p(\gamma)$ is contained in a rotation disk of (h_k^*, P_k^*) , say D, with the center a. Notice that there is an annulus $A \subset \Delta$ such that:

• the inner boundary of A is $f^p(\gamma)$,

• the outer boundary of A is a (h_k^*, P_k^*) -periodic curve in the marked set P_k^* ,

• $A \cap P_k^{\star} = \emptyset.$

We define $\sigma_t(\gamma)$ to be the modulus of A. By definition, $\sigma_t(\gamma) = \sigma_t(f(\gamma)) = \cdots$. In this way, for all curves $\gamma \in \Sigma^*$, the quantity $\sigma_t(\gamma)$ is well defined.

Lemma 6.4.2. When t is large enough, the function $\sigma_t : \Sigma^* \to \mathbb{R}^+$ satisfies: 1. For any $\gamma \in \Gamma^*$, $\sigma_t(\gamma) \leq t \cdot \rho(S_{\gamma}, \pi(\gamma)) \cdot v(\pi(\gamma))$.

2. For every $\gamma \in \Sigma$, suppose that $\pi^{-1}(\gamma) = \{\alpha_{\gamma}, \beta_{\gamma}\}$. Then

$$\sigma_t(\alpha_{\gamma}) + \sigma_t(\beta_{\gamma}) \le \begin{cases} tv(\gamma), & \text{if } \gamma \in \Gamma, \\ mod(A_{\gamma}), & \text{if } \gamma \in \Gamma_0, \end{cases}$$

where A_{γ} is the rotation annulus of (f, P) that contains γ if $\gamma \in \Gamma_0$.

3. For every $\gamma \in \Gamma^*$, if $\gamma \in \partial_1(S^*_{\gamma})$, then we have the following inequality:

$$\sum_{\beta \in \Gamma^{\star}} \sum_{\alpha \sim \gamma, \alpha \subset S^{\star}_{\gamma}} \frac{\sigma_t(\beta)}{\deg(f: \alpha \to \beta)} < \sigma_t(\gamma)$$

where the second summation is taken over all components of $f^{-1}(\beta)$ contained in S^*_{γ} and homotopic to γ in $\overline{\mathbb{C}} - P$.

Proof. 1. Notice that if $\gamma \in \Gamma^*$, then $\gamma \in \partial_1(S^*_{\gamma}) \cup \partial_2(S^*_{\gamma})$. If $\gamma \in \partial_1(S^*_{\gamma})$ or $S^*_{\gamma} \in \{S^*_1, \cdots, S^*_n\}$, then by definition, $\sigma_t(\gamma) = t\rho(S_{\gamma}, \pi(\gamma))v(\pi(\gamma))$. Now suppose $\gamma \in \partial_2(S^*_{\gamma})$ and $S^*_{\gamma} \in \mathcal{S}^* - \{S^*_1, \cdots, S^*_n\}$. Let $p \ge 1$ be the first integer such that $f^p_*(S^*_{\gamma}) \in \{S^*_1, \cdots, S^*_n\}$. There is a largest number $k \in \{0, \cdots, p\}$ such that $f^j(\gamma) \in \partial_2(f^*_j(S^*_{\gamma}))$ for $0 \le j < k$. Thus we have

$$\sigma_t(\gamma) = \frac{\sigma_t(f(\gamma))}{\deg(f|_{\gamma})} = \dots = \frac{\sigma_t(f^k(\gamma))}{\deg(f^k|_{\gamma})}$$

If $f^k(\gamma) \in \partial_0(f^k_*(S^{\star}_{\gamma}))$, then $\sigma_t(f^k(\gamma))$ is a constant independent of t, thus $\sigma_t(\gamma) \leq t\rho(S_{\gamma}, \pi(\gamma))v(\pi(\gamma))$ when t is large. If $f^k(\gamma) \in \partial_0(f^k_*(S^{\star}))$, then

If $f^k(\gamma) \in \partial_1(f^k_*(S^{\star}_{\gamma}))$, then

$$\sigma_t(\gamma) = \frac{t \cdot \rho(S_{f^k(\gamma)}, \pi(f^k(\gamma))) \cdot v(\pi(f^k(\gamma)))}{\deg(f^k|_{\gamma})}.$$

By the choice of the numbers $\{\rho(S_{\gamma}^+, \gamma), \rho(S_{\gamma}^-, \gamma); \gamma \in \Gamma\}$, we see that for any curve $\beta \in \Gamma - \Gamma_1 = \bigcup_{n \geq 2} \Gamma_n$,

$$\frac{v(f(\beta))\rho(S^{\omega}_{f(\beta)}, f(\beta))}{\deg(f|_{\beta})} < v(\beta)\rho(S^{\omega}_{\beta}, \beta), \ \omega \in \{\pm\}.$$
Since for each $\gamma \in \Gamma^{\star}$, $\deg(f|_{\gamma}) = \deg(f|_{\pi(\gamma)})$, we have that

$$\sigma_t(\gamma) < \frac{t\rho(S_{f^{k-1}(\gamma)}, \pi(f^{k-1}(\gamma)))v(\pi(f^{k-1}(\gamma)))}{\deg(f^{k-1}|_{\gamma})} < \cdots$$
$$< \frac{t\rho(S_{f(\gamma)}, \pi(f(\gamma)))v(\pi(f(\gamma)))}{\deg(f|_{\gamma})} < t\rho(S_{\gamma}, \pi(\gamma))v(\pi(\gamma))$$

If $f^k(\gamma) \in \partial_2(f^k_*(S^*_{\gamma}))$, in this case, we have k = p by the choice of k and

$$\sigma_t(\gamma) = \frac{t\rho(S_{f^p(\gamma)}, \pi(f^p(\gamma)))v(\pi(f^p(\gamma)))}{\deg(f^p|_{\gamma})}.$$

With the same argument as above, we have $\sigma_t(\gamma) < t\rho(S_{\gamma}, \pi(\gamma))v(\pi(\gamma))$.

- 2. It follows from 1 and the definition of σ_t .
- 3.

$$\begin{split} &\sum_{\beta \in \Gamma^{*}} \sum_{\alpha \sim \gamma, \alpha \subset S^{*}_{\gamma}} \frac{\sigma_{t}(\beta)}{\deg(f : \alpha \to \beta)} \\ &= \sum_{\beta \in \Gamma^{*}} \sum_{\alpha \sim \gamma, \alpha \subset S^{*}_{\gamma} \setminus \gamma} \frac{\sigma_{t}(\beta)}{\deg(f : \alpha \to \beta)} + \frac{\sigma_{t}(f(\gamma))}{\deg(f|_{\gamma})} \\ &\leq \sum_{\beta \in \Gamma^{*}} \sum_{\alpha \sim \gamma, \alpha \subset S^{*}_{\gamma} \setminus \gamma} \frac{\sigma_{t}(\beta)}{\deg(f : \alpha \to \beta)} + \frac{t\rho(S_{f(\gamma)}, \pi(f(\gamma)))v(\pi(f(\gamma)))}{\deg(f|_{\gamma})} \quad (By \ 1) \\ &= \sum_{\delta \in \Gamma} \sum_{\zeta \in \pi^{-1}(\delta)} \sum_{\alpha \sim \gamma, \alpha \subset S^{*}_{\gamma} \setminus \gamma} \frac{\sigma_{t}(\zeta)}{\deg(f : \alpha \to \beta)} + \frac{t\rho(S_{f(\gamma)}, \pi(f(\gamma)))v(\pi(f(\gamma)))}{\deg(f|_{\gamma})} \\ &= \sum_{\delta \in \Gamma} \sum_{\alpha \sim \pi(\gamma), \alpha \subset S_{\gamma} \setminus \pi(\gamma)} \frac{\sum_{\zeta \in \pi^{-1}(\delta)} \sigma_{t}(\zeta)}{\deg(f : \alpha \to \delta)} + \frac{t\rho(S_{f(\gamma)}, \pi(f(\gamma)))v(\pi(f(\gamma)))}{\deg(f|_{\gamma})} \\ &\leq \sum_{\delta \in \Gamma} \sum_{\alpha \sim \pi(\gamma), \alpha \subset S_{\gamma} \setminus \pi(\gamma)} \frac{tv(\delta)}{\deg(f : \alpha \to \delta)} + \frac{t\rho(S_{f(\gamma)}, \pi(f(\gamma)))v(\pi(f(\gamma)))}{\deg(f|_{\gamma})} \quad (By \ 2) \\ &< t\rho(S_{\gamma}, \pi(\gamma))v(\pi(\gamma)) = \sigma_{t}(\gamma). \quad (By \ the \ choice \ of \ the \ number \ \rho) \\ \\ & \Box \end{split}$$

Holomorphic Models

We first decompose \mathcal{S}^* into $\mathcal{S}_0^* \sqcup \mathcal{S}_1^* \sqcup \cdots$, where

$$\mathcal{S}_0^{\star} = \{ f_*^j(S_{\nu}^{\star}); 0 \le j < p_{\nu}, 1 \le \nu \le n \},$$
$$\mathcal{S}_k^{\star} = \{ S^{\star} \in \mathcal{S}^{\star}; k \text{ is the first integer such that } f_*^k(S^{\star}) \in \mathcal{S}_0^{\star} \}, k \ge 1.$$

It's obvious that \mathcal{S}_0^{\star} consists of all f_* -periodic \mathcal{S}^{\star} -pieces.

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 140THEOREMS

Lemma 6.4.3. (Pre holomorphic models) Suppose that $(h_{\nu}^{\star}, P_{\nu}^{\star})$ is q.cequivalent to a rational map (R_{ν}, Q_{ν}) via a pair of quasiconformal maps (Φ_{ν}, Ψ_{ν}) for $1 \leq \nu \leq n$. Then for each \mathcal{S}^{\star} -piece S^{\star} , there exist a pair of quasiconformal maps $(\Phi_{S^{\star}}, \Psi_{S^{\star}}) : \mathbb{C}(S^{\star}) \to \mathbb{C}$ and a rational map $R_{S^{\star}}$ such that Φ_{S^*} is isotopic to Ψ_{S^*} rel $P(S^*)$ and the following diagram commutes:



Proof. Using the same argument as the proof of the sufficiency of Theorem 6.4.1 (see Section 6.4.2, step 1), one can show that for any $1 \leq \nu \leq n$ and any $0 \leq k < p_{\nu}$, there exist a quasiconformal map $\Phi_{f_{\pi}^{k}(S_{\pi}^{\star})}$ and a rational map $R_{f_{\star}^{k}(S_{\nu}^{\star})}$ such that the following diagram commutes

$$\overline{\mathbb{C}}(S_{\nu}^{\star}) \xrightarrow{H_{S_{\nu}^{\star}}} \overline{\mathbb{C}}(f_{*}(S_{\nu}^{\star})) \xrightarrow{H_{f_{*}(S_{\nu}^{\star})}} \cdots \longrightarrow \overline{\mathbb{C}}(f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})) \xrightarrow{\mathbb{C}} \overline{\mathbb{C}}(S_{\nu}^{\star}) \\
\Psi_{S_{\nu}^{\star}} = \Psi_{\nu} \left| \begin{array}{c} \Phi_{f_{*}(S_{\nu}^{\star})} \\
\overline{\mathbb{C}} \xrightarrow{R_{S_{\nu}^{\star}}} & \overline{\mathbb{C}} \xrightarrow{R_{f_{*}(S_{\nu}^{\star})}} & \cdots \longrightarrow \overline{\mathbb{C}} \xrightarrow{R_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})}} \\
\overline{\mathbb{C}} \xrightarrow{R_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})}} & \overline{\mathbb{C}} \xrightarrow{R_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})}} & \overline{\mathbb{C}} \end{array}\right.$$

We set $\Psi_{f_*^k(S_{\nu}^{\star})} = \Phi_{f_*^k(S_{\nu}^{\star})}$ for $0 < k < p_{\nu}$.

For each $S^* \in \mathcal{S}_1^*$, notice that $f_*(S^*) \in \mathcal{S}_0^*$, we pull back the standard complex structure of $\overline{\mathbb{C}}$ to $\overline{\mathbb{C}}(S^*)$ via $\Phi_{f_*(S^*)} \circ H_{S^*}$ and integrate it to get a quasiconformal map $\Phi_{S^*}: \overline{\mathbb{C}}(S^*) \to \overline{\mathbb{C}}$. Then $R_{S^*}:=\Phi_{f_*(S^*)} \circ H_{S^*} \circ \Phi_{S^*}^{-1}$ is a rational map. We set $\Psi_{S^*} = \Phi_{S^*}$.

By the inductive procedure, for each \mathcal{S}_n^{\star} -piece $(n = 2, 3, \cdots)$, we can get a pair of quasiconformal maps (Φ_{S^*}, Ψ_{S^*}) and a rational map R_{S^*} , as required.

Lemma 6.4.4. (Holomorphic model for periodic pieces) Fix a periodic piece $S^{\star} \in \mathcal{S}_{0}^{\star}$. Let p be the period of S^{\star} . Then for any large parameter t > 0, there exist a pair of quasiconformal maps $(\Phi^t_{S^\star}, \Psi^t_{S^\star}) : \overline{\mathbb{C}}(S^\star) \to \overline{\mathbb{C}}$ such that

1. $\Psi_{S^{\star}}^{t}$ is isotopic to $\Phi_{S^{\star}}^{t}$ rel $P(S^{\star})$. 2. $\Phi_{f_{\star}(S^{\star})}^{t} \circ f \circ (\Psi_{S^{\star}}^{t})^{-1}|_{\Psi_{S^{\star}}^{t}(E_{S^{\star}})} = R_{S^{\star}}|_{\Psi_{S^{\star}}^{t}(E_{S^{\star}})}$, where $R_{S^{\star}}$ is defined in Lemma 6.4.3.

3. The return map $f_i := R_{f_*^{i-1}(S^*)} \circ \cdots \circ R_{S^*} \circ R_{f_*^{p-1}(S^*)} \circ \cdots \circ R_{f_*^i(S^*)}$ is either a Siegel map or a Thurston map.

4. For each $i \geq 0$ and each curve $\gamma \in \partial(f^i_*(S^*))$, let β_{γ} be the unique boundary curve of $E_{f_i(S^*)}$ such that either $\gamma = \beta_{\gamma}$, or γ and β_{γ} bound an annulus in $S^* - P$. Then both $\Phi^t_{f_i(S^*)}(\gamma)$ and $\Psi^t_{f_i(S^*)}(\beta_{\gamma})$ are equipotentials in the same marked disk of f_i , with potentials

$$\varpi(\Phi_{f_*^i(S^\star)}^t(\gamma)) = \sigma_t(\gamma), \ \varpi(\Psi_{f_*^i(S^\star)}^t(\beta_\gamma)) = \frac{\sigma_t(f(\beta_\gamma))}{\deg(f|_{\beta_\gamma})}.$$

Proof. For each $\nu \in [1, n]$ and each $i \geq 0$, the critical values of $H_{f_{i}^{i}(S_{\nu}^{\star})}$ are contained in $P(f_{*}^{i+1}(S_{\nu}^{\star}))$ and $H_{f_{*}^{i}(S_{\nu}^{\star})}(P(f_{*}^{i}(S_{\nu}^{\star}))) \subset P(f_{*}^{i+1}(S_{\nu}^{\star}))$. Let $(\Phi_{f_{*}^{i}(S_{\nu}^{\star})}, \Psi_{f_{*}^{i}(S_{\nu}^{\star})}) : \overline{\mathbb{C}}(f_{*}^{i}(S_{\nu}^{\star})) \to \overline{\mathbb{C}}$ be the quasiconformal maps constructed in Lemma 6.4.3. Since $\Phi_{S_{\nu}^{\star}}$ is isotopic to $\Psi_{S_{\nu}^{\star}}$ rel $P_{\nu}^{\star} = P(S_{\nu}^{\star})$, there is a quasiconformal map $\phi_{f_{*}^{p\nu-1}(S_{\nu}^{\star})} : \overline{\mathbb{C}}(f_{*}^{p\nu-1}(S_{\nu}^{\star})) \to \overline{\mathbb{C}}$ isotopic to $\Phi_{f_{*}^{p\nu-1}(S_{\nu}^{\star})}$ rel $P(f_{*}^{p\nu-1}(S_{\nu}^{\star}))$ and $\Psi_{S_{\nu}^{\star}} \circ H_{f_{*}^{p\nu-1}(S_{\nu}^{\star})} = R_{f_{*}^{p\nu-1}(S_{\nu}^{\star})} \circ \phi_{f_{*}^{p\nu-1}(S_{\nu}^{\star})}$. Inductively, there is a sequence of quasiconformal maps $\phi_{f_{*}^{i}(S_{\nu}^{\star})} : \overline{\mathbb{C}}(f_{*}^{i}(S_{\nu}^{\star})) \to \overline{\mathbb{C}}$ for $i = p_{\nu} - 2, \cdots, 1$, such that $\phi_{f_{*}^{i}(S_{\nu}^{\star})}$ is isotopic to $\Phi_{f_{*}^{i}(S_{\nu}^{\star})}$ rel $P(f_{*}^{i}(S_{\nu}^{\star}))$ and the following diagram commutes:

$$\overline{\mathbb{C}}(f_{*}(S_{\nu}^{\star})) \xrightarrow{H_{f_{*}(S_{\nu}^{\star})}} \overline{\mathbb{C}}(f_{*}^{2}(S_{\nu}^{\star})) \xrightarrow{H_{f_{*}^{2}(S_{\nu}^{\star})}} \cdots \longrightarrow \overline{\mathbb{C}}(f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})) \xrightarrow{H_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})}} \overline{\mathbb{C}}(S_{\nu}^{\star}) \xrightarrow{\phi_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})}} \sqrt{\Phi_{S_{\nu}^{\star}}} \xrightarrow{\Phi_{f_{*}^{\star}(S_{\nu}^{\star})}} \sqrt{\Phi_{S_{\nu}^{\star}}} \xrightarrow{\Phi_{f_{*}^{*}(S_{\nu}^{\star})}} \overline{\mathbb{C}}(S_{\nu}^{\star}) \xrightarrow{\Phi_{f_{*}^{p_{\nu}-1}(S_{\nu}^{\star})}} \sqrt{\Phi_{S_{\nu}^{\star}}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \longrightarrow \overline{\mathbb{C}}(F_{F_{\nu}^{p_{\nu}-1}(S_{\nu}^{\star})}) \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \xrightarrow{\Phi_{F_{\nu}^{*}(S_{\nu}^{\star})}} \cdots \xrightarrow{\Phi_{F_{\nu}^{$$

This diagram together with the diagram in Lemma 6.4.3 implies that for any $1 \leq i < p_{\nu}$, the map $H_{f_*^{i-1}(S_{\nu}^{\star})} \circ \cdots \circ H_{S_{\nu}^{\star}} \circ H_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} \circ \cdots \circ H_{f_*^i(S_{\nu}^{\star})}$ is q.c-equivalent to $f_i = R_{f_*^{i-1}(S_{\nu}^{\star})} \circ \cdots \circ R_{S_{\nu}^{\star}} \circ R_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} \circ \cdots \circ R_{f_*^i(S_{\nu}^{\star})}$ via $(\Phi_{f_*^i(S_{\nu}^{\star})}, \phi_{f_*^i(S_{\nu}^{\star})})$. Notice that $f_i(\phi_{f_*^i(S_{\nu}^{\star})}(P(f_*^i(S_{\nu}^{\star})))) \subset \phi_{f_*^i(S_{\nu}^{\star})}(P(f_*^i(S_{\nu}^{\star})))$, so f_i is either a Siegel map or a Thurston map.

The relation $f_{i+1} \circ R_{f_*^i(S_\nu^\star)} = R_{f_*^i(S_\nu^\star)} \circ f_i$ with $f_{p_\nu} = R_\nu$ (here, R_ν is the rational map defined in Lemma 6.4.3) means that $R_{f_*^i(S_\nu^\star)}$ is a semi-conjugacy between f_{i+1} and f_i , so their Julia sets satisfy $J(f_i) = R_{f_*^i(S_\nu^\star)}^{-1}(J(f_{i+1}))$. One can check that $R_{f_*^i(S_\nu^\star)}$ maps the marked disks of f_i onto the marked disks of f_{i+1} , and maps the equipotentials of f_i to the equipotentials of f_{i+1} .

In the following, we will construct a pair of quasiconformal maps $(\Phi_{S^{\star}}^{t}, \Psi_{S^{\star}}^{t}) : \overline{\mathbb{C}}(S^{\star}) \to \overline{\mathbb{C}}$ that satisfy the required properties.

Step 1: Construction of $\Phi_{S_{\nu}^{\star}}^{t}$ and $\Phi_{f_{\nu}^{p\nu-1}(S_{\nu}^{\star})}^{t}$. We first modify $\Phi_{S_{\nu}^{\star}}$ to a new quasiconformal map $\Phi_{S_{\nu}^{\star}}^{t} : \overline{\mathbb{C}}(S_{\nu}^{\star}) \to \overline{\mathbb{C}}$ such that $\Phi_{S_{\nu}^{\star}}^{t}$ is isotopic to $\Phi_{S_{\nu}^{\star}}$ rel $P(S_{\nu}^{\star})$, and for each curve $\gamma \in \partial(S_{\nu}^{\star})$, the curve $\Phi_{S_{\nu}^{\star}}^{t}(\gamma)$ is the equipotential in a marked disk of $f_{p_{\nu}} = R_{\nu}$ with potential $\varpi(\Phi_{S_{\nu}^{\star}}^{t}(\gamma)) = \sigma_{t}(\gamma)$.

Then, we lift $\Phi_{S_{\nu}^{\star}}^{t}$ via $R_{f_{\ast}^{p_{\nu}-1}(S_{\nu}^{\star})}$ and $H_{f_{\ast}^{p_{\nu}-1}(S_{\nu}^{\star})}$ and get a quasiconformal map $\widehat{\Phi}_{f_{\ast}^{p_{\nu}-1}(S_{\nu}^{\star})}^{t}$ isotopic to $\Phi_{f_{\ast}^{p_{\nu}-1}(S_{\nu}^{\star})}$ rel $P(f_{\ast}^{p_{\nu}-1}(S_{\nu}^{\star}))$. See the following commutative diagram:



Now, we modify $H_{f_*^{p\nu-1}(S_{\nu}^{\star})}$ to another marked disk extension of $f|_{E_{f_*^{p\nu-1}(S_{\nu}^{\star})}}$, say $\widehat{H}_{f_*^{p\nu-1}(S_{\nu}^{\star})}$, such that for each curve $\gamma \in \partial_1(f_*^{p\nu-1}(S_{\nu}^{\star}))$, the curve $\Phi_{S_{\nu}^{t}}^{t}(\widehat{H}_{f_*^{p\nu-1}(S_{\nu}^{\star})}(\gamma))$ is an equipotential in some marked disk of $f_{p\nu} = R_{\nu}$. Since $\gamma \in \partial_1(f_*^{p\nu-1}(S_{\nu}^{\star}))$, the potential of $\Phi_{S_{\nu}^{t}}^{t}(\widehat{H}_{f_*^{p\nu-1}(S_{\nu}^{\star})}(\gamma))$ should be larger than $\varpi(\Phi_{S_{\nu}^{t}}^{t}(f(\beta_{\gamma}))) = \sigma_t(f(\beta_{\gamma}))$. It follows from Lemma 6.4.2 that $\deg(f|_{\beta_{\gamma}})\sigma_t(\gamma) > \sigma_t(f(\beta_{\gamma}))$ when t is large. So we designate $\varpi(\Phi_{S_{\nu}^{t}}^{t}(\widehat{H}_{f_*^{p\nu-1}(S_{\nu}^{t})}(\gamma)))$ to be $\deg(f|_{\beta_{\gamma}}) \cdot \sigma_t(\gamma)$.

Since both $H_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}$ and $\widehat{H}_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}$ are marked disk extensions of $f|_{E_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}}$, there is a quasiconformal map $\xi_{p_{\nu}-1}$: $\overline{\mathbb{C}}(f_*^{p_{\nu}-1}(S_{\nu}^{\star})) \to \overline{\mathbb{C}}(f_*^{p_{\nu}-1}(S_{\nu}^{\star}))$ isotopic to the identity map rel $E_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} \cup P(f_*^{p_{\nu}-1}(S_{\nu}^{\star}))$ such that $\widehat{H}_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} = H_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} \circ \xi_{p_{\nu}-1}.$

We set $\Phi_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}^t = \widehat{\Phi}_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}^t \circ \xi_{p_{\nu}-1}$. It's obvious that $\Phi_{S_{\nu}^{\star}}^t \circ \widehat{H}_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} = R_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})} \circ \Phi_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}^t$.

Step 2: Construction of $\Phi_{f_*(S_{\nu}^{\star})}^t$ for $i = p_{\nu} - 2, \cdots, 1$ and $\Psi_{S_{\nu}^{\star}}^t$. By the same argument as in Step 1, we can lift $\Phi_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}^t$ via $R_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}$ and $H_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}$ and $H_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}$ isotopic to $\Phi_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}$ rel $P(f_*^{p_{\nu}-2}(S_{\nu}^{\star}))$. Then we modify $H_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}$ to another marked disk extension of $f|_{E_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}}$, say $\widehat{H}_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})} = H_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})} \circ \xi_{p_{\nu}-2}$, where $\xi_{p_{\nu}-2} : \overline{\mathbb{C}}(f_*^{p_{\nu}-2}(S_{\nu}^{\star})) \to \overline{\mathbb{C}}(f_*^{p_{\nu}-2}(S_{\nu}^{\star}))$ is a quasiconformal map isotopic to the identity map rel $E_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})} \cup P(f_*^{p_{\nu}-2}(S_{\nu}^{\star}))$, such that for each $\gamma \in \partial_1(f_*^{p_{\nu}-2}(S_{\nu}^{\star}))$, the curve $\Phi_{f_*^{p_{\nu}-1}(S_{\nu}^{\star})}^t(\widehat{H}_{f_*^{p_{\nu}-2}(S_{\nu}^{\star})}(\gamma))$ is an equipotential of $f_{p_{\nu}-1}$ with potential

$$\varpi(\Phi^t_{f^{p\nu-1}_*(S^{\star}_{\nu})}(\widehat{H}_{f^{p\nu-2}_*(S^{\star}_{\nu})}(\gamma))) = \deg(f|_{\beta_{\gamma}})\sigma_t(\gamma).$$

We set $\Phi_{f_*^{p_{\nu-2}}(S_{\nu}^{\star})}^t = \widehat{\Phi}_{f_*^{p_{\nu-2}}(S_{\nu}^{\star})}^t \circ \xi_{p_{\nu-2}}.$

Inductively, we can get a sequence of new marked disk extensions $\widehat{H}_{f^i_*(S^{\star}_{\nu})}, i = p_{\nu} - 1, \cdots, 0$, and a sequence of quasiconformal maps $\Phi^t_{f^i_*(S^{\star}_{\nu})}, i = p_{\nu} - 1, \cdots, 1, \ \Psi^t_{S^{\star}_{\nu}}$ such that the following diagram commutes



Moreover, for each $i \in [0, p_{\nu} - 1]$ and each curve $\gamma \in \partial_1(f^i_*(S^*_{\nu}))$, we require

$$\varpi(\Phi_{f_*^{i+1}(S_{\nu}^{\star})}^t(\widehat{H}_{f_*^i(S_{\nu}^{\star})}(\gamma))) = \deg(f|_{\beta_{\gamma}})\sigma_t(\gamma).$$

Finally, we set $\Psi_{f_*^i(S_{\nu}^{\star})}^t = \Phi_{f_*^i(S_{\nu}^{\star})}^t$ for $1 \le i \le p_{\nu} - 1$.

Step 3: Prescribed potentials. In this step, we will show that for each $0 \le i \le p_{\nu} - 1$ and each curve $\gamma \in \partial(f_*^i(S_{\nu}^*))$,

$$\varpi(\Phi_{f_*^i(S_\nu^\star)}^t(\gamma)) = \sigma_t(\gamma), \ \varpi(\Psi_{f_*^i(S_\nu^\star)}^t(\beta_\gamma)) = \frac{\sigma_t(f(\beta_\gamma))}{\deg(f|_{\beta_\gamma})}.$$
(6.1)

Notice that for each curve $\gamma \in \partial(S^*_{\nu}) \cup \bigcup_{0 < i < p_{\nu}} \partial_0(f^i_*(S^*_{\nu}))$, the first equation of (6.1) holds by the evaluation of ϖ .

If $\gamma \in \partial_1(f_*^i(S_{\nu}^*))$ for some $0 < i < p_{\nu}$, then by construction, $\Phi_{f_*^{i+1}(S_{\nu}^*)}^t(\widehat{H}_{f_*^i(S_{\nu}^*)}(\gamma))$ is an equipotential in a marked disk (Δ_{i+1}, a) of f_{i+1} . Since $\Phi_{f_*^{i+1}(S_{\nu}^*)}^t \circ \widehat{H}_{f_*^i(S_{\nu}^*)}(\gamma) = R_{f_*^i(S_{\nu}^*)} \circ \Phi_{f_*^i(S_{\nu}^*)}^t(\gamma)$, we conclude that $\Phi_{f_*^i(S_{\nu}^*)}^t(\gamma)$ is also an equipotential of some marked disk of f_i , denoted by (Δ_i, b) . Then $R_{f_*^i(S_{\nu}^*)} : \Delta_i - \{b\} \to \Delta_{i+1} - \{a\}$ is a covering map of degree $\deg(f|_{\beta_{\gamma}})$. The potential of $\Phi_{f_*^i(S_{\nu}^*)}^t(\gamma)$ satisfies (Here, we use $A(\alpha, \beta)$ to denote the annulus bounded by α and β)

$$\varpi(\Phi_{f_*^i(S_{\nu}^{\star})}^t(\gamma)) = \operatorname{mod}\left(A(\partial\Delta_i, \Phi_{f_*^i(S_{\nu}^{\star})}^t(\gamma))\right) \\
= \operatorname{mod}\left(A(\partial\Delta_{i+1}, \Phi_{f_*^{i+1}(S_{\nu}^{\star})}^t(\widehat{H}_{f_*^i(S_{\nu}^{\star})}(\gamma)))\right) / \operatorname{deg}(f|_{\beta_{\gamma}}) \\
= \varpi(\Phi_{f_*^{i+1}(S_{\nu}^{\star})}^t(\widehat{H}_{f_*^i(S_{\nu}^{\star})}(\gamma))) / \operatorname{deg}(f|_{\beta_{\gamma}}) \\
= \sigma_t(\gamma).$$

Now we consider $\gamma \in \partial_2(f_*^i(S_\nu^*))$ for some $0 < i < p_\nu$. In this case, by the same argument as above, we can see that

$$\varpi(\Phi_{f_*^i(S_\nu^\star)}^t(\gamma)) = \frac{\varpi(\Phi_{f_*^{i+1}(S_\nu^\star)}^t(f(\gamma)))}{\deg(f|_{\gamma})}.$$

By the definition of σ_t , for $\gamma \in \partial_2(f_*^i(S_{\nu}^{\star}))$, we have

$$\sigma_t(\gamma) = \frac{\sigma_t(f(\gamma))}{\deg(f|_{\gamma})}.$$

Based on this observation, we conclude by induction that $\varpi(\Phi_{f_{i}^{i}(S^{\star})}^{t}(\gamma)) =$ $\sigma_t(\gamma).$

To finish, we show that the second equation of (6.1) holds. Since for each $i \in [0, p_{\nu} - 1]$ and each curve $\gamma \in \partial(f^i_*(S^{\star}_{\nu}))$, the curve $\Phi^t_{f^{i+1}_*(S^{\star}_{\nu})}(f(\beta_{\gamma}))$ is an equipotential, it follows from the relation

$$\Phi^t_{f^{i+1}_*(S^{\star}_{\nu})} \circ f \circ (\Psi^t_{f^{i}_*(S^{\star}_{\nu})})^{-1}|_{\Psi^t_{f^{i}_*(S^{\star}_{\nu})}(E_{f^{i}_*(S^{\star}_{\nu})})} = R_{f^{i}_*(S^{\star}_{\nu})}|_{\Psi^t_{f^{i}_*(S^{\star}_{\nu})}(E_{f^{i}_*(S^{\star}_{\nu})})}$$

that $\Psi_{f^i(S^*)}^t(\beta_{\gamma})$ is also an equipotential. Using a similar argument as above, we obtain

$$\varpi(\Psi_{f_*^i(S_\nu^\star)}^t(\beta_\gamma)) = \frac{\varpi(\Phi_{f_*^{i+1}(S_\nu^\star)}^t(f(\beta_\gamma)))}{\deg(f|_{\beta_\gamma})} = \frac{\sigma_t(f(\beta_\gamma))}{\deg(f|_{\beta_\gamma})}.$$

The proof is completed.

Now, we deal with the strictly pre-periodic \mathcal{S}^{\star} -pieces. Let $S^{\star} \in \mathcal{S}_{k}^{\star}$ for some $k \geq 1$. Then $f_*^k(S^*)$ is a f_* -periodic \mathcal{S}^* -piece. Notice that for $0 \leq i < k, H_{f_*^i(S^\star)}(P(f_*^i(S^\star))) \subset P(f_*^{i+1}(S^\star))$ and each critical value of $H_{f^i_*(S^\star)}$ is contained in $P(f^{i+1}_*(S^\star))$, we have that $R_{f^i_*(S^\star)} \circ \Phi_{f^i_*(S^\star)}(P(f^i_*(S^\star))) \subset \mathbb{C}$ $\Phi_{f^{i+1}_*(S^\star)}(P(f^{i+1}_*(S^\star)))$ and every critical value of $R_{f^{k-1}_*(S^\star)} \circ \cdots \circ R_{S^\star}$ is contained in $\Phi_{f_*^k(S^\star)}(P(f_*^k(S^\star))) = \Phi_{f_*^k(S^\star)}^t(P(f_*^k(S^\star)))$, here $R_{f_*^i(S^\star)}$ and $\Phi_{f_*^i(S^\star)}$ are defined in Lemma 6.4.3. For any marked point $a \in P(S^*) \cap (\overline{\mathbb{C}}(S^*) - S^*)$, the point $R_{f_*^{k-1}(S^*)} \circ \cdots \circ R_{S^*}(\Phi_{S^*}(a))$ is the center of some marked disk (Δ, q) of some f_j , where f_j is a return map defined in Lemma 6.4.4. The component $\Delta_{\Phi_{S^{\star}}(a)}$ of $(R_{f^{k-1}_{\star}(S^{\star})} \circ \cdots \circ R_{S^{\star}})^{-1}(\Delta)$ that contains $\Phi_{S^{\star}}(a)$ is also a disk. We call $(\Delta_{\Phi_{S^{\star}}(a)}, \Phi_{S^{\star}}(a))$ a marked disk of $R_{f^{k-1}_{*}(S^{\star})} \circ \cdots \circ R_{S^{\star}}$.

By the same argument as in the proof of Lemma 6.4.4, we can show that

Lemma 6.4.5. For any $k \geq 1$, any $S^{\star} \in \mathcal{S}_k^{\star}$ and any large parameter t > 0, there exist a pair of quasiconformal maps $\Phi_{S^{\star}}^{t} = \Psi_{S^{\star}}^{t} : \overline{\mathbb{C}}(S^{\star}) \to \overline{\mathbb{C}}$ such that 1. $\Phi_{f_{\star}(S^{\star})}^{t} \circ f \circ (\Psi_{S^{\star}}^{t})^{-1}|_{\Psi_{S^{\star}}^{t}(E_{S^{\star}})} = R_{S^{\star}}|_{\Psi_{S^{\star}}^{t}(E_{S^{\star}})}$, where $R_{S^{\star}}$ is defined in

Lemma 6.4.3.

2. For each curve $\gamma \in \partial(S^*)$, let β_{γ} be the unique curve in $\partial(E_{S^*})$ homotopic to γ in $\mathbb{C} - P$. Then both $\Phi^t_{S^*}(\gamma)$ and $\Phi^t_{S^*}(\beta_{\gamma})$ are equipotentials in the same marked disk of $R_{f_*^{k-1}(S^\star)} \circ \cdots \circ R_{S^\star}$, with potentials

$$\varpi(\Phi_{S^{\star}}^{t}(\gamma)) = \sigma_{t}(\gamma), \ \varpi(\Phi_{S^{\star}}^{t}(\beta_{\gamma})) = \frac{\sigma_{t}(f(\beta_{\gamma}))}{\deg(f|_{\beta_{\gamma}})}.$$

_	-	
	н	
	н	
	л.	
_	-	

We decompose \mathcal{E}^{\star} into $\mathcal{E}_{ess}^{\star} \sqcup \mathcal{E}_{A}^{\star} \sqcup \mathcal{E}_{D}^{\star}$, where

• $\mathcal{E}_{ess}^{\star} = \{ E_{S^{\star}}; S^{\star} \in \mathcal{S}^{\star} \}$, it consists of all \mathcal{E}^{\star} -pieces parallel to some \mathcal{S}^{\star} -piece;

• \mathcal{E}_A^* is the collection of all \mathcal{E}^* -pieces E^* contained essentially in an annular component of $S^* - E_{S^*}$ for some \mathcal{S}^* -piece S^* (Here 'essential' means at least one boundary curve of E^* is non-peripheral in $\overline{\mathbb{C}} - P$);

• $\mathcal{E}_D^{\star} = \mathcal{E}^{\star} - (\mathcal{E}_{ess}^{\star} \sqcup \mathcal{E}_A^{\star})$. One may verify that each \mathcal{E}_D^{\star} -piece is contained in a disk component of $S^{\star} - E_{S^{\star}}$ for some \mathcal{S}^{\star} -piece S^{\star} .



Figure 6.5: Different types of \mathcal{E}^* -pieces: Here, S^* is a \mathcal{S}^* -piece with boundary $\partial S^* = \gamma_1 \cup \gamma_2 \cup \gamma_3$. E_0^* is a \mathcal{E}_{ess}^* -piece. E_5^* and E_6^* are \mathcal{E}_A^* -pieces. E_1^*, E_2^*, E_3^* and E_4^* are \mathcal{E}_D^* -pieces.

In the following, for every \mathcal{E}_A^* -piece E^* , we will construct a holomorphic model for $f|_{E^*}$. Given an \mathcal{E}_A^* -piece E^* , first notice that E^* has no intersection with the marked set P. As we did before, we also associate a Riemann sphere $\overline{\mathbb{C}}(E^*)$ for E^* . We mark a point in each component of $\overline{\mathbb{C}}(E^*) - E^*$, and let $P(E^*)$ be the collection of all these marked points. We can get a marked disk extension of $f|_{E^*}$, say $H_{E^*}: \overline{\mathbb{C}}(E^*) \to \overline{\mathbb{C}}(f(E^*))$, such that $H_{E^*}|_{E^*} = f|_{E^*}$, $H_{E^*}(P(E^*)) \subset P(f(E^*))$ and all critical values (if any) of H_{E^*} are contained in $P(f(E^*))$. Let $\Phi_{E^*}^t: \overline{\mathbb{C}}(E^*) \to \overline{\mathbb{C}}$ be a quasiconformal map such that $R_{E^*} :=$ $\Phi_{f(E^*)}^t \circ H_{E^*} \circ (\Phi_{E^*}^t)^{-1}$ is holomorphic. We remark that if we change $\Phi_{f(E^*)}^t$ to another quasiconformal map $\Phi_{f(E^*)}^{t_1}$ isotopic to $\Phi_{E^*}^t$ rel $P(f(E^*))$, such that $R_{E^*} = \Phi_{f(E^*)}^{t_1} \circ H_{E^*} \circ (\Phi_{E^*}^{t_1})^{-1}$. This means that once we get the holomorphic map R_{E^*} , we can always assume that it is independent of the parameter t. We set $\Psi_{E^*}^t = \Phi_{E^*}^t$.

Notice that the \mathcal{E}^{\star}_{A} -piece E^{\star} has exactly two boundary curves α and

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 146 THEOREMS

 β which are non-peripheral and homotopic to each other in $\overline{\mathbb{C}} - P$. By the choice of $\Phi_{S^*}^t$ for $S^* \in \mathcal{S}^*$, both $\Phi_{f(E^*)}^t(f(\alpha))$ and $\Phi_{f(E^*)}^t(f(\beta))$ are equipotentials in the marked disks of some f_j (defined in Lemma 6.4.4) or some $R_{f_*^{k-1}(S^*)} \circ \cdots \circ R_{S^*}$. We denote the marked disk that contains $\Phi_{f(E^*)}^t(f(\alpha))$ (resp. $\Phi_{f(E^*)}^t(f(\beta))$) by (Δ_a, a) (resp. (Δ_b, b)). It can happen that $(\Delta_a, a) = (\Delta_b, b)$. Let Δ_α (resp. Δ_β) be the component of $R_{E^*}^{-1}(\Delta_a)$ (resp. $R_{E^*}^{-1}(\Delta_b)$) that contains $\Phi_{E^*}^t(\alpha)$ (resp. $\Phi_{E^*}^t(\beta)$). Then Δ_α (resp. Δ_β) contains a marked point in $P(E^*)$, say z_α (resp. z_β). The marked disks $(\Delta_\alpha, z_\alpha)$ and (Δ_β, z_β) are called the marked disks of R_{E^*} . They are independent of the choice of t. Clearly, $\Phi_{E^*}^t(\alpha)$ is an equipotential in the marked disk $(\Delta_\alpha, z_\alpha)$

$$\varpi(\Phi_{E^{\star}}^{t}(\alpha)) = \frac{\varpi(\Phi_{f(E^{\star})}^{t}(f(\alpha)))}{\deg(f|_{\alpha})} = \frac{\sigma_{t}(f(\alpha))}{\deg(f|_{\alpha})} = \frac{\sigma_{t}(f(\alpha))}{\deg(f|_{\alpha})} = \frac{\sigma_{t}(f(\alpha))}{\deg(f|_{\beta})} = \frac{\sigma_{t}(f(\beta))}{\deg(f|_{\beta})} = \frac{\sigma_{t}(f(\beta))}{\deg(f|$$

We denote by $A(E^*) \subset \overline{\mathbb{C}}(E^*)$ the annulus bounded by α and β . By Reversed Grötzsch Inequality (See Theorem 2.2.3, also Lemma B.1 in [CT1]), there is a constant $C(E^*)$, independent of the parameter t, such that

$$\operatorname{mod}(\Phi_{E^{\star}}^{t}(A(E^{\star}))) \leq \frac{\sigma_{t}(f(\alpha))}{\operatorname{deg}(f|_{\alpha})} + \frac{\sigma_{t}(f(\beta))}{\operatorname{deg}(f|_{\beta})} + C(E^{\star}).$$



Figure 6.6: A \mathcal{S}^* -piece S^* with boundary $\partial S^* = \gamma \cup \gamma_1 \cup \gamma_2 \cup \gamma_3$. Here, E_0^* is the \mathcal{E}_{ess}^* -piece parallel to S^* , E_1^* and E_2^* are two \mathcal{E}_A^* -pieces between γ and β_{γ} .

For any \mathcal{S}^* -piece S^* and any $\gamma \in \partial_1(S^*)$, let $A_{S^*}^{\gamma}$ be the annulus bounded by γ and β_{γ} . By the construction of $\Phi_{S^*}^t, \Psi_{S^*}^t : \overline{\mathbb{C}}(S^*) \to \overline{\mathbb{C}}$, both $\Phi_{S^*}^t(\gamma)$ and $\Psi_{S^{\star}}^{t}(\beta_{\gamma})$ are equipotentials. We denote the annulus between $\Phi_{S^{\star}}^{t}(\gamma)$ and $\Psi_{S^{\star}}^{t}(\beta_{\gamma})$ by $A^{t}(S^{\star},\gamma)$. It's obvious that

$$\operatorname{mod}(A^t(S^*,\gamma)) = \varpi(\Phi^t_{S^*}(\gamma)) - \varpi(\Psi^t_{S^*}(\beta_{\gamma})) = \sigma_t(\gamma) - \frac{\sigma_t(f(\beta_{\gamma}))}{\operatorname{deg}(f|_{\beta_{\gamma}})}.$$

Then we have the following

Lemma 6.4.6. (Large parameter implies Grötzsch inequality) When t is large enough, for any S^* -piece S^* and any $\gamma \in \partial_1(S^*)$, we have

$$\sum_{\substack{\mathcal{E}_{A}^{\star}\ni E^{\star}\subset\overline{A_{S^{\star}}^{\gamma}}}} \operatorname{mod}(\Psi_{E^{\star}}^{t}(A(E^{\star}))) < \operatorname{mod}(A^{t}(S^{\star},\gamma)),$$

where the summation is taken over all the \mathcal{E}_A^{\star} -pieces contained in $\overline{A_{S^{\star}}^{\gamma}}$.

Proof. It suffices to show that when t is large enough,

$$\sum_{\mathcal{E}_A^{\star} \ni \widehat{E} \subset \overline{A_{S^{\star}}^{\gamma}}} \left(\frac{\sigma_t(f(\alpha_{E^{\star}}))}{\deg(f|_{\alpha_{E^{\star}}})} + \frac{\sigma_t(f(\beta_{E^{\star}}))}{\deg(f|_{\beta_{E^{\star}}})} + C(E^{\star}) \right) + \frac{\sigma_t(f(\beta_{\gamma}))}{\deg(f|_{\beta_{\gamma}})} < t \cdot \rho(S_{\gamma}, \pi(\gamma)) \cdot v(\pi(\gamma)),$$

where α_{E^*} and β_{E^*} are the boundary curves of E^* , homotopic to γ in $\overline{\mathbb{C}} - P$. One can verify that

$$\sum_{\mathcal{E}_A^\star \ni E^\star \subset \overline{A_{S^\star}^\gamma}} \left(\frac{\sigma_t(f(\alpha_{E^\star}))}{\deg(f|_{\alpha_{E^\star}})} + \frac{\sigma_t(f(\beta_{E^\star}))}{\deg(f|_{\beta_{E^\star}})} \right) + \frac{\sigma_t(f(\beta_\gamma))}{\deg(f|_{\beta_\gamma})} = \sum_{\beta \in \Sigma^\star} \sum_{\alpha \sim \gamma, \alpha \subset S_\gamma^\star} \frac{\sigma_t(\beta)}{\deg(f:\alpha \to \beta)}$$

Since $\Sigma^{\star} = \Gamma_0^{\star} \cup \Gamma^{\star}$, we can decompose the summation into two parts:

$$I = \sum_{\beta \in \Gamma^{\star}} \sum_{\alpha \sim \gamma, \alpha \subset S^{\star}_{\gamma}} \frac{\sigma_t(\beta)}{\deg(f : \alpha \to \beta)}, \ II = \sum_{\beta \in \Gamma^{\star}_0} \sum_{\alpha \sim \gamma, \alpha \subset S^{\star}_{\gamma}} \frac{\sigma_t(\beta)}{\deg(f : \alpha \to \beta)}$$

It follows from the proof of Lemma 6.4.2 that $I \leq t\omega(\gamma)$, where

$$\omega(\gamma) := \frac{\rho(S_{f(\gamma)}, \pi(f(\gamma)))v(\pi(f(\gamma)))}{\deg(f|_{\gamma})} + \sum_{\delta \in \Gamma} \sum_{\pi(\gamma) \sim \alpha \subset S_{\gamma} \setminus \pi(\gamma)} \frac{v(\delta)}{\deg(f : \alpha \to \delta)},$$

if $f(\gamma) \in \Gamma^*$ (or equivalently $\gamma \in \Gamma_2^* \cup \Gamma_3^* \cup \cdots$); and

$$\omega(\gamma) := \sum_{\delta \in \Gamma} \sum_{\pi(\gamma) \sim \alpha \subset S_{\gamma} \setminus \pi(\gamma)} \frac{v(\delta)}{\deg(f : \alpha \to \delta)},$$

if $f(\gamma) \in \Gamma_0^*$ (or equivalently $\gamma \in \Gamma_1^*$). For the second term, we have

$$II \leq \sum_{A \in \mathcal{A}} \sum_{\alpha \in f^{-1}(\Gamma_0) \setminus \Gamma_0} \frac{\operatorname{mod}(A)}{\operatorname{deg}(f|_{\alpha})},$$

where \mathcal{A} is the collection of all rotation annuli of (f, P), Γ_0 is the collection of (f, P)-invariant curves defined in Section 6.2.

So if we choose t large enough such that for any $\gamma \in \bigcup_{S^{\star} \in \mathcal{S}^{\star}} \partial_1(S^{\star})$,

$$\sum_{E^{\star} \in \mathcal{E}_{A}^{\star}} C(E^{\star}) + \sum_{A \in \mathcal{A}} \sum_{\alpha \in f^{-1}(\Gamma_{0}) \setminus \Gamma_{0}} \frac{\operatorname{mod}(A)}{\operatorname{deg}(f|_{\alpha})} < t \Big(\rho(S_{\gamma}, \pi(\gamma)) \cdot v(\pi(\gamma)) - \omega(\gamma) \Big),$$

then the conclusion follows (notice that by the choice of the number ρ , we have $\rho(S_{\gamma}, \pi(\gamma)) \cdot v(\pi(\gamma)) - \omega(\gamma) > 0$ for all $\gamma \in \Gamma^{\star}$).

Now, we are ready to complete the proof of Theorem 6.4.1. Here is a fact used in the proof, which is equivalent to the Grötzsch inequality, and we will state it as follows. Let $A, B \subset \overline{\mathbb{C}}$ be two annuli. We say that B can be embedded into A essentially and holomorphically if there is a holomorphic injection $\phi: B \to A$ such that $\phi(B)$ separates the two boundary components of A.

Fact Let $A, A_1, \dots, A_n \subset \overline{\mathbb{C}}$ be annuli, then A_1, \dots, A_n can be embedded into A essentially and holomorphically such that the closures of the images of A_i 's are mutually disjoint if and only if

$$\sum_{i=1}^{n} \operatorname{mod}(A_i) < \operatorname{mod}(A).$$

Proof of the sufficiency of Theorem 6.4.1, assuming $\Gamma \neq \emptyset$

The idea of the proof is to glue the holomorphic models together along the stable multicurve Γ . Here is the detailed proof:

Recall that for each \mathcal{S}^* -piece S^* , we use S to denote the \mathcal{S} -piece that contains S^* . For each curve $\gamma \in \Sigma$, A^{γ} is the annular neighborhood of γ such that A^{γ} avoids $P \cup f(P - \cup \mathcal{A})$ and $f(A^{\gamma}) = A^{f(\gamma)}$, where \mathcal{A} is the collection of all rotation annuli of (f, P).

For each \mathcal{S}^* -piece S^* , we extend $\Phi_{S^*}^t : S^* \to \Phi_{S^*}^t(S^*)$ to a quasiconformal homeomorphism $\Phi_S : S \to \Phi_S(S)$ such that Φ_S is holomorphic in $(S - S^*) \cap (\cup \mathcal{A})$.

We first choose t large enough such that Lemma 6.4.6 holds. In this case, one can embedded $\Psi_{E^{\star}}^{t}(E^{\star})$ holomorphically into the interior of $\Phi_{S}(S)$ for each \mathcal{E}_{A}^{\star} -piece E^{\star} contained in S according to the original order of their non-peripheral boundary curves so that the embedded images are mutually disjoint. In other words, there is a quasiconformal homeomorphism $\psi_S: S \to \Phi_S(S)$ such that

• $\psi_S|_{\partial S} = \Phi_S|_{\partial S}$ and ψ_S is isotopic to Φ_S rel $\partial S \cup (S \cap P)$. Moreover $\psi_S|_{S \cap (\cup A)} = \Phi_S|_{S \cap (\cup A)}$.

- $\psi_S|_{E_{S^\star}} = \Psi_{S^\star}^t|_{E_{S^\star}}$, where E_{S^\star} is the unique \mathcal{E}^\star -piece parallel to S^\star .
- For each curve $\gamma \in \partial_1(S)$, $\Phi_S(S \cap A^{\gamma}) = \psi_S(S \cap A^{\gamma})$.

• For every \mathcal{E}^{\star}_{A} -piece E^{\star} with $E^{\star} \subset S$, the map $\Psi^{t}_{E^{\star}} \circ \psi^{-1}_{S}$ is holomorphic in $\psi_{S}(E^{\star})$.

We define a subset \mathcal{E}_A of \mathcal{E} by $\mathcal{E}_A = \{E; E^* \in \mathcal{E}_A^*\}$. Let $\mathcal{D}(S)$ be the collection of all disk components of $S - E_S \cup (\cup_{\mathcal{E}_A \ni E \subset S} E)$, here E_S is the unique \mathcal{E} -piece parallel to S. For each $D \in \mathcal{D}(S)$, we construct a quasiconformal homeomorphism $\zeta_D : D \to \psi_S(D)$, whose Beltrami coefficient satisfies

$$\mu_{\zeta_D}(z) = \sum_{\mathcal{E} \ni E \subset D} \chi_E(z) \mu_{\Phi_{f(E)} \circ f}(z),$$

here the summation is taken over all \mathcal{E} -pieces contained in D. We further require $\zeta_D(p) = \psi_S(p)$ if D contains a marked point $p \in P$.

Let Γ_S be the collection of all boundary curves of $\bigcup_{D \in \mathcal{D}(S)} D$. For each $\gamma \in \Gamma_S$, notice that $f(\gamma) \in \Sigma$. Let A^{γ} be the component of $f^{-1}(A^{f(\gamma)})$ that contains γ . It's obvious that A^{γ} is an annular neighborhood of γ . We define a quasiconformal homeomorphism $\Psi_S : S \to \Phi_S(S)$ by

$$\Psi_{S}(z) = \begin{cases} \zeta_{D}(z), & z \in D, D \in \mathcal{D}(S), \\ \psi_{S}(z), & z \in S - (\bigcup_{D \in \mathcal{D}(S)} D) \cup (\bigcup_{\gamma \in \Gamma_{S}} A^{\gamma}), \\ q.c \ interpolation, & z \in \bigcup_{\gamma \in \Gamma_{S}} A^{\gamma} - \bigcup_{D \in \mathcal{D}(S)} D. \end{cases}$$

The map Ψ_S satisfies:

• $\Psi_S|_{\partial S} = \Phi_S|_{\partial S}$ and Ψ_S is isotopic to Φ_S rel $\partial S \cup (S \cap P)$. Moreover $\Psi_S|_{S \cap (\cup A)} = \Phi_S|_{S \cap (\cup A)}$.

• For every $\mathcal{E}_{ess}^{\star} \cup \mathcal{E}_{A}^{\star}$ -piece $E^{\star} \subset S$, the map $\Phi_{f(E)} \circ f \circ \Psi_{S}^{-1}$ is holomorphic in $\Psi_{S}(E^{\star})$.

• For every \mathcal{E} -piece $E \subset \bigcup_{D \in \mathcal{D}(S)} D$, the map $\Phi_{f(E)} \circ f \circ \Psi_S^{-1}$ is holomorphic in $\Psi_S(E)$.

Now, we define a quasiconformal map $\Theta : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ by $\Theta|_S = \Psi_S^{-1} \circ \Phi_S$ for all $S \in \mathcal{S}$. It's obvious that Θ is isotopic to the identity map rel P. Moreover, for each curve $\gamma \in \Gamma$, we have $\Theta(\gamma) = \gamma$ and $A^{\gamma} \subset \Theta^{-1}(A^{\gamma})$. Let $\Phi : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ be a quasiconformal map whose Beltrami coefficient satisfies

$$\mu_{\Phi}(z) = \sum_{S \in \mathcal{S}} \chi_S(z) \mu_{\Phi_S}(z), \ z \in \overline{\mathbb{C}}.$$

Set $\Psi = \Phi \circ \Theta^{-1}$. Then (f, P) is q.c-equivalent to the Herman map $(g, Q) := (\Phi \circ f \circ \Psi^{-1}, \Phi(P))$ via (Φ, Ψ) .

One can verify that g is holomorphic outside $X := \Psi(\bigcup_{\gamma \in \Gamma \cup (\bigcup_{S \in S} \Gamma_S)} A^{\gamma})$. To see this, notice that if $E^* \in \mathcal{E}_{ess}^* \cup \mathcal{E}_A^*$ and E^* is contained in some S-piece S, then the decomposition

$$g|_{\Psi(E^{\star})} = (\Phi \circ \Phi_{f(E)}^{-1}) \circ (\Phi_{f(E)} \circ f \circ \Psi_S^{-1}) \circ (\Phi_S \circ \Phi^{-1})|_{\Psi(E^{\star})}$$

implies that g is holomorphic in $\Psi(E^*)$ since each factor is holomorphic. If $E \in \mathcal{E}$ and $E \subset D \in \mathcal{D}(S)$, then

$$g|_{\Psi(E)} = (\Phi \circ \Phi_{f(E)}^{-1}) \circ (\Phi_{f(E)} \circ f \circ \zeta_D^{-1}) \circ (\Phi_S \circ \Phi^{-1})|_{\Psi(E)}$$

so g is holomorphic in $\Psi(E)$.

The last step is to apply quasiconformal surgery. For each curve $\gamma \in \Gamma$, let $\iota(\gamma)$ be the first integer $p \geq 1$ such that $f^p(\gamma) \in \Gamma_0$ and $L = \max_{\gamma \in \Gamma} \iota(\gamma)$. One may verify by induction that for any $j \geq 1$,

$$g^{-j}(\Psi(\cup\mathcal{A})) = \Psi((\Theta \circ f)^{-j}(\cup\mathcal{A})) \supset \Psi(\cup_{\gamma \in \Gamma, \iota(\gamma) \le j} A^{\gamma}).$$

In particular, $g^{-L-1}(\Psi(\cup \mathcal{A})) \supset X$. Let σ_0 be the standard complex structure in $\overline{\mathbb{C}}$. Define a (g, Q)-invariant complex structure σ by

$$\sigma = \begin{cases} (g^k)^*(\sigma_0), & \text{ in } g^{-k}(\Psi(\cup\mathcal{A})) \setminus g^{-k+1}(\Psi(\cup\mathcal{A})), \ k \ge 1, \\ \sigma_0, & \text{ in } \overline{\mathbb{C}} - \cup_{k \ge 1} (g^{-k}(\Psi(\cup\mathcal{A})) \setminus g^{-k+1}(\Psi(\cup\mathcal{A}))). \end{cases}$$

Since (g, Q) is holomorphic outside X, the Beltrami coefficient μ of σ satisfies $\|\mu\|_{\infty} < 1$. By Measurable Riemann Mapping Theorem, there is a quasiconformal map $\zeta : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that $\zeta^*(\sigma_0) = \sigma$. Let $R = \zeta \circ g \circ \zeta^{-1}$, then R is a rational map and (f, P) is q.c-equivalent to $(R, \zeta \circ \Phi(P))$ via $(\zeta \circ \Phi, \zeta \circ \Psi)$.

6.5 Decomposition part II

In the following three sections, we will prove the following:

Theorem 6.5.1. Let (f, P) be a non-parabolic map with $n_A(f) > 0$, then there exist a (f, P)-stable multicurve \mathcal{Y} and a finite collection of Herman maps, or Siegel maps, or Thurston maps, say $\{(h_k, P_k), k \in \Lambda\}$, such that

1. (f, P) has no Thurston obstructions if and only if $\lambda(\mathcal{Y}, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) has no Thurston obstructions.

2. (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\mathcal{Y}, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is q.c-equivalent to a rational map.

This theorem can be read as

Non-parabolic = multicurve + Herman + Siegel + Thurston.

The proof of Theorem 6.5.1 is based on Cui-Tan's repelling system theory [CT1].

6.5.1 The hole-filling operator

Let (f, P) be a non-parabolic map with $n_A(f) > 0$.

A bordered surface $S \subsetneqq \overline{\mathbb{C}}$ is the Riemann sphere $\overline{\mathbb{C}}$ minus at most finitely many open quasidisks whose closures are mutually disjoint. The set of all boundary curves of S is denoted by $\partial(S)$.

Let S be a bordered surface with $\partial S \cap P = \emptyset$. The topological complexity of S with respect to the marked set P is defined by

$$\mathcal{T}_P(S) = \#\partial(S) + \#(S \cap P).$$

We say S is of disk type if S is a closed disk containing at most one point in P; of annular type if S is a closed annulus disjoint from P. Otherwise, we say S is of complex type. One may verify that S is of complex type if and only if $\mathcal{T}_P(S) \geq 3$.

A surface puzzle $\mathbf{S} = S_1 \sqcup \cdots \sqcup S_k$ is a finite union of disjoint bordered surfaces. Each S_i is called an **S**-piece.

Let S be a bordered surface. We define $\mathcal{D}(S)$ by

 $\mathcal{D}(S) = \{U; U \text{ is a component of } \overline{\mathbb{C}} - S \text{ with } \#(U \cap P) \le 1\}.$

The hole-filling of S, denoted by $\mathcal{H}(S)$, is defined by

$$\mathcal{H}(S) = S \cup (\cup_{V \in \mathcal{D}(S)} V).$$

The hole-filling of the surface puzzle $\mathbf{S} = S_1 \sqcup \cdots \sqcup S_k$, denoted by $\mathcal{H}(\mathbf{S})$, is defined by

$$\mathcal{H}(\mathbf{S}) = \bigcup_{1 \le j \le k} \mathcal{H}(S_j).$$

One can verify that the hole-filling operator satisfies the following properties:

1. Let S_1 and S_2 be two bordered surfaces. If $S_1 \subset S_2$, then $\mathcal{H}(S_1) \subset \mathcal{H}(S_2)$. If $S_1 \subseteq S_2$, then $\mathcal{H}(S_1) \subseteq \mathcal{H}(S_2)$.

2. Let S_1 and S_2 be two disjoint bordered surfaces. Then either $\mathcal{H}(S_1) \cap \mathcal{H}(S_2) = \emptyset$, or $\mathcal{H}(S_1) \subseteq \mathcal{H}(S_2)$, or $\mathcal{H}(S_2) \subseteq \mathcal{H}(S_1)$.

3. For any bordered surface S, we have $f^{-1} \circ \mathcal{H}(S) \subset \mathcal{H} \circ f^{-1}(S)$ and $\mathcal{H} \circ \mathcal{H}(S) = \mathcal{H}(S)$.



Figure 6.7: The hole-filling of E (Suppose that each boundary curve of S is non-peripheral in $\overline{\mathbb{C}} - P$).

6.5.2 Surface puzzle of constant complexity

Now, we denote by $\mathcal{A}(f)$ the union of all attracting cycles of f in P'_f . For each point $z \in \mathcal{A}(f)$, there is a small disk neighborhood U_z of z such that

- 1. ∂U_z is a quasicircle, and $\partial U_z \cap P = \emptyset$.
- 2. $f^{-1}(\overline{\mathbb{C}} \bigcup_{z \in \mathcal{A}(f)} U_z)$ is contained in the interior of $\overline{\mathbb{C}} \bigcup_{z \in \mathcal{A}(f)} U_z$.
- 3. f is holomorphic in $\bigcup_{z \in \mathcal{A}(f)} U_z$.

Let $\mathbf{S}_0 = \overline{\mathbb{C}} - \bigcup_{z \in \mathcal{A}(f)} U_z$. For $n \geq 1$, we define the surface puzzle \mathbf{S}_n inductively in the following way:

$$\mathbf{S}_n = \mathcal{H} \circ f^{-1}(\mathbf{S}_{n-1}) = \cdots = (\mathcal{H} \circ f^{-1})^n(\mathbf{S}_0).$$

One can verify by induction that $\mathbf{S}_{n+1} \Subset \mathbf{S}_n$ for all $n \ge 0$.

Definition 6.5.1. (Parallel) Let S, E be two bordered surfaces, with $E \Subset S$ and S is of complex type. We say that E is parallel to S if each component of S - E is either a disk containing at most one point in P or an annulus containing no point in P (notice that in the latter case, one boundary curve of the annulus is on ∂S and the other is on ∂E).

Notice that if S is of complex type and E is parallel to S, then $\mathcal{T}_P(E) \geq \mathcal{T}_P(S) \geq 3$. This means E is also of complex type.

Definition 6.5.2. (Constant complexity) Let $\mathbf{S} = S_1 \sqcup \cdots \sqcup S_k$ be a surface puzzle with $f^{-1}(\mathbf{S}) \subseteq \mathbf{S}$ and $\partial \mathbf{S} \cap P = \emptyset$. We say \mathbf{S} is of constant complexity, if either there is no complex type \mathbf{S} -piece, or every complex type \mathbf{S} -piece Scontains a $f^{-1}(\mathbf{S})$ -piece E which is parallel to S. One may compare with [CT1] for the definitions of 'hole-filling', 'parallel' and 'constant complexity'. However, our definitions on these objects are slightly different from Cui-Tan's original ones, in order to make Theorem 6.6.2 sharp.



Figure 6.8: $f^{-1}(S) = E_1 \cup E_2$ (two shadow regions). In the left, E_1 is parallel to S and S is of constant complexity. In the right, neither E_1 nor E_2 is parallel to S, so S is not of constant complexity.

Proposition 6.5.1. When n is large enough, S_n is of constant complexity.

Proof. 1. We claim that: for every $n \ge 0$, and every annular type or complex type \mathbf{S}_n -piece S, all boundary curves of S are non-peripheral in $\overline{\mathbb{C}} - P$.

It's obvious for n = 0. So we just consider the case when $n \ge 1$. Let S be a \mathbf{S}_n -piece, either of annular type or of complex type. Since the hole-filling of two disjoint bordered surfaces either are disjoint or one contains the other, we conclude that there is a unique $f^{-1}(\mathbf{S}_{n-1})$ -piece T such that $S = \mathcal{H}(T)$. Notice that the hole-filling operator satisfies $\mathcal{H} \circ \mathcal{H} = \mathcal{H}$, we have $S = \mathcal{H}(S)$. This implies all boundary curves of S are non-peripheral in $\overline{\mathbb{C}} - P$.

2. For every $n \ge 0$, one can verify that $\mathbf{S}_n \cap P$ has finitely many connected components, and $\mathbf{S}_{n+1} \cap P \subset \mathbf{S}_n \cap P$. We can choose n_0 large enough such that for all $n \ge n_0$, the number of connected components of $\mathbf{S}_n \cap P$ remains constant.

3. For any $n \geq 0$, let k_n be the number of homotopy classes of nonperipheral curves of $\bigcup_{0\leq j\leq n}\partial \mathbf{S}_j$ in $\overline{\mathbb{C}} - P$. It's obvious that $k_n \leq k_{n+1}$. Since $\partial \mathbf{S}_0 \cup (\mathbf{S}_0 \cap P)$ has finitely many components, there exists $n_1 \geq n_0$, such that for all $n \geq n_1$, $k_n = k_{n_1}$.

4. By the choice of n_1 , for any $n > n_1$ and any non-peripheral curve γ of $\partial \mathbf{S}_{n+1}$, there is a curve $\alpha \subset \bigcup_{0 \le j \le n_1} \partial \mathbf{S}_j$ homotopic to γ in $\overline{\mathbb{C}} - P$. Let $A(\gamma, \alpha)$

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 154 THEOREMS

be the annulus bounded by γ and α . First notice that there is a unique \mathbf{S}_{n+1} piece, say S_{n+1} , containing γ as a boundary curve. The piece S_{n+1} is either of annular type or of complex type. Since $\mathbf{S}_{n+1} \in \mathbf{S}_n$, there is a unique \mathbf{S}_n -piece S_n containing S_{n+1} as a proper subset. The piece S_n is either of annular type or of complex type. By 1, each boundary curve of S_n is non-peripheral in $\overline{\mathbb{C}} - P$. It follows that there is a unique boundary curve β of S_n contained in $A(\gamma, \alpha)$. Moreover, β is homotopic to γ in $\overline{\mathbb{C}} - P$, and the annulus $A(\gamma, \beta)$ bounded by γ and β is contained in S_n .

Claim: Every complex type \mathbf{S}_n -piece S_n contains at most one complex type \mathbf{S}_{n+1} -piece, say S_{n+1} . Each component of $S_n - S_{n+1}$ is an annulus.

To see this, let S_{n+1} and S'_{n+1} be two \mathbf{S}_{n+1} -pieces contained in S_n with S_{n+1} of complex type. By the previous argument, each component of $S_n - S_{n+1}$ is an annulus in $\overline{\mathbb{C}} - P$. So there exist $\gamma_n \subset \partial S_n$ and $\gamma_{n+1} \subset \partial S_{n+1}$ such that $S'_{n+1} \subset A(\gamma_n, \gamma_{n+1}) \subset \overline{\mathbb{C}} - P$, where $A(\gamma_n, \gamma_{n+1})$ is the annulus bounded by γ_n and γ_{n+1} . This implies that S'_{n+1} is either of disk type or of annular type.

Let c_n be the number of complex type \mathbf{S}_n -pieces for $n \ge 0$. Notice that every complex type \mathbf{S}_{n+1} -piece is contained in a unique complex type \mathbf{S}_n -piece, we have that for $n > n_1$, $c_{n+1} \le c_n$. So there is $n_2 \ge n_1$ such that for all $n \ge n_2$, $c_n = c_{n_2}$.

To finish, we show that for any $n \ge n_2$, \mathbf{S}_n is of constant complexity. Let S_n be a complex type \mathbf{S}_n -piece. Then there is a unique \mathbf{S}_{n+1} -piece S_{n+1} that is parallel to S_n . Since $c_n = c_{n+1}$, there is a unique $f^{-1}(\mathbf{S}_n)$ -piece, say T, such that $S_{n+1} = \mathcal{H}(T)$. We have

$$S_n - T = (S_n - S_{n+1}) \cup (S_{n+1} - T) = (S_n - S_{n+1}) \cup (\bigcup_{V \in \mathcal{D}(T)} V).$$

This means that T is parallel to S_n .

From the proof of Proposition 6.5.1, we see that when $n \ge n_2$, \mathbf{S}_n is of constant complexity. Fix some $n \ge n_2$, the surface puzzle $\mathbf{S} = \mathbf{S}_n$ satisfies:

• For each annular type or complex type **S**-piece S, each boundary curve γ of S is non-peripheral in $\overline{\mathbb{C}} - P$.

• For every complex type **S**-piece S, there is a unique $f^{-1}(\mathbf{S})$ -piece E_S parallel to S. And $f(E_S)$ is also a complex type **S**-piece.

Let \mathbf{S}_D be the union of all disk type **S**-pieces, \mathbf{S}_A the union of all annular type **S**-pieces and \mathbf{S}_C the union of all complex type **S**-pieces. If $\mathbf{S}_C \neq \emptyset$, then for every \mathbf{S}_C -piece S, there is a unique $f^{-1}(\mathbf{S})$ -piece E_S parallel to S, and $f(E_S)$ is also a complex type **S**-piece. We define a map f_* from all \mathbf{S}_C -pieces to themselves by

$$f_*(S) = f(E_S)$$

where E_S is the unique $f^{-1}(\mathbf{S})$ -piece that is parallel to S. Since \mathbf{S}_C has finitely many components, every \mathbf{S}_C -piece is eventually periodic under f_* . Let $\mathcal{Y} \subset \{\text{all boundary curves of } \mathbf{S}_A \cup \mathbf{S}_C\}$ be a multicurve that represents all homotopy classes of boundary curves of $\mathbf{S}_A \cup \mathbf{S}_C$ in $\overline{\mathbb{C}} - P$. Such multicurve \mathcal{Y} is unique up to homotopy.

The constant complexity of \mathbf{S} implies:

Lemma 6.5.1. The multicurve \mathcal{Y} is (f, P)-stable.

6.5.3 Marked disk extension

We assume in this section that $\mathbf{S}_C \neq \emptyset$. This section is an analogue of Section 6.2. For each \mathbf{S}_C -piece S, we denote by $\overline{\mathbb{C}}(S)$ the Riemann sphere containing S. We always assume that different \mathbf{S}_C -pieces are embedded into different copies of Riemann spheres.

Since each component U_i of $\overline{\mathbb{C}}(S) - E_S$ either is contained in S or contains a unique component V_j of $\overline{\mathbb{C}}(S) - S$. In the former case, if U_i contains a marked point $p \in P$, then we get a marked disk (U_i, p) ; else, we don't mark any point in U_i . In the latter case, we mark a point $p \in V_j \subset U_i$, and get two marked disks (V_i, p) and (U_i, p) .

We define

 $P(S) = (P \cap S) \cup \{ \text{all marked points in } \overline{\mathbb{C}}(S) - S \}.$

We call $(\overline{\mathbb{C}}(S), P(S))$ a marked sphere of S.

Now we extend $f|_{E_S}$ to a branched covering $H_S : \overline{\mathbb{C}}(S) \to \overline{\mathbb{C}}(f_*(S))$. For each component U_i of $\overline{\mathbb{C}}(S) - E_S$, we define $H_S|_{U_i}$ in the following way:

a). If U_i contains a marked point $p \in P(S)$, then H_S maps the marked disk (U_i, p) to the marked disk (W_j, q) . Here W_j is the component of $\overline{\mathbb{C}}(f_*(S)) - f_*(S)$ whose boundary curve is $f(\partial U_i)$. We require further that $H_S(p) = q$ and p is the only possible critical point, with local degree deg $(f|_{\partial U_i})$.

b). If U_i contains no marked point in P(S), then H_S maps U_i to the marked disk (W_j, q) , such that $\deg(H_S|_{U_i}) = \deg(f|_{\partial U_i})$ and q is the only possible critical value. Here, W_j is the component of $\overline{\mathbb{C}}(f_*(S)) - f_*(S)$ whose boundary curve is $f(\partial U_i)$.

In this way, for each \mathbf{S}_C -piece S, we can get an extension H_S of $f|_{E_S}$. It satisfies $H_S(P(S)) \subset P(f_*(S))$.

Notice that every \mathbf{S}_C -piece is eventually periodic under the map f_* . We list all periodic \mathbf{S}_C -pieces in the following:

$$S_{\nu} \mapsto f_*(S_{\nu}) \mapsto \dots \mapsto f_*^{p_{\nu}-1}(S_{\nu}) \mapsto f_*^{p_{\nu}}(S_{\nu}) = S_{\nu}, \quad 1 \le \nu \le n,$$

where n is the number of f_* -periodic cycles, S_{ν} is a representative in the ν -th cycle and p_{ν} is the period of the piece S_{ν} .

Set

$$h_{\nu} = H_{f_*^{p_{\nu}-1}(S)} \circ \cdots \circ H_{f_*(S)} \circ H_S, \ P_{\nu} = P(S_{\nu}), \ 1 \le \nu \le n.$$

Then $h_{\nu} : \overline{\mathbb{C}}(S_{\nu}) \to \overline{\mathbb{C}}(S_{\nu})$ is a branched covering with $h_{\nu}(P_{\nu}) \subset P_{\nu}$. There are four types of the resulting map (h_{ν}, P_{ν}) :

• S_{ν} contains at least one rotation annulus of (f, P). In this case, (h_{ν}, P_{ν}) has at least one cycle of rotation annulus, and each boundary cycle of which contains at least one critical point of h_{ν} . So (h_{ν}, P_{ν}) is a Herman map.

• S_{ν} contains no rotation annulus of (f, P) but contains at least one rotation disk of (f, P). In this case, (h_{ν}, P_{ν}) has at least one cycle of rotation disk whose boundary cycle contains at least one critical point of h_{ν} . So (h_{ν}, P_{ν}) is a Siegel map.

• S_{ν} contains neither rotation annulus nor rotation disk of (f, P) and $\deg(h_{\nu}) > 1$. In this case, (h_{ν}, P_{ν}) is a Thurston map.

• S_{ν} contains neither rotation annulus nor rotation disk of (f, P) and $\deg(h_{\nu}) = 1$. In this case, (h_{ν}, P_{ν}) is a homeomorphism and all points in P_{ν} are periodic.

Let Λ be the index set consisting of all $\nu \in \{1, \dots, n\}$ such that $\deg(h_{\nu}) > 1$. For each $\nu \in \Lambda$, (h_{ν}, P_{ν}) is either a Herman map, or a Siegel map, or a Thurston map. Let Λ^* be the set of all indices $\nu \in \{1, \dots, n\}$ such that $\deg(h_{\nu}) = 1$.

We use the following notation to record the decomposition procedure:

$$\operatorname{Dec}_0(f, P) = \left(\bigoplus_{\nu \in \Lambda \cup \Lambda^*} (h_\nu, P_\nu)\right)_{\mathcal{Y}}.$$

By the same argument as Lemma 6.2.4, one may show that if $\lambda(\mathcal{Y}, f) < 1$, then the following holds:

1. For any $1 \leq \nu \leq n$, every point in $(\overline{\mathbb{C}}(S_{\nu}) - S_{\nu}) \cap P_{\nu}$ is eventually mapped to a periodic critical point of (h_{ν}, P_{ν}) .

2. $\Lambda^* = \emptyset$.

3. If (h_{ν}, P_{ν}) is a Thurston map, then the signature of the orbifold of (h_{ν}, P_{ν}) is not (2, 2, 2, 2).

6.6 Combinatorial part II

In this section, we will prove

Theorem 6.6.1. Let (f, P) be a non-parabolic map with $n_A(f) > 0$. and

$$\operatorname{Dec}_0(f, P) = \left(\bigoplus_{\nu \in \Lambda \cup \Lambda^*} (h_{\nu}, P_{\nu})\right)_{\mathcal{Y}}.$$

Then (f, P) has no Thurston obstructions if and only if $\lambda(\mathcal{Y}, f) < 1$ and for each $\nu \in \Lambda$, (h_{ν}, P_{ν}) has no Thurston obstructions.

Here is the idea of the proof:

We first show that any (f, P)-stable multicurve induces a (F, Q)-stable multicurve (Here, (F, Q) is a suitable restriction of (f, P), which will be called a 'repelling system' of (f, P), see Section 6.6.1), and vice versa. These two multicurves have the same leading eigenvalues of their respective transition matrices. This implies that (f, P) has no Thurston obstructions if and only if (F, Q) has no Thurston obstructions (see Theorems 6.6.2 and 6.6.3).

We then show that every (F, Q)-stable multicurve can be decomposed into a ' \mathcal{Y} -part' and a (h_{ν}, P_{ν}) -stable multicurve for each $\nu \in \Lambda \cup \Lambda^*$. Conversely, every (h_{ν}, P_{ν}) -stable multicurve can generate a (F, Q)-stable multicurve. This enables us to prove that (F, Q) has no Thurston obstructions if and only if $\lambda(\mathcal{Y}, f) < 1$ and for each $\nu \in \Lambda$, the map (h_{ν}, P_{ν}) has no Thurston obstructions (Theorems 6.6.4 and 6.6.5).

Then Theorem 6.6.1 follows by combining the above two steps.

6.6.1 Multicurves for repelling system

Let $\mathbf{B} = f^{-1}(\mathbf{S}) \Subset \mathbf{S}$, $F = f|_{\mathbf{B}}$, $Q = \mathbf{S} \cap P$. We call (F, Q) a repelling system of (f, P).

A Jordan curve $\gamma \subset \mathbf{S} \setminus Q$ is called *null-homotopic* (resp. *peripheral*, *non-peripheral*) in $\mathbf{S} \setminus Q$ if it is null-homotopic (resp. peripheral, non-peripheral) in $\overline{\mathbb{C}} \setminus P$.

We say that $\Gamma = \{\gamma_1, \dots, \gamma_n\}$ is a *multicurve* in $\mathbf{S} \setminus Q$ if each γ_i is a nonperipheral Jordan curve in $\mathbf{S} \setminus Q$, and they are mutually disjoint and no two homotopic in $\mathbf{S} \setminus Q$. Its (F, Q)-transition matrix $W_{\Gamma} = (a_{ij})$ is defined by

$$a_{ij} = \sum_{\alpha \sim \gamma_i} \frac{1}{\deg(F : \alpha \to \gamma_j)},$$

where the summation is taken over all components α of $F^{-1}(\gamma_j)$ homotopic to γ_i in $\mathbf{S} \setminus Q$.

Notice that if two Jordan curves are homotopic in $\mathbf{S} \setminus Q$, then they are necessarily homotopic in $\overline{\mathbb{C}} \setminus P$. But the converse is not true.

A multicurve Γ in $\mathbf{S} \setminus Q$ is called (F, Q)-stable if each component of $F^{-1}(\gamma)$ for $\gamma \in \Gamma$ is either null-homotopic or peripheral, or homotopic to a curve of Γ in $\mathbf{S} \setminus Q$.

We say a multicurve Γ in $\mathbf{S} \setminus Q$ is a *Thurston obstruction* of (F, Q) if Γ is (F, Q)-stable and the leading eigenvalue $\lambda(\Gamma, F)$ of its transition matrix W_{Γ} satisfies $\lambda(\Gamma, F) \geq 1$.



Figure 6.9: Suppose that the annulus $A(\gamma_1, \gamma_2)$ bounded by γ_1 and γ_2 contains no point in P, then γ_1 and γ_2 are homotopic in $\overline{\mathbb{C}} - P$. But they are not homotopic in $\mathbf{S} \setminus Q$.

For convention, an empty set $\Gamma = \emptyset$ is also considered as a (F, Q)-stable multicurve with $\lambda(\Gamma, F) = 0$.

Lemma 6.6.1. Given two different **S**-pieces S_1 and S_2 , each is either of annular type or of complex type. If there are two non-peripheral curves $\gamma_i \subset S_i, i = 1, 2$, homotopic to each other in $\overline{\mathbb{C}} - P$, then γ_i is homotopic in S_i to a boundary curve of $S_i, i = 1, 2$.

Proof. We consider the annulus $A(\gamma_1, \gamma_2)$ bounded by γ_1 and γ_2 . Since γ_1 is homotopic to γ_2 in $\overline{\mathbb{C}} - P$, we have that $A(\gamma_1, \gamma_2) \cap P = \emptyset$. Since S_i is either of annular type or of complex type, each boundary curve of S_i is non-peripheral in $\overline{\mathbb{C}} - P$. This implies that the closure of $A(\gamma_1, \gamma_2)$ necessarily contains a unique boundary curve of S_i , i = 1, 2. The conclusion follows immediately. \Box

Given a (f, P)-stable multicurve Γ , suppose that each curve $\gamma \in \Gamma$ is contained in $\mathbf{S}_A \cup \mathbf{S}_C$. Obviously, Γ is not necessarily a (F, Q)-stable multicurve in $\mathbf{S} - Q$. But we can use Γ to generate a (F, Q)-stable multicurve as follows:

Take a non-peripheral curve $\gamma \in f^{-1}(\Gamma)$ that is not homotopic to any curve of Γ in $\mathbf{S} - Q$ (if any), then γ is contained in some $\mathbf{S}_A \cup \mathbf{S}_C$ -piece, say S. Since Γ is (f, P)-stable, there is a curve $\delta \in \Gamma$, homotopic to γ in $\overline{\mathbb{C}} - P$. We conclude by Lemma 6.6.1 that there is a unique boundary curve $\beta(\gamma)$ of S such that γ is homotopic to $\beta(\gamma)$ in S - Q. We define

> $\Gamma_1 = \Gamma \cup \{\beta(\gamma); \gamma \in f^{-1}(\Gamma) \text{ is non-peripheral}$ and not homotopic to any curve of Γ in $\mathbf{S} - Q\}.$

For n > 1, we define Γ_n inductively in the following way: Let Σ_n be the set of all non-peripheral curves in $f^{-n}(\Gamma)$ that are not homotopic to any curve of Γ_{n-1} in $\mathbf{S} - Q$. For each $\gamma \in \Sigma_n$, let $\beta(\gamma)$ be the unique boundary curve of \mathbf{S} homotopic to γ in $\mathbf{S} - Q$. Notice that given two different curves $\gamma_1, \gamma_2 \in \Sigma_n$, it may happen $\beta(\gamma_1) = \beta(\gamma_2)$. We define

$$\Gamma_n = \Gamma_{n-1} \cup \{\beta(\gamma); \gamma \in \Sigma_n\}.$$

One may verify that for each $n \geq 1$, the set Γ_n is a multicurve in $\mathbf{S} - Q$. Since $\partial \mathbf{S}_A \cup \partial \mathbf{S}_C \cup (\mathbf{S}_C \cap P)$ has finitely many components, there exists a positive integer N such that $\#\Gamma_n = \#\Gamma_N$ for all $n \geq N$. We define $\mathbf{G}(\Gamma) = \Gamma_N$. By the choice of N, one can verify that $\mathbf{G}(\Gamma)$ is a (F, Q)-stable multicurve. We call $\mathbf{G}(\Gamma)$ a (F, Q)-stable multicurve generated by Γ .

Now, let Σ be a (F, Q)-stable multicurve. We define an equivalent relation for the curves in Σ : $\alpha \sim \beta$ if α and β are homotopic in $\overline{\mathbb{C}} - P$. In this way, we can decompose Σ into finitely many equivalent classes $\Sigma_1 \sqcup \cdots \sqcup \Sigma_k$. For each equivalent class Σ_j , we choose a representative $\gamma_j \in \Sigma_j$. We define a multicurve $\pi(\Sigma)$ as follows:

$$\pi(\Sigma) = \{\gamma_1, \cdots, \gamma_k\}.$$

It's easy to check that $\pi(\Sigma)$ is a (f, P)-stable multicurve.

Theorem 6.6.2. For any (f, P)-stable multicurve Γ , suppose that each curve $\gamma \in \Gamma$ is contained in **S**, then $\lambda(\Gamma, f) = \lambda(\mathbf{G}(\Gamma), F)$. Conversely, for any (F, Q)-stable multicurve Σ , we have $\lambda(\Sigma, F) = \lambda(\pi(\Sigma), f)$.

Proof. For any (f, P)-stable multicurve Γ , we decompose $\mathbf{G}(\Gamma)$ as follows: $\mathbf{G}(\Gamma) = \bigcup_{\gamma \in \Gamma} \Gamma_{\gamma}$, where $\Gamma_{\gamma} = \{\delta \in \mathbf{G}(\Gamma); \delta \text{ is homotopic to } \gamma \text{ in } \overline{\mathbb{C}} - P\}$. Let $W_{\Gamma} = (a_{\alpha\beta})$ be the (f, P)-transition matrix of Γ and $W_{\mathbf{G}(\Gamma)} = (b_{\xi\eta})$ be the (F, Q)-transition matrix of $\mathbf{G}(\Gamma)$. Then

$$a_{\alpha\beta} = \sum_{\delta \in \Lambda_{\alpha\beta}} \frac{1}{\deg(f:\delta \to \beta)}, \ b_{\xi\eta} = \sum_{\zeta \in \Omega_{\xi\eta}} \frac{1}{\deg(F:\zeta \to \eta)},$$

where $\Lambda_{\alpha\beta}$ is the collection of components of $f^{-1}(\beta)$ which are homotopic to α in $\overline{\mathbb{C}} - P$ and $\Omega_{\xi\eta}$ is the collection of components of $F^{-1}(\eta)$ which are homotopic to ξ in $\mathbf{S} - Q$.

We claim that for any $\alpha, \beta \in \Gamma$ and any $\eta \in \Gamma_{\beta}$,

$$a_{\alpha\beta} = \sum_{\xi \in \Gamma_{\alpha}} b_{\xi\eta}.$$

To prove this, we first assume $\eta = \beta$. One may check that

$$\cup_{\xi\in\Gamma_{\alpha}}\Omega_{\xi\beta} = \{\delta\subset F^{-1}(\beta) \text{ is homotopic to a curve of } \Gamma_{\alpha} \text{ in } \mathbf{S} - Q\} \\ \subset \{\delta\subset F^{-1}(\beta) \text{ is homotopic to } \alpha \text{ in } \overline{\mathbb{C}} - P\} = \Lambda_{\alpha\beta}.$$

Conversely, for any curve $\delta \subset F^{-1}(\beta)$ which is homotopic to α in $\overline{\mathbb{C}} - P$, since $\mathbf{G}(\Gamma)$ is (F, Q)-stable, there is a curve $\xi \in \mathbf{G}(\Gamma)$, homotopic to δ in $\mathbf{S} - Q$. This implies $\Lambda_{\alpha\beta} \subset \bigcup_{\xi \in \Gamma_{\alpha}} \Omega_{\xi\beta}$. So we have $\bigcup_{\xi \in \Gamma_{\alpha}} \Omega_{\xi\beta} = \Lambda_{\alpha\beta}$ and

$$a_{\alpha\beta} = \sum_{\delta \in \cup_{\xi \in \Gamma_{\alpha}} \Omega_{\xi\beta}} \frac{1}{\deg(f:\delta \to \beta)} = \sum_{\xi \in \Gamma_{\alpha}} \sum_{\delta \in \Omega_{\xi\beta}} \frac{1}{\deg(F:\delta \to \beta)} = \sum_{\xi \in \Gamma_{\alpha}} b_{\xi\eta}.$$

If $\eta \neq \beta$, we can replace η by β in Γ . This replacement doesn't change the transition matrix W_{Γ} . So the claim holds.

Now for any $p \times q$ real matrix $A = (a_{ij})$, we define $||A|| = \sum_{ij} |a_{ij}|$. It's obvious that if p = q, then ||A|| is the Banach norm of A. In the following, we will show that:

There exist two constants $C_2 \ge C_1 > 0$ such that for all $n \ge 1$,

$$C_1 \|W_{\Gamma}^n\| \le \|W_{\mathbf{G}(\Gamma)}^n\| \le C_2 \|W_{\Gamma}^n\|.$$

To prove this relation, we make a block decomposition of $W_{\mathbf{G}(\Gamma)}$: $W_{\mathbf{G}(\Gamma)} = (W_{\alpha\beta})_{\alpha,\beta\in\Gamma}$, where $W_{\alpha\beta}$ is the (α,β) -th block. Moreover, $W_{\alpha\beta}$ is a $\#\Gamma_{\alpha} \times \#\Gamma_{\beta}$ matrix, whose entries are $b_{\xi\eta}, \xi \in \Gamma_{\alpha}, \eta \in \Gamma_{\beta}$. One can verify that

$$||W_{\alpha\beta}|| = \sum_{\xi \in \Gamma_{\alpha}} \sum_{\eta \in \Gamma_{\beta}} b_{\xi\eta} = \#\Gamma_{\beta} \cdot a_{\alpha\beta}$$

We set $C_1 = \min\{\#\Gamma_{\gamma}; \gamma \in \Gamma\}, C_2 = \max\{\#\Gamma_{\gamma}; \gamma \in \Gamma\}$. Then we have

$$\|W_{\mathbf{G}(\Gamma)}\| = \sum_{\alpha,\beta\in\Gamma} \|W_{\alpha\beta}\| = \sum_{\alpha,\beta\in\Gamma} \#\Gamma_{\beta}a_{\alpha\beta} \begin{cases} \geq C_1 \|W_{\Gamma}\|, \\ \leq C_2 \|W_{\Gamma}\|. \end{cases}$$

For any $n \geq 1$, the (α, β) -th block of $W^n_{\mathbf{G}(\Gamma)}$ is $\sum_{\alpha_1, \dots, \alpha_{n-1}} W_{\alpha \alpha_1} W_{\alpha_1 \alpha_2} \cdots W_{\alpha_{n-1}\beta}$. One can prove by induction that

$$\left\|\sum_{\alpha_1,\cdots,\alpha_{n-1}} W_{\alpha\alpha_1}W_{\alpha_1\alpha_2}\cdots W_{\alpha_{n-1}\beta}\right\| = \#\Gamma_\beta \cdot \sum_{\alpha_1,\cdots,\alpha_{n-1}} a_{\alpha\alpha_1}a_{\alpha_1\alpha_2}\cdots a_{\alpha_{n-1}\beta}.$$

It follows that

$$\|W_{\mathbf{G}(\Gamma)}^{n}\| = \sum_{\alpha,\beta\in\Gamma} \#\Gamma_{\beta} \cdot \sum_{\alpha_{1},\cdots,\alpha_{n-1}} a_{\alpha\alpha_{1}}a_{\alpha_{1}\alpha_{2}}\cdots a_{\alpha_{n-1}\beta} \begin{cases} \geq C_{1}\|W_{\Gamma}^{n}\|, \\ \leq C_{2}\|W_{\Gamma}^{n}\|. \end{cases}$$

The proof of the claim is completed. This implies

$$\operatorname{sp}(W_{\Gamma}) = \lim_{n \to \infty} \sqrt[n]{\|W_{\Gamma}^n\|} = \lim_{n \to \infty} \sqrt[n]{\|W_{\mathbf{G}(\Gamma)}^n\|} = \operatorname{sp}(W_{\mathbf{G}(\Gamma)}).$$

By Perron-Frobenius Theorem, $\lambda(\Gamma, f) = \lambda(\mathbf{G}(\Gamma), F)$. The second statement follows from the same argument.

6.6.2 Reduction of no Thurston obstructions

We first prove the following

Theorem 6.6.3. (f, P) has no Thurston obstructions if and only if (F, Q) has no Thurston obstructions.

Proof. Let Γ be a (f, P)-stable multicurve. Using the same argument as in Lemma 6.3.1, we can show that there is a (f, P)-stable multicurve Γ_* , such that

- Γ_* is homotopically contained in Γ .
- Each curve of Γ_* is contained in a **S**-piece.
- $\lambda(\Gamma_*, f) = \lambda(\Gamma, f).$

The multicurve Γ_* can generate a (F, Q)-stable multicurve $\mathbf{G}(\Gamma_*)$. By Theorem 6.6.2, $\lambda(\mathbf{G}(\Gamma_*), F) = \lambda(\Gamma_*, f) = \lambda(\Gamma, f)$. Thus that (F, Q) has no Thurston obstructions implies that (f, P) has no Thurston obstructions.

On the other hand, given a (F, Q)-stable multicurve Σ , the set $\pi(\Sigma)$ is a (f, P)-stable multicurve, and $\lambda(\Sigma, F) = \lambda(\pi(\Sigma), f)$. Thus if (f, P) has no Thurston obstructions, then (F, Q) has no Thurston obstructions either. \Box

Definition 6.6.1.

We say that (F,Q) is $\begin{cases}
a \text{ disk-covering, if } \mathbf{S}_C = \mathbf{S}_A = \emptyset, \\
an \text{ annular-covering, if } \mathbf{S}_C = \emptyset \text{ but } \mathbf{S}_A \neq \emptyset, \\
a \text{ complex-covering, if } \mathbf{S}_C \neq \emptyset.
\end{cases}$

Notice that if $F : (\mathbf{E}, Q) \to (\mathbf{S}, Q)$ is a disk-covering or an annularcovering, then (f, P) has no rotation domains.

Let \mathcal{B} be a multicurve of (F, Q), consisting of all homotopy classes the boundary curves of $\mathbf{S}_A \cup \mathbf{S}_C$ in $\mathbf{S} - Q$. In fact, \mathcal{B} can be written as

$$\mathcal{B} = \{ \gamma \in \partial(S); S \text{ is an } \mathbf{S}_C \text{-piece} \} \cup \{ \gamma_S; S \text{ is an } \mathbf{S}_A \text{-piece} \},\$$

where γ_S is a boundary curve of S if S is an \mathbf{S}_A -piece (notice that for each annular piece S, \mathcal{B} contains only one boundary curve of S).

Lemma 6.6.2. The multicurve \mathcal{B} is (F,Q)-stable and $\lambda(\mathcal{B},F) = \lambda(\mathcal{Y},f)$.

Proof. The constant complexity of **S** implies that \mathcal{B} is (F, Q)-stable. Notice that \mathcal{Y} is identical to $\pi(\mathcal{B})$ up to homotopy. The equality $\lambda(\mathcal{B}, F) = \lambda(\mathcal{Y}, f)$ follows from Theorem 6.6.2.

Theorem 6.6.4. (Decomposition of stable multicurve) Suppose that (F,Q) is a complex-covering and let Υ be a (F,Q)-stable multicurve. We define

$$\Upsilon_{\mathcal{B}} = \{ \gamma \in \Upsilon; \gamma \text{ is homotopic to a curve } \delta \in \mathcal{B} \text{ in } \mathbf{S} - Q \},\\ \Sigma_{\nu} = \{ \gamma \in \Upsilon - \Upsilon_{\mathcal{B}}; \gamma \text{ is contained in } S_{\nu} \}, \nu \in \Lambda \cup \Lambda^* = [1, n].$$

Then $\Upsilon_{\mathcal{B}}$ is a (F, Q)-stable multicurve, Σ_{ν} is a (h_{ν}, P_{ν}) -stable multicurve for $1 \leq \nu \leq n$, and we have the following identity:

$$\lambda(\Upsilon, F) = \max\left\{\lambda(\Upsilon_{\mathcal{B}}, F), \sqrt[p_1]{\lambda(\Sigma_1, h_1)}, \cdots, \sqrt[p_n]{\lambda(\Sigma_n, h_n)}\right\}$$

The proof of Theorem 6.6.4 is the same as that of Theorem 6.3.2, we omit the details.

Suppose that (F, Q) is a complex-covering. For each $1 \leq \nu \leq n$, let Σ_{ν} be a (h_{ν}, P_{ν}) -stable multicurve. Up to homotopy, we may assume that each curve of Σ_{ν} is contained in S_{ν} .

In the following, we will use $\Sigma_1, \dots, \Sigma_n$ to generate a (F, Q)-stable multicurve.

Let $\Gamma_k \subset F^{-k}(\Sigma_1 \cup \cdots \cup \Sigma_n)$ be a multicurve in $\mathbf{S} - Q$, such that Γ_k represents all homotopy classes of non-peripheral curves of $F^{-k}(\Sigma_1 \cup \cdots \cup \Sigma_n)$.

Lemma 6.6.3. For any pair of curves $\gamma_i \in \Gamma_i$, $\gamma_j \in \Gamma_j$ with $0 \le i < j$, if γ_i is not homotopic to γ_j in $\mathbf{S} - Q$, then γ_i and γ_j are homotopically disjoint.

Proof. Since γ_i is not homotopic to γ_j in $\mathbf{S} - Q$, either γ_i and γ_j are contained in two different $\mathbf{S}_A \cup \mathbf{S}_C$ -pieces, or γ_i and γ_j are contained in the same \mathbf{S}_C piece, say S. If we are in the former case, then the proof is done. So we just consider the latter case. In this case, if one of the curves γ_i, γ_j is homotopic to a boundary curve of S, then they are obviously homotopically disjoint.

We assume that neither of γ_i, γ_j is homotopic to a boundary curve of S. We further assume by contradiction that γ_i and γ_j intersect homotopically. There is $\nu \in [1, n]$ such that $f^i(\gamma_i)$ and $f^i(\gamma_j)$ are contained in S_{ν} . Moreover, neither of $f^i(\gamma_i), f^i(\gamma_j)$ is homotopic to a boundary curve of S_{ν} . It follows that $f^i(\gamma_j)$ is contained in the unique $F^{i-j}(S_1 \cup \cdots \cup S_n)$ -piece that is parallel to S_{ν} . This implies $i \equiv j \mod p_{\nu}$, where p_{ν} is the f_* -period of S_{ν} . Since Σ_{ν} is (h_{ν}, P_{ν}) stable, $f^i(\gamma_j)$ is homotopic to either a curve $\delta \in \Sigma_{\nu}$, or a boundary curve of S_{ν} . But neither is possible since $f^i(\gamma_j)$ intersects $f^i(\gamma_i) \in \Sigma_{\nu}$ homotopically. \Box It follows from Lemma 6.6.3 that for any $k \ge 1$, there is a multicurve Δ_k in $\mathbf{S} - Q$ such that $\Sigma_1 \cup \cdots \cup \Sigma_n \subset \Delta_k$ and Δ_k represents all homotopy classes of non-peripheral curves in $\Gamma_0 \cup \cdots \cup \Gamma_k$. Obviously, Δ_k is homotopically contained in Δ_{k+1} . Since $\partial \mathbf{S}_A \cup \partial \mathbf{S}_C \cup (\mathbf{S}_C \cap P)$ has finitely many components, there exists a positive integer N such that $\#\Delta_k = \#\Delta_N$ for all $k \ge N$.

By the choice of N, the set Δ_N is a (F, Q)-stable multicurve. We denote Δ_N by $\mathbf{G}_F(\Sigma_1, \dots, \Sigma_n)$ since it is generated by $\Sigma_1, \dots, \Sigma_n$. We define

$$\mathcal{B}_F(\Sigma_1, \cdots, \Sigma_n) = \{ \gamma \in \mathbf{G}_F(\Sigma_1, \cdots, \Sigma_n); \gamma \text{ is homotopic to a curve of } \mathcal{B} \text{ in } \mathbf{S} - Q \}.$$

Obviously $\mathcal{B}_F(\Sigma_1, \dots, \Sigma_n)$ is a (F, Q)-stable multicurve. One may check that for all $\nu \in [1, n]$,

$$\Sigma_{\nu} = \{ \gamma \in \mathbf{G}_F(\Sigma_1, \cdots, \Sigma_n) - \mathcal{B}_F(\Sigma_1, \cdots, \Sigma_n); \gamma \subset S_{\nu} \}.$$

Theorem 6.6.1 is essentially equivalent to the following:

Theorem 6.6.5. We have that

1. If (F,Q) is a disk-covering, then (f,P) has no Thurston obstructions.

2. If (F,Q) is an annular-covering, then (f,P) has no Thurston obstructions if and only if $\lambda(\mathcal{Y}, f) < 1$.

3. If (F,Q) is a complex-covering, then (f,P) has no Thurston obstructions if and only if $\lambda(\mathcal{Y}, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) has no Thurston obstructions.

Proof. 1. Since (F, Q) is a disk-covering, any (F, Q)-stable multicurve \mathcal{X} is an empty set. So we have $\lambda(\mathcal{X}, F) = 0$ and (F, Q) has no Thurston obstructions. It follows from Theorem 6.6.3 that (f, P) has no Thurston obstructions either.

2. For any (F, Q)-stable multicurve \mathcal{X}, \mathcal{X} is homotopically contained in \mathcal{B} . So $\lambda(\mathcal{X}, F) \leq \lambda(\mathcal{B}, F)$. This implies that (F, Q) has no Thurston obstructions if and only if $\lambda(\mathcal{B}, F) < 1$. It follows from Lemma 6.6.2 and Theorem 6.6.3 that (f, P) has no Thurston obstructions if and only if $\lambda(\mathcal{Y}, f) < 1$.

3. The 'sufficiency' follows from Theorem 6.6.3 and 6.6.4. We just prove the 'necessity'. We may assume that (F, Q) has no Thurston obstructions by Theorem 6.6.3. It follows immediately $\lambda(\mathcal{B}, F) < 1$ and $\Lambda^* = \emptyset$. Let Σ_{ν} be a (h_{ν}, P_{ν}) -stable multicurve for $\nu \in \Lambda = [1, n]$. By homotopic deformations, we may assume that each curve of Σ_{ν} is contained in the piece S_{ν} . Let $\mathbf{G}_F(\Sigma_1, \cdots, \Sigma_n)$ be the (F, Q)-stable multicurve generated by $\Sigma_1, \cdots, \Sigma_n$. Again by Theorem 6.6.4, for each $\nu \in \Lambda$,

$$\lambda(\Sigma_{\nu}, h_{\nu}) \leq \lambda(\mathbf{G}_F(\Sigma_1, \cdots, \Sigma_n), F)^{p_{\nu}} < 1.$$

So (h_{ν}, P_{ν}) has no Thurston obstructions for each $\nu \in \Lambda$.

6.7 Surgery part II

The section is a sketch of Cui-Tan's work [CT1], with slight modifications.

In the pervious section, we call (F, Q) a repelling system of (f, P). In this section, to begin with, we give a more general definition of repelling systems:

Definition 6.7.1. We say that a map $H : \mathbf{U} \to \mathbf{V}$ is a repelling system, if

• Both U and V are surface puzzles, with $U \subseteq V$;

• The map H maps every U-piece U properly onto a V-piece V as a quasiregular map;

• The orbit $\{H^n(c); n \ge 0\}$ of every critical point c of H is disjoint from the boundary of **U**.

Notice that in the definition, we don't require that $H(\mathbf{U}) = \mathbf{V}$.

For example, a polynomial-like map is a repelling system.

Let C(H) be the critical set of the repelling system $H : \mathbf{U} \to \mathbf{V}$, the postcritical set P(H) is defined as the closure of $\{H^n(c) \in \mathbf{V}; c \in C(H), n \geq 1\}$. A marked set of the repelling system $H : \mathbf{U} \to \mathbf{V}$ is a compact set $M \subset \mathbf{V}$ such that $H(M \cap \mathbf{U}) \subset M$ and M - P(H) is a finite set. We will use $H : (\mathbf{U}, M) \to (\mathbf{V}, M)$ to denote a marked repelling system. If there is no confusion, the marked repelling system $H : (\mathbf{U}, M) \to (\mathbf{V}, M)$ is also denoted by (H, M).

Definition 6.7.2. We say that two marked repelling systems $H_1 : (\mathbf{U}_1, M_1) \rightarrow (\mathbf{V}_1, M_1)$ and $H_2 : (\mathbf{U}_2, M_2) \rightarrow (\mathbf{V}_2, M_2)$ are q.c-equivalent if there is a pair of quasiconformal homeomorphisms $\Phi, \Psi : \mathbf{V}_1 \rightarrow \mathbf{V}_2$, such that

- $\Psi(\mathbf{U}_1) = \mathbf{U}_2, \ \Psi(M_1) = M_2;$
- Ψ is isotopic to Φ rel $\partial \mathbf{V}_1 \cup M_1$;
- $\Phi \circ H_1 = H_2 \circ \Psi.$

In this case, we say that (H_1, M_1) is q.c-equivalent to (H_2, M_2) via (Φ, Ψ) . Let $H : (\mathbf{U}, M) \to (\mathbf{V}, M)$ be a marked repelling system. We say that a **U**-piece U is of *disk type*, if one boundary curve of U bounds a disk Δ_U such that $U \subset \Delta_U \subset \mathbf{V}$ and Δ_U contains at most one point in M.

Let \mathbf{U}_0 the union of all disk type **U**-pieces. A repelling system H_1 : $(\mathbf{U}_1, M_1) \rightarrow (\mathbf{V}_1, M_1)$ is called a *sub-marked repelling system* of (H, M), if \mathbf{U}_1 is the union of some **U**-pieces, \mathbf{V}_1 is the union of some **V**-pieces, and $M_1 = \mathbf{V}_1 \cap M$.

Lemma 6.7.1. Let $H : (\mathbf{U}, M) \to (\mathbf{V}, M)$ be a marked repelling system and $H_1 : (\mathbf{U}_1, M_1) \to (\mathbf{V}_1, M_1)$ be a sub-marked repelling system of (H, M), with $\mathbf{U} - \mathbf{U}_1 \subset \mathbf{U}_0$. Then (H, M) is q.c-equivalent to a holomorphic marked repelling system if and only if (H_1, M_1) is q.c-equivalent to a holomorphic marked repelling system. *Proof.* Suppose that (H, M) is q.c-equivalent to a holomorphic marked repelling system via (Φ, Ψ) , then (H_1, M_1) is q.c-equivalent to a holomorphic marked repelling system via $(\Phi|_{\mathbf{V}_1}, \Psi|_{\mathbf{V}_1})$.

On the other hand, suppose that (H_1, M_1) is q.c-equivalent to a holomorphic marked repelling system via (Φ_1, Ψ_1) . We first extend Φ_1 to a quasiconformal map $\Phi : \mathbf{V} \to \Phi(\mathbf{V}) \subset \overline{\mathbb{C}}$. Then we define a quasiconformal homeomorphism $\Psi : \mathbf{V} \to \Phi(\mathbf{V})$ isotopic to Φ rel $\partial \mathbf{V} \cup M$, as follows:

Given a V-piece V, if V is a disk containing at most one point in M, then we choose a closed disk V' such that $V \cap \mathbf{U} \Subset V' \Subset V$ and $V' \cap M = V \cap M$. By Measurable Riemann Mapping Theorem, there is a q.c homeomorphism $\varphi: V' \to \Phi(V')$ with Beltrami coefficient:

$$\mu_{\varphi}(z) = \sum_{U \subset V \cap \mathbf{U}} \chi_U(z) \mu_{\Phi \circ F}(z),$$

where the summation is taken over all U-pieces contained in V. We further require that $\varphi(p) = \Phi(p)$ if V contains a marked point $p \in M$.

We define a quasiconformal map $\Psi: V \to \Phi(V)$ by

$$\Psi(z) = \begin{cases} \Phi(z), & z \in \partial V, \\ \varphi(z), & z \in V', \\ q.c \ interpolation, & z \in V - V' \end{cases}$$

Now, suppose that V is either an annulus or $\#\partial(V) + \#(V \cap M) \geq 3$. Let $\mathcal{U}(V)$ be the collection of all $(\mathbf{U} - \mathbf{U}_1)$ -pieces that are contained in V. Notice that for $U_1, U_2 \in \mathcal{U}(V)$, either $\Delta_{U_1} \cap \Delta_{U_2} = \emptyset$ or one of Δ_{U_i} contains another. We denote all components of $\bigcup_{U \in \mathcal{U}(V)} \Delta_U$ by $\{D_i\}_{i \in \Lambda_V}$, where Λ_V is an index set induced by V. We thicken each D_i a little bit along the boundary ∂D_i to get a larger disk U_i . Take a quasiconformal homeomorphism $\phi_i : D_i \to \psi(D_i)$ (here, we set $\psi = \Psi_1$ if $V \subset \mathbf{V}_1$ and $\psi = \Phi$ if $V \subset \mathbf{V} - \mathbf{V}_1$) such that

$$\mu_{\phi_i}(z) = \sum_{\mathcal{U}(V) \ni U \subset D_i} \chi_E(z) \mu_{\Phi \circ F}(z).$$

We further require that $\phi_i(p) = \Phi(p)$ if D_i contains a marked point $p \in M$. Now, we define

$$\Psi(z) = \begin{cases} \psi(z), & z \in V - \bigcup_{i \in \Lambda_V} U_i, \\ \phi_i(z), & z \in D_i, i \in \Lambda_V, \\ q.c \ interpolation, & z \in \bigcup_{i \in \Lambda_V} (U_i - D_i). \end{cases}$$

In this way, we can construct a quasiconformal map $\Psi : \mathbf{V} \to \Phi(\mathbf{V})$, isotopic to Φ rel $\partial \mathbf{V} \cup M$. One may verify that (H, M) is q.c-equivalent to a holomorphic marked repelling system via (Φ, Ψ) . Let (f, P) be a non-parabolic map with $n_A(f) > 0$, and (F, Q) the marked repelling system of (f, P) defined in Section 6.6.1.

Theorem 6.7.1. (f, P) is q.c-equivalent to a rational map if and only if (F, Q) is q.c-equivalent to a holomorphic marked repelling system.

Proof. Suppose that (f, P) is q.c-equivalent to a rational map (R, M) via (ϕ_0, ϕ_1) , then we can construct a sequence of quasiconformal maps $\phi_n, n \ge 0$ such that $R\phi_{n+1} = \phi_n f$ and ϕ_n is isotopic to ϕ_{n+1} rel $f^{-n}(P \cup N)$, where N is a neighborhood of all attracting cycles of (f, P) in P'_f . By the construction of the repelling system $F : (\mathbf{E}, Q) \to (\mathbf{S}, Q)$, when n is large enough, $\overline{\mathbb{C}} - \mathbf{S} \subset f^{-n}(N)$, thus ϕ_{n+1} and ϕ_n are identical on the boundary $\partial \mathbf{S}$. We set $\Phi = \phi_n |_{\mathbf{S}}, \Psi = \phi_{n+1}|_{\mathbf{S}}$ for such large n. Then (F, Q) is q.c-equivalent to the holomorphic marked repelling system $R|_{\Psi(\mathbf{E})} : (\Psi(\mathbf{E}), \Psi(Q)) \to (\Psi(\mathbf{S}), \Psi(Q))$ via (Φ, Ψ) .

Sufficiency. It follows from [CT1], Proposition 2.4.

Theorem 6.7.2. Suppose that (f, P) is a quasiregular map, then

1. If (F,Q) is a disk-covering, then (f,P) is q.c-equivalent to a rational map. This rational map is unique up to Möbius conjugation.

2. If (F, Q) is an annular-covering, then (f, P) is q.c-equivalent to a rational map (R, M) if and only if $\lambda(\mathcal{Y}, f) < 1$. The rational realization (R, M) is unique up to Möbius conjugation.

3. If (F, Q) is a complex-covering, then (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\mathcal{Y}, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is q.c-equivalent to a rational map.

Proof. 1. If (F, Q) is a disk-covering, then all **E**-pieces are of disk type. It follows from Lemma 6.7.1 that (F, Q) is q.c-equivalent to a holomorphic marked repelling system. By Theorem 6.7.1, (f, P) is q.c-equivalent to a rational map, say (R, M). The uniqueness of the rational realization follows from Lemma 6.9.2 below and the fact that J(R) has zero Lebesgue measure.

2. Let \mathbf{E}_A be the union of all \mathbf{E} -pieces E which are contained essentially in the \mathbf{S}_A -pieces (here, 'essentially' means that E separates the two boundary curves of some \mathbf{S}_A -piece). One may verify that $\mathbf{E}_A \Subset \mathbf{S}_A$ and each \mathbf{E}_A -piece is mapped properly onto some \mathbf{S}_A -piece. Let $G = F|_{\mathbf{E}_A}$, then $G : \mathbf{E}_A \to \mathbf{S}_A$ is a repelling system.

One may check that \mathcal{B} is also a stable multicurve for $G : \mathbf{E}_A \to \mathbf{S}_A$, with $\lambda(\mathcal{B}, G) = \lambda(\mathcal{B}, F)$. Since $\lambda(\mathcal{B}, F) < 1$, it follows from Lemma 6.2 in [CT1] that $G : \mathbf{E}_A \to \mathbf{S}_A$ is q.c-equivalent to a holomorphic repelling system. Notice that each $(\mathbf{E} - \mathbf{E}_A)$ -piece is of disk type, we deduce by Lemma 6.7.1 that (F, Q) is q.c-equivalent to a holomorphic marked repelling system.

By Theorem 6.7.1, (f, P) is q.c-equivalent to a rational map, say (R, M).

The uniqueness of the rational realization follows from Lemma 6.9.2 and the fact that J(R) has zero Lebesgue measure.

3. The 'necessity' part is essentially the same as the proof of 'necessity' of Theorem 6.4.1 (See Section 6.4.1). The 'sufficiency' part is essentially due to Cui Guizhen and Tan Lei [CT1]. Even if they deal only with maps without rotation domains, their proof applies equally well to our situations. For more details, one may refer Cui-Tan's paper ([CT1], Section 7 'Proof of Theorem 5.4 for a cycle of complex pieces' and Section 8 'Proof of Theorem 5.4') \Box

6.8 Proof of Theorem 6.1.2, the first two parts

Let (f, P) be a non-parabolic map. Notice that if (f, P) itself is a Siegel map or a Thurston map, then Theorem 6.1.2 follows immediately (We may take $\Gamma = \emptyset$, and the resulting map as (f, P) itself). If (f, P) is a Herman map or a non-parabolic map without rotation annulus, then Theorem 6.1.2 follows from Theorem 6.2.1 and Theorem 6.5.1.

So in the following, we need only consider the case when $n_A(f) > 0$, $n_{RD}(f) \ge 0$ and $n_{RA}(f) > 0$. By Theorem 6.5.1, there is a (f, P)-stable multicurve Γ_0 , and finitely many branched coverings $(h_k, P_k), k \in \Lambda$, each is either a Herman map, or Siegel map, or Thurston map, such that

• (f, P) has no Thurston obstructions if and only if $\lambda(\Gamma_0, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) has no Thurston obstructions.

• (f, P) is q.c-equivalent to a rational map if and only if $\lambda(\Gamma_0, f) < 1$ and for each $k \in \Lambda$, (h_k, P_k) is q.c-equivalent to a rational map.

Suppose that there are exactly n Herman maps in the resulting maps. We may relabel them such that $(h_1, P_1), \dots, (h_n, P_n)$ are Herman maps. By Theorem 6.2.1, for each $k \in [1, n]$, there is a decomposition

$$\operatorname{Dec}(h_k, P_k) = \left(\bigoplus_{j \in \Lambda_k \cup \Lambda_k^*} (h_{k,j}, P_{k,j})\right)_{\Sigma_k},$$

such that: (h_k, P_k) has no Thurston obstructions if and only if $\lambda(\Sigma_k, h_k) < 1$ and for each $j \in \Lambda_k$, $(h_{k,j}, P_{k,j})$ has no Thurston obstructions; (h_k, P_k) is q.cequivalent to a rational map if and only if $\lambda(\Sigma_k, h_k) < 1$ and for each $j \in \Lambda_k$, $(h_{k,j}, P_{k,j})$ is q.c-equivalent to a rational map.

Now let (F, Q) be the repelling system of (f, P) defined as in Section 6.6.1. We may assume that for each $k \in [1, n]$, each curve of Σ_k is contained in the **S**-piece S_k . In this way, Σ_k can be considered as a multicurve of (F, Q). We use $\Sigma_1, \dots, \Sigma_n$ to generate a (f, P)-stable multicurve $G_F(\Sigma_1, \dots, \Sigma_n)$ (see Section 6.6.2). Let $\Gamma_1 = \pi \circ G_F(\Sigma_1, \dots, \Sigma_n)$, then Γ_1 is a (f, P)-stable multicurve. We decompose Γ_1 into two submulticurves $\Gamma_{1,0} := \{\gamma \in \Gamma_1; \gamma \text{ is homotopic to a curve in } \Gamma_0\}$ and $\Gamma_{1,1} = \Gamma_1 - \Gamma_{1,0}$. One may check that $\Gamma_{1,0}$ is (f, P)-stable and the (f, P)-transition matrix W_1 (resp. $W_{1,0}, W_{1,1}$) of Γ_1 (resp. $\Gamma_{1,0}, \Gamma_{1,1}$) satisfies:

$$W_1 = \left(\begin{array}{cc} W_{1,0} & * \\ O & W_{1,1} \end{array}\right).$$

Thus we have $\lambda(\Gamma_1, f) = \max\{\lambda(\Gamma_{1,0}, f), \lambda(\Gamma_{1,1}, f)\}.$

Define $\Gamma = \Gamma_0 \cup \Gamma_{1,1}$. Then Γ is a (f, P)-stable multicurve and

$$\lambda(\Gamma, f) = \max\{\lambda(\Gamma_0, f), \lambda(\Gamma_{1,1}, f)\} = \max\{\lambda(\Gamma_0, f), \lambda(\Gamma_1, f)\}.$$

By Theorem 6.6.2,

$$\lambda(G_F(\Sigma_1, \cdots, \Sigma_n), F) = \lambda(\Gamma_1, f),$$

$$\lambda(\mathcal{B}_F(\Sigma_1, \cdots, \Sigma_n), F) = \lambda(\pi \circ \mathcal{B}_F(\Sigma_1, \cdots, \Sigma_n), f)$$

Since $\pi \circ \mathcal{B}_F(\Sigma_1, \dots, \Sigma_n)$ is homotopically contained in Γ_0 , we have $\lambda(\pi \circ \mathcal{B}_F(\Sigma_1, \dots, \Sigma_n), f) \leq \lambda(\Gamma_0, f)$. It follows from Theorem 6.6.4 that

$$\lambda(\Gamma, f) = \max\left\{\lambda(\Gamma_0, f), \sqrt[p_1]{\lambda(\Sigma_1, h_1)}, \cdots, \sqrt[p_n]{\lambda(\Sigma_n, h_n)}\right\}$$

This implies $\lambda(\Gamma, f) < 1$ if and only if $\lambda(\Gamma_0, f) < 1$ and for each $k \in [1, n]$, $\lambda(\Sigma_k, h_k) < 1$.

The proof is completed if we take the (f, P)-stable multicurve as Γ and the resulting maps as $(h_{k,j}, P_{k,j}), k \in [1, n], j \in \Lambda_k$ and $(h_k, P_k), k \in \Lambda - [1, n]$.

6.9 Analytic part

In this section, we will discuss the rational-like maps, renormalizations of rational maps and prove the analytic part of Theorem 6.1.2.

6.9.1 Rational-like maps

A rational-like map $g: U \to V$ is a proper and holomorphic map between two multi-connected domains such that $\overline{U} \subset V \subset \overline{\mathbb{C}}$ and the complementary set $\overline{\mathbb{C}} - X$ of $X \in \{U, V\}$ consists of finitely many topological disks. In our discussion, we always assume $V \neq \overline{\mathbb{C}}$ and the degree of g is at least two. The filled Julia set is defined by $K(g) = \bigcap_{n \geq 1} g^{-n}(V)$, the Julia set is defined by $J(g) = \partial K(g)$. The filled Julia set K(g) is not necessarily a full set. This implies that J(g) is not necessarily connected even if K(g) is connected. Two rational-like maps g_1 and g_2 are hybrid equivalent if there is a quasiconformal conjugacy ϕ between g_1 and g_2 , defined in a neighborhood of their respective filled Julia sets, such that $\overline{\partial}\phi = 0$ on $K(g_1)$. We call ϕ a hybrid conjugacy between g_1 and g_2 . These definitions are simply the generalizations of Douady-Hubbard's definitions of polynomial-like maps.

The following is an analogue of Douady-Hubbard's straightening theorem for polynomial-like maps.

Theorem 6.9.1. (Straightening Theorem) Let $g: U \to V$ be a rationallike map of degree $d \ge 2$, then

1. The map g is hybrid equivalent to a rational map R of degree d.

2. If K(g) is connected, then g is hybrid equivalent to a rational map R of degree d, which is postcritically finite outside $\phi(K(g))$. Here ϕ is the hybrid conjugacy. Such R is unique up to Möbius conjugation.

Remark 6.9.1. 1. A rational-like map $g: U \to V$ can be hybrid equivalent to a rational map of degree greater than d.

2. Even if K(g) is connected, the rational-like map g can be hybrid equivalent to a rational map of degree greater than d, which is postcritically finite outside $\phi(K(g))$. Such example can be found in the family of rational maps: $f_{\lambda}(z) = z^n + \lambda/z^n, n \ge 3$, where λ is a complex parameter. We denote by B_{λ} the immediate attracting basin of ∞ . We assume that each critical point of the form $\sqrt[2n]{\lambda}$ has an orbit meeting an attracting cycle other than ∞ . In this case, the Julia set is connected since the map is postcritically finite, and f_{λ} is strictly expanding on ∂B_{λ} . There is an annular neighborhood A of ∂B_{λ} such that $f_{\lambda}|_{A} : A \to f_{\lambda}(A)$ is a proper map of degree n. (Such annulus can be chosen as the union of puzzles pieces that intersect with ∂B_{λ} , see the previous chapter). The rational-like map $f_{\lambda}|_{A}$ can be hybrid equivalent to the power map $z \mapsto z^n$, whose degree is lower than that of f_{λ} .

3. If K(g) is connected and $\overline{\mathbb{C}} - K(g)$ consists of two components, then there are two annuli U', V' such that $K(g) \subset U' \subset V' \subset V$ and the restriction $g|_{U'}: U' \to V'$ is a rational-like map. One may show that K(g) is a quasicircle by quasiconformal surgery.

Proof. 1. The proof is a standard surgery procedure. By shrinking V a little bit, we may assume that each boundary curve of U and V is a quasicircle. We then extend $g: U \to V$ to a quasiregular branched covering $G: \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that G is holomorphic in $\overline{\mathbb{C}} - \overline{V}$ and G maps each component U_k of $\overline{\mathbb{C}} - U$ onto a connected component V_j of $\overline{\mathbb{C}} - V$, with degree equal to deg $(g|_{\partial U_k})$. Such extension keeps the degree. By pulling back the standard complex structure σ_0 on $\overline{\mathbb{C}} - V$ via G, we get a G-invariant complex structure

$$\sigma = \begin{cases} (G^k)^*(\sigma_0), & \text{ in } G^{-k}(\overline{\mathbb{C}} - V), \ k \ge 1, \\ \sigma_0, & \text{ in } K(g). \end{cases}$$

The Beltrami coefficient μ of σ satisfies $\mu|_{K(g)} = 0$ and $\|\mu\|_{\infty} < 1$. Let ϕ solve the Beltrami equation $\overline{\partial}\phi = \mu\partial\phi$. Then $R = \phi \circ G \circ \phi^{-1}$ is a rational map and ϕ is a hybrid conjugacy between g and R.

2. By a hole-filling process, we can find a suitable restriction $g|_{U'}: U' \to V'$ of g with $K(g) \subseteq U' \subseteq V' \subseteq V$ such that

a). All postcritical points of $g|_{U'}$ in V' are contained in K(g).

b). Each connected component of $V' - \overline{U'}$ is either an annulus or a disk.

Notice that such V' can be chosen arbitrarily close to the filled Julia set K(g). (To see this, one may replace V' by $g^{-k}(V')$ for some large k, and a), b) still holds.)

In this way, each component U_i of $\overline{\mathbb{C}} - \overline{U'}$ either is contained in V' or contains a unique component V_j of $\overline{\mathbb{C}} - \overline{V'}$. In the former case, we mark a point $p \in U_i$ and get a marked disk (U_i, p) ; in the latter case, we mark a point $p \in V_j$, and get two marked disks (V_j, p) and (U_i, p) . We extend $g|_{U'}$ to a quasiregular branched covering $G : \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that

a). For each component U_i of $\overline{\mathbb{C}} - \overline{U'}$, G maps the marked disk (U_i, p) to the marked disk (V_k, q) , where V_k is the component of $\overline{\mathbb{C}} - \overline{V'}$ whose boundary is $g(\partial U_i)$. We require that G(p) = q and p is the only possible critical point, with local degree equal to deg $(g|_{\partial U_i})$.

b). We further require that G is holomorphic in $\overline{\mathbb{C}} - \overline{V'}$.

By pulling back the standard complex structure on $\overline{\mathbb{C}} - V'$, we can get a *G*invariant complex structure whose Beltrami coefficient μ satisfies $\mu|_{K(g)} = 0$ and $\|\mu\|_{\infty} < 1$. Let ϕ solve the Beltrami equation $\overline{\partial}\phi = \mu\partial\phi$. Then $f = \phi \circ G \circ \phi^{-1}$ is a rational map, postcritically finite outside $\phi(K(g))$, as required.

To prove the uniqueness, we need investigate some mapping properties of R, a rational map of degree d, to which $g|_{U'}$ is hybrid equivalent via ϕ , and postcritically finite outside $\phi(K(g))$. We assume V' is sufficiently close to K(g) such that ϕ is defined on V'. Then $g|_{U'}$ induces a suitable restriction $R|_{\phi(U')}$. Let \mathcal{X}_1 be the collection of all components of $\overline{\mathbb{C}} - \phi(K(g))$ which intersect with the boundary curves of $\phi(V')$ and \mathcal{X}_2 be the collection of all components of $\phi(U')$. It's obvious that $\mathcal{X}_1 \subset \mathcal{X}_2$. Since the degree of R is equal to d (This is very important), we have that

{U is a component of $R^{-1}(X); X \in \mathcal{X}_1$ } = \mathcal{X}_2 .

Thus for each $X \in \mathcal{X}_2$, $R(X) \in \mathcal{X}_2$. This implies that each $X \in \mathcal{X}_2$ is eventually periodic under the map R. Let $X \in \mathcal{X}_2$ be a periodic element, with period p. Since R is poscritically finite outside $\phi(K(g))$, $R^p|_X : X \to X$ is proper and each critical point in X has finite orbit. Thus $R^p|_X$ is conformally conjugate to $z \mapsto z^d$, where $d = \deg(R^p|_X) \ge 2$ (For a proof of this fact, see [DH2] Lemma 4.1). It follows that for all $X \in \mathcal{X}_2$, the proper map $R|_X : X \to$ R(X) has only one possible critical point, which is eventually mapped to a superattracting cycle. Base on these observations, we are now ready to prove the uniqueness part of the theorem.

Suppose that R_1 and R_2 are two rational maps of degree d, both are hybrid equivalent to $g|_{U'}$ and poscritically finite outside $\phi_1(K(g))$ and $\phi_2(K(g))$, respectively. Here, ϕ_i is a hybrid conjugacy between $g|_{U'}$ and R_i , i = 1, 2. We assume that V' is sufficiently close to K(g) such that ϕ_i is defined on U'. Then $g|_{U'}$ induces two restrictions $R_i|_{\phi_i(U')}$, i = 1, 2 and a hybrid conjugacy $\phi = \phi_2 \circ \phi_1^{-1}$ between them. One can construct a pair of quasiconformal maps $\varphi_0, \varphi_1: \overline{\mathbb{C}} \to \overline{\mathbb{C}}$ such that

a). $\varphi_0 \circ R_1 = R_2 \circ \varphi_1$ on $\overline{\mathbb{C}}$.

b). φ_0, φ_1 are isotopic rel $\phi_1(K(g)) \cup P_{R_1}$ and $\varphi_0|_{\phi_1(U')} = \varphi_1|_{\phi_1(U')} = \phi|_{\phi_1(U')}$.

c). φ_0, φ_1 are holomorphic and identical in a neighborhood N of all superattracting cycles of R_1 in $\overline{\mathbb{C}} - \phi_1(K(g))$.

By Thurston algorithm, there is a sequence of quasiconformal maps $\{\varphi_n, n \geq 0\}$ such that $\varphi_n \circ R_1 = R_2 \circ \varphi_{n+1}$ and φ_n is isotopic to φ_{n+1} rel $R_1^{-n}(\phi_1(U') \cup P_{R_1} \cup N)$. The quasiconformal map φ_n satisfies $\overline{\partial}\varphi_n = 0$ on $\phi_1(K(g)) \cup R_1^{-n}(N)$. The sequence $\{\varphi_n\}$ has a limit quasiconformal map $\varphi = \lim \varphi_n$. Since the Lebesgue measure of $\overline{\mathbb{C}} - \phi_1(K(g)) \cup R_1^{-n}(N)$ tends to zero as $n \to \infty$, the map φ satisfies $\overline{\partial}\varphi = 0$ outside a zero measure set. It is in fact a holomorphic conjugacy between R_1 and R_2 .

6.9.2 Renormalizations of rational maps

Let f be a rational map. Its Julia set, critical set and postcritical set are denoted by J(f), Ω_f and P_f , respectively. Let P'_f be the accumulation set of P_f . A Julia component is a connected component of J(f).

We say f^p is renormalizable if there exist two multi-connected domains U, V such that $f^p : U \to V$ is a rational-like map of degree at least two, and the filled Julia set $K(f^p|_U)$ is connected. The triple (f^p, U, V) is called a renormalization of f. A renormalization (f^p, U, V) is of annular type if $\overline{\mathbb{C}} - K(f^p|_U)$ consists of two components (In this case, $K(f^p|_U)$ is necessarily a quasicircle and we may assume that U, V are annuli, see Remark 6.9.1). By Theorem 6.9.1, the rational-like map $f^p : U \to V$ is hybrid equivalent to a unique rational map R via some quasiconformal map ϕ , such that $\deg(R) = \deg(f^p|_U)$ and R is postcritically finite outside $\phi(K(f^p|_U))$. Such R is called

the canonical straightening map of $f^p: U \to V$. In this section, we will prove:

Theorem 6.9.2. (Renormalization) Let f be a rational map of degree at least two. Suppose that $J(f) \cap P_f$ is contained in finitely many Julia components and P'_f contains attracting cycles. Then f admits finitely many nonannular type renormalizations $(f^{p_i}, U_i, V_i), i \in \Lambda$ (Λ is a finite index set) which satisfy

1. For every $i \in \Lambda$, V_i contains no attracting cycles of f in P'_f .

2. The Julia set J(f) has zero Lebesgue measure (resp. carries no invariant line fields) if and only if for each $i \in \Lambda$, $J(f^{p_i}|_{U_i})$ has zero Lebesgue measure (resp. carries no invariant line fields).

Remark 6.9.2. Theorem 6.9.2 also holds when f is a rational-like map.

The proof is based on the following ([McM1]):

Theorem 6.9.3. (Ergodic or attracting) Let f be a rational map of degree at least two, then either

• $J(f) = \overline{\mathbb{C}}$ and the action of f on $\overline{\mathbb{C}}$ is ergodic, or

• the spherical distance $d(f^n(z), P_f) \to 0$ for almost every $z \in J(f)$ as $n \to \infty$.

Proof of Theorem 6.9.2. Let $\mathcal{A}(f)$ be the union of all attracting cycles of f in P'_f . For each point $z \in \mathcal{A}(f)$, there is a disk neighborhood U_z of z such that 1. ∂U_z is a quasicircle, and $\partial U_z \cap P_f = \emptyset$.

2. $f^{-1}(\overline{\mathbb{C}} - \bigcup_{z \in \mathcal{A}(f)} U_z)$ is contained in the interior of $\overline{\mathbb{C}} - \bigcup_{z \in \mathcal{A}(f)} U_z$.

Let $\mathbf{S}_0 = \overline{\mathbb{C}} - \bigcup_{z \in \mathcal{A}(f)} U_z$, $P = P_f$. For $n \ge 1$, we define the surface puzzle \mathbf{S}_n inductively in the following way:

$$\mathbf{S}_n = \mathcal{H} \circ f^{-1}(\mathbf{S}_{n-1}) = \dots = (\mathcal{H} \circ f^{-1})^n(\mathbf{S}_0).$$

Since $J(f) \cap P_f$ is contained in finitely many Julia components, the same proof as Lemma 6.5.1 yields that when n is large enough, \mathbf{S}_n is of constant complexity. We set $\mathbf{S} = \mathbf{S}_n$ for such large n and $\mathbf{E} = f^{-1}(\mathbf{S})$. Let $\mathbf{S}_D, \mathbf{S}_A, \mathbf{S}_C$ be the union of all disk pieces, annular pieces and complex pieces of \mathbf{S} , respectively.

If $\mathbf{S}_C = \emptyset$, then each **S**-piece is either of disk type or of annular type. In this case, $P_f \cap J(f)$ is a finite set. This implies that the map f has neither indifferent cycles nor rotation domains. So the orbit of every point $z \in P_f \cap J(f)$ meets a repelling cycle. It follows that the Julia set J(f) has Lebesgue measure zero. The conclusion follows if we set $\Lambda = \emptyset$. If $\mathbf{S}_C \neq \emptyset$, then each \mathbf{S}_C -piece is eventually periodic under the map f_* . We list all f_* -periodic cycles of \mathbf{S}_C -pieces in the following:

$$S_{\nu} \mapsto f_*(S_{\nu}) \mapsto \dots \mapsto f_*^{p_{\nu}-1}(S_{\nu}) \mapsto f_*^{p_{\nu}}(S_{\nu}) = S_{\nu}, \quad 1 \le \nu \le n,$$

where n is the number of f_* -periodic cycles, S_{ν} is a representative in the ν -th cycle and p_{ν} is the period of the piece S_{ν} . For $i \in [1, n]$, let $V_i = S_i$, and U_i be the unique component of $f^{-p_i}(S_i)$ that is parallel to V_i (Recall that 'parallel' means each component of $V_i - U_i$ is either a disk containing at most one point in P, or an annulus in $\overline{\mathbb{C}} - P$ containing a boundary curve of V_i). Then $(f^{p_i}, U_i, V_i), i \in [1, n]$ are the renormalizations of f. (To see this, one should prove that $\deg(f^{p_i}|_{U_i}) \geq 2$, this follows from the fact $\lambda(\partial(V_i), f^{p_i}|_{U_i}) < 1$, with the same argument as Lemma 6.2.4.)

We claim that

The Lebesgue measure of J(f) is zero if and only if for each $i \in [1, n]$, the Lebesgue measure of $J(f^{p_i}|_{U_i})$ is zero.

Let \mathcal{E}_C be the collection of all **E**-pieces that are parallel to the \mathbf{S}_C -pieces. $\mathcal{E}_{AD} = \{D; D \text{ is an } \mathbf{E}\text{-piece contained in } \mathbf{S}_A \cup \mathbf{S}_D \cup (\mathbf{S}_C - \bigcup_{E \in \mathcal{E}_C} E)\}$. Each element $E \in \mathcal{E}_{AD}$ contains at most one point in the postcritical set P_f . Let $\mathcal{E} = \mathcal{E}_C \cup \mathcal{E}_{AD}$. For each $E \in \mathcal{E}$, the boundary of E is contained in the Fatou set F(f).

Notice that $J(f) \subset \bigcap_{k>0} f^{-k}(\bigcup_{E \in \mathcal{E}} E)$, we can define an itinerary map by:

iter :
$$\begin{cases} J(f) \to \mathcal{E}^{\mathbb{N}}, \\ z \mapsto (E_0(z), E_1(z), E_2(z), \cdots). \end{cases}$$

where $E_k(z)$ is the unique element in \mathcal{E} that contains $f^k(z)$.

For simplicity, we denote $J_i = J(f^{p_i}|_{U_i})$ for $i \in [1, n]$. Given a point $z \in J(f)$ with itinerary $iter(z) = (E_0(z), E_1(z), E_2(z), \cdots)$, one can verify that $z \in \bigcup_{k\geq 0} f^{-k}(J_1\cup\cdots\cup J_n)$ if and only if there is an integer N (depending on z) such that for all $k \geq N$, $E_k(z) \in \mathcal{E}_C$. Moreover, $\bigcup_{k\geq 0} f^{-k}(J_1\cup\cdots\cup J_n)$ contains all possible parabolic cycles, Cremer cycles and the boundaries of rotation domains, together with their preimages.

This implies that if $z \in J(f) - \bigcup_{k\geq 0} f^{-k}(J_1 \cup \cdots \cup J_n)$, then there exists a sequence of integers $\{n_j; j \geq 1\}$ such that $E_{n_j} \in \mathcal{E}_{AD}$ for all $j \geq 0$. We consider the sequence $\{F_j(z); j \geq 1\}$, where $F_j(z) = E_{n_j}(z)$. It contains a subsequence $\{F_{j_i}(z); i \geq 1\}$ that satisfies either of the following three cases:

1. $F_{j_i}(z) \cap P_f = \emptyset$ for all $i \ge 1$.

2. For all $i \ge 1$, $F_{j_i}(z) \cap P_f \ne \emptyset$ and $F_{j_i}(z)$ contains a point in P_f which is contained either in the Fatou set or in the grand orbit of a repelling cycle.

3. For all $i \ge 1$, $F_{j_i}(z) \cap P_f \ne \emptyset$ and $F_{j_i}(z)$ contains a point in P_f whose orbit accumulates at $P'_f \cap J(f)$.

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 174 THEOREMS

In the first two cases, one may easily check that $\limsup d(f^n(z), P_f) > 0$.

In the last case, the set $\{F_{j_i}(z); i \geq 1\}$ can be rewritten as $\{E_1, \dots, E_m\}$, which is a finite subset of \mathcal{E}_{AD} . Since each E_k is contained either in a disk component of $\mathbf{S}_C - \bigcup_{E \in \mathcal{E}_C} E$ or in a \mathbf{S}_D -piece, there is an integer M > 0 such that $f^{-M}(E_1 \cup \cdots \cup E_m) \cap P_f = \emptyset$. If $\limsup d(f^n(z), P_f) = 0$, then there exists a sequence of integers $\{\ell_j\}$ such that $d(f^{\ell_j}(z), (E_1 \cup \cdots \cup E_m) \cap P_f) \to 0$ as $j \to \infty$. It follows that $f^{\ell_j - M}(z) \in f^{-M}(E_1 \cup \cdots \cup E_m)$ for all large j. Since the boundary of each component of $f^{-M}(E_1 \cup \cdots \cup E_m)$ is contained in the Fatou set F(f), there is an integer $\varepsilon(z) > 0$ such that $d(f^{\ell_j - M}(z), P_f) \ge \varepsilon(z)$ for all large j, which is contradiction. So in this case, we also have $\limsup d(f^n(z), P_f) > 0$.

Thus, for any $z \in J(f) - \bigcup_{k \ge 0} f^{-k} (J_1 \cup \cdots \cup J_n)$, we have

 $\limsup d(f^n(z), P_f) > 0.$

It follows from Theorem 6.9.3 that the Lebesgue measure of $J(f) - \bigcup_{k\geq 0} f^{-k}(J_1 \cup \cdots \cup J_n)$ is zero. This means Leb(J(f)) = 0 if and only if for each $k \in \Lambda$, $Leb(J_k) = 0$.

Now we set Λ as the indices $i \in \{1, ..., n\}$ such that the renormalization (f^{p_i}, U_i, V_i) is not of annular type.

1. Notice that the Julia set of an annular type renormalization is a quasicircle, whose Lebesgue measure is zero. It follows that Leb(J(f)) = 0 if and only if for each $k \in \Lambda$, $Leb(J_k) = 0$.

2. Suppose that J(f) carries an invariant line field. That is, there is a measurable Beltrami differential μ supported on a positive measure subset Eof J(f) such that $f^*\mu = \mu$ a.e, and $|\mu| = 1$ on E. Let $\mu_k = \mu|_{J_k}$ for $k \in \Lambda$. It follows from 1 that there exists $\ell \in \Lambda$ such that $Leb(J_\ell \cap E) > 0$, then μ_ℓ is an invariant line field for $f^{p_\ell}|_{U_\ell}$ since $(f^{p_\ell}|_{U_\ell})^*\mu_\ell = \mu_\ell$. On the other hand, suppose that μ_ℓ is an invariant line field for $f^{p_\ell}|_{U_\ell}$, then the Beltrami differential defined by $\mu = (f^k)^*\mu_\ell$ on $f^{-k}(J_\ell), k \geq 1$ is an invariant line field for f.

Remark 6.9.3. (Hyperbolic rational maps) It follows from Theorem 6.9.1 and Theorem 6.9.2 that: For every hyperbolic rational map f, either

• each Julia component is a single point or a quasicircle, or

• It admits finitely many rational maps f_1, \dots, f_n as renormalizations, and for each $1 \leq i \leq n, 3 \leq \#P_{f_i} < \infty$.

Let f be a rational map of degree d that satisfies the condition of Theorem 6.9.2. Here is a question concerning the number $#\Lambda$ of non annular type renormalizations of f, posed by Cui Guizhen and Tan Lei:

Question 6.9.1. (Cui-Tan) Is there a constant C = C(d) depending on d, such that $\#\Lambda \leq C(d)$?
The answer is yes if f is a polynomial. In fact, we can take C(d) = d - 1 in the polynomial case. However, for general rational maps, we don't know much.

6.9.3 Herman-Siegel renormalization

In [Sh1], Shishikura developed a surgery which transfers a rational map with Herman rings into finitely many rational maps with Siegel disks. These resulting maps can be considered as the renormalizations of the original map. However, we will see in the following that this kind of renormalization does not fit our definition in Section 6.9.2.

To begin with, we restate the 'Herman-Siegel surgery' following Shishikura. Let f be a rational map with Herman rings. For our purpose, we assume that $P_f \cap J(f)$ is contained in finitely many Julia components (This assumption enables us to obtain a stable multicurve).

Let \mathcal{A} be the collection of all Herman rings of f. For each $A \in \mathcal{A}$, we choose an analytic curve $\gamma_A \subset A$ such that $\gamma_A \cap P_f = \emptyset$ and $f(\gamma_A) = \gamma_{f(A)}$. Let $\Gamma_0 = \{\gamma_A; A \in \mathcal{A}\}, P = P_f \cup \bigcup_{A \in \mathcal{A}} \overline{A}$. By the same argument as in Lemma 6.2.1, we can use Γ_0 to generate a (f, P)-stable multicurve Γ such that:

1. For any $\gamma \in \Gamma$, $f(\gamma) \in \Gamma \cup \Gamma_0$.

2. Γ represents all homotopy classes of non-peripheral curves of $\bigcup_{k\geq 1} f^{-k}(\Gamma_0) - \Gamma_0$ in $\overline{\mathbb{C}} - P$.

Here, we follow the notations in Section 6.2. Recall that \mathcal{S} is the set of all closures of connected components of $\overline{\mathbb{C}} - \cup (\Gamma \cup \Gamma_0)$, \mathcal{E} is the set of all closures of connected components of $\overline{\mathbb{C}} - \cup f^{-1}(\Gamma \cup \Gamma_0)$. For each $S \in \mathcal{S}$, let $E_S \in \mathcal{E}$ be the unique \mathcal{E} -piece that is parallel to S. We define a map $f_* : \mathcal{S} \to \mathcal{S}$ by $f_*(S) = f(E_S)$. Since there are finitely many \mathcal{S} -pieces, every \mathcal{S} -piece is eventually periodic under the map f_* .

We list all periodic cycles of \mathcal{S} -pieces in the following:

$$S_{\nu} \mapsto f_*(S_{\nu}) \mapsto \dots \mapsto f_*^{p_{\nu}-1}(S_{\nu}) \mapsto f_*^{p_{\nu}}(S_{\nu}) = S_{\nu}, \quad 1 \le \nu \le n,$$

where S_{ν} is a representative of the ν -th cycle and p_{ν} is the period of S_{ν} .

For $i \in [1, n]$, let $V_i = S_i$ and U_i be the unique component of $f^{-p_i}(S_i)$ that is contained in S_i and parallel to S_i . The triple (f^{p_i}, U_i, V_i) can be considered as a renormalization of f. In general, U_i is not contained in the interior of V_i (For example, if there is a boundary curve $\gamma \in \partial(V_i)$ such that $\gamma \in \Gamma_0$, then γ is necessarily a boundary curve of U_i). For this reason, we call (f^{p_i}, U_i, V_i) a Herman-Siegel (HS for short) renormalization of f.

We should show that $\deg(f^{p_i}|_{U_i}) \ge 2$ for all $i \in [1, n]$. If $\deg(f^{p_i}|_{U_i}) = 1$ for some $i \in [1, n]$, then $\#\partial(U_i) = \#\partial(V_i)$. There are two possibilities:

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 176 THEOREMS

a). $\partial(V_i) - \Gamma_0 \neq \emptyset$. In this case, for each curve $\gamma \in \partial(V_i) - \Gamma_0$, $(f^{p_i}|_{U_i})^{-1}(\gamma)$ is a curve in $\partial(U_i) - \Gamma_0$. Conversely, each curve $\alpha \in \partial(U_i) - \Gamma_0$ is homotopic to a curve in $\partial(V_i) - \Gamma_0$. Thus $\partial(V_i) - \Gamma_0$ contains a Levy cycle of f^{p_i} . But this contradicts Theorem 6.4.2.

b). $\partial(V_i) - \Gamma_0 = \emptyset$. In this case, $U_i = V_i$ and $f(U_i) = U_i$. This implies that U_i is contained in the Fatou set of f, which is again a contradiction.

So in either case, $\deg(f^{p_i}|_{U_i}) \ge 2$.

The filled Julia set of the HS renormalization (f^{p_i}, U_i, V_i) is defined by $K(f^{p_i}|_{U_i}) = \bigcap_{k \ge 0} (f^{p_i}|_{U_i})^{-k}(U_i)$, and the Julia set is defined by $J(f^{p_i}|_{U_i}) = K(f^{p_i}|_{U_i}) \cap J(f)$ (Notice that $\partial K(f^{p_i}|_{U_i})$ is not a reasonable definition of the Julia set because $\partial K(f^{p_i}|_{U_i})$ may contain a curve in the Herman ring of f). One may check that $K(f^{p_i}|_{U_i})$ is connected. Moreover, if $\partial(V_i) \cap \Gamma_0 = \emptyset$, then $K(f^{p_i}|_{U_i})$ is contained in the interior of V_i and $J(f^{p_i}|_{U_i}) = \partial K(f^{p_i}|_{U_i})$.

We say (f^{p_i}, U_i, V_i) is hybrid equivalent to a rational map R, if there is a quaiconformal map ϕ defined in a neighborhood N of $K(f^{p_i}|_{U_i})$ such that $N \subset U_i, \ \overline{\partial}\phi = 0$ on $K(f^{p_i}|_{U_i})$ and $\phi \circ f^{p_i}|_N = R \circ \phi$. Notice that ∂N may intersect with $\partial K(f^{p_i}|_{U_i})$.

Theorem 6.9.4. (HS Renormalization) Let f be a rational map with Herman rings, assume that $P_f \cap J(f)$ is contained in finitely many Julia components. Let $(f^{p_i}, U_i, V_i), i \in [1, n]$ be all the HS renormalizations defined as above. Then

1. For each $i \in [1, n]$, the HS renormalization (f^{p_i}, U_i, V_i) is hybrid equivalent to a rational map R_i of degree $\deg(f^{p_i}|_{U_i})$ which is postcritically finite outside $\phi(K(f^{p_i}|_{U_i}))$. Here ϕ is the hybrid conjugacy. Such R_i is unique up to Möbius conjugation.

2. The Julia set J(f) has zero Lebesgue measure (resp. carries no invariant line fields) if and only if for each $i \in [1, n]$, the Julia set $J(f^{p_i}|_{U_i})$ has zero Lebesgue measure(resp. carries no invariant line fields).

The proof of the first statement follows from the same line as Theorem 6.9.1, the proof of the second statement is essentially the same as the proof of Theorem 6.9.2. We omit the details here.

6.9.4 Q.c-equivalence vs Möbius conjugation

Lemma 6.9.1. (Q.c-equivalence implies q.c-conjugacy) Let (f, P) and (g, Q) be two non-parabolic rational maps, and $J(f) \neq \overline{\mathbb{C}}$. If (f, P) and (g, Q) are q.c-equivalent via a pair of q.c maps (ϕ_0, ϕ_1) , then they are q.c conjugate. That is, there is a quasiconformal map ϕ , holomorphic in the Fatou set F(f), such that $\phi f = g\phi$.

Proof. By the definition of q.c-equivalence, ϕ_0 and ϕ_1 are holomorphic and identical in the union of all rotation domains R of f (if any) and a neighborhood N_A of all attracting cycles in P'_f (if any).

If f has a superattracting cycle $\langle z_0, z_1, \cdots, z_{p-1} \rangle \subset P_f - P'_f$, then we can modify ϕ_0 and ϕ_1 such that they are holomorphic and identical near these superattracting cycles. The modification is as follows:

First, notice that for any $\zeta \in \langle z_0, z_1, \cdots, z_{p-1} \rangle$, $\phi_0(\zeta)(=\phi_1(\zeta))$ is a superattracting point of g. We can choose a neighborhood U_{ζ} of ζ (resp. $V_{\phi_0(\zeta)}$ of $\phi_0(\zeta)$), a Böttcher coordinate $B^f_{\zeta} : U_{\zeta} \to \mathbb{D}$ (resp. $B^g_{\phi_0(\zeta)} : V_{\phi_0(\zeta)} \to \mathbb{D}$), such that the following diagram commutes:

$$\begin{array}{c|c} U_{\zeta} & \xrightarrow{B_{\zeta}^{f}} \mathbb{D} \overset{B_{\phi_{0}(\zeta)}^{g}}{\longrightarrow} V_{\phi_{0}(\zeta)} \\ f & \downarrow & \downarrow_{z \mapsto z^{d_{\zeta}}} & \downarrow^{g} \\ U_{f(\zeta)} & \xrightarrow{B_{f(\zeta)}^{f}} \mathbb{D} \overset{s}{\underset{B_{\phi_{0}(f(\zeta))}}{\longrightarrow}} V_{\phi_{0}(f(\zeta))} \end{array}$$

where d_{ζ} is the local degree of f at ζ . By a suitable choice of the neighborhoods U_{ζ} and a suitable choice of the Böttcher coordinates, we can modify ϕ_0, ϕ_1 such that $\phi_0|_{U_{\zeta}} = \phi_1|_{U_{\zeta}} = (B^g_{\phi_0(\zeta)})^{-1} \circ B^f_{\zeta}$. A suitable modification elsewhere guarantees $\phi_0 f = g\phi_1$.

In this way, ϕ_0 and ϕ_1 can be made holomorphic in a neighborhood N_{SA} of all superattracting cycles in $P_f - P'_f$ (if any). Then we construct a sequence of q.c maps $\{\phi_n; n \geq 0\}$ by $\phi_n f = g\phi_{n+1}$ so that ϕ_n is isotopic to ϕ_{n+1} rel $f^{-n}(P \cup N_A \cup N_{SA})$. The sequence ϕ_n has a unique limit ϕ , which is holomorphic in $\bigcup_{n\geq 0} f^{-n}(R \cup N_A \cup N_{SA}) = F(f)$, as required. \Box

Lemma 6.9.2. Suppose that the non-parabolic map (f, P) is q.c-equivalent to a rational map (R, Q) with $J(R) \neq \overline{\mathbb{C}}$. Then the rational realization (R, Q)is unique up to Möbius conjugation if and only if J(R) carries no invariant line field.

Proof. It's obvious that if J(R) carries an invariant line field, then the rational realization (R, Q) is not unique up to Möbius conjugation. Conversely, let (R_1, Q_1) be another rational realization of (f, P). Then it follows from Lemma 6.9.1 that (R, Q) and (R_1, Q_1) are q.c conjugate via some q.c map ϕ , which is holomorphic in F(R). This implies $R^*\mu_{\phi} = \mu_{\phi}$ on J(R). Since J(R) carries no invariant line field, $\mu_{\phi} = 0$ almost everywhere on $\overline{\mathbb{C}}$. This implies that ϕ is a Möbius transformation.

Proof of the analytic part of Theorem 6.1.2. It follows from Theorem 6.9.2 Theorem 6.9.4 and Lemma 6.9.2.

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 178 THEOREMS

Q.c-equivalence is a special case of c-equivalence. It follows that the rigidity of c-equivalence always implies the rigidity of q.c-equivalence. On the other hand, c-equivalences between two non-parabolic rational maps without rotation domains can always be promoted to q.c-equivalences, this is because the pair of c-equivalences are holomorphic in a neighborhood of all attracting cycles. For two non-parabolic rational maps with Siegel disks, to the author's knowledge, whether the promotion works depends on the boundary regularity of the Siegel disks.

It's known from Gaofei Zhang [Zh1] that the boundary of every bounded type Siegel disk of a rational map must be a quasicircle containing at least one critical point.

In [Zh2], Zhang showed that given a rational map R with a fixed Siegel disk and postcritically finite outside this Siegel disk, then the Lebesgue measure of the Julia set J(R) is zero. He told the author that his method also works for more general case:

Given a rational map R with Siegel disks, all with bounded type rotation numbers and postcritically finite outside these Siegel disks, then the Julia set J(R) has zero Lebesgue measure.

Based on Zhang's Theorems and Theorems 6.9.4, we have

Theorem 6.9.5. (Rigidity) Let (f, P) be a non-parabolic rational map with rotation domains (i.e. Siegel disk or Herman ring), and the rotation number of each rotation domain is of bounded type, then the Lebesgue measure of the Julia set J(f) is zero. Thus if two such rational maps are c-equivalent, then they are conformally conjugate.

6.10 Applications

Besides of the independent interest, Theorem 6.1.2 enables us to extend Thurston's Theorem beyond postcritically finite cases, and give characterizations of hyperbolic rational maps, and rational maps with rotation domains (Siegel disks and Herman rings).

6.10.1 Characterization of hyperbolic rational maps

Theorem 6.10.1. (Cui-Tan [CT1], Jiang-Zhang [JZ]) Let (f, P) be a non-parabolic map without rotation domains. Then (f, P) is c-equivalent to a rational map (R, Q) if and only if (f, P) has no Thurston obstructions. The rational map (R, Q) is unique up to Möbius conjugation.

Remark 6.10.1. In fact, the family of non-parabolic rational maps without rotation domains is slightly larger than the family of hyperbolic rational maps. It is conjectured to be dense in the parameter space.

Proof. The proof here in fact follows from Cui-Tan's original one [CT1]. If (f, P) is a disk covering or annular covering, then the proof is done by Theorem 6.6.5 and Theorem 6.7.2. Else, by the Decomposition Theorem, (f, P) admits finitely many decompositions $(h_k, P_k), k \in [1, n]$ along some stable multicurve Γ . Since (f, P) has no rotation domains, all of these resulting maps are Thurston maps. If (f, P) has no Thurston obstructions, then each resulting map has no Thurston obstructions and the signature of its orbifold is not (2, 2, 2, 2) (Lemma 6.2.4). By Marked Thurston Theorem, all (h_k, P_k) have rational realizations, so does (f, P). The uniqueness of the rational realization of (f, P) follows from Lemma 6.9.2 the fact that any such rational realization has a Julia set of zero Lebesgue measure.

6.10.2 Characterization of rational maps with Siegel disks

For rational maps with Siegel disks, Zhang [Zh2] proved the following:

Theorem 6.10.2. (Zhang) Let (f, P) be a non-parabolic map, with only one rotation disk cycle which is of period one and has rotation number of bounded type, and without rotation annulus. Then (f, P) is c-equivalent to a rational map (R, Q) if and only if (f, P) has no Thurston obstructions. Moreover, the Lebesgue measure of the Julia set J(R) is zero, and (R, Q) is unique up to Möbius conjugation.

Zhang's Theorem requires that the non-parabolic map has only one rotation disk, and it is postcritically finite outside the rotation disk. It's possible to generalize Zhang's Theorem to a more general setting without the assumptions of

- the postcritical finiteness outside the rotation disks, and
- the number of rotation disk cycles.

But for compensation, we usually need a separate condition for these rotation disks.

For a non-parabolic map (f, P) with $n_A(f) > 0$, let \mathcal{A} be the union of all attracting cycles. The filled Julia set K_f of f is defined by

$$K_f = \{ z \in \overline{\mathbb{C}}; \limsup d(f^n(z), \mathcal{A}) > 0 \},\$$

where $d(\cdot, \cdot)$ is the spherical distance. K_f is a compact subset of $\overline{\mathbb{C}}$.

Theorem 6.10.3. Let (f, P) be a non-parabolic map, with $n_A(f) > 0$, $n_{RD}(f) > 0$ and $n_{RA}(f) = 0$. Suppose that the rotation numbers of all rotation disk cycles are of bounded type, and all rotation disks are contained in different components of the filled Julia set K_f . Then (f, P) is c-equivalent to a rational map (R, Q) if and only if (f, P) has no Thurston obstructions. Moreover, the Lebesgue measure of the Julia set J(R) is zero, and (R, Q) is unique up to Möbius conjugation.

Proof. By the Decomposition Theorem, (f, P) admits finitely many decompositions (h_k, P_k) , $k \in [1, n]$. Since different rotation disks are contained in different components of the filled Julia set K_f , each resulting Siegel map has only one rotation disk cycle, of period one and with bounded type rotation number (Thus the number of these Siegel maps is $n_{RD}(f)$). Then the conclusion follows from the Decomposition Theorem, Thurston's Theorem, Zhang's Theorem and Theorem 6.9.2.

6.10.3 Characterization of rational maps with Herman rings

As another application, we can give a characterization of a class of rational maps with Herman rings, as follows:

Theorem 6.10.4. Let (f, P) be a non-parabolic map, with only one rotation annulus cycle which is of period one and has rotation number of bounded type, and without rotation disks. Then (f, P) is c-equivalent to a rational map (R, Q) if and only if (f, P) has no Thurston obstructions. Moreover, the Lebesgue measure of the Julia set J(R) is zero, and (R, Q) is unique up to Möbius conjugation.

Proof. By the decomposition procedure, (f, P) admits finitely many decompositions $(h_k, P_k), k \in [1, n]$. Two are Siegel maps and the rest are Thurston maps. The theorem follows from the Decomposition Theorem, Thurston's Theorem, Zhang's Theorem and Theorem 6.9.4.

We can further generalize Thurston's Theorem to rational maps with many Herman ring cycles but satisfying 'sperate configuration'.

Let (f, P) be a Herman map, \mathcal{A} be the collection of all rotation annuli of f. For each $A \in \mathcal{A}$, we associate an analytic curve γ_A such that $\gamma_A \cap P_f = \emptyset$ and $f(\gamma_A) = \gamma_{f(A)}$. Let $\Gamma_f = \{\gamma_A; A \in \mathcal{A}\}$. We call Γ_f a (f, P)-invariant curve system.

Definition 6.10.1. (Sperate configuration) We say that a Herman map (f, P) satisfies 'sperate configuration', if there is a (f, P)-invariant curve system Γ_f such that for any two different rotation annuli A_{α} and A_{β} , there is a curve $\gamma_{\alpha\beta} \in \bigcup_{k\geq 1} f^{-k}(\Gamma_f)$ that separates A_{α} and A_{β} .

Theorem 6.10.5. Let (f, P) be a Herman map without rotation disk cycle and the rotation numbers of all rotation annuli cycles are of bounded type. Suppose (f, P) satisfies sperate configuration. Then (f, P) is c-equivalent to a rational map (R, Q) if and only if (f, P) has no Thurston obstructions. Moreover, the Lebesgue measure of the Julia set J(R) is zero, and (R, Q) is unique up to Möbius conjugation.

Proof. By the decomposition procedure, (f, P) admits finitely many decompositions $(h_k, P_k), k \in [1, n]$. The sperate configuration implies that each Siegel map has only one rotation disk cycle, of period one and with bounded type rotation number. The theorem then follows from the Decomposition Theorem, Marked Thurston Theorem, Zhang's Theorem and Theorem 6.9.4.

6.11 No Thurston obstructions vs rational realization

In this section, we will exhibit many examples of non-parabolic maps which have no Thurston obstructions but are not c-equivalent to rational maps. By Theorem 6.10.1, such non-parabolic map necessarily has at least one cycle of rotation domain. We first construct a Siegel map by mating two quadratic Siegel polynomials and show that it has no Thurston obstructions but is not c-equivalent to a rational map. Then we will use it to construct more such non-parabolic maps by surgery.

We begin with the definition of the mating of two quadratic polynomials. The notations here follow from [YaZ]. Let \bigcirc denote the complex plane \mathbb{C} compactified by adjoining a circle of directions at infinity $\{\infty \cdot e^{2\pi i t}; t \in \mathbb{R}/\mathbb{Z}\}$ with the natural topology. Each $f_i = z^2 + c_i$ extends continuously to a copy of \bigcirc_i , acting as the squaring map $z \mapsto z^2$ on the circle at infinity. Gluing the disks \bigcirc_i together via the equivalence relation \sim_{∞} identifying the point $\infty \cdot e^{2\pi i t} \in \bigcirc_1$ with $\infty \cdot e^{-2\pi i t} \in \bigcirc_2$, we obtain a 2-sphere $(\bigcirc_1 \sqcup \bigcirc_2)/\sim_{\infty}$. The well-defined map $f_1 \sqcup f_2$ on this sphere given by f_i on \bigcirc_i is a degree 2 branched covering of the sphere with an invariant equator. We shall refer to this map as the formal mating of f_1 and f_2 .

For any quadratic polynomial $z \mapsto e^{2\pi i\theta} z + z^2$, we can conjugate it to the normal form

$$f_{\theta}(z) = z^2 + c_{\theta}, \ c_{\theta} = \frac{e^{2\pi i\theta}}{2} \Big(1 - \frac{e^{2\pi i\theta}}{2} \Big).$$

We may assume that θ is an irrational number of bounded type. It's known that f_{θ} has a Siegel disk whose boundary is a quasicircle containing the critical point 0. The mating $f = f_{\theta} \sqcup f_{-\theta}$ is a Siegel map. Let D_{θ} and $D_{-\theta}$ be two rotation disks of f, where D_{θ} (resp. $D_{-\theta}$) inherits the complex structure of the Siegel disk of f_{θ} (resp. $f_{-\theta}$).

Lemma 6.11.1. The Siegel map $(f, P) := (f_{\theta} \sqcup f_{-\theta}, \overline{D_{\theta}} \cup \overline{D_{-\theta}})$ has no Thurston obstructions but is not c-equivalent to a rational map.

Remark 6.11.1. One the other hand, given two irrational numbers θ_1 and θ_2 of bounded type, with $\theta_1 + \theta_2 \neq 0 \mod \mathbb{Z}$, Yampolsky and Zakeri [YaZ] proved that the mating $f_{\theta_1} \sqcup f_{\theta_2}$ is c-equivalent to a unique quadratic rational map up to Möbius conjugation.

Proof. Let Γ be a (f, P)-stable multicurve. If Γ is non-empty, then Γ necessarily consists of one curve $\gamma \subset \overline{\mathbb{C}} - \overline{D_{\theta}} \cup \overline{D_{-\theta}}$. Consider the annulus A bounded by γ and the boundary of the rotation disk D_{θ} . The preimage $f^{-1}(A)$ is an annulus B, and deg $(f : B \to A) = 2$. One boundary curve of B is a figure eight curve while the other boundary curve δ of B is the preimage of γ , and deg $(f : \delta \to \gamma) = 2$. Since δ is homotopic to γ in $\overline{\mathbb{C}} - P$, we have that $\lambda(\Gamma, f) = \frac{1}{2}$. Thus (f, P) has no Thurston obstructions.

If (f, P) is c-equivalent to a rational map, say (R, Q). Then (R, Q) has three fixed points. Two fixed points are the centers of Siegel disks, with multipliers $e^{2\pi i\theta}$ and $e^{-2\pi i\theta}$. The third fixed point is necessarily a repelling fixed point, since a quadratic rational map has at most two non-repelling cycles. We denote the multiplier of the repelling fixed point by λ . Then by holomorphic index formula ([M1]),

$$\frac{1}{1 - e^{2\pi i\theta}} + \frac{1}{1 - e^{-2\pi i\theta}} + \frac{1}{1 - \lambda} = 1.$$

It follows that $\lambda = \infty$, which is a contradiction.

In the following, we will use the Siegel map $(f_{\theta} \sqcup f_{-\theta}, \overline{D_{\theta}} \cup \overline{D_{-\theta}})$ to produce more non-parabolic maps without Thurston obstructions but not c-equivalent to rational maps.

Theorem 6.11.1. Given nonnegative integers n_A , n_{RD} , n_{RA} , d satisfying

 $n_A + n_{RD} + 2n_{RA} \le 2d - 2, \ 1 \le n_{RA} \le d - 2, \ n_{RD} + n_{RA} \ge 2.$

There exists a non-parabolic map (f, P) of degree d, such that

1. $n_A(f) = n_A, n_{RD}(f) = n_{RD}, n_{RA}(f) = n_{RA}$, and the rotation number of each rotation cycle is of bounded type.

- 2. (f, P) has no Thurston obstructions.
- 3. (f, P) is not c-equivalent to a rational map.

The idea of the proof of Theorem 6.11.1 is to glue some well-chosen rational maps with the Siegel map (f, P) defined as in Lemma 6.11.1. To obtain these rational maps as candidates, we need a result of Shishikura ([Sh1]):

Theorem 6.11.2. (Shishikura) Given nonnegative integers n_{AB} , n_{PB} , n_{SD} , n_{HR} , n_{cremer} and d satisfying

 $n_{AB} + n_{PB} + n_{SD} + 2n_{HR} + n_{cremer} \le 2d - 2, \ n_{HR} \le d - 2,$

there exists a rational function f of degree d, such that the numbers of attracting cycles, parabolic cycles, Siegel disk cycles, Herman ring cycles and Cremer cycles are n_{AB} , n_{PB} , n_{SD} , n_{HR} and n_{cremer} , respectively.

Proof of Theorem 6.11.1

We first consider the case $n_{RD} \geq 1$. Then Shishikura's Theorem and quasiconformal surgery guarantee the existence of a non-parabolic rational map g with $(n_A(g), n_{RD}(g), n_{RA}(g), \deg(g)) = (n_A, n_{RD}, n_{RA} - 1, d - 1)$. Choose a Siegel disk cycle $D_0 \mapsto D_1 \mapsto \cdots \mapsto D_p = D_0$ of g. For each D_i we choose an analytic curve $\gamma_i \subset D_i$ such that $g(\gamma_i) = \gamma_{i+1}$ and the disk $\Delta_i \subset D_i$ bounded by γ_i contains at most one point in the postcritical set P_g (Notice that the only possible point in $\Delta_i \cap P_g$ is necessarily the center of the Siegel disk D_i). Let θ be the rotation number of the Siegel disks cycle.

We consider the Siegel map $(S_{\theta}, P_{\theta}) := (f_{\theta} \sqcup f_{-\theta}, \overline{D_{\theta}} \cup \overline{D_{-\theta}})$. We can view it as the composition of p maps $S_{\theta} = f_{p-1} \circ \cdots \circ f_0$ with $f_i : \overline{\mathbb{C}}_i \to \overline{\mathbb{C}}_{i+1}, 0 \leq 1 \leq p$ and $f_0 = S_{\theta}, f_1 = \cdots = f_{p-1} = id, \overline{\mathbb{C}}_p = \overline{\mathbb{C}}_0$. Let δ_0 be a S_{θ} -invariant curve in $D_{-\theta}$ and $\delta_i = f_i \circ \cdots \circ f_0(\delta_0)$ for $1 \leq i \leq p-1$. We cut $\Delta_0 \cup \cdots \cup \Delta_{p-1}$ off for g and cut the disk $U_i \subset f_i \circ \cdots \circ f_0(D_{-\theta})$ bounded by δ_i for $f_i, 0 \leq i \leq p-1$. Then we glue each $f_i|_{\overline{\mathbb{C}}_i - U_i}$ with $g|_{\overline{\mathbb{C}} - \Delta_0 \cup \cdots \cup \Delta_{p-1}}$ along the boundary curves δ_i and γ_i . We can assume that

a). The gluing procedure preserves the complex structure of the rotation domains.

b). The center of the rotation disk $f_i \circ \cdots \circ f_0(D_{-\theta})$ replace the center of the Siegel disk D_i .

In this way, we get a non-parabolic map f with $(n_A(f), n_{RD}(f), n_{RA}(f), \deg(f)) = (n_A(g), n_{RD}(g), n_{RA}(g) + 1, \deg(g) + 1) = (n_A, n_{RD}, n_{RA}, d)$, as required.

Now we consider the case $n_{RD} = 0$. In this case, $n_{RA} \ge 2$. First, it follows from Shishikura's Theorem that there is a non-parabolic rational map g with

$$(n_A(g), n_{RD}(g), n_{RA}(g), \deg(g)) = \begin{cases} (n_A, 1, n_{RA} - 2, d - 2), & \text{if } n_A = 0, 1, \\ (n_A - 1, 1, n_{RA} - 2, d - 2), & \text{if } n_A \ge 2. \end{cases}$$

Let θ be the rotation number of the Siegel disk cycle of g. We mate the Siegel map $(S_{\theta}, P_{\theta}) = (f_{\theta} \sqcup f_{-\theta}, \overline{D_{\theta}} \cup \overline{D_{-\theta}})$ with the quadratic Siegel polynomial $f_{-\theta}$

CHAPTER 6. DECOMPOSITION THEOREM AND THURSTON-TYPE 184 THEOREMS

in the following way: Cut the S_{θ} -fixed disk $\Delta_{\theta} \in D_{\theta}$ off for S_{θ} and cut the $f_{-\theta}$ -fixed disk $U_{\theta} \in D_{-\theta}$ off for $f_{-\theta}$, then we glue $S_{\theta}|_{\overline{\mathbb{C}}-\Delta_{\theta}}$ and $f_{-\theta}|_{\overline{\mathbb{C}}-U_{\theta}}$ along the boundary curves $\partial \Delta_{\theta}$ and ∂U_{θ} . We get a Herman map (H_{θ}, Q_{θ}) . This Herman map has one fixed rotation disk with rotation number $-\theta$ and one rotation annulus with rotation number θ . Moreover, $\deg(H_{\theta}) = \deg(S_{\theta}) + \deg(f_{-\theta}) - 1 = 3$. By performing the same mating procedure as above, we can mate (g, P_g) with (H_{θ}, Q_{θ}) and obtain a non-parabolic map (f, P) with $(n_A(f), n_{RD}(f), n_{RA}(f), \deg(f)) = (n_A(g), n_{RD}(g) + n_{RD}(H_{\theta}) - 2, n_{RA}(g) + n_{RA}(H_{\theta}) + 1, \deg(g) + \deg(H_{\theta}) - 1) = (n_A(g), 0, n_{RA}, d)$. If $n_A = 0, 1$, then $n_A(g) = n_A$, then map (f, P) is as required; if $n_A \geq 2$, then $n_A(f) = n_A - 1$. In this case, notice that the superattracting fixed point ∞ of $f_{-\theta}$ descends to a 'superattracting' cycle of (f, P), we can change this superattracting cycle of (f, P) to be locally holomorphic and attracting whose multiplier satisfies $0 < |\lambda| < 1$ by quasiconformal surgery and get the required map.

To finish, we will show that in either case, (f, P) has no Thurston obstructions but is not c-equivalent to a rational map. By Decomposition Theorem, there is a (f, P)-stable multicurve Γ and finitely many Siegel maps or Thurston maps $(h_k, P_k), k \in [1, n]$ whose combinatorics and rational realizations dominate the original one. The construction of (f, P) guarantees that

• The (f, P)-stable multicurve Γ is in fact a stable multicurve of $g|_{\overline{\mathbb{C}}-\Delta_0\cup\cdots\cup\Delta_{p-1}}$. So it follows from Marked McMullen Theorem that $\lambda(\Gamma, f) = \lambda(\Gamma, g) < 1$.

• One of the resulting maps of $(h_k, P_k), k \in [1, n]$ is the Siegel map $(S_\theta, P_\theta) = (f_\theta \sqcup f_{-\theta}, \overline{D_\theta} \cup \overline{D_{-\theta}})$ while the rest resulting maps all have rational realizations.

Then by the Decomposition Theorem, (f, P) has no Thurston obstructions but is not c-equivalent to a rational map.

Bibliography

- [Ahl] L. V. Ahlfors. Lectures on quasiconformal mappings, volume 38 of University Lecture Series, Amer. Math. Soc., 2006. (Cited on pages 40 and 90.)
- [AY] M. Aspenberg and M. Yampolsky. Mating non-renormalisable quadratic polynomials, Comm. Math. Phys. 287 (2009), 1-40. (Cited on page 33.)
- [Ast] K. Astala. Area distortion of quasiconformal mappings. Acta Math, 173:37-60. (Cited on page 17.)
- [BFH] B. Bielefield, Y. Fisher, J. H. Hubbard, The classification of critically preperiodic polynomials as dynamical systems. Jour. Amer. Math. Soc. 5, 721-762 (1992) (Cited on pages 5 and 108.)
- [BH] B. Branner and J. H. Hubbard, *The iteration of cubic polynomials*, Acta Math. 169(1992), 229-325. (Cited on pages 2, 51, 80 and 93.)
- [Bro] D. Brown, Thurston equivalence without postcritical finiteness for a family of polynomial and exponential mappings. Manuscript (Cited on pages 5 and 108.)
- [BCT] X. Buff, G. Cui, L. Tan. Teichmüller spaces and holomorphic dynamics. Handbook of Teichmüller theory, Vol. III, ed. Athanase Papadopoulos, EMS, 2011. (Cited on page 112.)
- [BE] X. Buff and A. Epstein. A parabolic Pommerenke-Levin-Yoccoz inequality. Fundamenta Mathematicae, 72, (2002), 249-289. (Cited on page 11.)
- [CG] L. Carleson and T. Gamelin, Complex Dynamics, Springer-Verlag, New York, 1993. (Cited on page 85.)
- [CJY] L. Carleson, P. Jones, and J.-C. Yoccoz, Julia and John, Bol. Soc. Brasil mat.(N.S.) 25(1994), 1-30. (Cited on pages 2, 51 and 91.)
- [C] G. Cui. Dynamics of rational maps, topology, deformation and bifurcation. Preprint. 2002. (Cited on page 12.)
- [CT1] G. Cui, L. Tan. A characterization of hyperbolic rational maps, Invent math. DOI 10.1007/s00222-010-0281-8 (Cited on pages 5, 6, 12, 108, 112, 134, 146, 151, 153, 164, 166, 167, 178 and 179.)

- [CT2] G. Cui, L. Tan. *Hyperbolic-parabolic deformations of rational maps*. Manuscript in preparation (Cited on pages 5 and 108.)
- [CJ] G. Cui and Y. Jiang. Geometrically finite and semi-rational branched coverings of the two-sphere, Trans. Amer. Math. Soc., to appear (Not cited.)
- [DDI] B. Derrida, L. De Seze, C. Itzykson. Fractal structure of zeros in hierarchinal models, J. Stat. Phys. 33, 559-569 (1983). (Cited on pages 7 and 31.)
- [D1] R. Devaney, Cantor Webs in the Parameter and Dynamical Planes of Rational Maps, Fields Institute Communications, 53(2008), 105-123. (Cited on pages 2, 51 and 100.)
- [D2] R. Devaney, Intertwined Internal Rays in Julia Sets of Rational Maps, Fund. Math. 206(2009), 139-159. (Cited on pages 2, 51, 58 and 61.)
- [DK] R. Devaney and L. Keen, Complex Dynamics: Twenty Five Years After the Appearance of the Mandelbrot Set. American Mathematical Society, Contemporary Math 396, 2006. (Cited on pages 2, 3, 51 and 52.)
- [DLU] R. Devaney, D. Look and D. Uminsky, The Escape Trichotomy for Singularly Perturbed Rational Maps, Indiana University Mathematics Journal 54(2005), 1621-1634. (Cited on pages 2, 51 and 55.)
- [DS] T. C. Dinh and N. Sibony, Sur les endomorphisms holomorphes permutables, Math. Annalen 324 (2002), 33-70. (Cited on page 24.)
- [Dou1] A. Douady, *Disques de Siegel et anneaux de Herman*, Sém. Bourbaki 39 (1986°C1987), 151°C172. (Cited on page 86.)
- [Dou2] A. Douady, Topological entropy of unimodal maps. In: Proceedings of the NATO Adv. Study Inst. on Real and Complex Dynamical Systems. NATO ASI Series, vol. 464 (1993) (Cited on page 5.)
- [Dou3] A. Douady. Systèmes dynamiques holomorphes. Séminaire Bourbaki, Volume 1982-83, exposé no. 599, Astérisque, 105-106, 1983, 39-63. (Not cited.)
- [DH1] A. Douady, J.H. Hubbard, A proof of Thurston's topological characterization of rational functions. Acta Math. 171, 263-297 (1993) (Cited on pages 5 and 107.)

- [DH2] A. Douady and J. H. Hubbard, Etude dynamique des polynômes complexes, Publications mathématiques d'Orsay, 1984. (Cited on pages 2, 18, 36, 51 and 171.)
- [DH3] A. Douady and J. H. Hubbard, On the dynamics of polynomial-like mappings, Ann. Sci. École Norm. Sup. 18(1985), 287-343. (Cited on pages 31 and 83.)
- [Eps] A.L. Epstein, Transersality in holomorphic dynamics (Preliminary Version). (Not cited.)
- [Ere] A. Eremenko, Some functional equations connected with iteration of rational functions, Algebra and Analysis 1 (1989), 102-116 (Russian). Leningrad Math. J. 1 (1990), 905-919 (English). (Cited on page 24.)
- [EL] A. Eremenko, M. Lyubich. The dynamics of analytic transformations, Leningrad Math. J. 1, 563-634 (1990). (Cited on pages 7 and 31.)
- [GJW] F. Gardiner, Y. Jiang, Z. Wang. Holomorphic motions and related topics, Geometry of Riemann Surfaces, London Mathematical Society Lecture Note Series, No. 368, 2010, 156-193 (Cited on page 17.)
- [Gey1] L. Geyer, *Linearizability of saturated polynomials*, Preprint, 2004. (Cited on page 86.)
- [Gey2] L. Geyer, Sharp bounds for the valence of certain harmonic polynomials. Proc. Amer. Math. Soc. 136 (2008), no. 2, 549-555. (Cited on page 108.)
- [G] S. Godillon, Construction de fractions rationnelles à dynamique prescrite, Thesis, Université de Cergy-Pontoise, 2010. (Not cited.)
- [F] O. Foster. Lectures on Riemann Surfaces. Springer-Verlag, New York, Heidelberg, Berlin, 1981. (Cited on page 44.)
- [GS] J. Graczyk and G. Swiatek, *The Real Fatou Conjecture*, Ann. of Math. Studies 144, Princeton Univ. Press, Princeton, NJ (1998). (Cited on pages 2 and 51.)
- [HL] B. Hu, B. Lin. Yang-Lee zeros, Julia sets, and their singularity spectra, Phys. Review A 39, 4789-4796 (1989). (Cited on page 48.)
- [Hu] J. H. Hubbard, Local connectivity of Julia sets and bifurcation loci: three theorems of J.-C. Yoccoz, in Topological Methods in Modern Mathematics, 467-511 (L. R. Goldberg and A. V. Phillips, eds.), Publish or Perish Houston, TX, 1993. (Cited on pages 2, 51 and 80.)

- [HS] J.H. Hubbard, D. Schleicher, The spider algorithm, complex dynamical systems. In: Devaney, R. (ed.) The Mathematics behind the Mandelbrot and Julia Sets. Proceedings of Symposia in Applied Mathematics, vol. 49, pp. 155-180. AMS, Providence (1994) (Cited on pages 5 and 108.)
- [HSS] J.H. Hubbard, D. Schleicher, M. Shishikura, Exponential Thurston maps and limits of quadratic differentials. Jour. Amer. Math. Soc. 22, 77-117 (2009) (Cited on pages 5 and 108.)
- [JZ] Y. Jiang, G. Zhang, Combinatorial characterization of sub-hyperbolic rational maps. Adv. Math. 221, 1990-2018 (2009) (Cited on pages 5, 6, 108, 112 and 178.)
- [KL] J. Kahn, M. Lyubich, Quasi-Additivity Law in Conformal Geometry. Ann. Math. 169(2009), 561-193. (Cited on page 14.)
- [Kiwi] J. Kiwi, Real laminations and the topological dynamics of complex polynomials. Adv. Math. 184(2), 207-267 (2004) (Cited on pages 2, 5, 51 and 108.)
- [Ly] M. Lyubich, Dynamics of quadratic polynomials, I-II, Acta Math. 178(1997), 185-297. (Cited on pages 2 and 51.)
- [LM] M. Lyubich and Y. Minsky, Laminations in holomorphic dynamics, J. Differential Geom. 47(1997), 17-94. (Cited on page 104.)
- [McM1] C. McMullen, Complex Dynamics and Renormalization, Ann. of Math. Studies 135, Princeton Univ. Press, Princeton, NJ, 1994. (Cited on pages 5, 7, 18, 23, 24, 27, 28, 49, 86, 108, 129 and 172.)
- [McM2] C. McMullen. The Mandelbrot set is universal. In: The Mandelbrot set, Theme and Variations, edited by Tan Lei, 1-17, London Math. Soc. Lecture Note Ser., No 274, Cambridge University Press, Cambridge, 2000. (Cited on page 32.)
- [McM3] C. McMullen. Automorphisms of rational maps. In Holomorphic Functions and Moduli I, pages 31-60. Springer-Verlag, 1988. (Cited on page 53.)
- [M1] J. Milnor, Dynamics in One Complex Variable, Vieweg, 1999, 2nd edition, 2000. (Cited on pages 2, 27, 97, 103, 109 and 182.)

- [M2] J. Milnor, Local Connectivity of Julia Sets: Expository Lectures, in The Mandelbrot Set, Theme and Variations (Tan Lei, ed.), London Math. Soc. Lecture Note Series 274, 67-116, Cambridge Univ. Press, Cambridge, 2000. (Cited on pages 2, 3, 51, 52, 78, 80 and 93.)
- [M3] J. Milnor, On Lattes maps, in Dynamics on the Riemann sphere: A Bodil Branner Festschrift, Eds. P.G. Hjorth and C.L. Peterson, European Mathematical Society, 2006, 9-43. (Cited on page 24.)
- [MT] J. Milnor and W. Thurston, On iterated maps of the interval, in Dynamical systems (College Park, MD, 1986-87), 465-563, Lecture Notes in Math., 1342, Springer, Berlin, 1988. (Cited on page 108.)
- [O] A. Osbaldestin. 1/s-expansion for generalized dimensions in a hierarchical s-state Potts model. J. Phys. A 28, 5951-5962 (1995). (Cited on pages 7, 31 and 48.)
- [P] C. L. Petersen, Local connectivity of some Julia sets containing a circle with an irrational rotation, Acta Math. 177(1996), 163-224. (Cited on pages 5 and 108.)
- [Pi] K. Pilgrim, Combinations of Complex Dynamical Systems. Lecture Notes in Mathematics, vol. 1827. Springer, Berlin (2003). (Not cited.)
- [PT1] K. Pilgrim, L. Tan, Combining rational maps and controlling obstructions. Erg. Th. and Dyn. Sys. 18(1), 221-245 (1998) 27. (Cited on pages 5 and 108.)
- [PT2] K. Pilgrim, L. Tan, Rational maps with disconnected Julia set. Astérisque 261, 349-384 (2000) 28. (Not cited.)
- [PQRTY] W. Peng, W. Qiu, P. Roesch, L. Tan and Y. Yin, A tableau appproach of the KSS nest, Conformal geometry and dynamics(AMS electronic journal), 14(2010), 35-67. (Cited on page 80.)
- [Poi] A. Poirier, On postcritically finite polynomials, part I: critical portraits. Stony Brook IMS preprint 1993/5. math.DS/9305207 29. (Cited on pages 5 and 108.)
- [QG] J. Qiao, J. Gao. Jordan domain and Fatou set concerning diamond-like hierarchical models, Nonlinearity, 40, 119-131(2007). (Cited on pages 7 and 31.)
- [QL] J. Qiao, Y. Li. On connetivity of Julia sets of Yang-Lee zeros, Comm. Math. Phys. 222,319-326 (2001). (Cited on pages 7, 31, 32 and 42.)

- [QY] W. Qiu and Y. Yin, Proof of the Branner-Hubbard conjecture on Cantor Julia sets, Science in China(Series A) 52(2009), 45-65. (Cited on page 80.)
- [Rees1] M. Rees, Realization of matings of polynomials as rational maps of degree two. Manuscript (1986) (Cited on pages 5 and 108.)
- [Rees2] M. Rees, Views of parameter space: topographer and resident, Astérisque 288 (2003) (Cited on pages 5 and 108.)
- [Rees3] M. Rees, Components of degree two hyperbolic rational maps. Invent. Math., 100 (1990), 357-382. (Cited on page 33.)
- [RvS] L. Rempe and S. van Strien, Absence of line fields and Mañé's theorem for non-recurrent transcendental functions. Trans. Amer. Math. Soc. 363 (2011) 203-228. (Cited on page 25.)
- [Ritt] J. F. Ritt, Permutable rational functions, Trans. Amer. Math. Soc. 25 (1923), 398-448. (Cited on page 24.)
- [Ro1] P. Roesch, On local connectivity for the Julia set of rational maps: Newton's famous example, Ann. of Math. 168 (2008), 1-48. (Cited on pages 2, 3, 4, 51, 53 and 80.)
- [Ro2] P. Roesch. On capture zones for the family f(z) = z² + λ/z², in Dynamics on the Riemann sphere: A Bodil Branner Festschrift, Eds. P.G. Hjorth and C.L. Peterson, European Mathematical Society, 2006, 121-129. (Cited on page 36.)
- [Ro3] P. Roesch. Topologie locale des méthodes de Newton cubiques. PhD Thesis, Ecole normale supérieure de Lyon. 1997. (Cited on page 36.)
- [RY] P. Roesch and Y. Yin, Bounded critical Fatou components are Jordan domains, for polynomials. ArXiv 0909.4598v2. (Cited on page 80.)
- [Ru] D. Ruelle, Repellers for real analytic maps. Erg. Th. & Dyn. Sys(1982),2,99-107. (Cited on page 18.)
- [Sh1] M. Shishikura, On the quasiconformal surgery of rational functions, Ann. Sci. École Norm. Sup. 20(1987) 1-29. (Cited on pages 6, 85, 112, 114, 115, 175 and 183.)
- [Sh2] M. Shishikura, Trees associated with the configuration of Herman rings, Ergod. Th. Dynam. Sys. 9(1989), p.543-560. (Not cited.)

- [ST] M. Shishikura, L. Tan, A family of cubic rationalmaps and matings of cubic polynomials. Exp. Math. 9, 29-53 (2000) (Cited on pages 5 and 108.)
- [Slo] Z. Slodkowski. Holomorphic motions and polinomial hulls. Proc. Amer. Math. Soc. 111, 347-355,1991. (Cited on pages 17, 46, 49 and 50.)
- [Smi] S. Smirnov. Dimension of quasicircles. Acta Math. 205, 189-197(9), 2010. (Cited on pages 17 and 49.)
- [Tan1] L. Tan, Matings of quadratic polynomials. Erg. Th and Dyn. Syst. 12, 589-620 (1992) (Cited on pages 5 and 108.)
- [Tan2] L. Tan, Branched coverings and cubic Newton maps. Fund. Math. 154(3), 207-260 (Cited on pages 5 and 108.)
- [TY] L. Tan and Y. Yin, Local connectivity of the Julia set for geometrically finite rational maps, Science in China (Serie A) 39 (1996), 39-47. (Cited on pages 2, 51, 98, 100 and 104.)
- [T] V. Timorin. External boundary of M2, Proceedings of conference dedicated to Milnor's 75th birthday, Proceedings of the Fields Institute 53 (2006), 225-267. (Cited on page 33.)
- [W] X. Wang. Rational maps admitting meromorphic invariant line fields.
 Bull. Aust. Math. Soc. (2009), 80: 454-461. (Cited on page 25.)
- [WQY] X. Wang, W. Qiu, Y. Yin. Local connectivity of Julia set: McMullen maps. Submitted. (Not cited.)
- [WQYQG] X. Wang, W. Qiu, Y. Yin, J. Qiao, J. Gao. Connectivity of the Mandelbrot set for the family of renormalization transformations. Sci China Math. 2010, 53(3): 849-862. (Cited on page 33.)
- [Wh] G. Whyburn, Analytic Topology, AMS Colloq. Publ. 28, 1942. (Cited on page 104.)
- [Yin] Y. Yin. On the Julia sets of quadratic rational maps. Complex variables. 18, 141-147,1992. (Cited on page 33.)
- [YZ] Y. Yin and Y. Zhai, No invariant line fields on Cantor Julia sets, Forum Math. 22(2010), 75-94. (Not cited.)
- [YaZ] M. Yampolsky and S. Zakeri, Mating Siegel quadratic polynomials. J. AMS. 14(1), 25-78. (Cited on pages 181 and 182.)

- [Yo] J.-C. Yoccoz, Petits diviseurs en dimension 1, Astérisque No. 231 (1995), 3-88. (Cited on page 86.)
- [Zhai] Y. Zhai, Dynamics of rational maps with Cantor Julia sets, Thesis(in Chinese), Zhejiang University. 2008 (Cited on page 14.)
- [Zh1] G. Zhang, All bounded type Siegel disks of rational maps are quasidisks, Inventiones Mathematicae, pp. 1-46, 2011. (Cited on page 178.)
- [Zh2] G. Zhang, Dynamics of Siegel rational maps with prescribed combinatorics. arXiv:0811.3043v1. (Cited on pages 5, 6, 108, 112, 178 and 179.)